Integrated Functional Adult Literacy Education for Empowerment and Sustainable Development in Ethiopia

A Comparative Study of Tigray and Amhara Regions

FESSEHA ABADI WELDEMICHAEL



Faculty of Educational Sciences, Department of Education

UNIVERSITY OF OSLO 2018

Integrated Functional Adult Literacy Education for Empowerment and Sustainable Development in Ethiopia

A Comparative Study of Tigray and Amhara Regions

FESSEHA ABADI WELDEMICHAEL

Faculty of Educational Sciences, Department of Education

UNIVERSITY OF OSLO

2018

© Fesseha Abadi Weldemichael

2018

Integrated Functional Adult Literacy Education for Empowerment and Sustainable Development in Ethiopia: A Comparative Study of Tigray and Amhara Regions

Fesseha Abadi Weldemichael

http://www.duo.uio.no/

University of Oslo

Acknowledgements

Writing this dissertation would not have been realized had it not been for the contribution of many people and organizations. I would like to acknowledge just some.

I am very grateful to the Norwegian Government and the Institutional Collaboration between Mekelle and Hawassa Universities and the Norwegian University of Life Sciences for making my study possible. I am thankful to the Norwegian government for funding my living expenses through Lånekassen within the Quota Scheme during my PhD stay in Oslo for five years. The Institutional Collaboration Project deserves special thanks because it has covered all the finance required for this PhD research. The Faculty of Educational Sciences at UiO, in general, and the Department of Education, in particular, including Ulf Tero Grefsgård and Olga Mukhina, as well as the Student Information and Communications Office at UiO, especially Michele Nysæter, the executive officer, deserve distinctive acknowledgements for their unreserved support for the smooth progress of my PhD study.

I greatly acknowledge my supervisors Professor Lene Buchert and Dr Wim Hoppers for giving me unceasing and meticulous advice to make the research a better piece of work. Professor Lene's pleasant advice, untiring efforts to read several drafts of my dissertation, critical comments, dynamic discussions, and care have been beyond what words can express. Dr Wim's critical remarks and questions, as well as challenging feedback have made a difference in improving the quality of the dissertation. Even more, I have learned a lot from both of them as regards what supervision should be like. I look forward to our continued collaboration in publishing, research, and more.

I also would like to thank Professor Tone Kvernbekk, my philosophy of science course instructor, since her ideas during and after the course helped me to develop the analytical framework for my study. I thank Dr Teklu Bekele and Dr Lihong Huang who read my PhD manuscript as midterm and final reviewers for their supportive comments and constructive feedback. I extend my appreciation to Associate Professor Emeritus Thor Arnfinn Kleven for validating the data analysis of the survey and providing valuable comments.

My special thanks go to my colleagues and friends Dr Kinfe Abraha, Ato Habtu Nigus, Dr Melaku Abera, Dr G/Hawaria G/Her and Senait Luel for their support in the data definition, data entry, and data analysis of the survey data. Heartfelt thanks go to Ato Tesfamariam Abay, Ato Yemane Kidanu, Ato Tewelde Tesfai, Ato Desta H/Selassie, Dr Dereje Assefa, Dr

G/Eeyesus Teklu, Dr Yideg Alemayhu, Dr Tesfatsion Sahlu, Ato Abebe Haile, Ato Tekleberhan Araya, Dr Abel Gizaw, Dr Desalegn Chala, Dr Shewit Kalayu, Dr Mulugeta Tsegai, Ato Haile G/Medhin, Ato Atakilti Mekonnen, Weizero Gimbo Mekonnen, Ato Haile Asmamaw, Ato Abrahaley Berhe, Ato Yibabe Tilahun, Ato Sahle Sumur, Weizero Rahel Belay, Dr Messele Terecha, Ato Dawit Girmay, Weizero Mahder Mekonnen, Ato Teame Kiros, and Caro Kirsebom and family. They all provided me with technical, material and moral inspiration, motivation, support, peer pressure and mentoring along the PhD journey.

I also extend my gratitude to the staff in the MoE, Tigray Bureau of Education, Eastern Zone of Tigray Social Affairs Bureau, Saesae-Tsaeda Emba and Kilte-Awlaelo Wereda education offices, Amhara Bureau of Education, Northern-Wello Education Office, Meqhet and Wadla Wereda education offices, DVV International (Head Office, Addis Ababa), ANFEAE (Addis Ababa), UNESCO (Addis Ababa branch), Pact-Ethiopia and all other interview and survey participants who served as sources of information for the study.

I would like to extend my heart-felt gratitude to my colleagues and friends for collecting the interview and observation data as well as transcribing and translating the data; namely, Dr Abraha Tesfay, Dr Tesfaye Messele, Dr Mehari Yimulaw, Dr Mesfin Eyob, Ato Kiflom Teklemariam, Ato Teklu Equar, and Ato Tsegay Girmay. I would also like to thank Bisrat Kindeya, Mahlet Hailu, Medhin Halefom, Merhawit Tsegay, Samrawit Niguse, Abebech Hailemariam, Hiwot Tekle, Meseret Alem, Mulu Belay, Tsehay Mekuriaw, and Ato Wondwessen Fetene for conducting quality interviews under challenging fieldwork conditions. I wish to express my sincere thanks to Ato Atakilti Hailu and Ato Kahsay Tadelle for their contributions as drivers and as research assistants during the data collection.

Finally, I am indebted to my family for their assistance in different respects. I am greatly thankful to my wife Genet who took the twin responsibility of a father and a mother during my absence for the study. My niece Eyerusalem, her support to the family, especially to little Silvana, and her presence in the family in my absence deserve high recognition. I also appreciate her technical support to my PhD study. I would like to thank my sons Matusala and Biniam, and my daughter Silvana for their patience and discipline throughout my PhD study. My Mom, Alefech Gebrewahid, and my sisters Kiros and Haregeweini with their children provided me both moral and technical support during my study. I thank them all!

Fesseha Abadi Weldemichael 2018 Oslo, Norway

Dedication

To my nephew Gebrehiwot Tetemke Abadi (nicknamed *Mahta*)—in honour of his contributions and efforts.

Abstract

The implementation and contributions of adult literacy programs in Ethiopia during the three governments since Emperor Haile-Selassie and until now differ. Ethiopia promulgated a new Education and Training Policy (ETP) in 1994 that ignored adult literacy education as a means to provide the rural illiterate majority with their right to basic education. However, in 2008 the Ministry of Education developed the National Adult Education Strategy focusing on Integrated Functional Adult Literacy Education (IFALE). The strategy aims to develop the knowledge and practices of adult participants to support accelerated and sustained development efforts to make Ethiopia a middle-income country by 2025. The purpose of the current study is to explore the implementation of IFALE and its contributions to learner empowerment and sustainable development (SD) efforts in the country. It is a regional, comparative, rural-focused study of Tigray and Amhara regions, using a mixed methods design. Data collection through interviews, observations, document analysis and a survey took place in 2013.

The results from observations and interviews indicate that the implementation of the IFALE policy in terms of the strategies of a relevant curriculum and building an efficient institutional system was unsuccessful in Tigray but partly successful in Amhara. The comparison between IFALE participants and non-participants, based on the survey results, indicates that IFALE results in significant contributions in limited areas of the political, economic, socio-cultural and psychological empowerment dimensions in both regions. The IFALE participants in Tigray have gained more in political empowerment than those in Amhara, whereas the participants in Amhara gained more in socio-cultural achievement than those in Tigray. The findings indicate that economic empowerment gains have generally been constrained in both regions but more so in Amhara than in Tigray. The findings that IFALE participants had significantly improved saving practices in both regions and participated in income generation schemes in Tigray are important seen in light of previous studies. IFALE seems to have improved self-efficacy but not improved self-concept in both regions. Therefore IFALE empowered participants in limited areas, through improving learning, capabilities and livelihoods at personal, family, and/or societal levels which can ultimately support SD in some ways.

The results also show that in Amhara IFALE embraces most of the major features of education for sustainable development (ESD) as defined by UNESCO (2006b) while in Tigray it embraces only half of the features, suggesting a difference in the extent of support for SD in the two regions. According to the findings, in Amhara IFALE had partly successful implementation of a relevant curriculum and an efficient institutional system, led to significant empowerment gains in limited areas, and fulfilled most of the major ESD features. This implies that IFALE gave moderate support to SD in its own right. By contrast, in Tigray, the findings that the implementation of a relevant curriculum and an efficient institutional system was not successful, the empowerment gains were limited, and half of the ESD features were not embraced suggest that IFALE's support to SD was restricted.

For both regions, the result that IFALE appears to not have led to substantial gains in most indicators of the empowerment dimensions and that it did not fulfill at least three of the ESD features may have negatively affected its potential contribution to the four components of SD (environment, economy, society and culture). The precondition to doing so is political will. The regional comparison contributes to identifying relevant contextual factors that can help future implementation of IFALE.

Table of Contents

Acknowledgements	IV	
DedicationV		
Abstract	VII	
Table of Contents	VIII	
List of Tables.	XIII	
List of Figures	XV	
List of Acronyms/Abbreviations		
Chapter 1 Introduction	1	
1.1 Background		
1.2 Rationale of the Study		
1.3 Concepts of Adult Education, Adult Literacy and Life Long Learning	6	
1.4 Understanding Adult Literacy Education within the Context of		
Empowerment and Sustainable Development	9	
1.4.1 Adult Literacy Education and its Models	9	
1.4.2 The Effects of Literacy		
1.5 Research Questions and Organization of the Study		
Chapter 2 Adult Literacy Education in Ethiopia		
2.1 The Introduction of Education into Ethiopia		
2.2 Adult Literacy Education during Imperial Rule (1930-1974)		
2.2.1 Early Adult Literacy Education Efforts, 1948-1964		
2.2.2 The Work-Oriented Adult Literacy Program (WOALP), 1968-1973		
2.2.3 Summary		
2.3 Adult Literacy Education during the Derg Military Rule (1974-1991)		
2.3.1 The Development through Cooperation, Knowledge and Work		
Campaign (Zemecha), 1974-1976		
2.3.2 The National Literacy Campaign (NLC), 1979-1991		
2.3.3 Summary		
2.4 Adult Literacy Education during the Ethiopian Peoples' Revolutionary		
Democratic Front (EPRDF), 1991-now		
2.4.1 Adult and Non-formal Education (NFE) in the Education Sector		
Development Plan (ESDP I) and the Education Sector Development		
Plan II (ESDP II), 1997/8-2004/5		
2.4.2 Adult Literacy in ESDP III (2005/6-2009/10) and the National Adult		
Education Strategy (NAES), 2008		
2.4.3 Links between IFALE, Learning for Life: A Master Plan for Adult		
Education in Ethiopia (2010-2020) and ESDP IV (2010/11-2014/15)		
2.4.4 IFALE for Sustainable Development		
2.5 Key Issues in the Literacy Programs of the Three Governments		
Chapter 3 Review of Literacy Studies and of the Concepts of Empowerment and		
Sustainable Development		
3.1 Empirical Studies: Design, Implementation, and Impact Assessment		
Studies of Adult Literacy Education		
3.1.1 Design of Adult Literacy Education		
3.1.2 Implementation of Adult Literacy Education		
3.1.3 Adult Literacy Education Impact Assessment Studies		
3.1.4 Conclusion		
3.2 Theoretical Studies: Literacy Models		

3.2.1 The Functional Model	
3.2.2 The Critical Model	
3.2.3 The Social-Practices Model	
3.2.4 Conclusion	
3.3 Empowerment and Sustainable Development in the Context of Adult	
Literacy	58
3.3.1 The Concept of Empowerment	
3.3.2 The Concept of Sustainable Development	
3.4 Conclusion	
Chapter 4 Causation and Evidence-based Policy as the Foundation for the	/ 1
Analytical Model of the Study	74
4.1 Understandings of Causation.	
4.2 The Notions of Support Factors, Causal Cakes and Their Ingredients	
4.3 A Policy for Better Results	
4.4 Understanding the Causation of IFALE, Empowerment and SD in Ethiopia	
4.5 The Causation of IFALE.	
4.5.1 The Strategy of a Relevant Curriculum.	
4.5.2 The Strategy of an Efficient Institutional System	
4.6 The Causation of Learner Empowerment and SD	
4.7 Analytical Model for the Study	
Chapter 5 Research Design and Methodology	
5.1 Realism as the Epistemological Stance	
5.2 Comparative Case Study and Study Locations	
5.3 Exploratory Sequential Mixed Methods Design	
5.4 Sampling and Instruments of Data Collection 5.4.1 Semi-structured Interviews	
5.4.2 Observations	
5.4.3 Survey	
5.4.4 Document Analysis	
5.5 Validity and Reliability	
5.6 Ethical Issues	
5.7 Conclusion	
Chapter 6 IFALE and the Implementation of a Relevant Curriculum and an	
Efficient Institutional System	
6.1 Implementing the Strategy of a Relevant Curriculum	
6.1.1 Understanding of a Relevant Curriculum	
6.1.2 Primers and Other Materials	
6.1.3 Qualification of Officials, Experts and Facilitators	
6.1.4 Use of Active Learning Methods	109
6.1.5 Learner Motivation and Access to IFALE	
6.1.6 Appropriate Environment and Time for Learning	
6.1.7 Summary	
6.2 The Strategy of Building an Efficient Institutional System	
6.2.1 Understanding of an Efficient Institutional System	
6.2.2 Leadership	
6.2.3 Staff Motivation	
6.2.4 Policy Direction and Understanding of IFALE	
6.2.5 Collaboration among Stakeholders	
6.2.6 Working Conditions	
6.2.7 Budget	117

6.2.8 Summary	118
6.3 IFALE Participants' Proposed Solutions	
6.4 Conclusion	
Chapter 7 IFALE's Contribution to Learner Empowerment	121
7.1 IFALE and Political Empowerment	
7.1.1 Importance and Practice of Participation in Voting	
7.1.2 Interest in Becoming an Elected Representative	
7.1.3 Participation in Decision Making in Public Institutions	123
7.1.4 Role in Local 2013 Election	
7.1.5 Knowledge of Minimum Voting Age and Political Parties	125
7.1.6 Confidence to be a Locally Elected Representative	
7.1.7 Knowledge of a Law on Electing Representatives	
7.1.8 Membership of Community Organizations	
7.1.9 Knowledge of Community Development Committees	
7.1.10 Participation in Community Development Activities	
7.1.11 Sources of Political Knowledge and Practice	
7.1.12 Summary	
7.2 IFALE and Economic Empowerment	
7.2.1 Importance and Practice of Agricultural Technologies	
7.2.2 Importance and Practice of Environmental Protection	
7.2.3 Importance and Practice of Saving	
7.2.4 Sources of Knowledge and Practices of Agricultural Technologies	
and Environmental Protection Measures	133
7.2.5 Financial Contributions to Productive Investment and/or Mutual	
Support	134
7.2.6 Sources of Knowledge and Practice of Financial Contributions	135
7.2.7 Access to Credit	135
7.2.8 Income Generation Schemes	136
7.2.9 Summary	137
7.3 IFALE and Socio-Cultural Empowerment	
7.3.1 Gender Equality in Decision Making	138
7.3.2 Understanding Gender Equality for Development	
7.3.3 Knowledge of Laws against Gender Discrimination	140
7.3.4 Knowledge of Laws Protecting Women and Children	140
7.3.5 Knowledge Sources of Gender Equality, and Laws against Gender	
Discrimination and for Protection of Women and Children	141
7.3.6 Importance and Practice of Family Cohesion	
7.3.7 Importance and Practice of Specific Traditional Activities	
7.3.8 Importance and Practice of Wanting an Additional Child	
7.3.9 Importance and Practice of Family Planning	
7.3.10 Importance and Practice of Personal Hygiene	
7.3.11 Importance and Practice of Maternal and Child Care	
7.3.12 Importance and Practice of Measures of HIV/AIDS Control	
7.3.13 Importance and Practice of Social Justice	
7.3.14 Importance and Practice of Democratic Culture	
7.3.15 Importance and Practice of Conflict Resolution	
7.3.16 Sources of Socio-Cultural Empowerment Gains	
7.3.17 Summary	
7.4 IFALE and Psychological Empowerment	
7.4.1 Importance and Practice of Self-Concept	156

7.4.2 Importance and Practice of Self-Efficacy	157
7.4.3 Sources of Knowledge and Practice of Self-Concept and Self-Efficacy	
7.4.4 Summary	
7.5 Conclusion	
Chapter 8 IFALE as Education for Sustainable Development (ESD)	
8.1 IFALE and ESD	163
8.1.1 IFALE and the Design of an Interdisciplinary and Holistic	
Curriculum	164
8.1.2 IFALE and Critical Respect for Others, for Difference and	
Diversity, for the Environment, and for the Resources of the	1.00
Planet We Inhabit	
8.1.3 IFALE and Sharing the Values and Principles Underpinning SD	16/
8.1.4 IFALE and Endeavors to Address Critical Thinking and Problem	160
Solving Skills of Its Participants	108
Teaching and Learning Process	160
8.1.6 IFALE and Implementation of Participatory Decision Making in	109
Teaching and Learning.	170
8.1.7 IFALE and Integration of Learning Experiences in Daily Life	
8.1.8 IFALE and Local Relevance: Addressing Local and Global Issues	
and the Language of the Participants	
8.2 Conclusion	
Chapter 9 Major Findings, Key Conclusions, and Significance of the Study	
9.1 Implementation and Contribution of IFALE	
9.1.1 Implementation of IFALE	178
9.1.2 IFALE and Empowerment	
9.1.3 IFALE for Empowerment and Sustainable Development	
9.2 Answering the Research Questions	182
9.2.1 How successful is the implementation of IFALE in terms of the	
strategy of a relevant curriculum?	182
9.2.2 How successful is the implementation of IFALE in terms of the	100
strategy to build an efficient institutional system?	
9.2.3 What is the contribution of IFALE to political empowerment?	
9.2.4 What is the contribution of IFALE to economic empowerment?	
9.2.5 What is the contribution of IFALE to socio-cultural empowerment?9.2.6 What is the contribution of IFALE to psychological empowerment?	
9.2.6 What is the contribution of IFALE to psychological empowerment? 9.2.7 How does IFALE support SD?	
9.3 Reflections on Conclusion	
9.4 Significance of the Study	
9.4.1 Significance for Academia	
9.4.2 Significance for Policy Makers	
References.	
Appendices	
A. Interview Protocols	
1. Interview Protocol for IFALE Participants	206
2. Interview Protocol for IFALE Facilitators	
3. Interview Protocol for Wereda IFALE Officials and Supervisors	
4. Interview Protocol for Regional IFALE Officials and Focal Persons	217
5. Interview Protocol for Policy Makers	
6. Interview Protocol for Collaborators of Adult Literacy Education	223

B. IFALE Session Observation Protocol	
C. IFALE Center Observation Protocol	
D. Individual IFALE Participant Observation Protocol	
E. Survey for IFALE Participants and Non-Participants	
F. Research Permit	
G. University of Oslo's Confirmation Letter for Data Collection	
H. Mekelle University's Confirmation Letter for Data Collection	
I. Informed Consent Form for Interview	
J. Letter from Kilte-Awlaelo Health Office (Tigray)	
K. Letter from Meghet Agriculture and Rural Development Office	
(Amhara)	
L. Table on gender equality in decision making in the family by number	
of IFALE participants and non-participants and by gender	

List of Tables

Table 2.1: Enrollment in literacy programs, 1964-1971	
Table 2.2: Enrollment in adult and non-formal education, 2003/4	
Table 2.3: Conditions and challenges for ALE under different governments	
in Ethiopia compared to the world ALE trend	
Table 3.1: Purpose, focus and research methods in selected literacy models	
Table 3.2: A livelihoods framework for monitoring the impact of literacy	
interventions	
Table 5.1: Basic facts of Tigray and Amhara	91
Table 5.2: Sampling procedures for study area and population	
Table 5.3: Category and number of interviewees	
Table 5.4: Center and participant observation by region and number	
Table 5.5: Number of survey participants in Tigray and Amhara	
Table 5.6: Domains of empowerment by indicator, measure and value	
Table 5.7: Sources for document analysis by type and number	
Table 6.1: Time allotted by method in IFALE classes in Tigray and Amhara by	
center	109
Table 6.2: Enrollment, completion, drop out and dropout rate of IFALE learners	
in Tigray and Amhara, 2011/12-2012/13	110
Table 6.3: Adult and non-formal basic education enrollments during ESDP IV,	
2010/11-2014/15	
Table 6.4: Support factors for the strategy of a relevant curriculum	119
Table 6.5: Support factors for the strategy of an efficient institutional system	120
Table 7.1: Self-reported political empowerment gains of survey participants by	
mean and statistical significance	123
Table 7.2: Survey participants' self-reported participation in the local election in	
2013 by frequency and percent	124
Table 7.3: Self-reported political empowerment gains of survey participants by	
frequency, percent and level of significance	126
Table 7.4: Self-reported sources of survey participants' knowledge and practice	
of political activities by frequency and percent	128
Table 7.5: Self-reported economic empowerment gains of survey participants	
by mean and statistical significance	131
Table 7.6: Survey participants' self-reported sources of knowledge and practice	
of agricultural technologies and environmental protection measures	
by frequency and percent	133
Table 7.7: Self-reported credit access gains and financial contributions of survey	
participants by frequency, percent, and statistical significance	134
Table 7.8: Self-reported sources of survey participants' knowledge and practice	
of financial contributions by frequency and percent	135
Table 7.9: Survey participants' self-reported sources of credit by frequency and	
percent	136
Table 7.10: Survey participants' self-reported involvements in income	
generation schemes by frequency and percent	137
Table 7.11: Married survey participants' self-reports on gender equality in	
decision making in the family by frequency and percent	139
Table 7.12: Self-reported socio-cultural empowerment gains of survey	
participants by frequency, percent, and statistical significance	140

Table 7.13: Survey participants' self-reported knowledge of gender equality,	
and laws against gender discrimination and protection of women and	
children by frequency and percent	. 141
Table 7.14: Self-reported socio-cultural empowerment gains of survey	
participants by mean and statistical significance	. 147
Table 7.15: Survey participants' self-reported sources of knowledge and	
practice of socio-cultural gains by frequency and percent	. 154
Table 7.16: Self-reported psychological empowerment gains of survey	
participants by mean and statistical significance	. 158
Table 7.17: Survey participants' self-reported sources of knowledge and practice	
of self-concept and self-efficacy by frequency and percent	. 159
Table 7.18: IFALE's contribution to empowerment in Tigray and Amhara	. 161
Table 8.1: ESD features embraced in IFALE in Tigray and Amhara	. 177

List of Figures

Figure 2.1 Map of Ethiopia	16
Figure 3.1: Dimensions of functional literacy and their outcomes	61
Figure 4.1: A causal cake for improved reading scores	75
Figure 4.2: Causal connection of the IFALE policy to SD	77
Figure 4.3: A cake for IFALE in Ethiopia: curriculum and support factors	
Figure 4.4: A cake for improved IFALE: institutional system and support factors	
Figure 4.5: A cake for the IFALE policy and support factors for learner	
empowerment and SD.	83
Figure 4.6: A cake for learners' empowerment and SD: FTCs and support factors	
Figure 4.7: A model of IFALE causal pathways for learner empowerment and SD	85
Figure 5.1: Study locations in Tigray and Amhara	
Figure 6.1: IFALE session (Center G) in Tigray conducted under trees after a	
terracing chore	112
Figure 6.2: IFALE session in Amhara (Center A) conducted in a classroom	113
Figure 8.1: Kitchen items in the house of an IFALE participant in Tigray	172
Figure 8.2: Kitchen items in the house of an IFALE participant in Amhara	
Figure 8.3: IFALE participant's latrine in rural Tigray	173
Figure 8.4: IFALE participant's latrine in rural Amhara	174
Figure 8.5: IFALE participant's house in Tigray with a separate shelter for	
domestic animals	174
Figure 8.6: IFALE participant's house in Amhara partitioned between the	
family and domestic animals	175
Figure 9.1: Summary of results of the implementation and contributions of	
IFALE to empowerment and SD	186
1	

List of Acronyms/Abbreviations

ABE	Amhara Bureau of Education
AE	Adult Education
ALE	Adult Literacy Education
BDEC	Basic Development Education Center
CBAE	Competency-Based Adult Education
CSTC	Community Skills Training Center
DESD	Decade of Education for Sustainable Development
EFA	Education for All
ESD	Education for Sustainable Development
EWLP	Experimental World Literacy Program
FAL	Functional Adult Literacy
HIV/AIDS	Human Immunodeficiency Virus/Acquired Immuno-Deficiency Syndrome
IALS	International Adult Literacy Survey
IFALE	Integrated Functional Adult Literacy Education
IFALP	Integrated Functional Adult Literacy Program
LLL	Lifelong Learning
MoE	Ministry of Education
NAES	National Adult Education Strategy
NDR	National Democratic Revolution
NFE	Non-Formal Education
NGO	Non-Governmental Organization
NLC	National Literacy Campaign
NLCC	National Literacy Coordinating Committee
NLCCC	National Literacy Campaign Coordinating Committee
NLCO	National Literacy Campaign Organization
OECD	Organisation for Economic Cooperation and Development
OED	Operations Evaluation Department
PA	Peasant Association
PASDEP	Plan for Accelerated and Sustained Development to End Poverty
PHMC	Philadelphia Public Management Corporation
PLO	Provincial Literacy Officers
RCT	Randomized Controlled Trial
REFLECT	Regenerated Freirean Literacy through Empowering Community Techniques
SD	Sustainable Development
SDGs	Sustainable Development Goals
TBE	Tigray Bureau of Education
UNESCO	United Nations Educational Scientific and Cultural Organization
WOALP	Work-Oriented Adult Literacy Program

WOALP Work-Oriented Adult Literacy Program

Chapter 1 Introduction

1.1 Background

The field of adult literacy education (ALE), also called adult basic education and adult literacy and numeracy, has received increasing attention in many countries as an indispensable part of education. According to Tett and St. Clar (2010), there are three reasons. The first one is the inspiration from the International Adult Literacy Survey of the mid-1990s (and the less influential Adult Literacy and Life Skills Survey of the next decade) which for the first time compared the skills of adults in different countries. The second reason is the advance towards the information society that considers literacy as the foundation for success. The final one is the assumption of agencies, such as the World Bank and United Nations Educational Scientific and Cultural Organization (UNESCO), that education is a key to prosperity. Scholars, for example Rogers (2001), also believe that literacy serves as an instrument for life improvement and minimizing poverty among poor people.

The importance of ALE has been clear at the global level. Expectations have been high for ALE to meet the needs of adults previously denied of basic education. The Dakar Framework for Action stipulated six Education for All (EFA) goals that aimed to meet the learning needs of all children, youth and adults by 2015 (UNESCO, 2000a). One of the goals was to achieve a 50% improvement in levels of adult literacy by 2015, especially for women, and equitable access to basic and continuing education for all adults. Goal 4 of the UN post-2015 agenda on Sustainable Development Goals (SDG) aims to ensure that all youth and a substantial proportion of adult men and women achieve literacy and numeracy by 2030 (UN, 2015). In addition, Goal 16 of the same agenda aims to 'promote peaceful and inclusive societies for sustainable development, provide access to justice for all and build effective, accountable and inclusive institutions at all levels' (UN, 2015). The literacy rate is one of the indicators for sustainable development (SD) (United Nations Economic Commission for Africa, 2005, p. 19, p. 23).

By engaging in adult literacy, the Ethiopian government is attempting to fulfill national, regional and international goals. As stated in its Plan for Accelerated and Sustained Development to End Poverty (PASDEP), Ethiopia strongly and ambitiously envisions being

an economically competitive, middle-income country by the year 2025. To achieve the desired social and economic transformation, the country recognizes that education plays a central role (MoE, 2010a). The Ministry of Education (MoE) believes that a high increase of the literacy rate supports other development goals of the country (MoE, 2010b, p. 12). In addition, efforts to expand ALE seem to be a way of both fulfilling the regional indicator of SD, and of the international commitments to universal primary education and increased literacy and numeracy rates for all youth and a higher percentage of adults by 2030.

Many studies show that Ethiopia has made much effort to expand primary, secondary, technical and tertiary education through successive Education Sector Development Plans (ESDPs) based on the Education and Training Policy (ETP) of the country which was published in 1994. Despite the high attention to Universal Primary Education (UPE), reports indicate that the absence of a comprehensive adult education (AE) policy has resulted in difficulty in terms of coordinating provisions and providers and in getting basic data on AE (Werqneh, 2011, p. 245). As UNESCO (2009/10) and World Bank (2014) reports show, about 27% of the total population in the country were non-literate adults (23.2 million of 87.1 million) in 2007. This indicates the country's low focus on ALE. Kebede (2005, p. 27) argues that ALE in Ethiopia 'remained peripheral to education policies' and minimal resources had been allocated due to the government's lack of political will. The adult literacy rate (15 years and older) in Ethiopia was 39% in 2007, one of the lowest rates in Africa (UNESCO Institute for Statistics, 2012, p. 13).

However, the Ethiopian government recently started to focus on ALE. Together with its partner institutions, the MoE designed the National Adult Education Strategy (NAES) in 2008. A number of successive ALE documents were produced. Among these, Learning for Life: Master Plan for Adult Education in Ethiopia (2010/11-2019/20) and the Education Sector Development Plan IV (2010/11-2014/15)—which includes the implementation of the Integrated Functional Adult Literacy Education (IFALE)¹ Program—are the most crucial. MoE (2008) considers that adult education, i.e., IFALE, can contribute as an integral part of the implementation of other development policies, strategies and packages to achieve accelerated and sustained development to end poverty. In ESDP IV, the initial target was to enroll 36.4 million 15-60 year old adults in a two-year IFALE program. This target was

¹ Different Minstry of Education (MoE) documents have different names for the current literacy program in Ethiopia, e.g. Functional Adult Literacy (FAL) and Integrated Functional Adult Literacy (IFAL). In this study, Integrated Functional Adult Literacy Education (IFALE) is used throughout.

adjusted in 2012 to instead reach 19.4 million youth and adults. MoE (2008) underlines that IFALE seeks to link writing, reading and numeracy skills to livelihoods and skills training in areas such as agriculture, health, civics and cultural education. The ambition appears to be to use IFALE as a tool for empowering underprivileged communities and eventually supporting SD.

This dissertation explores the implementation of IFALE and its role in participants' empowerment and SD in Ethiopia. This is done by comparing the Tigray and Amhara regions. This is an important area of study for both policy makers and the academic community. In terms of policy-making, the study contributes to global knowledge based on identifying best practices from the local context in a specific nation. It can thereby also assist future adult education policy and practice. In terms of academia, the study contributes to the scholarly debate on the link between an empowering adult education and sustainable development; it helps promoting understanding that adult education must be addressed in a socially comprehensive way; and it contributes to filling the gap on reliable data on adult education in Ethiopia.

1.2 Rationale of the Study

There are many arguments concerning the importance of adult education and adult literacy both as a field of study and as a part of government policies to sustain development efforts. It is thus important for both theory and practice.

Being comparative in nature, adult education can draw 'best practices' from international experiences and provide lessons to improve policy and practice at home (Torres, 2013). Yet the process of adoption or adaptation of a given learning experience or 'best practice' within a country and among countries must consider careful analysis of the context if it is expected to produce good results (Cartwright & Hardie, 2012; Torres, 2013). Within countries, studies must also be contextualized in order to 'lend and borrow' 'best practices'. For Arnove (2013), comparative data help guide theory to reach reasonable propositions about the outcomes and workings of lifelong learning systems understood in their social and historical contexts.

According to Torres (2013), adult learning education has three dimensions: scientific (contributing to theory building, research and practice); pragmatic (discovering what can be learned to improve policy and practice); and international or global (reaching a new importance with respect to the changes in the world system). Arnove (2013) claims that these

dimensions are essential to comparative and international education. As Torres (2013) suggests, they must be combined in order for adult education to be better informed and more effective in promoting educational policy and practice.

However, despite the agreed understanding of the importance of adult education and literacy for development and despite world-wide efforts, illiteracy continues to exist in diverse social contexts. According to Stromquist (2009, p. 71), literacy programs suffer from a shortage of resources, failure to gain a deeper understanding of adult learners, inadequacy to invent instructional methodologies suiting the learners' realities and needs, and inability to equip facilitators with skills required in order to address the demands of adult learners. Similarly, Betts (2002) underlines that recent developments raised a number of questions concerning literacy policy and practice: How can provision respond to the daily realities of the poor, i.e. to their needs, practices, hopes and aspirations? How can it avoid stigmatizing those who have not had the opportunity to learn how to read and write? How can literacy be made more relevant, accessible, equitable and extended to the very hardest to reach?

These questions are all relevant to the current study on Ethiopia. One reason for their relevance in Africa, according to Aitchison and Alidou (2009), is that literacy teachers in Africa are not only the least qualified, but also the lowest paid of all adult educators. In Ethiopia, adult and non-formal education is insufficient (low accessibility and inadequate distribution), of low level of relevance and quality, and with no organizational system to coordinate and lead programs and integrate the endeavors of different ministries and other partners (MoE, 2008; 2010b). ALE programs also encounter difficulties, such as inappropriate contents; absence of active learning methods in the teaching and learning process; low retention of literacy skills and practices; and lack of learner motivation (Reder, 2009; Stromquist, 2009; MoE, 2008). Other challenges are: lack of commitment at policy and application level; scarce financial resources; failure to reach marginalized illiterates; absence of coherent language policies and planning, coordination and partnership mechanisms; challenging processes of monitoring and evaluation; and low quality of literacy programs (Ito, 2013, p. 3).

Another reason, according to Stromquist (2009, p. 71), is that many adult literacy programs continue to emphasize coding and decoding of basic texts and fail to expose learners to more encompassing and sustained literacy practices. Programs should instead be developed that also focus on the retention of various skills since adult learners' literacy proficiency alone

may not be the most appropriate measure of program impact and other dimensions, such as engagement in everyday literacy and numeracy practices (Reder, 2009).

Scholars, such as Lauglo (2001) and Kanukisya (2012), recognize adult education as one of the factors playing a role in political, social, cultural and economic development in its own right. To Stromquist (2009, p. 65), adult literacy is important for both individual and social transformation as reflected in the concept of empowerment. It is political in nature. Therefore, literacy programs should go beyond mastery of literacy and numeracy skills, to include political, economic socio-cultural and psychological empowerment of learners in order to support a nation's SD. Literacy must, therefore, relate to other sectors of development than education and to civil society, particularly non-governmental organizations (NGOs), in order to produce collective empowerment of a nation.

SDG 16 aims to 'promote peaceful and inclusive societies for sustainable development, provide access to justice for all and build effective, accountable and inclusive institutions at all levels' (UN, 2015). This thinking should inspire scholarly debates on the links between adult education and sustainable societies. Education, in general, and adult education, in particular, enables learners, among others, to be active citizens, have respect for other humans, and acquire the values and principles underpinning SD. As the focus of this study is on IFALE's contributions to SD in Ethiopia, it can hopefully contribute to this kind of debate.

However, so far, normative expectations about literacy as a means to produce active citizens (a form of civic participation that engages individuals in organized political action beyond voting) seem not to have been put into practice (Stromquist, 2009, p. 73). UNESCO (2005a) and the Global Campaign for Education (GCE) (2009) recommend that literacy programs should have sustained post-literacy activities, and be coordinated with efforts to provide micro-credit for the poor. Addressing literacy in a socially comprehensive way will reduce the barriers to literacy participation by marginalized adults (Stromquist, 2009, p. 73).

Even if national governments, international institutions, and multilateral agencies readily endorse a normative view of literacy as a human right for all, they often disregard their own commitments to offering literacy programs of quality, indicating a noticeable discrepancy between intentions and genuine responses (Stromquist, 2009, p. 74). According to Kenea (2014, p. 256), the presence of legislation and official strategies in Ethiopia do not guarantee the success of a literacy program, suggesting that real commitment from both federal and local government bodies in the country is crucial. In Ethiopia, 'free will' to attend ALE will have no effect in the absence of government interventions to create a conducive learning atmosphere (Kenea, 2014, p. 256). Kebede (2005, p. 27) claims that political will and explicit policy is only meaningful when supported by the allocation of adequate resources.

What is clear, therefore, is the necessity to conduct studies on ALE in order for literacy programs to play their role in supporting learners and wider development efforts in a nation. Comings and Soricone (2007, p. 5) argue that 'research in adult literacy has the promise of having a positive impact on the lives of millions of adults'. However, several authors claim that there are few adult literacy studies fulfilling the standards of scientific research (Beder, 1999; Kruidenier, 2002; Comings, Soricone, & Santos, 2006; Comings & Soricone, 2007). Comings and Soricone (2007) also claim that there are few studies on design and evaluation of literacy programs. According to MoE (2010b), there is a lack of adequate and reliable information on ALE in Ethiopia.

In conclusion, Ethiopia has an explicit strategy to improve and expand adult literacy, particularly through IFALE, in order to support political, socio-economic and cultural transformation and become a middle-income country by 2025. The purpose of this study is to investigate whether this strategy is implemented in practice, taking into consideration the importance of adult education and literacy for empowerment at personal, community and national levels, and understanding potential barriers in light of the factors identified above. Before doing so, a deeper conceptual discussion of the key concepts for adult education is presented.

1.3 Concepts of Adult Education, Adult Literacy and Life Long Learning

The field of adult education refers to several important concepts including the terms education, literacy and learning. *Adult literacy education* (ALE) which is set within the perspective of lifelong learning (LLL) is a part of AE. The concept of AE is narrower. According to Merriam and Brockett (1997, p. 13), AE is a set of 'activities intentionally designed for the purpose of bringing about learning among those whose age, social roles, or self-perception define them as adults'. In contrast, UNESCO (1976) defines AE more widely, capturing the dimensions of personal and national development that are also essential to Stromquist, as discussed earlier.

The term 'adult education' denotes the entire body of organized educational process, whatever the content, level, method, whether formal or otherwise, whether they

prolong or replace initial education in schools, colleges and universities as well as in apprenticeship, whereby persons regarded as adults by the society to which they belong develop their abilities, enrich their knowledge, improve their technical or professional qualifications or turn in a new direction and bring about changes in their attitudes or behavior in a twofold perspective of full personal development and participation in balanced and independent social, economic and cultural development (p. 2).

In this understanding, any educational program, be it formal or non-formal, is part of AE if it targets adults. Programs might focus on agricultural extension, in-service training, literacy, out-of-school education, vocational education, community development, and cooperative learning (Youngman, 1998). According to Nafukho, Amutabi and Otunga (2005, p. 7), parallel degree programs, self-sponsored degree programs, mature-entry programs, privately sponsored degree programs, prison education, non-formal education, informal education, distance education, human resource development, Acquired Immuno-Deficiency Syndrome (AIDS) awareness education, and midwives' education are also used to denote AE. In the Ethiopian context, adult and non-formal education, i.e. Integrated Functional Adult Literacy, aims at providing basic education opportunities to adults who are 15 years and above (MoE, 2008, 2010c). AE is part of LLL (Nafukho et al., 2005; Torres, 2006) and can contribute to 'the development of a LLL culture' (Kanukisya, 2012, p. 10).

However, 'repositioning AE within LLL requires a shared philosophy of the purposes and benefits of adult learning' (UNESCO, 2009b, p. 23). UNESCO (2009b) suggests a philosophy for LLL in accordance with Sen's (1999) capabilities approach². This approach is also supported and expanded by Nussbaum (2000). In this approach, the principal objective of development is freedom based on the expansion of human capabilities. It embraces not only the economic dimension and the pursuit of well-being, but also the notions of affiliation, for instance the capability of social interaction and political participation.

Lifelong education and LLL have been subject to debate among scholars since the 1970s. The attention gradually shifted to the latter because learning takes place through systematized formal interventions, and through experience and knowledge acquired by non-formal and informal forms of learning. In the European Employment Strategy, LLL is an 'all purposeful learning activity, undertaken on an ongoing basis with the aim of improving knowledge, skills

² Amartya Sen's (1999) capability approach to development entails much more than increasing income and wealth. Poverty is understood as a deprivation of basic capabilities in the form of high mortality, significant under-nourishment, morbidity and widespread illiteracy. Poverty is, thus, a limitation on freedom. For Sen, the enhancement of human freedom is both the main objective and the means of development. The bases of human freedom are economic facilities, political freedom, social opportunities, transparency guarantees, and protective security. People must be actively involved in shaping their own destinies, and not be passive recipients of development programs. In this context, AE is important to enable and empower communities to realize social, political and economic freedoms.

and competence' (European Commission, 2000, p. 3). According to Torres (2013, p. 67), LLL was developed to provide 'a unified and compelling vision of the future' and to produce change.

Delors (1996, pp. 86-91) argues that LLL is founded on four pillars of education: learning to know (mastering instruments of learning instead of acquiring itemized or structured knowledge); learning to do (equipping people for the types of work needed now and in the future including innovation and adaptation of learning to future work environments); learning to live together and with others (peacefully resolving conflict, discovering other people and their cultures, fostering community capability, individual competence and capacity, economic resilience, and social inclusion); and learning to be (education contributing to a person's complete development: mind; body; intelligence; sensitivity, aesthetic appreciation; and spirituality). UNESCO (2005c) suggests that the principles from the four pillars should be cautiously followed in the design of carefully planned and culturally relevant adult literacy programs.

Ouane (2008) has suggested an additional fifth pillar—learning to change and to take risk. Learning to change refers to proactively directing or redirecting change for human well-being and development (Torres, 2003). This means to take account of past as well as future contexts for LLL, linking learning at all levels of education with national and human development aspirations (Preece, 2013). In the view of Hoppers and Yekhlef (2012), LLL is lifelong, voluntary, and self-motivated pursuit of knowledge for either personal or professional purposes. The notion of 'accidental learning' throughout life could perhaps be added to the concept of LLL (Power & Maclean, 2011). According to Walters, Yand, and Roslander (2012), LLL is multi-dimensional and has three components. The first is life-wide learning that relates to the breadth of learning across family, cultural settings, communities, work and leisure. The second is life-deep learning that relates to contemplative, meditative, spiritual learning practices. The third one is lifelong learning that relates to the four stages of life: childhood (includes children and young people who are dependent); productive age (when people can enhance productivity and prosperity); maturity age (older people who are still active and engaged); and old age (when people are dependent due to old age). In short, LLL in its broadest sense is a continuous process that embraces all kinds of learning from early to late life.

To Walters et al. (2012), LLL includes learning behaviors and obtaining knowledge, understanding, attitudes, values and competencies for personal growth, social and economic

well-being, democratic citizenship, cultural identity and employability. In addition to social inclusion, active citizenship and personal development, LLL also aims to enhance competitiveness and employability (Baku, Asiegor & Boakyi-Yiadom, 2011). Formal, non-formal and informal forms of learning are often considered as inseparable for which reason there is growing recognition of 'the porous boundaries between different forms of education and learning' (Walters et al., 2012, p. 26; Walters and Cooper, 2011).

Torres (2006), Walters et al. (2012), and Nafukho, Amutabi and Otunga (2005, p. 6) claim that LLL can potentially assist people in sustainable socio-economic development. Nafukho et al. (2005, p. 6) argue that adult education and LLL are crucial for Africa to achieve meaningful and sustainable development in the future. Preece (2013, p. 103) argues that 'LLL agendas have to harness the *local knowledge, talent* and *wisdom* of Africa by drawing on its own context specific heritage—but within a modern frame of reference' (Preece, 2013, p. 103). However, positioning AE within the LLL perspective is challenging, especially for African countries such as Ethiopia, because of low levels of human, material and financial capacities.

Considering the fact that AE should be situated within the perspective of LLL, it is important to have a general understanding of ALE as a principal component of AE. The following discussion is on ALE in relation to the two important concepts of the study: empowerment; and SD.

1.4 Understanding Adult Literacy Education within the Context of Empowerment and Sustainable Development

1.4.1 Adult Literacy Education and its Models

According to Stromquist (2009, p. 75), the social relevance of literacy can enhance the empowerment of learners. In other words, ALE that considers the local context and responds to the needs of learners is empowering. Furthermore, empowering ALE is important to support SD.

Scholars have defined literacy differently. Okech (2009, p. 17) states: 'to talk meaningfully about literacy, it is important to adopt a common basis of understanding the concept of literacy'. He defines literacy as:

The meaningful acquisition, development and use of reading and writing (also for numeracy purposes) in everyday life, as a tool for self-expression, information, communication, LLL, work and civic participation, and as a means to improve one's life and to contribute to family, community and national transformation and development (p. 17).

This definition implies that literacy, improvement of the learner's life, and social and national development are interrelated.

ALE is marked by a high level of diversity in terms of structure, delivery, and philosophy and performs different roles in different parts of the world, whether industrialized Europe or a developing country (Tett & St. Clar, 2010). Adult literacy can be carried out in diverse settings (in formal schools, church basements, workplaces, or community settings), can be delivered by professionally qualified facilitators or unqualified volunteers, and can be provided to employed or unemployed, women or men, and full-time or part-time learners. Each context is supposed to require the design and implementation of a distinctive literacy education.

According to Papen (2005), there are different ways of conceptualizing adult literacy, each relying on different understandings of learning. The major ones include the functional, the critical, and the social-practices models. The *functional model of literacy* is the ability to use skills to perform certain tasks in domestic or work life. In this model, skills are the foundation for other functions. As a result, they are taught in a generic fashion and transfer across situations is understood largely as unproblematic (Culligan, 2005). The functional, vocational approach resulting in a discourse of literacy as technical skills and vocational competence has achieved more popularity globally. The *critical literacy*, which is associated with the Brazilian educator Paulo Freire, refers to the potential of literacy as both 'reading the word' and 'reading the world' (Freire and Macedo, 1987). Its pedagogy aims at making participants understand their world in terms of justice and injustice, power and oppression, and transformation (Papen, 2005). The proponents of critical literacy believe that it is the only means to empower learners. The *social-practices model* on literacy looks at reading not only as a skill, but also as a social activity that is always situated in particular cultural and historical contexts (Barton, Hamilton, & Ivanic, 2000; Papen, 2005).

Each literacy model has drawbacks. Targeting only the mastery of literacy and numeracy skills makes the functional model reductionist. The claim that only critical literacy can attain psychological empowerment of learners implies that this is not possible through functional and social practices models. The social practices advocates focus on local context without

concerns for the capacity of literacy to achieve improved participation of learners in the economic, social and political spheres. And just like the understandings of literacy and its purposes differ in different schools of thought, so does the understandings of its outcomes.

1.4.2 The Effects of Literacy

There are different understandings of the outcomes of adult literacy. Lind (1985), Gee (2007) and Freire (1970) discuss its negative effects, for example, as an instrument for social control (Lind 1985; Gee, 2007) and negatively contributing to development (Lind, 1985). Freire (1970) argues that literacy conducted by oppressors dehumanizes, domesticates and reinforces social injustice in the learners. As cited in UNESCO (2005c, p. 138), Graff (1987) argues that literacy programs and written materials can be a mechanism to indoctrinate people to participate uncritically in a political system. Literacy education delivered in official languages can lead to the extinction of local languages.

According to UNESCO (2005c, p. 138), it is difficult to provide 'a systematic, evidence based account of the benefits of literacy' for several reasons. First, most studies fail to separate whether the benefits emanate from attending a school or a literacy program or both. Second, adult literacy education is under researched and available studies focus primarily on women. Third, literacy studies emphasize literacy impact on the individual, not on the family/community/national/international arena. Fourth, some effects of literacy, including those on culture, are inherently difficult to define and measure.

Despite the various challenges associated with adult literacy studies, many scholars report the benefits of literacy programs. Adult literacy enables learners to master basic reading, writing and calculating. Literacy not only improves the lots of the needy, but also contributes to nations' overall social and economic development (Coombs, 1985). Literacy promotes effectiveness in addressing economic, socio-political, educational, cultural and environmental issues (Davidson, 1990). Literacy can thus liberate the poor from ignorance, disease and hunger.

Gray (1990) has also identified what gives a literate person an advantage over one who is not, i.e. the empowering capacity of literacy. According to Gray, literacy helps to meet many of the literates' practical needs of daily life and to improve their standard of living, their economic status through engaging in vocations, and their individual and group activities that require knowledge of reading and writing. This eventually helps them gain social prestige, understand community activities, and meet their civic obligations, such as voting.

International organizations and scholars have also recognized other benefits of literacy. According to UNESCO (2005c, p. 22), participation in literacy can be linked to wide-ranging benefits, for example individual self-esteem, confidence and personal empowerment, increased civic engagement (including in labor unions, community activities or politics), and knowledge of health and family planning. Literacy is a highly powerful facilitator of cultural and societal transformation (Olson, 1977; Goody, 1977; UNESCO, 2005c). Greaney (1996) argues that there is an important link between literacy rates and overall development levels of a country in terms of increase in the Gross National Product (GNP), primary and secondary school enrollment, life expectancy, and newspaper circulation, etc. Psacharopoulos and Woodhall (1985) also claim that the literacy rate is important to enhance human capital that facilitates economic development. Adults who learn to read benefit not just themselves but also the whole community as they become examples for their children and grandchildren and can better serve their community (Selber, 2004). In addition, for Levine (1994), literacy can be an instrument to integrate the poor and the marginal in society or as a way to participatory democracy. Equally important, it is less costly to train literate people since their post-literacy empowerment becomes easier (Ayodele and Adedokun, 2012).

As Scribner and Cole (1981, p. 236) explain, 'literacy is not simply knowing how to read and write a particular script but applying this knowledge for specific purposes in specific contexts of use'. Through its scope for enhancing learner capacities and practices (to access and use information, critically engage with issues and institutions relevant to one's life), and having the confidence and space to make one's voice heard, literacy has proved itself to be a powerful aspect of the development response to poverty, vulnerability and isolation (Betts, 2002). According to UNESCO (2005c, p. 143), literacy education participants, particularly women, can increase their voice in household discussions because of their experience of speaking in the 'public' space of the class. For Rogers and Street (2012, p. 84), literacy can sometimes result in a 'sense of belonging, confidence to play a larger role in the public spheres, a new sense of identity'.

All benefits discussed so far can be associated with what Stromquist (2009, pp. 64-66) calls literacy's empowering potentials, namely improvement of the cognitive, economic, political and psychological perspectives of the learners. The benefits also imply the positive effects of literacy for socio-cultural empowerment of learners. According to Stromquist, the impacts of ALE can be studied indirectly, usually by recording the learners' feelings about self-esteem, self-efficacy and self-confidence that are the foundations for the development of citizenship,

and by observing learners' behavioral change in political and community participation, decision making etc.

Based on these discussions, one can argue that the contribution of ALE to learner empowerment is an essential ingredient of SD. Empowerment in relation to ALE refers to the set of feelings, knowledge, and skills that permits a learner to participate in his/her social environment and affect the political system (Stromquist, 2009, p. 64). According to the UNESCO World Conference on Education for Sustainable Development (2009a), Education for Sustainable Development (ESD) addresses the ideals and principles of sustainability and includes issues, such as poverty reduction, sustainable livelihoods, environmental protection and restoration, natural resource conservation, climate change, gender equality, social cohesion, and protection of indigenous cultures in an integral way. ESD constitutes a comprehensive approach to the quality of education and learning, including ALE. It comprises all sectors of society and all forms of education—formal, non-formal and informal—in a lifelong process (UNESCO World Conference on Education for Sustainable Development, 2009a). This suggests that ALE as non-formal education can be delivered within the ESD framework.

In the same document, it is also claimed that the provision of ALE is a fundamental human right and a means to developing the capabilities of marginalized adults. When addressing the particular circumstances of the learners, it can positively contribute to their lives. Political will and overall capacity of the providers may however affect implementation. The central argument is that a well planned and implemented ALE program can empower adult learners by educating them, changing their behavior, and developing their competencies to act. It can also positively contribute to SD—developing a sustainable environment, economy, society and culture.

With this in mind, the current study explores IFALE's implementation and contributions to empowerment and SD in two regions of Ethiopia, namely Tigray and Amhara regions. The regional comparison may help to identify contextual factors that contribute to achieving these goals. The study systematically investigates the contributions of literacy education to the lives of rural participants of both sexes, identifying literacy benefits at individual, family and societal levels. The study fills the void of knowledge on Ethiopian adult education specifically and also contributes to the general understanding of the function and outcomes of adult education programs which is essential to the field of comparative and international education.

1.5 Research Questions and Organization of the Study

The general purpose of the study is to investigate the implementation of Integrated Functional Adult Literacy Education (IFALE) and its contributions to rural learners' empowerment and SD in Ethiopia through a comparison of Tigray and Amhara regions.

This is done through seeking answers to the following specific research questions:

- 1. How successful is IFALE in terms of implementing the strategy of a relevant curriculum?
- 2. How successful is IFALE in terms of implementing the strategy to build an efficient institutional system?
- 3. What is the contribution of IFALE to political empowerment?
- 4. What is the contribution of IFALE to economic empowerment?
- 5. What is the contribution of IFALE to socio-cultural empowerment?
- 6. What is contribution of IFALE to psychological empowerment?
- 7. How does IFALE support SD in the country?

In order to answer these research questions, the study uses a case study exploratory mixedmethods design.

The study has ten chapters. Following this introductory chapter, Chapter Two provides the background and context of ALE in Ethiopia in a historical perspective. Chapter Three reviews existing ALE research, including empirical studies, and discusses theoretical lenses of relevance to the study. It focuses in particular on ALE design, implementation and impact studies, as well as theoretical discussions of empowerment, SD (focusing on ESD), and literacy models. Chapter Four discusses evidence for use and develops the analytical model for the study based on the previously outlined empirical studies and theoretical discussions.

Chapter Five presents realism as the philosophical foundation for the study followed by the research design and methods of the study. It explains the comparative case study design and study locations, the exploratory sequential mixed methods design employed, and the sampling and instruments of data collection. It also discusses the techniques of data analysis, reliability and validity issues, and ethical concerns of the study.

This is followed by the comparative analysis of the original data collected on IFALE in Chapter Six to Chapter Nine. Chapter Six explores the implementation of IFALE, particularly the strategies of delivering a relevant curriculum and building an efficient institutional system. Chapters Seven and Eight examine the way IFALE contributes to learner empowerment and supports SD. Chapter Nine discusses the key findings of the study, reflects on the major conclusion and presents the implications for advancement of knowledge and for policy-making.

Chapter 2 Adult Literacy Education in Ethiopia

Ethiopia is the oldest sovereign nation in Africa. In 2015, the population of the Federal Democratic Republic of Ethiopia was approximately 99 million (World Bank, 2016a) of whom more than 80% live in rural areas. The country is composed of nine national, regional states as well as two city administration states, Addis Ababa and Dire-Dawa (Figure 2.1). The national, regional states and the two administrative cities are sub-divided into eight hundred weredas and around 15,000 kebeles (5,000 urban and 10,000 rural).

Located on a massive and rugged mountainous plateau in Eastern Africa, Ethiopia covers about 1.12 million square kilometers. It is the tenth largest of the 54 African countries. While being a natural fortress that discouraged many foreign aggressors, the mountainous terrain has also posed difficulties for communication, transportation, and the spread of education.

This chapter presents the development of education during different historical periods with a particular focus on adult education. It is presented in the context of changing governments and of world trends in adult education.



Figure 2.1 Map of Ethiopia *Source*: https://www.crwflags.com/fotw/flags/et(.html) (image copied 27 July, 2018).

2.1 The Introduction of Education into Ethiopia

Ethiopia had its own alphabet in the fourth century. During this period, King Ezana of Axum opened church schools to perpetuate Christianity, marking the beginning of formal education. The church schools achieved their "golden age" of expansion between the 13th and 16th centuries (Hable-Selassie & Tamrat, 1970). The schools helped to create a unique Ethiopian literary tradition. The Kibre Negest (Glory of Kings) script of the 14th century that starts with an account of Queen Sheba and King Solomon and links the Ethiopian kings to the House of Israel (Budge, 2001; Dejene, 2008) is a vital example. According to Levine (1974), this manuscript was the script for the Tigrayan and Amhara societies for many centuries. A small minority—who had studied Ge'ez, an ancient liturgical language used for religious services—could understand the script.

Until the start of the 20th century, education in Ethiopia was restricted to religious institutions. Throughout the country's long history, the Orthodox Church schools have taken the responsibility for producing literates in the Ge'ez language. The schools taught learners who filled the religious system and provided church services. Later, the schools offered religious education not only to the children of the aristocracy, but also to the sons of the tenant farmers and servants of the elite families. Even today, these schools teach religious education to many young learners in rural as well as urban areas of the country in order to prepare them for church positions and services and/or to get them to sustain their Orthodox Christian faith. Since the seventh century, the Koran schools have provided religious education in Arabic to followers of the Muslim religion, but the Orthodox Church organized most of the education.

According to Kebede (2006) and Molla (2008), literacy and basic education were instruments of development and modernity in the 1890s. Emperor Minilik II issued a declaration in 1893/4 stating that every child, male or female, should get an education after the age of six. However, parents did not know that they had to send their children to modern secular education nor were there schools to enroll learners. The fact that the church education system could not satisfy the demands of those who worked in government, business, and external relations resulted in the opening of government funded secular education at the beginning of the 20th century (Teferra & Altbach, 2003).

A major stride in the history of secular education occurred with the opening of the Menelik II School in Addis Ababa in 1908 (Pankhurst 1962; Negash, 1996) and the gradual expansion of

schools in Ankober. The Orthodox Church saw the start of secular education as a challenge to traditional Ethiopian religious values. To resolve the discontent of the Church, Emperor Menelik (1889-1913) agreed with Abuna Matwos—the Patriarch of the Ethiopian Orthodox Church—that the modern Egyptian Coptic teachers, who were introducing modern methods of schooling and teaching young Ethiopian students the Orthodox Christian faith, had to run the modern schools. According to Pankhurst (1998), teaching was for a limited number of urban students and combined foreign languages (including French, English and Arabic), elementary mathematics, and rudimentary science with Amharic and religious subjects. French was the main medium of instruction as it was the principal foreign language of the country.

Each government, from the rule of Emperor Haile-Selassie to the current government, has made its own efforts to expand modern education. The education system of both the imperial and the Derg military systems has been considered rigid, unjust, inefficient, and irrelevant to the life of the people (Bishaw & Lasser, 2012). Nowadays, there are significant efforts to expand general education but its quality is low.

2.2 Adult Literacy Education during Imperial Rule (1930-1974)

The literacy endeavors of the imperial period marked the beginning of adult literacy efforts through state initiatives. There were two main efforts: the earlier initiatives, and the Work-Oriented Adult Literacy Program of UNESCO.

2.2.1 Early Adult Literacy Education Efforts, 1948-1964

Before the end of the Italian invasion in 1941, adult basic literacy initiatives were unrelated to an official policy (Kenea, 2014). The 1944 Memorandum on Educational Policy that stipulates free education for all at all levels of education was an important milestone even if the national capacity was too weak to carry out mass education, and the awareness of people concerning education was low. The Memorandum emphasized the importance of both adult education and adult literacy (Kenea, 2014). Scholars, such as Inquai (1998) and Kebede (2005), believe that adult education started in 1948 after the establishment of the emperor financed institute aimed at supporting the education of adults, the Berhaneh Zare New (which means *your light is today*). According to Bekele (1966) and Sandhaas (2009), this was the time when evening shift classes started in almost all schools of Addis Ababa and in the Addis Ababa University College.

The 1944 memorandum merged formal and non-formal education, including literacy education. Niehoff and Wilder (1974) argue that there was no apparent special attention to literacy education separate from formal education until 1955. In November 1955, Emperor Haile-Selassie made an announcement in an Amharic newspaper (Kenea, 2014; Werqneh, 2011) which read:

We charge every illiterate Ethiopian between the ages of eighteen and fifty to learn in the time left over from his daily task such fundamental education³ as will enable him to know Amharic reading and writing, either at government schools or private, existing in the neighborhood, or by employing a private teacher in his particular village or district (Niehoff & Wilder, 1974, p. 12).

The announcement gave responsibility for coordination to the MoE and demanded that both the clergy and government employees support the process of implementation together with government, private, charity and religious organizations (Kenea, 2014).

Soon after the announcement was official, there was an agreement with UNESCO to launch a center that trained 'fundamental educators'⁴ to be engaged in rural areas with the main responsibility of teaching literacy skills (Niehoff & Wilder, 1974; Tekleyesus, 2004). The center opened in Debre-Berhan in 1958. The program offered training to all family members of the future fundamental educators: men had training in methodology, agriculture, community development principles, health, etc. whereas women and children learned home economics and conventional primary education, respectively. According to Niehoff and Wilder (1974), the program taught 179 fundamental educators with their families until 1960 when it terminated for unknown reasons. Furthermore, fundamental schools that included adult education changed into primary schools for children between three and five years old.

According to Kenea (2014), the announcement showed no strong commitment since most of the responsibility was left with the illiterate adults themselves. It did not appear in the national *Negarit Gazetta*, a newspaper publishing binding legal provisions or proclamations (Kenea, 2014). According to Wagaw (1979), the country did not have the capability to implement it. The announcement shows that the majority of Ethiopians were to learn basic literacy in

³ Fundamental education refers to the preparation of children or adults without opportunity for effective participation in community life through instruction in basic facts and skills, for example literacy, agriculture, homemaking, hygiene, citizenship (Merriam-Webster Online Dictionary, 2006).

⁴ Fundamental educators are people who teach literacy skills.

Amharic despite the range of local languages in Ethiopia. This all had a negative effect on both the continuity and the effectiveness of the literacy programs.

A voluntary association known as Biherawi Yefidel Serawit (Association of the National Army for the Alphabet) was set up in 1954. Its purpose was to expand literacy education to the broader population under the guiding motto 'Let everyone learn' (Sineshaw, 1996; Kebede, 2006). It aimed at enabling illiterate Ethiopians aged 18-50 to read and write in Amharic during their spare time (Mammo, 1982, p. 194). This organization, also known as the National Literacy Campaign Organization (NLCO), whose finances came from donations and international development organizations, was directed by a board of high officials under the patronage of Emperor Haile-Selassie. NLCO relied on volunteer efforts of students and their teachers during the summer vacation, and supplied instructional materials and training programs to its volunteer teachers. NLCO enabled the public to understand the hugeness of the education problem. The program was, however, ineffective because of lack of systematic organization and problems of resource allocation, inadequate training, and lack of commitment of teachers to conduct teaching. Critiques claimed that learners who had not acquired the necessary literacy skills received completion certificates anyway (Mammo, 1982; Niehoff & Wilder, 1974).

In 1967, the Directorate of Adult Education and Literacy (DAEL) was set up under the MoE with the aim of coordinating literacy efforts, collecting statistics, and setting standards. The Directorate prepared instructional materials, instructors' manuals, follow-up readers, and posters besides publishing a monthly newspaper for newly literate people. It also designed a syllabus that covered literacy skills, numeracy, health, civics, and other basic information (Niehoff & Wilder, 1974; Amare, 2006). Based on the contents of the syllabus, the DAEL formulated a standard test to complete the program. The test was open to all learners in any literacy program, be it government or privately run.

The DAEL had a relatively small staff corresponding to its limited duties of coordination and support. The program was carried out by Provincial Literacy Officers (PLO) whose responsibility included organizing programs in their respective province, promoting participation, hiring teachers upon the recommendations of the school principals, conducting training sessions for the teachers, and handling finances.

In 1970-71, there were 901 literacy training centers run by the MoE in the entire country, with 2,140 classes taught by 2,192 teachers, most of whom were volunteers (Niehoff & Wilder,
1974). The literacy teachers received a short training by the PLO or the PLO delegate. They taught four to five times a week for one and a half to two hours per night. They received 40 birr (19.30 United States Dollars/USD)⁵ for teaching 80 hours per month. Learning in the literacy program was four months for learners who spoke Amharic as their mother tongue and six months for learners of other mother tongues. However, both types of learners used the same material that was prepared in Amharic. This means that the majority of the learners could not use their mother tongue.

According to Niehoff and Wilder (1974), this literacy effort was significant since it produced a higher number of literates than formal schools. Table 2.1 shows the enrollment in the literacy program during 1964-71. As can be seen enrollment fluctuated, being higher in 1964 than during 1965-68, but generally increasing during 1965-70. The highest enrollment was in 1970. There were always more male than female participants. After successful completion, the graduates were encouraged to continue education in formal education. In 1969, 74,813 (about 33%) of the 228,877 literacy graduates passed the MoE test of whom 44,076 (i.e. more than half) came from three of the fourteen provinces: Tigray (26,564); Gojam (10,353); and Wello (7,106).

Year	Males	Females	Total
1964	158,778	35,172	193,950
1965	107,660	14,747	122,407
1966	126,057	19,817	145,874
1967	129,527	23,630	153,157
1968	130,753	42,703	173,456
1969	183,535	45,332	228,867
1970	238,000	59,079	297,079
1971	119,292	38,071	157,363
Total	1,193,602	278,551	1,472,153

 Table 2.1: Enrollment in literacy programs, 1964-1971

Source: Directorate of Adult Education and Literacy, MoE as cited in Niehoff and Wilder, 1974: 171.

2.2.2 The Work-Oriented Adult Literacy Program (WOALP), 1968-1973

After agreements with the Imperial Government of Emperor Haile-Selassie, UNESCO introduced WOALP in early 1968, and Ethiopia joined the countries that were implementing

⁵ Haile (2008) citing the World Development Report and National Bank of Ethiopia (2005) states that between 1973 and 1991, 1 dollar was equivalent to 2.07 birr. This rate has been used in the study.

the Experimental World Literacy Program (EWLP), which included WOALP, financed by the United Nations Development Program (UNDP) (Sjöström & Sjöström, 1982).

WOALP planned to teach 78,000 learners in five years using new materials and methods (Niehoff & Wilder, 1974). The UNESCO experts in Ethiopia used the 'Action Research' approach to test and then decide on the best method. WOALP had two types of teachers: literacy teachers trained in teaching the Amharic language for a total of fifteen days by the central staff of UNESCO; and specialist teachers trained for two weeks in the work-oriented components. WOALP classes took place in schools, homes, factories and, in some cases, in classes constructed by peasants.

While the earlier adult literacy efforts primarily emphasized mastering reading, writing and arithmetic (the three R's), WOALP introduced the idea of functional literacy (Kenea, 2014). More importance was given to functional contents that related to the day-to-day activities of the learners. By June 1973, WOALP had almost 25,000 learners taught by 783 instructors in 885 classes. According to Sjöström and Sjöström (1982), the training of instructors, preparation of vocational-technical materials for adults, and relating functional literacy to agriculture or home economic demonstration programs and pre-vocational training were areas in which experience was developed. WOALP raised awareness about the importance of education not only within the government, but also among the public (Kebede, 2006).

The weakness of the program was that it only reached the provincial education offices despite the assignment of a Director General who coordinated the program at the ministry level and that the MoE had neither the resources nor the authority to coordinate other line ministries (sectors) that were to collaborate (Sandhaas, 2009; Kenea, 2014). According to Niehoff and Wilder (1974), the program was too expensive compared to other literacy programs⁶.

2.2.3 Summary

The start of adult literacy provision raised the public's awareness about the importance of adult education. However, the shortcomings of the literacy education endeavors of the imperial period were many. In Kenea's (2014) view, access to adult literacy education was restricted to the ones who voluntarily joined the program, which means the aristocratic class and males. During the imperial period, adult literacy policies never considered provision in the learners' mother tongue because of the use of Amharic to maintain a unified country

⁶ The US Special Funds was to contribute USD 1,357,957. The Imperial Government was to allocate USD 149,200 in cash and USD 2,058,800 in kind (Niehoff & Wilder, 1974: 178).

(Kenea, 2014), and the government showed no strong political will to carry out literacy initiatives (Amare, 2006).

This, as a whole, resulted in low achievements in both the basic literacy and the functionally oriented programs. According to the World Bank, in 1973 the literacy rate of the population aged 15 and above was 7%, one of the lowest in the world (Sjöström & Sjöström, 1982, p. 35). During the imperial rule, AE played no significant role in empowering people to master skills for participating in activities and issues that affected them.

2.3 Adult Literacy Education during the Derg Military Rule (1974-1991)

After the overthrow of Emperor Haile-Selassie, the Derg Military Regime launched the National Democratic Revolution (NDR) in 1974. According to Bhola (1987), the NDR promised to destroy the soul and structure of the old feudo-bourgeois system and transform culture and technology by eradicating ignorance, ill health, and exploitation. The mass of the population was mobilized through a campaign and adult literacy was essential to realize the goals of the new revolution (Bhola, 1987). There were two kinds of adult literacy campaigns: the Development through Cooperation, Knowledge and Work (or Zemecha), 1974-1976; and the National Literacy Campaign, 1979-1991.

2.3.1 The Development through Cooperation, Knowledge and Work Campaign (Zemecha), 1974-1976

In the Development through Cooperation, Knowledge and Work Campaign, literacy was one of nine programs in the entire country with the greatest focus on rural, including the remote and neglected, areas (Kebede, 2006). The government deployed around 60,000 young campaigners, mostly teachers and students, all over the countryside to perform various development activities, including literacy education, for a two-year term service (Mammo, 1982; Amare, 2006). According to Mammo (1982):

The campaigners formed over 21,000 peasant associations with a total number of 5 million members. About a million people participated in the literacy program; over two hundred clinics were built in the rural areas with the assistance of the Ministry of Health technicians. During the campaign, 200,000 persons were vaccinated against tuberculosis and 300,000 against smallpox. During the first round of the campaign, nearly four and a half million functional literacy primers were produced in a total of five national languages (p. 194).

The campaign developed scripts, and designed and used literacy texts in four languages that were unwritten at the time. It continued to do translations and text preparations in other Ethiopian languages after the campaign stopped (Kebede, 2006).

Despite its success, Amare (2006) claims that the lack of time to organize the campaign carefully and the organizers' insufficient experience created challenges for the campaigners in terms of understanding the objectives of the campaign, the traditions of the people in the literacy education sites, and the expected pace of change. The campaign ended because of internal problems and the border war with Somalia (Mammo, 1982).

2.3.2 The National Literacy Campaign (NLC), 1979-1991

The launch of the NLC took place on 8 June 1979. It was 'a multi-purpose venture aimed at raising the level of material and cultural life of the people' (Amare, 2006, p. 104). The Campaign had a number of objectives aiming at transforming the lives of the mass of the population. These included to emancipate the masses from illiteracy without hampering their day-to-day activities and in the languages they best understood; to strengthen and upgrade the skills gained through literacy and lay down a strong foundation for further education; and to encourage and promote the development of political consciousness, economic prosperity and social maturity (MoE, 1986, p. 3).

The NLC was to teach adults both literacy and functional skills to increase their productivity from the fields (rural learners) and in factories (factory-workers). It originated in socialist thinking. Of the total population (estimated in 1974 at 26.6 million), the NLC targeted 18.5 million illiterates aged 15-45 years⁷.

The objective of the NLC was to eradicate illiteracy from the urban areas by 1982 and from the rural areas by 1987. This did not materialize partly because the total population was higher than first predicted and because of the severe drought and famine in the early 1980s, and the ongoing civil war. The target for the eradication of illiteracy was postponed until 1984 for urban areas and 1992 for rural areas (Bhola, 1987)⁸.

The NLC Teaching and Learning Process

According to Werqneh (2011), the core contents of the campaign were the 3Rs (reading, writing, and arithmetic). There were three types of classes (beginners, remedial and post-

⁷ According to some sources, it also targeted learners aged 8-45 (Werqneh, 2011). Bhola (1987) states that the official age for joining NLC was 8 to 60 years but in reality, it ranged from 3 years old to the very old.

⁸ Werqneh (2011) claims that the target date was changed from 1986 to 1994.

literacy) with different time allotment and contents. The beginners' classes had six hours a week and 3 hours a day (Bhola, 1987, p. 13). Adults learned how to read and write in 312 hours of instruction, of which 240 hours were for basic skills and 72 hours for practice on functional readers (Bhola, 1987). It included the alphabet, forming words, Arabic numerals and the four mathematical operations (Werqneh, 2011). The literacy classes also included the teaching of the socialist ideology (Amare, 2006). Teaching was by rote: learners repeated the alphabet chart which, according to Bhola (1987), was what learners preferred. The program relied on volunteer teachers and recruits (e.g., 12th grade students, school teachers, civil servants, and retirees).

The remedial classes had 288 hours for learners who failed to gain a certificate at the beginners' class. The post-literacy classes were follow-up programs in the literacy center for learners who had already earned a certificate (Bhola, 1987). The post-literacy reading, writing and computation materials had contents on environmental education, basic agriculture, health, home science, appropriate technology, cooperatives and socialist education, and political/ideological education (Werqneh, 2011). There were 120 hours of instruction in the post-literacy classes based on post-literacy readers. This served as a basis for continuing education (Bhola, 1987).

According to Bhola (1987), the literacy classes enabled learners to do some literacy activities, such as reading newspapers, magazines and posters, calculating land areas and crop yields, making a budget, and communicating in writing with friends, family, and local and other government agencies.

Training, Support, Institution Building, and Networking

According to many government sources, it was compulsory for both the management and the teaching staff to join NLC (Werqneh, 2011). Literacy teachers were trained for 2-3 days (instead of the originally planned 7-9 days) in how to use the materials and radio programs, how to do peer teaching and how to teach agricultural skills, health, hygiene, family and guidance, and political education.

NLC started in urban centers and spread to remote rural centers. There were 1.5 million people volunteers and more than 22 million reading booklets and 9 million primers were distributed to beginners and post-literacy learners, respectively (Abate, 1993). There were follow up courses equivalent to grade four enabling many learners to enroll in formal education. The total financing for NLC was 517.88 million Ethiopian birr (around USD 209

million) up to the 19th round in 1989 provided by the government, local communities, local and international organizations, and Ethiopian communities living abroad (Kebede, 2006; Werqneh, 2011, p. 80).

In order to strengthen institution building and networking, a separate Department of Adult Education (DAE) was set up in the MoE with a highly articulated planning and implementation system for NLC (Bhola, 1987; Amare, 2006; Kebede, 2006). DAE became the secretariat of the National Literacy Campaign Coordinating Committee (NLCCC)—a policy-making body with 28 representatives from government agencies, mass organizations, and professional and religious institutions. The NLCCC committee met once or twice a year with the Minister of Education as the Chair, while the executive committee of the NLCCC met each fortnight to handle the recurrent problems and decisions.

The National Literacy Coordinating Committee (NLCC) had four sub-committees in the MoE: the Educational Materials, Procurement and Distribution Committee; the Recruitment, Training and Placement Committee; the Propaganda and Aid Coordinating Committee; and the Data Collection, Supervision and Certification Committee. Sub-committees were also set up in the fourteen major regions, in the capital of Addis Ababa, in 106 awrajas (sub-regions) and in 588 weredas. At the grassroots level, 20,000 Peasant Associations (PA) and Urban Dwellers' Associations (UD) supported the NLC by employing literacy campaign officers that looked after the literacy centers, reading rooms and listening forums and supervised literacy instructors in literacy centers and sub-centers located in small and remote communities. There were in all nine layers from the top (MoE) to the bottom (the center) required to work under what was known as bureaucratic centralism (Kebede, 2006; Werqneh, 2011).

The NLC was implemented in collaboration with organizations such as Awraja Pedagogical Centers (APCs), Basic Technology Centers (BTCs), Community Skill Training Centers (CSTCs) and Basic Development Education Centers (BDECs) (Bohla, 1987). The campaign also worked in partnership with the educational radio, distance (or correspondence) education and night schools.

Benefits, Problems and Challenges

According to Kebede (2006) and UNESCO (2005c), the NLC taught 22 million people (of whom 52% were females). Over 20 million adults (51% females) passed the beginners' literacy test (UNESCO, 2005c) and received certificates for basic skills in reading and writing

(Negash, 1996). This meant that the literacy rate in the country increased from less than 10% during the Emperor Haile-Selassie regime to over 60% in 1990/91 (Abate, 1993).

Both the use of the mother tongue for literacy education and the opportunity to continue training in formal schools were significant benefits. The NLC employed fifteen different languages for both teaching and material preparation in the literacy and post-literacy training, meaning that 93% of the learners could use their mother tongue (Kebede, 2006; Abate, 1993; Molla, 2008). This led to a cultural rebirth for many nationalities whose languages had been unwritten. Even more, the campaign provided thousands of people with the opportunity to join formal education (Kebede, 2006; Molla, 2008). Successful completion of literacy education in the NLC was equivalent to grade two in formal primary school and provided access to grade three in formal education (Amare, 2006).

The increased literacy levels seem to have led to learners' higher awareness of their political environment and to skills that were functional for them both as citizens and as producers (Bhola, 1987). Amare (2006, p. 109) argues that literate parents understood the importance of education and contributed actively, for example, to setting up Community Reading Rooms and assisting teachers in the CSTCs. The campaign has also been viewed as a tool of domination since the promotion of the socialist ideology among the learners led to lack of questioning rather than developing critical, inquisitive and independent thinking of the learners (Woldemeskel 2005; Amare, 2006; Kenea, 2014).

The NLC gradually suffered from 'campaign burn-out' among its organizers as the campaign extended over a period of ten years. The forced use of teaching staff and shortage of instructional materials in many places led to relapse into illiteracy among many neo-literates, while graduates who enrolled in formal education faced challenges because of the use of Amharic instead of their mother tongue as the medium of instruction. Added to this were the severe famine of 1985 and the continually expanding civil war after mid-1988.

2.3.3 Summary

Despite positive features, the benefits from the Derg's adult literacy programs were temporary and did not make most learners participants in the economic, social and political domains. The Development through Cooperation, Knowledge and Work Campaign (1974-1976) introduced the use of other mother tongues than Amharic as the medium of instruction and for texts used in adult literacy education. It also mobilized tens of thousands campaigners who amongst others provided educational opportunities for much wider groups of the rural population.

However, the campaign encountered two basic problems: poor preparation; and lack of experience among campaigners. The NLC, which depended on voluntary unpaid teaching, prepared primers in fifteen languages and liberated tens of millions from illiteracy. This was, however, done in strict accordance with socialist ideology which deprived learners of their own free thinking. The campaign suffered from an inadequate government budget and eventually from 'campaign fatigue'. Above all, insufficient efforts were made to prevent neo-literates from relapsing into illiteracy.

2.4 Adult Literacy Education during the Ethiopian Peoples' Revolutionary Democratic Front (EPRDF), 1991-now

In May 1991, EPRDF ousted the Derg regime and a Transitional Government including all the opposition parties led the country from July 1991 to May 1995. The new government came into power at a time when the world witnessed the end of the cold war, the demand particularly from multilateral organizations (the World Bank and IMF) for structural adjustment, and the publication of the World Declaration on EFA. The EFA policy reinforced the focus globally on primary education which led to massive donor support for primary education in the developing world, including in Ethiopia (Lind, 2008).

In April 1994, the Transitional Government of Ethiopia issued a new Education and Training Policy (ETP) that led to a reform of the education system. The ETP and especially the ESDPs I-II (1997-2005) addressed problems of access, equity, quality and relevance in formal education, with most attention given to primary education. Adult education and adult literacy were not addressed. Neither did they form part of the Agricultural-Development-Led-Industrialization (ADLI) strategy that was first articulated in 1993. The lack of government emphasis meant that NGOs were virtually the sole provider of adult literacy programs which were conducted outside of the general policy context.

2.4.1 Adult and Non-formal Education (NFE) in the Education Sector Development Plan (ESDP I) and the Education Sector Development Plan II (ESDP II), 1997/8-2004/5

The ETP states that NFE is to be taught and incorporated with basic education at all levels of formal education. Adult and NFE was a three year cycle of basic education (equivalent to formal basic education grades 1-4) taught to out-of-school children aged 7-14, and to youth and adults aged 15 and above. In 2000/01, the former group included 320,581 children, and the latter group had 1,049,061 individuals. During the ESDP I period (1997/1998-2001/2002), the dropout rate for youth and adults was 16% (MoE, 2002, p. 23).

In ESDP II, the MoE set targets for the Adult and NFE program that were to be achieved through participation and contribution of communities and NGOs (MoE, 2002). They concerned the setting up of new skills training and basic literacy centers, reaching more out-of-school children and youth and adults, training of adult trainers in skills, and ensuring that the first cycle of formal primary education (1-4 grades) and alternative non-formal education programs were linked.

Table 2.2 shows enrollment in adult and non-formal education in 2003/4 across most of the regions in Ethiopia. Oromia, Amhara, SNNP and Tigray were particularly successful. As in the earlier period, there were fewer female than male learners.

	7-14 years			15 years and above			
Regions	Females	Males	Total	Females	Males	Total	Grand total by region
Tigray	14,316	17,528	31,844	5,886	10,549	16,435	48,279
Amhara							212,337 (F=99,712; M=112,625)
Oromia	207,689	268,508	476,197	108,350	243,862	352,212	828,409
Benshangul-Gumuz	1,626	2,650	4,276	1,232	3,650	4,882	9,158
SNNP	37,875	44,893	82,768	35,648	39,300	74,948	157,716
Gambella	855	926	1,781	2,476	2,589	5,065	6,846
Harari				672	2,825	3,497	3,497
Addis Ababa	11,612	6,043	17,655	17,972	9,136	27,108	44,763
Grand total	273,973	340,548	614,521	172,236	311,911	484,147	
Source: Mat 2005 p. 7: Kabada 2006 p. 16							

Table 2.2: Enrollment in adult and non-formal education, 2003/4

Source: MoE 2005, p. 7; Kebede 2006, p. 16

Despite some achievements, for example concerning the preparation of the curriculum and improved methods of teaching adults and youth, most problems continued to exist.

2.4.2 Adult Literacy in ESDP III (2005/6-2009/10) and the National Adult Education Strategy (NAES), 2008

Unlike the first two ESDPs, ESDP III included the Functional Adult Literacy (FAL) program for youth and adults aged 15 and above despite the fact that there was no adult education strategy until 2008. The program aimed at enhancing the participation of communities in national development and poverty reduction strategies in addition to making adults more productive and self-reliant. The program was to address the problem of gender disparity, reach 5.2 million adults, and train 143,500 adults in different skills in the existing 287 CSTCs through the Basic Skills Training Program (MoE, 2005). The MoE (2005) required support from all stakeholders (partner ministries, NGOs, local governments, communities).

In 2008, the MoE together with its partners prepared the National Adult Education Strategy (NAES) whose focus is to impart knowledge and skills to the adult population and to facilitate conditions for the provision of the Functional Adult Literacy Programs (FALP) (MoE, 2008). It also calls for an expansion of quality, relevant and equitable adult education programs accessible to all. It argues that each human being—irrespective of color, religious background and economic status—has the right to basic education. The strategy seeks to develop the knowledge, competencies and skills, attitudes and behaviors that would allow adult learners to contribute to all aspects of Ethiopia's development. Adult education, more specifically Integrated Functional Adult Literacy (IFAL) should enable adult learners to acquire knowledge and skills in areas such as agriculture, health, civics and culture to achieve accelerated and sustained development (MoE, 2008, p. 2). The national adult education task force also developed the National IFALE Curriculum Framework, the FAL Implementation Guidelines, the FAL Facilitators Training Manual, and the FAL Benchmark manuals.

An Adult and Non-formal Education Unit was set up under the MoE. A Memorandum of Understanding was signed between the MoE and five other collaborating ministries ensuring the creation of an Adult Education Board and a technical committee that were to also be set up throughout the decentralized structure down to the wereda level. Each region assigned an adult education focal person responsible for follow up of the adult literacy education programs.

However, the new targets were not met and most potential learners had no access to ALE that could help them gain the attitudes, knowledge and skills for a better life. The NAES did lead to new plans and programs, in particular Learning for Life: A Master Plan for Adult Education in Ethiopia (2010-2020). NAES was also crucial for the inclusion of Integrated Functional Adult Education (IFALE) in ESDP IV and ESDP V.

2.4.3 Links between IFALE, Learning for Life: A Master Plan for Adult Education in Ethiopia (2010-2020) and ESDP IV (2010/11-2014/15)

The Master Plan for Adult Education in Ethiopia (2010-2020) was developed particularly with support from Institut für Internationale Zusammenarbeit des Deutschen Volkshochschul-Verbandes (Institute for International Cooperation of the German Adult Education Association, IZZ/DVV-international) while other stakeholders played minor roles (MoE, 2010a). According to MoE (2010b), the plan is a general framework that assesses the main challenges and opportunities in adult education in Ethiopia and presents a shared strategic direction for adult education. The plan enables all stakeholders to deliver good quality adult education using existing resources, mobilizing new ones, and strengthening planning, governance and financial management practices (MoE, 2010a). The document has a clear vision, purpose, founding principles, and pillars for adult education.

The vision is 'to create a literate, well-informed, knowledgeable and skilled population of youth and adults who realize their own potential and maximize their contribution to society, Ethiopia's economy and future generations' (MoE, 2010a, p. 2). The purpose is to provide the necessary structures, programs and tools in order for every youth and adult in Ethiopia to have access to quality adult education services whose special focus is integrated functional education (MoE, 2010a).

The plan identifies eight founding principles and six major pillars of adult education. The founding principles consider adult education a part of strategic plans and policy directions. It is part of LLL, and should be based on a quality framework, be learner-centred and address the needs of the learner. The core focus is on IFALE (MoE, 2010a). The pillars address issues of content of adult education programs and approaches, governance, training, assessment, monitoring, evaluation and reporting, research and knowledge transfer, and costing and financing.

The plan also discusses the nature, delivery and relevance of Ethiopian adult education with a special focus on IFALE. Adult literacy should build on indigenous knowledge and link writing, reading and numeracy skills to livelihoods and skills training in areas, such as

agriculture (including off-farm activities), health, civics, and cultural education. This requires collaboration between the government and NGOs. The literacy skills should be meaningful to the learners since relevance and utility are the keys to accelerate skills acquisition and help learners expand their knowledge base through other learning opportunities. Livelihood and life skills are essential in a successful Integrated Functional Adult Literacy Program (IFALP).

ESDP IV (2010/11-2014/15) (MoE, 2010b) addressed four expected outcomes and nine key outcome targets for adult education with a special focus on IFALP. These were to improve democratic and good governance culture through expansion of adult education; to make economic development more sustainable through quality and relevant adult and non-formal education; to create an efficient institutional system for adult and non-formal education; and to attain significant improvement in adult literacy. The outcome targets included, amongst others, to reach 95% of the 36.4 million (later adjusted to 19.4 million) illiterate adults and reduce regional and gender gaps, to undertake adult literacy training aimed at linking literacy to livelihoods, to increase the number of trained facilitators and instructors, and to identify and share best practices and materials. The Master Plan for Integrated Adult and NFE should be utilized and implemented by all stakeholders and the National Adult Education Board be made operational.

The government budgeted about 12 billion Birr (approximately 571 million USD) constituting 9% of the budget allocated for the education sector and 90% of the total cost for IFALP. This is equivalent to 20 USD per participant per year (MoE, 2010b). The amount is considerably higher than the budget for ESDP III (288 million birr, 0.5% of the total budget for the education sector). All other levels of education had a decreased proportion of the total budget compared to the ESDP III. IFALE is therefore essential to the current government ambition for development of the country.

2.4.4 IFALE for Sustainable Development

According to NAES (2008):

To enable the adult to competently participate in the country's development efforts it is vital to design an adult education program, which complies with development strategies and packages (agricultural and rural development, health, women and youth development packages) and serves as a tool for accelerated and sustained utilization of these development activities (p. 4).

This suggests that exclusion of adults from educational opportunities hampers development. According to MoE (2008, p. 5), adults should contribute to the eradication of poverty and achievement of SD. Adult education should help develop the productive force of society, build a democratic and good governance culture and bring about change in social life (MoE, 2008, pp. 11-12).

The IFALE national curriculum framework document (MoE, 2010c) is consistent with the vision of the Master Plan for Adult Education (2010-2020) and the wider vision for society. It promotes values, such as love for hard work, respect for humanity, concern for individual and group rights, concern for common interest despite diversity, and capacities and skills for environmental protection (MoE, 2010c). These values promote learning for sustainable life as main features of SD.

According to the MoE (2010c), the life skills contents address agriculture, health, income generation, critical thinking, civics and ethics, environmental care and protection, gender and social issues. The teaching of life skills should be done holistically rather than as a separate subject, linking basic literacy teaching to life related issues (MoE, 2010c). For example, when discussing *traditional practices* as a life-related issue, learners can at the same time learn the letters in the phrase and calculate the number of letters in the words 'traditional' and 'practice'. This would motivate adults to participate in the program and simultaneously develop their knowledge and skills in order to address poverty and bring about SD.

The duration of the IFALE program is two years—equivalent to three years in formal education—and sessions should be suited to the learners' occupations (MoE, 2010c). Teaching and learning should be in small sessions, currently three sessions of two hours each per week (i.e. 6 hours per week). Learners can join primary education or technical and skills training after completion of IFALE (MoE, 2010c).

The major tasks for IFALE at the national level, according to MoE (2012), are to allocate adequate finance, create an institutional system with qualified personnel, prepare and distribute primers, and recruit, train and assign facilitators to the centers. The teaching and learning should take place in collaboration with relevant government institutions, including from agriculture and rural development, health, Farmer Training Centers (FTC) and with NGOs. Different MoE documents underline the importance attached to adult education and, particularly to IFALE, and are the foundation on which the current study is undertaken.

2.5 Key Issues in the Literacy Programs of the Three Governments

The ALE efforts of the different governments were done based on different understandings of program effectiveness. The understandings partly followed international trends (Table 2.3). The early ALE programs of the Emperor Haile-Selassie regime focused on the eradication of illiteracy or adult learners' mastery of reading and writing in Amharic, the official national language. In these programs, the traditional approach to literacy was the dominant philosophy. In Rogers' (1999) view, this approach neither motivates learners to participate nor helps them to apply what they learn in their everyday life. Besides, the programs suffered from the lack of a strong institutional set up, lack of resources, inadequate training of teachers, lack of teacher commitment, and inability to enroll the poor and women. As the wider literature shows, the majority of the adult literacy learners during the emperor's time had difficulties because their mother tongues were not the medium of instruction.

The work-oriented functional literacy approach also used Amharic as the medium of instruction for all adult learners in the country. It was implemented during 1965-1973 by EWLP consistent with the human capital theory to 'find ways of turning literacy into an effective instrument for social and economic development' (Lind, 2008, p. 49). As a UNESCO pilot project, the program focused on functional contents linked to specific projects in industry or agriculture. There was an endeavor to empower learners to contribute to economic and social development. However, the MoE faced difficulties due to lack of authority and resources to work with collaborating ministries, lack of a strong institutional system, and shortage of experts which affected the effectiveness and empowering capacity of the program.

During the Derg regime (1974-1991), the ALE perspective in both the Development through Cooperation, Knowledge and Work Campaign and NLC linked functional literacy to mass campaigns. They aimed to free people from illiteracy with the help of their mother-tongue, strengthen and upgrade the skills they acquired from literacy for use in their everyday life, enhance their socialist political consciousness, and enable many to pursue further education. In reality, the liberation and social justice agenda was excluded from ALE (Kenea, 2014) even if, internationally, the period 1974-1980 was the time when functional literacy extended beyond economic development to include political, social and cultural dimensions and was conceived as a means of liberation (Lind, 2008). Moreover, in the 1980s at the global level,

attention was given to mass campaigns to eradicate illiteracy on which Paulo Freire's liberating education and pedagogy had a strong influence.

Rulers		a ALE trena World trend of ALE*	Conditions of ALE programs in	Challenges of
Nuivi 3			Ethiopia	implementation in Ethiopia
Emperor Haile-	1945- 1964	Reading and writing in the mother tongue	Reading and wrting in the national official language; Establishment of Berhaneh Zare New Institute in 1948; Formation of Yefidel Serawit (literally Army of Alphabets) in 1954.	Absence of mother tongue education.
Selassie (1930- 1974)	1965- 1973	Work-oriented functional literacy	Public notice on fundamental education/ adult literacy in 1955; Work-oriented functional literacy introduced in 1968; Reading and wrting in the national official language.	Absence of mother tongue education; Lack of experts and data base.
Derg Military	1974- 1980	Literacy as a means of liberation	Development through Cooperation, knowledge and Work campaign, 1974- 1976; National Literacy Campaign (NLC), 1979- 1991; Mass campaign to eradicate literacy; Literacy texts in the mother tongue of the learners.	Exclusion of liberation agenda; No legal provision for the longer campaign, just directives.
Rule (1974-1991)	1980s	Mass campaigns for the eradication of illiteracy and social justice; Governments expected to take a lead	NLC operating; Derg Regime in charge of mass campaign to eradicate literacy through mother tongue education.	Exclusion of Social Justice agenda; Impostion to learn and teach in adult literacy program; Campaign fatigue; Post-literacy and the integration of literacy with development not given due attention; Civil war associated problems especially in northern Ethiopia; Relapse into illiteracy.
	1990s	Vagueness and NGO- ization of adult adult literacy programs despite the defined basic learning needs in Jomtien	State of neglect by government; ALE left to NGOs	Absence of ALE policy discourse; No strong efforts to deliver literacy education to the millions of rural adults.
EPRDF	2001- 2009	Contradictions and inconsistencies: lifelong learning, literacies, and literate societies (Dakar to 2008)	State of neglect by government; Government's focus on basic education for school children; ALE the business of NGOs; NAES formulated in 2008 focussing on functional literacy.	No strong alternative to adult literacy in the policy discourses, except NAES focus on functional adult literacy.
Government (1991- present)	2010 onwards	Literacy for all (UNESCO, 2009c); working towards 'literate families', 'literatecommunities' and 'literatesocieties' made up of independent, confident and effective lifelong learners (UNESCO, 2017)	Government engagement in IFALE; Master Plan for Adult Education in Ethiopia (2010-2020), National IFALE Curriculum Framework and other guidelines written; Regions started to publish teaching materials in the mother tonge of learners; Centers expanded and millions of adults enrolled; Empowerment of learners and support to sustained development efforts emphasized.	No adult education policy apart from the strategy.

Table 2.3: Conditions and challenges for ALE under different governments in Ethiopia compared to the world ALE trend

Source: Compiled from Inquai (1998); Kenea (2014); Lind (2008); Wirtu (2005); Kebede (2005); Woldemeskel (2005); UNESCO (2005c)

*Based on Lind 2008; UNESCO 2009c; UNESCO (2017)

The ALE of the NLC focused on mastery of the basic skills of literacy and numeracy, as well as a socialist orientation in which political empowerment elements appear to be very low and limited to participation in the government prescribed socialist system whereas, for example, freedom of speech and thought were unthinkable. Both teachers and learners had to participate in the NLC. This led temporarily to freeing millions of people from illiteracy, permitting them to master the basic skills of literacy and numeracy.

The EPRDF government, while giving utmost effort to the expansion of formal education, demolished the previous ALE institutional system, neglected ALE in the ETP and left the sub-sector as an NGO business until it promulgated a new national adult education strategy in 2008. According to Lind (2008), the international trend of ALE was one of vagueness and NGOization since governments ignored the provision of adult literacy even though the EFA vision embraced the right to education for all ages. Thus, until 2008, the EPRDF government had no strategy-oriented ALE program that aimed at empowering the majority of poor and marginalized learners in the country.

Internationally, UNESCO (2009) claimed the necessity of 'literacy for all'. Utmost efforts towards achieving literate families, communities and societies in order to produce independent, confident and effective lifelong learners was advanced to the top of the agenda (UNESCO, 2017) for international development post-2015. In the case of Ethiopia, IFALE is to achieve just this. It is to improve the low literacy rate in Ethiopia in accordance with the functional literacy model. IFALE is to educate millions of mostly rural-based illiterate people, empowering the learners in terms of their knowledge, competencies and skills, and attitudes and behaviors. It is to support the overall development efforts of Ethiopia to become a middle-income country by 2025.

An analysis of the implementation process and of learner empowerment in support of SD is imperative to judge whether government plans translate into reality. This is the purpose of this dissertation. As a basis for the study, a review of the wider literature on the implementation of other literacy programs and of factors that have promoted or impeded their implementation is presented. This review also includes a discussion of the key concepts of empowerment and SD.

Chapter 3 Review of Literacy Studies and of the Concepts of Empowerment and Sustainable Development

This chapter reviews design, implementation, and impact assessment studies of adult literacy education. It also discusses literacy models, as well as empowerment and SD concepts especially within the context of adult literacy education. The purpose is to understand the conceptualization and functioning of adult literacy in various contexts. This serves as the basis to develop a contextualized analytical model for IFALE's implementation and contribution to empowerment and SD in Ethiopia.

3.1 Empirical Studies: Design, Implementation, and Impact Assessment Studies of Adult Literacy Education

The review covers three sets of empirical studies that are presented below, namely design, implementation and impact assessment studies.

3.1.1 Design of Adult Literacy Education

ALE programs need to be carefully designed and executed if they are to generate benefits. Rabinowitz (2012) argues:

Adult literacy encompasses more than just reading and writing, and it's important to understand what the real literacy needs of your particular community are before you start planning a program. Once you've assessed the demand as well as your community's needs and assets, you can start planning a program. Consider what services you need to provide, and to whom; what entity or kind of entity will run the program; what the program's philosophical and educational assumptions are; and where funding might come from. When all of that is in place, it's time to actually establish the program in the community (Rabinowitz, 2012, p. 9).

As indicated by Rabinowitz, the design of an adult literacy program should include an assessment of the real needs of learners, the kind of service that is to be provided, the theoretical foundation of the program, and the source of funding. Program designs should consider not only the program demands of the government, but also the needs, aspirations and goals of its potential participants.

3.1.2 Implementation of Adult Literacy Education

In addition to the design, the implementation process is vital for the effectiveness of literacy programs. A review of available evaluation studies of adult literacy programs in developing countries concludes that poor program design and poor implementation are causes of inefficiency (Comings, Smith & Shrestha, 1995). Where programs are well designed and implemented, drop-out rates are between 30% and 50%, equivalent to drop-out rates in the first three years of primary school, and adult learners achieve a skill level equivalent to primary school grade 3 or higher (Comings et al., 1995). However, as Cameron and Cameron (2005, p. 2) claim, dropout rates between 25% and 30% from literacy programs are acceptable globally.

Comings et al. (1995) identified two problems regarding implementation failure. First, programs have suffered from time constraints and inadequate resources, resulting in high dropout rates and low skill acquisition. Second, the measures of skill retention have been inaccurate. These findings imply that the success of adult literacy can, amongst others, depend on the allocation of sufficient resources and time to design, try out (field-test) and implement the program.

Abadzi (2003a, p. v) reviewed adult literacy implementation for the Operations Evaluation Department (OED)—an independent unit within the World Bank. The review included countries financed by the World Bank during 1960-2000 and found failures related to poor access, completion and sustainable skills as well as shortage of resources. Government and donor interest in adult literacy activities reached the highest level during 1975-79.

As Abadzi (2003a) reports, in the 1960s and 1980s, several countries implemented stateadministered, top-down literacy campaigns with brief courses that lacked follow-up and in which only a small proportion of the eligible people participated. Only half of the learners were able to finish; of those who did, about half passed the literacy tests and of those passing the tests, around half relapsed into illiteracy. In her view, what generally characterized many of the campaigns in the 1970s was an efficiency rate of 12.5% with few learners gaining stable literacy skills.

The World Bank incorporated literacy components in 45 of the 304 education projects financed between 1963 and 1985 in countries such as Yemen, Somalia, Afghanistan, Ethiopia, and Malawi (Abadzi, 2003a). However, lending for literacy was only about 1%-3% of the education lending, and World Bank-financed components for any type of non-formal

education often accounted for less than 10% of the project costs. It appears that there was almost no support for reading materials, library services, or audiovisual aids and no attempts to strengthen implementation capacity. These figures were too low to fulfill the intended objectives.

According to Abadzi (2003a), seven stand-alone adult literacy projects of the 1990s (in Indonesia, Ghana, Bangladesh, Senegal) focusing on management depended on NGOs for implementation and instructional expertise and on the World Bank for funding. Generally speaking, the projects were implemented as planned. About 10.8 million people of both sexes were trained, showing the capacity of the projects to attract many learners. The sustainability was unclear, however, since the learners' capacity to read and maintain their literacy was doubtful.

Findings from the three World Bank sector documents published during 2000-2003 identify the major outcomes of adult literacy programs (financed by governments and other donors). Firstly, student attendance and performance data were scarce. The 32 literacy programs that provided data had a median completion rate of 78%, a median attendance (five programs only) of 62%, and a pass rate for the final test of 56%. Secondly, there was no information on participants' relapse into illiteracy. A few neo-literates could read some years after the instruction, such as in Uganda, but of all the literacy program graduates sampled from other programs, for example in Ajmer-India, Bangladesh, and Kenya, only 12%-60% met the literacy criteria. Thirdly, results imply that ALE graduates who remember the technicalities of reading probably comprehend little of what they read. In short, even though drop out has decreased and completion rates have increased, test achievement rates were almost similar to those of 1976 (Abadzi, 2003a, p. v-vi).

Some studies on adult literacy in the United States of America (USA) concern classroom practices and related issues. In a large-scale study of adult basic education that employed classroom observation, Mezirow, Darkenwald and Knox (1975) had the following results:

- Learners in the observed classes were quite diverse which impeded the formation of true groups, sharing of experience, and peer teaching. Learners arrived and left individually and largely failed to socialize even during class breaks.
- 2. Since there were high dropout rates, teachers were under pressure to maintain enrollment and were concerned with keeping accurate attendance records.

- 3. Traditional classroom conventions were relaxed, making learners arrive late and exhibit behavior not sanctioned by the teacher.
- 4. Mixed skill level and continuous enrollment were the norm.
- 5. Instructional interaction was similar to an elementary school of the 1920s, for example drill, recitation, blackboard work in groups, and routines.
- 6. There was substantial evidence of the present-recite/test-correct approach.
- 7. During the instruction, facts were emphasized and after the lessons question/answer 'recitations' were common.
- 8. Teachers attended to learners one by one, searching out those who had difficulties, and using peer help and classroom aides.
- 9. In order to reduce learner failure, teachers often broke tasks into their simplest components, rewarding successes with praise, and allowing learners to skip difficult assignments.

McCune and Alamprese (1985) conducted a study on the organization and needs of adult literacy services across the USA. The findings included general information about classroom instructional approaches and materials. Individualized instruction was the primary mode of instruction, with a significant portion of the sample using group instructional strategies. There was no consistent pattern in the programs' instructional materials, but most used commercially published materials interspersed with materials created by teachers.

Lerche (1985) identified particularly promising classroom practices. Successful instructional programs had clearly stated behavioral objectives: 'the most consistently successful programs are those that structure and systematize their instructional designs. [. . .] These programs individualize instructional plans to reflect learner strengths and to address learner deficiencies'. While acknowledging many ways to teach literacy, Lerche recommends that learners be involved in the decision-making process and be provided with opportunities to apply newly acquired skills.

According to Darkenwald (1986), there are three models in contemporary adult literacy education practice: competency-based education; tutorial approaches; and community-based approaches. Specific techniques, such as programed instruction, the language experience approach, computer-assisted instruction, and theme-based instruction are used.

Young, Fitzgerald and Morgan (1994) conducted a national evaluation of the federal adult literacy program funded by the Adult Education Act and assessed the extent to which competency-based adult education (CBAE) was being implemented. CBAE is rooted in the realities and interests of the learners and instructional components build on the competencies and goals of the learner. Learners can play a major role in defining the competencies they want to achieve and the way in which they will do so. Collins (1992) implemented a multimethod plan for data collection that used participant observation, interviews, and document analysis. In all but one of the classes observed, discrete skills-building instruction predominated.

Beder and Medina (2001) reviewed classroom behavior and instruction in adult literacy education. The large-scale quantitative studies employing surveys or focused interviews for data collection are primarily descriptive and do not contribute substantially to the development of theory or deep understanding. Although many studies portray instruction as primarily focused on discrete basic skills development, other studies, such as Jurmo (1987), Collins (1992), and Purcell-Gates, Degener and Jacobson (1998), suggest that more participatory forms of instruction also exist.

As Beder and Medina (2001) note:

Only two studies, Mezirow, Darkenwald and Knox (1975), and Collins (1992), are based on classroom observation and focus on classroom dynamics. Both studies agree that instruction is primarily focused on discrete basic skills and the conveyance of factual knowledge. Both also note that traditional classroom norms have been modified to accommodate adult learners, and both studies voice concern for mixed-level, continuous enrollment (p. 18).

The study shows that the major challenges related to ALE teaching and learning were using traditional techniques of teaching, teaching students at different levels in one class, and continuously welcoming new students.

Okech, Carr-Hill, Katahoire, & Kakooza (2001) did a comparative study of the evaluation of FAL and Action Aid's Regenerated Freirean Literacy through Empowering Community Techniques (REFLECT) programs in eight districts of Uganda. The results show that almost 3/4 of the sampled literacy graduates of both sexes and over 1/3 of the non-literates reported to have had some primary schooling, with a substantial minority having at least five years of primary education. The facilitators worked under conditions that were not ideal, for example most of them used open air spaces. Although most facilitators reported that they received the basic teaching materials, in some districts they had neither a blackboard nor a chalkboard. NGO supported facilitators had a better supply of the basic materials. The facilitators considered that the primers were relevant to the needs of the learners despite differences

among the primers of various languages in terms of the depth and quality of core issues, including health, water and sanitation, governance and gender relations.

According to the Ugandan study, the FAL program was run by volunteer facilitators. NGO center facilitators received some allowance. Most facilitators used the local language as their mother tongue and had high school education. The NGO center facilitators received better training compared to most (3/4) of the FAL facilitators. The facilitators reported that there was insignificant monitoring, supervision and support. The monitoring framework for FAL was top-down based on filling in complicated forms by unpaid volunteers who lacked incentives. This may have resulted in unreliable data. On average, the FAL programs were costed at about 4 USD per participant per year. By contrast, the REFLECT programs that provided incentives, and more intensive training and support was calculated at 9 USD per participant per year.

Ishaq and Ali (2012 pp. 95-98) suggest factors that are particularly important for effective implementation of literacy programs in Nigeria. These include: understanding the felt needs of the target community through situational analysis; planning both manpower and material resources; consulting the learners on the time for learning; allocating adequate funding; designing a relevant curriculum with appropriate methods and strategies of teaching; and setting appropriate educational objectives and SMART⁹ learning objectives. According to Philadelphia Public Management Corporation (PHMC) (2011), appropriate training of staff, commitment of staff, respect for adults, practical program content, assessment and evaluation of learners' progress, an appropriate learning environment, and relating the learning programs with the external community are the key characteristics of exemplary adult and literacy programs.

While acknowledging the various options available to decision makers when designing and implementing an adult literacy program, Oxenham (2008) suggests a number of factors that are essential to guarantee return on investment in basic education. These include: that educational planners consult local specialists on the education level appropriate to their societies, languages and writing systems; that literacy programs are accessible as a matter of right and that people are encouraged to participate; that methods addressing the learners' interests and lives are applied in the learners' mother tongue; that well-skilled and committed facilitators are ensured; that the lives of learners are addressed in the learning materials; and

⁹ SMART = Specific, Measurable, Achievable, Relevant and Time-oriented objectives.

that strategies to move from basic skills to fluency levels are designed, i.e. creating literate environments and carefully focused literacy programs that are flexible in their application and use of available communication technologies.

In all the cases, attention is particularly paid to factors directly related to the teaching and learning process rather than to the importance of an efficient institutional system. In contrast, in this dissertation both the delivery of a relevant curriculum and building an efficient institutional system are addressed in the analysis of the implementation of IFALE in Ethiopia. Following the identification of important factors for the implementation of literacy programs is a review of ALE impact assessments that address process and outcomes.

3.1.3 Adult Literacy Education Impact Assessment Studies

Beder (1999) understands outcomes as the changes that take place in learners because of their participation in adult literacy education, i.e. participation in adult literacy education can lead to measurable changes in knowledge, skills, attitudes, and behavior. Learners' participation in adult literacy education can also have an impact on the family, community, and society. According to UNESCO (2005c, pp. 138), existing empirical studies show that the successful completion of adult literacy programs offers not only cognitive outcomes similar to formal schooling but also more adult-specific outcomes, including political awareness, empowerment, critical reflection and community action that are not commonly linked to formal schooling. Credible measurement of outcomes and impacts of adult literacy education is critically important for program accountability and for program planning and improvement (Beder, 1999). However, Stromquist (2009, pp. 65-66) argues that it is difficult to conduct impact studies of ALE because of irregular attendance and (limited) time of enrollment, substantial difference in learners' prior knowledge of printed materials, and variation in content and intensity of literacy programs.

There are more qualitative than quantitative literacy impact studies (Stromquist, 2009). Many studies are based on small samples that could provide insights for the design of future ALE programs while some rely on control groups based on large samples. Some of the ALE impact studies employ both qualitative and quantitative methods for complementarity and triangulation purposes. The review here is of ALE empirical studies from different parts of the world irrespective of methodology.

Some studies have empowerment as a central point when they investigate the impact of literacy programs. Stromquist (2009, p. 66) argues that empowerment has usually been

assessed indirectly through self-reports by learners of their feelings and perceptions about their self-esteem, self-confidence and self-efficacy. Empowerment can also be studied by observing learner behaviors, for example new or improved practices regarding political participation at national and local levels, household decision making, and decisions concerning health and education of family members (Stromquist, 2009). Empowerment selfreporting studies can be based on indicators for various dimensions, e.g. cognitive, political, economic and psychological dimensions (Stromquist, 2009). Such studies can be vulnerable to cognitive illusion and social desirability responses (Abadzi, 2003a, p. 18), especially when applying structured questionnaire items with guided responses (Stromquist, 2009).

Impact studies from both developed and developing countries show that ALE plays a role in empowering the learners. This means that ALE has the capacity, amongst others, to change people from passive subjects to active citizens. McDonald & Scollay, 2009; Metcalf & Meadows, 2009; Bingham, 2009; and Mclachlan, Tett & Hall, 2009, as cited in Stromquist (2009), based on longitudinal studies of large samples found that literacy learners in the US and the UK had acquired feelings of self-esteem, self-efficacy, and self-confidence. Stromquist (2009) argues that the learners were more independent in performing several tasks on their own since they gained improved skills in communication, could employ information technology, could write formal letters and were more competent in dealing with government officials. The learners had more contact with local people, for example through involvement in community organizations. The findings derived from using either self-esteem scales or open-ended interview questions and the effects persisted several years after learners had completed the literacy program.

Besides, several studies conducted in Brazil (Stromquist, 1997), the United States (Young, Hipps, Hanberry, Hopstoch, & Golsmat, 1980, Young et al., 1994; Beder, 1999; Bingman & Ebert, 2000), and African and South Asian countries (Egbo, 2000; Lauglo, 2001; Abadzi, 2003b) indicate that adult literacy programs contribute to participants' improved self-esteem. Other examples of self-confidence gains after attending literacy training were reported by Bown (1990) in a review of forty-four studies on behavioral changes and by Canieso-Doronila (1996) in a study of the Philippines.

Kagitcibasi, Goksen and Gulgoz (2005) studied the impacts of a Functional Adult Literacy Program among women in Istanbul, Turkey. It comprised pre/post-test measurements, a comparison between program participants and non-participants, and a follow-up study after one year. The pre/post-tests used scale items on social participation, family cohesion, the value of children, household decision-making, self-concept and self-efficacy. The study found statistically significant positive changes in all variables, except for self-efficacy and household decision-making. There was significant variation in both social participation and self-efficacy between literates and non-literates.

A distinguishing characteristic of the study is that the participants had TV news recall and comprehension tests at the beginning and at the end of the program in which they recorded significant gains. As stated by the authors, 'learning to read initiates a chain reaction of changes in the cognitive processing system' (Kagitcibasi et al., 2005, p. 483). Stromquist (2009, p. 69) calls this argument 'an uncommon assertion given today's reluctance in several academic circles to consider that literacy can facilitate more complex reasoning and analysis'. What is more noticeable is that the follow up study after a year found that all gains identified at the end of the program period, except for family cohesion, were maintained or improved over time, revealing that the program's contribution to the participants' social integration, positive self-concept, and family cohesion was huge.

Ashe and Parrott (2001) investigated the impact of women's participation in NGO based Nepalese micro-credit programs. The programs comprised a variety of literacy components and lasted four years. It led to 6,500 savings and loan groups with over 130,000 women as members. According to the study, the participants reported a higher self-confidence and decision-making role compared to women who had not been exposed to the literacy components. The women attributed these achievements more to their teamwork than to their access to literacy and credit. Like participants with psychological empowerment gains in other literacy impact studies, the Nepalese female participants were also more inclined to send their children to school as a result of the program.

In a study on the impact of literacy programs in Nepal, Burchfield (1997) compared 400 women that completed the literacy program at least one year earlier to 100 non-literate women. Burchfield used about 40 indicators to investigate empowerment. They were categorized as follows: sense of economic security; ability to earn a living; control over income; household decision making; self-confidence and self-respect; ability to act effectively in the public sphere; participation in public protest or political campaigns; political awareness and participation; participation in non-family groups and organizational meetings; actions to improve children's future; and a vision of the future. Seventy-five percent of the indicators showed substantial difference in the neo-literate women compared to the non-literate ones. The neo-literate women felt more confident to express an opinion, discuss politics, talk about

men's drinking and beating of their wives, and check their children's attendance and progress in school.

Burchfield, Hua, Baral, and Rocha (2002a) and Burchfield, Hua, Saxo, and Rocha (2002b) conducted two longitudinal, primarily quantitative studies in Nepal and Bolivia for over two years to investigate a wide range of impacts related to integrated literacy programs. There were consistent patterns of behavior related to citizenship practices. The study of Nepali women assessed the government literacy program using a skills test of 49 items based on authentic measurement (real objects, pictures, posters found in Nepali villages) and a quasi-experimental (control) research design, including 773 women who participated for different amounts of time in the literacy programs, and 188 non-participants. Based on structured item responses, the study found that compared to non-literate participants, women who participated in the literacy program for a minimum of two years had more political knowledge regarding the minimum voting age and the names of political representatives, believed they could serve as political representatives, and were registered to vote in national elections. Because of their participation, the literate women joined a community group as members, participated in community decisions, and had awareness not only of girls' sex trafficking but also of domestic violence.

A Bolivian study on a women's literacy program run by an NGO involved 716 literacy participants and 224 non-participant women. It was both similar to and different from the Nepali study. First, the measurement of literacy combined self-reported skills and several measures of writing and reading skills. Second, the literacy program participants were a single group. Third, the questionnaire items comprised questions of awareness of laws concerning violence and discrimination, equal rights of women in municipal plans and the inclusion of women's needs in the plans. As in Nepal, the knowledge, attitudes, and practices of political and community participation of the literacy participating women were higher than for the non-participating women. The benefits continued years after completing the program.

Infante's (2000) study of seven countries in Latin America, as cited in Stromquist (2009, p. 66), compared social competencies by level of functional literacy proficiency. Self-esteem, autonomy and communicative competence ('ability to persuade others') were associated with abilities in reading, writing and math. Subjects with high levels of reading competence generally led to good communicative competence among learners.

Four qualitative studies found women's empowerment in the form of self-esteem and selfconfidence in different developing countries. First, in a case study research on women in Brazil, Stromquist (1997) found that the literacy education participating women reported deeper awareness of their environment, greater confidence about engaging in a variety of social activities, and greater self-esteem. Second, Egbo (2000) reported that literate women in Nigeria acquired more knowledge of their rights and became confident enough both to participate in community matters and to make autonomous decisions in the household. Third, Farrell (2004) found that neo-literates in Zambia developed a higher sense of competence considered as an expression of self-esteem by the researcher. Fourth, Archer and Cottingham (1996), in an evaluation study of REFLECT approach programs in El Salvador, Bangladesh, and Uganda, found that women in the literacy circles of Bangladesh improved their confidence and self-efficacy for action within the household but did not adequately participate in any civic organization.

Two ethnographic pieces of research on ALE also reported learners' empowerment gains. In a study based on literacy practices of a poor woman from Bangladesh called Mina, Maddox (2007) found that the woman was able to redefine her identity and modify power relations within the household by being able to have written records of household income and expenditures. Prins (2008)—who conducted an ethnographic study on literacy programs in El Salvador—found that both male and female participants in ALE programs were empowered in terms of their self-esteem, self-confidence, skills of formulating and expressing ideas, skills of participating in and influencing new spaces, and in relations with family members. However, there were no collective forms of empowerment because of the short program duration and the fragmentation of El Salvador in the aftermath of the civil war.

Vasumathi (2008) in her study on neo-literate adults of Malappuram district in Kerala, India, analyzed the personal, family, social, and cultural impacts of adult literacy programs. The study found that adult literacy programs contributed positively in a number of areas, for example learners' political awareness, knowledge of the importance of pre-natal and post-natal care, awareness of environmental pollution, awareness of renewable energy sources, attitude towards work, saving practices, hygienic practices, and participation in cultural activities. Literacy programs also made neo-literates engage in protests against under-age marriage of girls and injustice.

As UNESCO (2005c) states, several studies indicate that literacy has a positive contribution to earnings; however, studies of the impact of adult literacy programs are scarce. Some

studies have attempted to separate the impact of literacy on growth from that of education (UNESCO 2005c, p. 143; Cameron & Cameron, 2005). For example, Coulombe, Tremblay and Marchand (2004), based on data from the International Adult Literacy Survey (IALS), concluded that 55% of the differences in economic growth among OECD countries for 1960-94 was explained entirely by variations in average skill levels, suggesting that investments in increasing the average level of skills could produce large economic returns. Besides, the study revealed that literacy scores resulted in superior growth in terms of output per capita and per worker than indicators of years-of-schooling.

Some studies have explored the economic benefits of adult literacy based on a livelihoods framing of actual and potential literacy impacts (Cameron and Cameron, 2005). Oxenham, Diallo, Katahoire, Petkova-Mwangi, & Sall (2002), based on studies in several countries, stated that people who had finished literacy courses appeared to have more confidence and interest in taking initiatives for their livelihoods improvement. In Senegal, participants in a program that offered both literacy and technical training for agriculture and livestock raising were 6 per cent more productive than the comparison group (non-participant illiterate farmers), but it was difficult to determine whether the impact was caused more by the technical content or the literacy skills (Oxenham et al., 2001as cited in Cameron & Cameron, 2005). Participants in Uganda mentioned 'not being cheated' as one of the benefits of their participation in literacy education programs (Okech, Carr-Hill, Katahoire, Kakooza, & Ndidde, 1999), implying removing the advantage of the more literate party.

There were also other economic empowerments reports. Functional Adult Literacy participants from Rukungiri, Uganda, reported that their knowledge of fuel saving stoves gained from the program helped to reduce the amount of their wood fuel consumption (Katahoire, 2001 as cited in Cameron & Cameron, 2005). Adult education programs from Bangladesh, El Salvador and Uganda motivated participants to reassess and improve their use of land, water, crops as well as raise funding in order to have a tube well constructed (Archer and Cottingham, 1996). Literacy programs in Kenya and Tanzania enabled participants to spread modern agricultural techniques (Carron, Mwiria, & Righa, 1989; Carr-Hill, Kweka, Rusimbi, and Chengelele, 1991). Ashe and Parrott (2001) reported examples of group microcredit programs that were interconnected with literacy training (for instance the Women's Empowerment Program (WEP) in Nepal). The way WEP strengthened the strong saving and credit associations in Terai, Nepal helps explain the success of the program (Ashe & Parrott, 2001).

Studies that have focused on the macro-economic benefits of and economic returns to literacy are relatively rare. Barro (1991), in a study based on cross-country data for 1960-85, and Bashir and Darrat (1994), in a study of 32 Islamic developing countries during the same period, found that both adult literacy and school enrollment rates have positive contributions to growth. A panel data for 1970-90 for 44 African countries also identified literacy as one of the variables with a positive impact on GDP per capita growth (Naudé, 2004).

As Azariadis and Drazen (1990) and Sachs and Warner (1997) argue, the impact of literacy on economic growth depends on the initial literacy level. The former found a threshold effect, i.e. achieving at least 40% literacy rate for undertaking fast economic growth based on technology transfers. The latter identified 'a statistically significant S-shaped relationship with maximum effect when literacy rates were neither very high nor very low', implying that 'small changes at high and low levels might not affect economic growth, but small changes at the intermediate levels characteristic of many developing countries do have an important effect'.

Carron et al. (1989) conducted an impact study of a national literacy program in Kenya comparing literacy graduates who had received a completion certificate with non-literate adults, using structured item responses. In contrast to non-literates, literates showed improved political knowledge as regards ability to identify the ruling party and understand elections, election participation and local association membership. In a country evaluation of a literacy program in Uganda comparing literates and non-literates, Carr-Hill (2001) found that adult literates of both sexes tended to oppose domestic violence, accepted the idea of sharing household chores, considered women as good village leaders, and accepted women's rights to keep their own income. The study also revealed that neo-literate women saved more of their income than non-literate women, but displayed no change in childcare practices.

Okech et al. (2001), in the previously mentioned comparative study of FAL and REFLECT programs in various districts of Uganda, reported some outcomes to be evident. Test results indicated that all sample members were able to read and understand simple questions pertaining to their name, today's date, and who the president is. The average score for nine simple numeracy tests was over 70% and for six more difficult comprehensive questions 60%. The average score of the sample members for relatively simple writing tests was under 40%. Literacy education graduates scored better on the tests than primary school students of grades 3 and 4. There appeared to be little difference between FAL and REFLECT graduates after controlling for prior primary schooling.

In respect of items of functional knowledge attitudes, and practices, program participants showed better overall performances than the comparison group or non-literates. The length of exposure to literacy classes made an insignificant contribution to both knowledge and attitude scores, but contributed significantly to practice scores. REFLECT program participants generally showed significantly better performance than FAL program participants. Nevertheless, when years of primary schooling was controlled, FAL program participants with no or limited schooling performed better than their counterparts in the REFLECT program. By contrast, REFLECT program participants with three or more years of primary education had better performance than their counterparts in the FAL program. Approximately 80% of the literacy program participants claimed that they did some reading. Other participants reported to have practiced writing as well as mental and written calculations occasionally.

Most FAL and REFLECT participants included in the study were happy with reading, writing, and calculating. Other benefits of attending the programs include: improvements in family health, family income, and ability to pay children's school fees; decision making at the family and community levels; self-confidence, self-esteem and the capacity to avoid being cheated and manipulated. The graduates also spent more time discussing education and school matters with their children and looking at the children's homework than the non-literates. Several respondents also reported that participation in the programs helped them start incomegenerating projects.

Okech et al. (2001) draw some of the following conclusions: the literacy programs helped redress the imbalance of educational opportunities between men and women since most program participants were women; the programs appeared to be in danger of missing their primary clients, i.e. people that had no schooling; the majority of the program participants had achieved a level of reading, writing, and calculating that was better than the grade 4 primary school learners; and the quality of implementation was the major explanation for the variations in the learners' attainments.

In a study of 100 participants in the Agona District of Ghana, Arko and Addison (2009) identified a link between functional literacy and development. The beneficiaries of the functional literacy classes could read, write and calculate which improved their social and economic lives. The beneficiaries were also conscious of the need to attend to their health (e.g. knowing that they needed a hospital card to attend the hospital) and jobs. This helped to increase productivity and income levels (Arko & Addison, 2009).

Ayodele and Adedokun (2012) claim that adult functional literacy and adult education in Nigeria are necessary to have an enlightened citizenry that can help develop the nation by participation in decision making, expressing their voice and directing their own affairs. Trudell (2009) argues that local language literacy programs in Sub-Saharan Africa provide women with access to knowledge in a language they understand, leading to benefits, such as better family health, higher school enrollment rates for children, and higher average family income. Zents (2005) adds that literacy in the local language provides women with benefits, such as raising their level of confidence and sense of self-worth, and being assertive in public. They have an increased feeling of pride and personal accomplishment and are therefore more openly critical since they can learn and think for themselves.

3.1.4 Conclusion

The review of the ALE empirical studies shows that ill-designed and poorly implemented programs are important reasons for ALE ineffectiveness. The available research on efficiency and impact provides sufficient evidence to support the claim that adult literacy is not only a critical but also a viable contributor to empowerment of learners and a country's development. Studies on ALE impact clearly show that literacy programs can empower learners of both sexes as expressed in self-esteem and self-confidence, i.e. psychological empowerment. This empowerment enables women and other marginalized groups to break the chain of exclusion that hampers their opportunity for participatory practice. Moreover, knowledge of politics, participation in politics and community organizations, social integration, family cohesion, livelihoods improvement, and cognitive development are elements of empowerment that learners acquire from ALE.

All ALE programs have contributed to different dimensions of empowerment, irrespective of the approach, the cultural and national context, and the duration of delivery. According to Stromquist (2009, pp. 69-70), the Freirean argument that only literacy programs stimulating consciousness raising (particularly those utilizing problem-solving methods) can promote psychological empowerment can be challenged. 'It might be that the exposure of marginalized adults to safe and relaxed settings—such as those afforded by literacy classes—that offer moments of conviviality and information sharing end up producing a greater sense of individual worth' (Stromquist, 2009, p. 70).

Although studies have shown that ALE has an empowering potential, it seems that none has attempted to link ALE to SD via learner empowerment, or to study the impact of ALE on

empowerment of learners in Ethiopia specifically. This dissertation aims to fill this gap by seeking to understand how ALE (IFALE) contributes to SD in Ethiopia through the empowerment of its participants in the political, economic, socio-cultural and psychological dimensions. As will appear in Chapter 4, the study takes into consideration indicators and factors for the different empowerment dimensions that have been used in previous research. It also rests on assumptions developed in theoretical models for understanding adult literacy.

3.2 Theoretical Studies: Literacy Models

The concept of literacy broadly refers to 'a set of cognitive skills necessary to make meaning of print communication' (Stromquist, 2009, p. 64). These skills are improved in specific social contexts and practices (Cope & Kalantzis, 2000). They are applicable irrespective of time and space (Organisation for Economic Co-operation and Development, OECD, & Statistics Canada, 1995) both for efficient functioning in one's community environment and for individual and social transformation. This suggests that literacy can result in empowerment at individual and group levels and serves as a means to an end.

There are three important literacy models, namely the functional, the critical, and the socialpractices model.

3.2.1 The Functional Model

Functional literacy received more prominence since the International Adult Literacy Survey (IALS) was conducted in 22 countries from 1994 to 1998 (OECD, 2000). IALS assumes a link between individual skills and the overall performance of a nation in terms of modernization and economic productivity. This is especially true in OECD countries where literacy skills are key to unlocking the benefits of globalization (Tett & St. Clair, 2010). Many developing countries and international agencies also follow this model which they believe responds to their demands for development (Fransman, 2008; Tett & St. Clair, 2010).

The functional model (also known by many as the skills approach to literacy) has particular positions concerning the purpose of adult literacy. It considers reading and writing primarily as a set of autonomous skills or techniques applied to different contexts of use (Papen, 2005). Tett & St. Clair (2010) claim that in this model, literacy is a set of neutral, technical skills that have little to do with culture and society. Literacy is a skill that remains the same irrespective of the means of acquisition, type of use or contextual relevance to the users (Street, 1984). It serves to accomplish a broad range of activities related to participation of the individual in

society (Tett & St. Clair, 2010). The model relates to psychological and cognitive understanding of literacy and sees literacy from the perspective of individual ability that has relevance to participation at societal level and positively influences development.

According to this view, enabling an individual's access to information, developing his/her thinking, and improving his/her chances of employment and income are expected benefits of literacy. Skills can be monitored and assessed in a standardized quantitative way. Learning outcomes serve as a proxy just as achievement does in formal schooling. Therefore, the model is attractive to donors and policy makers (Fransman, 2008).

In this model, literacy is transferred through the autonomy of the text and the meanings it carries. Universal features of adult literacy as well as other semiotic sound systems are explored (Hamilton, Hiller & Tett, 2006; Tett & St. Clair, 2010). The model emphasizes individual deficits. It also considers literacy as a set of discrete skills that are universal and transferable to all types of circumstances demanding the use of the written language.

There are several critiques of the functional literacy model. First, the functional literacy definition (which labels people as either literate or illiterate depending on their basic reading, writing and arithmetic capacity) is widely acknowledged as unhelpful for policy and practice. It paints an artificial picture of the literacy challenges and results in unsustainable acquisition of competencies that do not respond to the demands of daily life (Fransman, 2008). Second, the model leads to a narrow, reductionist meaning of reading, writing and computing. It disregards learning features that are not manageable at the individual and cognitive level. Tett and St. Clair (2010) add that 'all too often the model can support a deficit view of literacy, where those with limited literacy engagements are seen to be lacking in some way, whether in ability or in education'. Fransman (2008) adds that the model:

gives credence to the understanding of adult literacy as 'second chance schooling' rather than as initiatives tailored to the demands of adults in particular contexts. It also suggests the goal of creating literate individuals rather than literate societies, while failing to recognize that literacy skills are neither relevant nor sustainable without a supportive environment in which to use and develop them (p. 1).

Third, many scholars including Barton (2007) are critical of the model and the assumption that there are clear and discrete stages in learning with individual skills learned in a linear order. In Barton's view, underlying this is the implicit idea that there is only one real way to learn to read, write or use numbers. Fourth, it ignores concerns that are essential to understand the reactions of the learner (Tett & St. Clair, 2010). Despite these challenges, the model has

received great attention since it focuses both on development and on global integration through learners' empowerment.

3.2.2 The Critical Model

In contrast, based on the critique of neo-liberal models of education and elements of the functional model, the critical model is inherently political and influenced by socialist thinking. Critical literacy, which is highly associated with the Brazilian educator Paulo Freire, means the potential to read the word and the world (Freire and Macedo, 1987). It stresses pedagogy that allows learners to understand their world in terms of justice and injustice, power and oppression and the ways of transforming it. The objective is 'conscientization' or becoming critically aware of social, political, economic, and historical forces that shape oppression and ultimately can lead to transformative action (Fransman, 2008). According to Freire (1995), speaking, reading and writing are interrelated elements of an active learning process and of social transformation. In Shor's (1993) view, the primary purpose of critical literacy is to be critical of the dominant culture and the existing power relationships between social groups rather than help learners move up the social ladder (as in the functional model).

In this approach, literacy acquisition means to explore the social and political dimensions of the learner experiences (Fransman, 2008). People need to master different skills—to make sense of literal and hidden text meanings and to discuss the position a text supports (Tett & St. Clair, 2010). The focus is on interpreting, reflecting on, interrogating, theorizing, investigating, exploring, probing, questioning, as well as acting on and dialogically transforming the social world. This model, with its focus on power and oppression, presents itself against the competencies position of the functional approach. It emphasizes the political and institutional dimensions rather than the personal and cultural aspects (Fransman, 2008).

3.2.3 The Social-Practices Model

The social-practices model varies from the functional and critical models particularly in emphasizing design over and above competence and critique (Fransman, 2008). There are various versions of the social-practices theory of adult literacy, numeracy, and language and, therefore, various approaches as well. The approach typified by the New Literacy Studies (Street, 1984; Barton et al., 2000) represents mostly ideas and methodologies from sociology, socio-linguistics, and anthropology. In contrast, Vygotsky and others focus more on active problem solving theory.

According to Fransman (2008), the social-practices model has at its core the notions of literacy events, practices and domains. Literacy events refer to 'the particular activities in which literacy has a role' (Barton, 1994). Literacy events draw on the general literacy practices, but are directly observable, for example reading a newspaper, writing a grocery list, writing a letter to a friend, reading a company memo or a pay slip, reading a textbook and writing a term paper (in school) (Merrifield, 1998).

Street (1984) uses the phrase 'literacy practices' to focus on the knowledge and power relations surrounding a literacy event. Literacy practices are 'general cultural ways of utilizing literacy' (Barton, 2007) that people draw upon in the varied contexts in which they live their lives—school, work, home, social groups, neighborhoods; they consist of behavior, meanings, values, and social relationships (Merrifield, 1998).

The term domain is defined by Merrifield (1998) as 'the extensive life contexts we engage in, including the family, the school, the church, and the various social groups that are molded in turn by the broader culture, by class, gender, ethnicity and regional variation'. When events, practices and domains are combined, they explain the notion of 'multiple-literacies' that may not only refer to schooling based literacy (Fransman, 2008). Tett and St. Clair (2010) claim that in addition to reading, speaking and writing, the use of new technologies is indispensable to the definition of literacy.

According to Hamilton et al. (2006), the social-practices model has moved away from the individually focused cognitive skills perspective to include the social practices associated with numbers, reading, and writing. The approach, while recognizing difference and diversity, and challenging the way differences are valued in society, gives emphasis to the various ways people engage with literacy instead of viewing literacy as something learners lack (Hamilton et al., 2006; Tett & St. Clair, 2010).

This is a shift from seeing literacy as a gift given to people to an ideological view that places literacy in the wider context of institutional purposes and power relationships (Street, 1995). While emphasizing the cultural practices within which written and spoken words are embedded, adult literacy is in this approach part of the range of social practices that are observable in events or moments and patterned by social institutions and power relationships (Tett & St. Clair, 2010).

This view sees literacy as historically and socially positioned and a part of the broader cultural and media engagement, not just as an individual activity. The functional approach emphasizes the self-sufficiency of the text and the sense it conveys. In contrast, socialpractices literacy, according to Hamilton et al. (2006) and Tett and St. Clair (2010), underlines the connections between the classroom and the community in which learners live, between learning and institutional power, and between print literacy and other media. The social-practices approach focuses on learning that takes place in the daily life of the learners as opposed to only the classroom, with meanings, values, and purposes located within a broader literacy framework than the texts themselves.

The social-practices model rests on two important principles: a two-way dialogue and movement between formal learning and the everyday world, and active learning (Tett & St. Clair, 2010). With regard to the first principle, Tett and St. Clair (2010) argue that situated cultures and practices cannot simply be accepted and imported into the classroom setting. As to the second principle, active learning is at the heart of the learning process in adult literacy. In the view of Hamilton et al. (2006), active learning has significant implications for relationships within the learning process and for reflective and questioning activity on the part of both learners and teachers. It is essential that teachers and learners participate in decision making and governance of the organization of learning, involvement in management committees, consultative bodies, and research and development activities based on which citizenship is modeled (Tett & St. Clair, 2010).

Brandt and Clinton (2002), and Collins and Blot (2003), however, argue that the approach does not adequately take into consideration the way external forces, such as economic globalization, missionaries and colonial powers, have been superimposed upon the local experiences of particular societies. According to Maddox (2001), proponents of the social-practices approach do not consider the potential of literacy to make people shift from local positions towards better economic, social and political involvement. In Fransman's (2008) view, the approach ignores the fact that sometimes people want basic competences (either for symbolic or instrumental purposes) and sometimes they want change or even to move away from their local context instead of merely continuing to function within it.

3.2.4 Conclusion

As indicated in Table 3.1, each literacy approach has a unique purpose, focus and method. The functional model focuses on gaining measurable competencies for national development. The critical model focuses on transformation of dominant power relations. The social practices model aims at contextually relevant literacy events, practices and domains. Each
model applies different research methods. The functional model is highly related to quantitative studies, the social practices model highly relies on ethnographic studies, and the critical model often adheres to participatory rural appraisal (PRA)¹⁰.

Model	Purpose	Focus	Methods
Autonomous (Functional)	Competition/ Development	Competences	Quantitative or quantifiable
Transformative (Critical)	Social transformation	Critique	Participatory rural appraisal
Social-practices	Contextual relevance	Design (events,	Ethnographic research
(ideological)		practices, domains)	

Table 3.1: Purpose, focus and research methods in selected literacy models

Source: Adapted from Fransman, 2008

In reality, the literacy provider or researcher may combine two or more models and, therefore, more than one method. UNESCO (2004) underlines that a pragmatic understanding of literacy should recognize important elements from different approaches to acknowledge the 'plurality of literacy', rather than taking a consensus position based on one or the other approach. With the shift of the conceptualization of literacy from a discrete set of technical skills, to human resource skills for economic growth, and to capabilities for socio-cultural and political change, international organizations admitted comprehensive understandings of literacy that 'encompass *conscientization*, literacy ratices, lifelong learning, orality, and information and communication technology literacy' (UNESCO; 2005, p. 159).

In this dissertation, conceptualizations from each literacy model will be combined in consideration of their relevance to the study. The functional model claims that individual abilities have relevance to the improvement of personal income, thereby enhancing participation at family and societal levels and making a positive contribution to overall development. In the critical model, critical thinking and problem-solving skills are essential for both individual and societal transformation. The social practices model stresses the use of active learning methods in the context of a curriculum. The assumption in the dissertation is that these aspects are particularly important if literacy is to empower learners by enhancing their confidence, knowledge and skills, changing their behavior and developing their capabilities to participate actively in the political, economic, and socio-cultural domains. All of this is essential to support SD. Understandings of empowerment and sustainable development are, however, contentious. The concepts are discussed in the following section in order to arrive at clarity of their use in this dissertation.

¹⁰ Chambers (1992) defines PRA as a technique intended to enable local communities to conduct their own analysis and to plan and take action. The objective of PRA is to help strengthen and/or assess the capacity of villagers to plan, make decisions, and take actions to improve their own situation (Cavestro, 2003).

3.3 Empowerment and Sustainable Development in the Context of Adult Literacy

3.3.1 The Concept of Empowerment

As a theory of social change, empowerment has attracted many scholars in both theoretical and practical terms (Stromquist, 2015). In Zimmerman's (1984) view, a single definition of empowerment may make it prescription-like. While the definition depends on the context, it should relate to a multi-dimensional social process that helps people gain control over their own lives. It is a process that fosters power in people—that is, the capacity to implement— by acting on issues that they define as important for their own lives for their communities, and for their society (Page & Czuba, 1999). According to Stromquist (2009), empowerment has been defined in different ways which all have 'a common occurrence with meaning-rich notions that include democracy and globalization'.

As Page and Czuba (1999) state, empowerment is multi-dimensional, social and a process. It is multi-dimensional as it has sociological, psychological, economic, and other dimensions and occurs at various levels, for example individual, group, and community levels. It is social since it occurs in relationship to others and a process that is similar to a path or a journey—it develops as one works through it. Batliwala (1994, p. 130) defines empowerment as 'a spiral process, changing consciousness, identifying areas to target for change, planning strategies, acting for change, and analyzing actions and outcomes'.

Easton, as quoted in UNESCO (2005a), claims that literacy—and informal or adult education in general—can be linked to empowerment at overlapping and intertwining micro and macro levels. At the micro level, literacy and NFE programs can be empowering when they turn learners into authors of their own learning, developers of their own knowledge, and partners in a dialogue about limiting situations and barriers imposed on them which are dehumanizing (Freire, 1970). This is psycho-cultural empowerment. At the macro level, literacy and NFE programs can contribute directly to large-scale processes of socio-economic empowerment in short, to the mastery that people, organizations and communities acquire over their own affairs and the control that they are able to exercise on their environment (Belloncle, 1993 as cited in UNESCO, 2005a; UNESCO 2005c).

In Sen's (1985) assertion, capabilities and functionings are essential for individual agency and self-determination to achieve well-being. Capabilities aggregate resources and human capital

to create the conditions under which individuals make choices. Functionings refer to the ways of being and doing of individuals. Building on Sen's capabilities, Nussbaum (2000) listed 10 essential capabilities¹¹ that provide a threshold for each citizen in order to have a decent life. Functional adult literacy can be a means to achieve two of the central capabilities that are essential elements of empowerment: senses, imagination and thought; and practical reasoning (Kagitcibasi et al., 2005).

It is claimed in this dissertation that literacy should also be a means to achieve three other capabilities that are essential for Nussbaum (2000): bodily health (having good health, including productive health); affiliation to others (have concern for and live with others in the environment); and control over one's environment (being able to participate in politics and being able to own property). Owning property may be associated with Sen's (1999) capabilities approach that addresses increased income and wealth. Stromquist (2015, p. 309), in her theory of empowerment, argues that education—that is closely linked to economic, political, cognitive and psychological conditions—remains a key instrument for achieving learners' empowerment. According to her, empowerment is inseparable from individual and collective action.

In this dissertation, ALE for empowerment is also understood within the LLL framework based on the four pillars of education identified by Delors (1996) to which Ouane (2008) added a fifth one, as discussed earlier: 1) learning to know (mastering instruments of learning); 2) learning to do (training people to work, invent and adapt for the present and future); 3) learning to live together and with others; 4) learning to be (supporting the complete development of learners' mind, intelligence, sensitivity, etc.); and 5) learning to change and to take risk (persevering for personal and national development).

ALE is believed to empower learners by improving their livelihoods. Cameron and Cameron (2005) state that the livelihoods framework generally connects economics and broader development studies. It is wide enough to embrace some of the problems that are likely to be disregarded in more conventional cost-benefit analysis. Development agencies, for example the Department for International Development (DFID) (2002), consider the framework as a vital means of linking literacy to the lives and aspirations of the poor.

¹¹ Nussbaum's (2000) list of capabilities include: life (being able to live); bodily health (being able to have good health, including productive health); bodily integrity (being able to go from place to place freely); senses, imagination and thought (being able to imagine, think, reason and use thought connection); emotions (being able to love, grieve, and angry); practical reasoning (being able to form a conception of the good and critically reflect on it); affiliation to other species (being able to have concern for and live with others in the environment); play; and control over one's environment (being able to participate in politics and being able to own property).

As stated by Cameron and Cameron (2005, p. 4):

In the livelihoods framework, all households are seen as utilising changing patterns of natural, produced, human, financial, and social wealth to create livelihoods:

- 1. people can develop their capabilities into skills whose expression over time as *human* wealth is both means and an end to long term development;
- 2. the relatively modified physical environment is a reservoir of *natural* wealth important to human well-being in itself and to self development;
- 3. human activity in the natural environment can generate *produced* wealth, such as equipment and cooking utensils, that has a physical life and productive potential beyond immediate human consumption;
- 4. some wealth is held in *financial* forms as money or near money, such as jewellery, due to properties of liquidity and high fungibility with other forms of wealth;
- 5. societies have collective histories of building trust, confidence and mutual security into relationships that constitute a *social* wealth.

An empowering literacy education can thus be a means to the human, natural, produced, financial, and social wealth of participants (Table 3.2).

	Social	Financial	Produced	Natural	Human
Human	More effective	Insurance and	Safe and efficient	Ownership/control of	Income raising
	and transparent	Income smoothing,	Use	natural resources	vocational skills and
	principal/agent	higher remittances		plus better health	more cost-effective
	relationships	from migration			schooling of next
					generation
Natural	Effective natural	Mortgage	Environmental	Improved fertility	
	resource user	indebtedness	Sustainability	and lower	
	groups	management		degradation/pollution	
Produced	Group sharing/	Investment	Technological		
	leasing/hiring	Financing	innovation with		
	arrangements		accurate		
			specifications		
Financial	Micro-credit	Cash flow		-	
	group stability	Management			
Social	More effective				
	co-operation for				
	advocacy and fairer				
	competition				

Table 3.2: A livelihoods framework for monitoring the impact of literacy interventions

Source: Cameron and Cameron, 2005, p. 5

Literacy interventions can be related to 'this framework through a matrix that considers each asset in itself and also allows consideration of paired relationships' (Cameron & Cameron, 2005, p. 4). Each of the linkages can be examined in terms of whether it works to improve livelihoods through either market forces or political advocacy (Cameron & Cameron, 2005). In this dissertation, it is claimed that ALE has the capacity to enable participants to develop some of Nussbaum's (2000) central capabilities and the paired links of natural, produced, human, financial, and social wealth patterns to create livelihoods, as argued by Cameron and Cameron (2005). Together, they can lead to political, economic, socio-cultural and psychological empowerment of learners and support the wider context essential to SD.

Olaleye and Adeyemo (2012) discuss the empowering capacity of adult literacy education with a focus on the four important areas of functional literacy (Figure 3.1). First, functional literacy for learner empowerment requires basic skills, including basic literacy skills, i.e. literacy, numeracy, graphics and measuring skills. Second, the life skills dimension of empowerment includes self-awareness development, analytical skills, decision-making skills, organizational skills and technical skills. Third, empowerment acquired from functional literacy includes the socio-economic aspects that comprise acquisition and improvement of vocational skills, sustainable income generating skills, and learning to learn skills. Fourth, an empowering functional literacy includes LLL skills for societal support, continuous learning, life skills and socio-economic skills.



Figure 3.1: Dimensions of functional literacy and their outcomes *Source*: Olaleye and Adeyemo, 2012, p. 14

Unlike Olalye and Adeyemo (2012), Stromquist (2009) defines empowerment clearly regardless of the literacy model:

In the context of adult literacy, in which non-literate adults reflect ongoing and cumulative processes of regional, ethnic, social class, or gendered marginalization, empowerment must mean the set of feelings, knowledge, and skills that produce the ability to participate in one's social environment and affect the political system (Stromquist, 2009, p. 64).

According to Stromquist (1995, 2009, p. 64), this ability embraces four dimensions: the cognitive; the economic; the political; and the psychological. Cognitive empowerment refers

to 'the knowledge of one's social reality and the mechanisms that make it function the way it does' (Stromquist, 2009, p. 64). Literacy programs can lead to cognitive empowerment depending on the contents of the programs and on the ability of the neo-literates to sustain their capacity by moving into some form of literacy habit (Stromquist, 2009, p. 70). Post-literacy initiatives can be a necessary condition for cognitive empowerment.

Cognitive empowerment

As argued by Stromquist (2009, p. 70), women in literacy programs develop greater understanding of gender discrimination and inequality in society. Literacy programs should also support adult male learners' understanding of this social phenomenon to change unfair practices. Some may be encouraged to be activists to fight gender discrimination and inequality. Furthermore, cognitive empowerment gained from ALE must enable learners of both sexes to engage in, for example, group or community development and environmental protection efforts, and should lead to attitudinal change of learners. Street (1988), however, contends that literacy may not result in stronger cognitive development than from oral discourses. In general, the contribution of literacy to wider areas of cognitive development is not well explored. The exception is the study by Kagitcibasi et al. (2005) that examines the role of ALE for media comprehension improvement. Their pre/post-test showed that literacy participants had significant gains on the TV recall and comprehension test.

Economic empowerment

Economic empowerment reflects 'access to independent means of support, helping make individuals more autonomous in their decisions' (Stromquist, 2009, p. 65). According to Lauglo (2001, p. 20), short term income gains from literacy are expected if adult basic education embraces income generating activities and is linked to credit access. This is also argued by Stromquist who states that the 'economic dimension of empowerment is less likely to emerge from literacy programs because economic improvement is highly predicated on access to credit and access to income generation—policies for which governmental support is essential' (2009, p. 70).

As Cameron and Cameron (2005, pp. 3-8) suggest, different measures of livelihoods improvement deriving from literacy education, for example knowledge of the measures of decreasing environmental degradation/pollution, use of new technologies for increasing agricultural production, improved saving practices, personal hygiene and maternal and child

care, practice of measures of HIV/AIDS, and confidence gains that lead to taking initiatives to develop participants' livelihood appear to be vital elements of economic empowerment.

Although there is data linking literacy and education to economic growth, the mechanisms are not well explained (UNESCO, 2005c). Windham (1999) reports major challenges in using conventional economic analysis to measure the benefits of literacy due to the difficulties of linking literacy gains and economic development. These, amongst others, include: the possible problems in determining the standards for calling an individual literate; and the risk that a considerable share of women's benefits from literacy may not to be calculated since it contributes either to household tasks or indirectly to other family members' economic activity. What is a more complicating factor, according to Cameron and Cameron (2005, p. 4), is that literacy training in practice is often interconnected with other development programs.

According to Lauglo (2001, p. 21), in the economics of education, it is challenging to isolate economic outcomes of literacy since these can also derive from other sources. The economic benefits of literacy education have generally not been supported by strong conclusions in economic analysis (Cameron & Cameron, 2005, p. 1). A careful research design is thus vital not only to identify whether the outcomes originate from literacy or from other sources, but also to justify the benefits based on appropriate economic analysis.

Political empowerment

The political dimension represents 'the skills of participating in and modifying institutions and policies of one's community or nation' (Stromquist, 2009, p. 65). UNESCO (2005c, p. 139) argues that the empowering potential of literacy can lead to improved political participation and quality of public policies as well as to the expansion of democracy. Political empowerment manifests itself in voting during elections or joining a union, discussing politics with friends or engaging in community self-help initiatives and community level decision-making. However, according to Hannum and Buchmann (2003), the nature of the link between education and democracy is unclear and hard to measure.

Neo-literates might participate in voluntary activities that aim to influence political decisions at all levels of the political system. These could be conformist participatory activities (for example, party membership or electoral campaigning) or non-conformist activities (such as strikes and demonstrations). However, the political effects of ALE do not, by and large, include 'collective action such as organizing and mobilizing around specific

objectives, a finding that is hardly surprising given the apolitical content of most literacy programs, their short duration, and the unsupportive economic climate that has placed many poor in a survival mode' (Stromquist, 2009, p. 70). When literacy programs embrace basic political issues in the teaching, their adult learners will better be ready to take further political action.

Psychological empowerment

The psychological dimension of empowerment, which is a major ingredient of social capital, encompasses 'feelings that individuals are competent, worthy of better conditions and capable of taking action on one's own behalf' (Stromquist, 2009, p. 65, 1995). These are mostly self-esteem and self-confidence. Psychological empowerment is associated with a high level of confidence among neo-literates that results in manifestations of social engagement and political behavior. Self-esteem or assessment of one's own worth precedes social and political action.

Stromquist (2009) also argues that the psychological dimension at the personal level often functions as a basic requirement for the other three dimensions although it is difficult to prove a linear sequence amongst them. For Olaleye and Adeyemo (2012, p. 212), an empowered individual has self-confidence and self-worth, critically analyzes his/her social and political environment, and is able to exercise control over decisions that affect his/her life. Stromquist (2009, p. 70) claims that psychological empowerment ought to produce new political actors with political will demanded for effective literacy design and implementation and sustainable literacy programs. Therefore, self-concept and self-efficacy should be vital elements of psychological empowerment that have a bearing on political, economic, social, and cultural gains.

Socio-cultural empowerment

Socio-cultural empowerment refers to an individual's ability to enhance social capital in terms of, for example, social networks and forms of cooperation (Vasumathi, 2008, pp. 121-122) and to transform traditional ways of life, such as beliefs, practices and life styles, into modern ones. Adult literacy education can enhance a range of capabilities, including maintaining good health and living longer, learning throughout life, controlling reproductive behavior, raising healthy children and educating them, and addressing gender equality (UNESCO, 2005c, pp. 141-142).

Literacy education can also provide access to new values, for example written culture, or promotes 'transformation of other values, attitudes and behaviors through critical reflection' and helps to 'challenge attitudes and behavioral patterns' (UNESCO, 2005c, p. 140). Literacy programs may have promotion of values, including equity, inclusion, respect for cultural diversity, peace and active democracy, as their major objective (UNESCO, 2005c). For Farah (2005), the contribution of a literacy program to individual or group cultural change may be slow, not easily accessible and difficult to identify as an effect of a program. In spite of this, adult literacy education can be instrumental to socio-cultural empowerment gains of participants.

The concept of empowerment as applied in this dissertation encompasses an analysis of the potential importance of IFALE to only four of the dimensions discussed above, i.e. the psychological, economic, political, and socio-cultural ones. The analysis takes into consideration the importance of other sources, such as agricultural and health extension workers as well as the local media, in the explanation of the possible impact of IFALE on the learners according to each of the dimensions. The outcomes of IFALE are understood at the personal, family and community levels and are assumed to contribute to development at the national level. In that sense, empowerment is understood as a precondition to SD.

3.3.2 The Concept of Sustainable Development

The concept of SD evolved out of the need for a world that is safe for life. In the 21st century, the dominant forms of political economy (capitalism and neo-liberalism) are unable to conserve ecological resources and services, guarantee economic stability, reduce social inequality, maintain cultural diversity, and protect people's physical and mental health (Huckle, 2008, p. 342). Ecological, economic, social, cultural and personal sustainability depends on a more sustainable political economy that includes a new method of global governance led by new forms of citizenship (Huckle, 2008).

SD is a continuously evolving concept (Walters et al., 2012). According to the United Nations' Brundtland World Commission on Environment and Development (1987),

SD is one that meets the needs of the present without compromising the ability of future generations to meet their needs. It contains within it two key concepts: the concept of 'needs', in particular the essential needs of the world's poor, to which overriding priority should be given; and the idea of limitations imposed by the state of technology and social organization on the environment's ability to meet present and future needs (p. 41).

SD gives special attention to the most disadvantaged and poor people of the developing world who are more exposed to an unsustainable life. Sustainability is a way of thinking about a future that balance environmental, societal, and economic considerations in the pursuit of development and an improved quality of life (UNESCO, 2006a, p. 10). Environment, society, and the economy are components of SD (UNESCO, 2006a, p. 10). Culture, referring to ethical values and respect for human rights and diverse cultures, is sometimes suggested as a fourth component (UNEP, 2004).

Ndoye and Walther (2011) suggest that SD for Africa must comprise four interrelated aspects:

i) protection of the environment and preservation of natural resources, and particularly the risks of drought and desertification threatening the continent due to global warming, ii) the promotion of a model of economic growth that is both accelerated and sustainable, because it is based on the rational and efficient use of resources, iii) the construction of inclusive societies founded upon the need to work hard to eradicate poverty and all sources of discrimination and marginalization, and iv) the strengthening of mutual cultural and spiritual knowledge and understanding in order to eliminate conflicts and civil wars and promote solidarity among peoples, peace and continental integration (p. 54).

In their view, in order to realize SD, African countries must engage in these issues through their sector development plans, including education and training—be it formal (pre-school, primary, secondary, technical and vocational, and tertiary education) or non-formal education (adult literacy education, alternative basic education programs, farmer training institutions).

Education for Sustainable Development

Research has shown that education is a key to a nation's ability to develop and achieve sustainability targets (UNESCO, 2006a). Education and training have the capacity to enhance productivity, the status of women, and environmental protection as well as reduce population growth rates (Walters et al., 2012). Walters et al. (2012, p. 23) claim that when education levels are low, economies are often limited to resource extraction and agriculture. In many developing countries, the level of education is low which severely hinders development options and plans for a sustainable future. Education and training are central to improving the quality of life and the educational attainment of the next generation, thereby raising its chances for economic and social well-being (UNESCO, 2006a).

SD implies that learning must respond to both present and future challenges (Hoppers & Yekhlef, 2012). Education must produce well-educated people who contribute to economic, social, cultural and political changes needed for SD (Ndoye & Walther, 2011). For Hoppers and Yekhlef (2012), 'this implies that education needs to ensure that people acquire a range of

skills, competencies, values and attitudes enabling them to act in their environment so as to enhance its quality'. Education, in general, and ALE, in particular, can enable learners to gain vital learning outcomes or skills and encourage them to apply the skills in their day-to-day activities.

Education should fully embrace initiatives linked to the UN's Decade of Education for Sustainable Development (DESD) 2005-2014 (Huckle, 2008, p. 342). The goal of DESD was 'to integrate the principles, values and practices of sustainable development into all aspects of education and learning' (UNESCO, 2005b, p. 6) in order to change the behavior required to produce a 'more sustainable future in terms of environmental integrity, economic viability, and a just society for the current and future generations' (UNESCO, 2005b, p. 6). DESD envisages that each person in the world takes advantage of education and acquires the values, behavior and lifestyle needed both for a sustainable future and for constructive societal change (UNESCO, 2005b, p. 6). DESD strives to make adult learners benefit from their learning, gain the basic skills, address the difficulties they experience, and achieve sustainable societal transformation for a better future life. ESD has both philosophical and ethical foundations.

While discussing the philosophical foundation of ESD, Huckle (2008) argues:

Central to the perspectives that ESD should develop is what Hartmann (1998) terms a social-ecological theory of reality and the values that stem from it. Rather than regarding nature and society as separate realms (modern dualism) we should acknowledge that reality is always the product of both ecological (bio-physical) and social relations and processes. The phenomena of global warming illustrates how the relations between objects in the bio-physical and social worlds enable ecological and social processes, how these processes affect one another constantly, and how our understanding of such phenomena can never be entirely neutral or objective because it is always partly a product of those social or power relations it needs to explain. The politics of sustainability is about the relations that humans are in with other human and non-human agents, how we understand these relations, and what we can do to ensure that they are more sustainable (p. 342).

ESD, which is at the heart of socio-ecological theory, should link ecology with social relations and processes in order to lead civilizations on to a more sustainable path. Global warming is concerned with the relation of living organisms to one another and to other physical surroundings, explaining a critical example of unsustainable life in this planet. The socio-ecological theory or reality, founded on dialectical materialism (Harvey, 1996) or the life sciences and systems theory (Capra, 2003), considers that humans are members of biological species depending on ecological resources and services to supply their needs;

however, they are capable of transforming their own nature and the nature surrounding them because of language and technology (Huckle 2008, p. 343).

As Huckle (2008, p. 343) argues, in order to have a sustainable life, there should be equilibrium between ecology and society. People must balance an ecocentric perspective¹² with an anthropocentric¹³ or technocentric perspective¹⁴. Huckle also argues that people have a moral obligation to act responsibly to keep the planet habitable for the human race and other living communities and care for the natural environment and its habitants despite the fact that we are more powerful:

In seeking sustainability, we should be guided by a weak anthropocentrism. This maintains that while humans are the only source of value, they are not the only bearers of value. In addition to valuing or caring for present and future generations of people, we should value and care for the rest of nature by recognizing its ecological, scientific, aesthetic and spiritual value alongside its economic value, and acknowledging its right to exist. In other words, we should balance our rights to self-determination and development, with responsibilities towards the rest of the human and biotic community (p. 343).

Huckle emphasizes that the capacity to think critically obliges people to act responsibly which is a rational view of maintaining life on our planet. The Earth Charter (UNESCO, 2000b) reflects the ethics of weak anthropocentrism by setting out the fundamental principles of SD. The charter recognizes respect and care for the community of life, ecological integrity, social and economic justice, and democracy, non-violence and peace as vital concerns. Principle 13 calls upon the world community to strengthen democratic institutions at all levels, provide transparency and accountability in governance, and have inclusive participation in decisionmaking and access to justice. Principle 14 advocates for the integration of ESD as part of formal and life-long learning, implying that ALE is important as well.

UNESCO, which has been leading the implementation of DESD worldwide, identified the main features of ESD as: design of an interdisciplinary and holistic curriculum; sharing the values and principles underpinning SD; critical thinking and problem solving skills; use of a multi-method approach in the teaching and learning; integrating learning experiences in personal and professional life; participatory decision-making of learners; local relevance: addressing local as well as global issues using the language(s) of the learners; and critical

¹²Ecocentric perspectives refer to a philosophy that places intrinsic value to all living organisms and their natural environment, regardless of their perceived usefulness or importance to human beings (Rowe, 1994).

¹³Anthropocentric is a philosophical viewpoint arguing that human beings are the central or most significant entities in the world. It also refers to interpreting or regarding the world in terms of human values and experiences (Brennan & Yeuk-Sze, 2016).

¹⁴Technocentric perspectives refer to a value system centered on technology and its ability to control and protect the environment (Wikipedia, The Free Encyclopedia, 2016).

respect for others (present and future generations), for difference and diversity, for the environment, and for the resources of the planet we inhabit (UNESCO, 2006b, pp. 4-5; UNESCO, 2007, p. 6).

ESD is shaped by perspectives on, for example, human rights, peace and human security, gender equality, cultural diversity and intercultural understanding, health, HIV/AIDS, governance, natural resources, climate change, rural development, sustainable urbanization, disaster prevention and mitigation, poverty reduction, corporate responsibility and accountability, and the market economy. These all have implications for the attainment of a more sustainable process of change perspectives which deal with all human development fields as well as the critical challenges of the world. These have implications for the attainment of a more sustainable process of change (UNESCO, 2006b).

The 2009 Bonn Declaration for SD, adopted at the UNESCO World Conference on ESD, reiterates that:

Through education and lifelong learning, we can achieve lifestyles based on economic and social justice, food security, ecological integrity, sustainable livelihoods, respect for all life forms and strong values that foster social cohesion, democracy and collective action. Gender equality, with special reference to the participation of women and girl children in education, is critical for enabling development and sustainability. Education for sustainable development is immediately necessary for securing sustainable life chances, aspirations and futures for young people (UNESCO, 2009a, p. 1).

Besides, the Bonn Declaration reaffirms that:

Building on the Jomtien, Dakar and Johannesburg promises, we need a shared commitment to education that empowers people for change. Such education should be of a quality that provides the values, knowledge, skills and competencies for sustainable living and participation in society and decent work. The Education for All agenda underlines that the availability of basic education is critical for sustainable development. It similarly emphasizes pre-school learning, education for rural people and adult literacy. Achievements in literacy and numeracy contribute to educational quality, and will also be critical to the success of ESD (p. 1).

This declaration implies that ESD and lifelong learning enhance learner empowerment and contribute to the realization of the comprehensive perspective of SD. As a prerequisite to the SD goals formulated in 2015, the members of the Conference reaffirmed the need for an effective education system and of adult literacy education to promote values, knowledge, skills and competencies that empower people.

Education, Literacy and the Sustainable Development Agenda

The 2030 agenda for sustainable development has 17 goals of which goal 4 envisages to ensure inclusive and equitable quality education and to promote lifelong learning opportunities for all (UN, 2015). This goal specifically aims to ensure the following targets for SD:

All learners acquire the knowledge and skills needed to promote sustainable development, including, among others, through education for sustainable development and sustainable lifestyles, human rights, gender equality, promotion of a culture of peace and non-violence, global citizenship and appreciation of cultural diversity and of culture's contribution to sustainable development by 2030 (UN, 2015).

Hoppers and Yekhlef (2012, p. 18) also argue that SD can influence education in various ways:

- 1. Paying attention to the environment, its exploitation and preservation as part of learning, and to the relationship between schools/learning centers and their environment;
- 2. Developing a fully inclusive approach to educational development so that all children, youth and adults have equitable access to relevant and appropriate basic education following the principle of LLL;
- Incorporating into the curriculum a range of basic skills, competencies, attitudes and values associated with living together in a non-discriminatory and non-racial manner; promoting democratic decision-making, prevention of conflict and respect for differences; building capacity and commitment to work for the betterment of society; enabling younger and older learners to deal with change;
- 4. Promoting application of taught skills and values as a basis for lifelong interaction between learning and living.

This suggests that ESD contributes to SD, but the goals for SD can also influence the content of formal and non-formal education, for example in terms of caring for the environment and providing fundamental skills, competencies and practices within a LLL perspective. ESD can have an enduring impact on education through reform of educational provision, the curriculum and the quality and relevance of the teaching and learning process (Hoppers & Yekhlef, 2012, p. 18). Capable curricula designers and teachers, adequate resources and building a strong institutional system are preconditions to addressing effectively issues of sustainability through education and training. Emphasis should be given to skills, values and perspectives that encourage active participation of all citizens (Walters et al., 2012; Trudell, 2009). These are often rooted in learner-centered approaches.

Research and experience indicate that literacy can be an important instrument to eliminate poverty and to achieve development. In Africa, local language literacy programs lead to a

wide range of development benefits and by their very nature can reach the populations most in need: the rural; the poor; women; and those who struggle to succeed in the formal education system. Whether carried out in formal or non-formal contexts, local-language literacy facilitates development in a variety of ways, and its inclusion in locally situated development efforts is key to their sustainability (Trudell, 2009).

Literacy education for the poor can serve two major purposes: increasing their income, and improving ideas and values that enable them to eradicate poverty (Yibing, 2000). As former UN Secretary-General Kofi Annan (2005) points out, 'literacy is a key lever of change and a practical tool of empowerment on each of the three main pillars of sustainable development: economic development, social development and environmental protection'. This claim supports the major assumption in this dissertation, namely that there is a causal connection among literacy, empowerment and SD.

3.4 Conclusion

The discussion of the empirical studies on ALE, the models of literacy, and the concepts of empowerment and SD indicates that well implemented ALE programs play an important role in enhancing learner empowerment and supporting SD efforts of a country.

The review shows that ALE programs should be based on needs assessment of potential participants and the community at large, appropriate choice of literacy model to be followed, and identification of services to be provided, and the source of funding if it is to satisfy the interests of target communities and governments. What is equally important is the implementation process. Scholars report that poor access to ALE, high dropout rates, low skills retention, shortage of relevant material supplies and qualified manpower, relapse into illiteracy, inappropriate use of teaching methods, poor program monitoring, supervision and support, and unfavorable learning and working conditions are negative results of both flawed design and poor implementation of ALE.

As indicated in the review, the precondition to successful implementation of ALE programs is, amongst others, a relevant curriculum with appropriate teaching methods, quality training for staff, favorable time and environment for learning, careful planning of manpower and human resources, adequate funding, and enhanced participant access to the programs (Ishaq and Ali, 2012; PHMC, 2011; Oxenham, 2008). However, these factors have been identified without due regard for an efficient institutional system. This dissertation claims that

implementing a relevant curriculum and building an efficient institutional system are equally important foundations of a successful literacy program.

The review also reports that effectively implemented ALE programs lead to measurable changes in knowledge, skills and attitudes of individual adult participants (outcomes) as well as impact on family, community and society. ALE scholars have studied outcomes and impacts (empowerment gains) from ALE through observation of behavior as well as self-report with indicators for political, economic, social, cultural and psychological dimensions. However, many found that measuring the contribution of ALE to empowerment gains is challenging, for example due to the difficulty of separating the contributions of ALE from prior formal/non-formal education. This dissertation agrees that empowerment gains in terms of knowledge and practice can also be obtained from other independent informal and non-formal sources and seeks to explore the contribution of, for example, local leaders, agricultural extension workers, and the local media.

According to the reviews, ALE enhances empowerment through the improvement of selfconcept, self-esteem, self-confidence, participatory practice, knowledge and practice in politics, social integration, family cohesion, and livelihoods improvement. However, impacts and outcomes or empowerment gains vary from place to place, partly because of differences in the literacy model followed and the focus of the program as related to contextual factors. Based on the review of the literacy models, improvement of individual abilities for development (functional literacy), use of critical thinking and problem-solving skills for individual and social transformation (critical literacy), and use of active learning methods in teaching and learning (social practices) are understood as vital factors to analyze IFALE for empowerment and SD in Ethiopia.

The discussion of the review may suggest that there is a direct link between literacy, empowerment and SD although it is hard to prove. Literacy for empowerment and SD is founded on the pillars of education, develops the capabilities of learners, and improves their livelihoods. It must be positioned within the ESD frame, or embrace the eight major features of ESD. The empirical studies also help to identify the most relevant factors, indicators, dimensions and thinking on adult literacy and its role for development at different levels of analysis.

The review is the basis for the development of the analytical model that guides the conduct of the study on the implementation and contributions of IFALE in the two regions of Ethiopia,

Tigray and Amhara. This model is presented at the end of Chapter 4. The model rests on the understanding that there may be a causal connection amongst adult literacy education, empowerment and SD. This assumption is supported by the theory of evidence for use (causation) that is first presented in Chapter 4.

Chapter 4 Causation and Evidence-based Policy as the Foundation for the Analytical Model of the Study

In this chapter, an analytical model is constructed which will guide the analysis of the links between IFALE, empowerment and SD in the selected regions in Ethiopia. As discussed in Chapter 3, there is general agreement that literacy and adult education contributes positively to a range of developments at different levels. But it is generally difficult to prove cause and effect and to isolate literacy or adult education as a variable.

In order to attempt to do so in this study, understandings of causality and evidence-based policy are first introduced. These understandings help identify the important factors and dimensions for successful implementation of adult literacy education programs in the analytical model. They concern two main areas. One is a relevant curriculum which is also pertinent in previous literature (e.g. Ishaq & Ali, 2012; MoE, 2008). The other is building an efficient institutional system which is less stressed in previous literature although apparent in the NAES (MoE, 2008).

4.1 Understandings of Causation

Philosophers have discussed the issue of causation for centuries without reaching an agreement on what it is (Wood, 2005; Lau & Chan, 2008; Scaffer, 2016). Many agree that there are causes and effects in the world but often use other terms to discuss them, e.g. source or reason for cause, and result or implication for effect. They also often use the terms role or contribution of particular causes when discussing their relation or connection with the effects (Schaffer, 2016; Cartwright & Hardie, 2012). The search for causes is important for two reasons: to explain and understand how or why things are the way they are, or behave and change in certain ways; and, to intervene in the course of nature or events and bring about effects or prevent others from occurring.

The discussion of causation is also apparent in the field of education. As Kvernbekk (2015, p. 56) claims, 'causation is an important ingredient in education, because causes make their effects happen....there is a focus on change and how it can be brought about, something

which surely lies at the heart of education...'. In Kvernbekk's understanding, school climate, educational leadership, high expectations of learners' achievements, clearly expressed goals, and assessment are frequently discussed causes that contribute to school effectiveness.

For the purpose of the analysis in this dissertation, the following notions from causation theory are particularly important: *support factors*, *causal cakes* and *their ingredients* (Cartwright & Hardie, 2012, pp. 61-75) and the *INUS condition* (an *insufficient but necessary* part of an *u*nnecessary but *sufficient* condition for a result) (Mackie, 1965). These are specific, practical and helpful for a more achievable policy and strategy.

4.2 The Notions of Support Factors, Causal Cakes and Their Ingredients

To discuss evidence-based policy, Cartwright and Hardie (2012, pp. 62-63) relate *support factors, causal cakes* and *their ingredients* to the metaphor of a cake. Cartwright and Hardie argue that homework can contribute to improved reading scores. In order to ensure a causal effect of introducing a policy of homework, the notion of a cake is introduced. The causal cake in Figure 4.1 identifies support factors (or ingredients) that are necessary for the policy of homework to have an effect on reading scores, e.g. student ability, study space, student motivation, homework, consistent lessons, supportive family, work feedback, and others.



Figure 4.1: A causal cake for improved reading scores

The policy of homework is, therefore, only one ingredient in the cake. It plays its causal part together with the other ingredients (support factors) to produce a contribution to the outcome

(improved reading scores). This implies that in the absence of the support factors, there is unlikely to be a significant contribution. Moreover, the presence of the whole cake cannot determine the value of the effect (reading scores) because it is likely to be only one of many to have an impact in a given situation (Cartwright & Hardie, 2012, p. 63).

The *INUS* condition is a related philosophical notion. Mackie (1965, p. 245) explains the cause of a fire in a house using fire experts' investigation about its causes. In his claim, a short-circuit and other positive and negative conditions, such as the presence of inflammable material, the absence of a suitably placed sprinkler, and others, create a complex condition that is sufficient for a house to catch fire: it is sufficient, yet not necessary, since other conditions could have produced the fire. Under this condition the short-circuit plays the decisive role (is necessary) since without the short-circuit, a combination of the other parts of this condition would not have produced the fire. Based on this, Mackie (1965) defines the *INUS condition* as follows:

In this case, then, the so-called cause is, and is known to be, an *insufficient but n*ecessary part of a condition which is itself *u*nnecessary but *sufficient* for the result. The experts are saying, in effect, that the short-circuit is a condition of this sort, that it occurred, that the other conditions which conjoined with it form a sufficient condition were also present, and that no other sufficient condition of the house catching fire was present on this occasion. I suggest that when we speak of the cause of some particular event, it is often a condition of this sort that we have in mind. In view of the importance of conditions of this sort in our knowledge of and talk about causation, it will be convenient to have a short name for them: let us call such a condition (from the initial letters of the words italicized above), an *INUS* condition (p. 245).

Cartwright and Hardie (2012) argue that the application of the notions of *support factors*, *causal cakes and their ingredients*, and the *INUS condition* can lead to better results from policies.

4.3 A Policy for Better Results

Cartwright and Hardie (2012, p. 61) argue that evidence-based policy is more likely to be successful than Randomized Controlled Trials (RCTs) which policy makers most commonly privileged as the best evidence for causal claims. RCTs can be used to identify a causal relation for a particular population in a study, but further assumptions are needed to extrapolate a causal relationship to another population in another study. RCTs show that a policy has produced a result 'somewhere', in one specific context, but does not guarantee that the same result can be produced 'here', in another context. In contrast, evidence-based policy is based on evidence for use, i.e. on an understanding of the kind of knowledge that can lead

to reliable predictions of whether policies will work as intended (Cartwright & Hardie, 2012, p. 61). There are two preconditions in the theory of evidence for policy effectiveness: the need for a 'causal model' to evaluate whether a policy will be effective for a targeted outcome; and an understanding of causes as *INUS* conditions (Cartwright & Stegenga, 2011, p. 308). The theory, therefore, implies that moving from the fact that 'a policy worked somewhere' to 'it will work here' requires critical thinking about support factors. This means identifying contextual support factors that must be in place for a policy to produce a contribution or have an impact.

4.4 Understanding the Causation of IFALE, Empowerment and SD in Ethiopia

In this dissertation, the assumption is that the policy of IFALE will be more productive in terms of empowering learners and supporting SD if it is based on the theory of evidence for use. The causal connection implies that the IFALE policy contributes to the empowerment of learners, and this effect, in turn, supports SD (Figure 4.2).

IFALE Policy ——>Learners' Empowerment ——> SD

Figure 4.2: Causal connection of the IFALE policy to SD

For the IFALE policy to bring about the expected result, its designers (policy makers) must analyze the context thoroughly, identify the key factors that support its improvement, work out the relations among factors, identify strategies, and ensure commitment for successful implementation of the IFALE policy. Policy makers should have adequate knowledge to design locally conditioned policies. A locally conditioned deficient policy will either fail or make less contribution to the expected outcome.

Cartwright and Hardie (2012, p. 64) emphasize that policy thinking is complex and needs to be implemented in consideration of the support factors, i.e. a causal cake that makes a contribution, and of other cakes that may make contributions. Identifying support factors and other cakes implies a *horizontal search*. The horizontal search takes into account whether the support factors in the population of the previous study are also found in the target population. But,

For the most part, though, predicting what will actually happen is way beyond your means, since you generally will have little knowledge of what other cakes will be operating. So mostly our advice is geared to helping you predict not what will actually

happen all told, but at least whether your proposed policy will make things better (Cartwright & Hardie, 2012, p. 64).

This implies that analyzing and applying a policy based on local facts is not foolproof. Nevertheless, a policy designed wisely can lead to better results than one that is not. In view of this, support factors are identified in this study that might enable the IFALE policy to generate its expected contribution to empowerment and SD in Ethiopia, and ways of implementing the policy are suggested. An effectively implemented IFALE is more likely to lead to empowerment of learners and the support of SD.

4.5 The Causation of IFALE

What does it take to have an IFALE policy in Ethiopia that can lead to learner empowerment? To answer this question, factors needed for effective implementation of adult literacy education have been adapted and contextualized from effective literacy programs in the literature, the key characteristics of exemplary adult literacy programs, and the models of literacy reviewed in the previous chapter. The success of IFALE is influenced by the political will at national and regional level, their relative autonomy, and the socio-cultural, economic and other conditions that vary from one region to another.

4.5.1 Strategy of a Relevant Curriculum

One common factor is the strategy of a relevant curriculum which was also announced by the MoE (2008, p. 14). In order for this strategy to play a causal role, it requires relevant support factors. In this study, factors that characterize model literacy programs in the literature (Ishaq & Ali, 2012; PHMC, 2011), features of the social-practices model of literacy (Tett & Clair, 2010; Hamilton et al., 2006), and inputs related to quality and relevant adult education stated in NAES (MoE, 2008) have been adapted and contextualized within the framework of evidence-based policy thinking.

The support factors for the strategy of a relevant curriculum are: supply of primers and additional materials; qualified officials, experts and facilitators; use of active learning methods; learner motivation and access to IFALE; adequate learning aids/facilities; appropriate environment and time for learning; and others. The factors are displayed in Figure 4.3 in accordance with Cartwright's and Hardie's (2012) philosophical notions of the cake and its ingredients or support factors.



Figure 4.3: A cake for IFALE in Ethiopia: curriculum and support factors

In Figure 4.3, the strategy of a relevant curriculum combines with the support factors to produce a contribution to the implementation of IFALE. Excluding one of the ingredients will lead to deficiency in the implementation of IFALE. The lesser the attention to any of the support factors, the lower the contribution of the strategy would be.

The strategy of a relevant curriculum must be put into practice taking a range of issues into consideration. The Adult and Non-formal Education Coordination Department in the MoE has already produced a country level or National IFALE Curriculum Framework that includes contents from the development packages of the country. However, each region should consider the local needs and conditions of the learners and produce relevant primers and other support materials in the local language of the learner. Officials, experts and facilitators should be qualified. They must continually receive relevant training to enhance their qualifications in accordance with the IFALE program. Each facilitator requires training at least in understanding the particularities of teaching adult learners and the way they can best learn literacy, the contents, and the use of learner-centered methods for facilitation of adult literacy classes. Each IFALE official and expert should receive capacity building training on issues of adult and non-formal education, including literacy education, literacy education organization and management, and ways of strengthening collaboration with stakeholders.

During the facilitation and learning process, facilitators must use active learning (or learnercentered) methods, such as group discussions, demonstrations, case studies, reflections and educational games, as well as receive follow-up training to strengthen this. With respect to the practice of learner motivation, the local adult literacy education officials at various levels, local leaders and community organizations must work hand-in-hand to convince learners to join IFALE, and particularly to make it accessible to rural adults. The knowledge, skills and commitment of the facilitators, as revealed during the facilitation process and the planning of the literacy classes, would influence the learners' motivation to learn. The wereda officials and facilitators—in collaboration with the local administration, women's association and other pertinent local social organizations—must encourage rural adults to participate in IFALE, for example by discussing its values, and by opening new IFALE centers near potential learners.

In order to support the strategy of a relevant curriculum, IFALE officials in the regional education bureaus and the wereda education offices must provide good, relevant and adequate facilities, such as chalkboards, charts and pictures, to the IFALE centers. Rural adult literacy classes can take place in nearby primary schools either early in the morning or in the evening before and after the primary school classes, rather than at noon and under trees where there is no shelter or benches. This would contribute to an appropriate learning environment and lesson delivery time. Learners must be fully involved in decisions of where and when literacy classes take place.

4.5.2 Strategy of an Efficient Institutional System

Another requirement for the effective implementation of IFALE in Ethiopia is the strategy of building an efficient institutional system in the country (Figure 4.4). This depends on strong leadership, motivated officials, experts and facilitators, a clear policy direction, shared understanding of IFALE (for example of its vision, values, principles, and objectives), collaboration among stakeholders, good working conditions, and an adequate budget. These factors are contextualized in this dissertation in accordance with exemplary literacy programs (Ishaq & Ali, 2012; PHMC, 2011), and the strategic directions in NAES (MoE, 2008) and other relevant Ministry documents (MoE, 2010a, 2010b, 2010c, 2012).

An efficient institutional system should address adult and non-formal education from the federal to the center level and have an effective monitoring, evaluation and reporting system. However, this will not result in the expected results from IFALE unless activities related to the support factors are undertaken.



Figure 4.4: A cake for improved IFALE: institutional system and support factors

Strong leadership is key to an effective adult literacy program. The program needs capable leadership (officials and experts) whose capacity is built over time and who rely on an enabling environment. There must be a clear strategic direction for implementing IFALE (why it is necessary, how it can be implemented, and what it is going to do). All major stakeholders of IFALE, including the Ministries of Agriculture, Health, and Women's Affairs, should be involved in the implementation.

The duties and responsibilities of the collaborators from the federal level to their employees at the IFALE centers should be defined according to IFALE tasks. The leaders and experts of IFALE as well as the various stakeholders should develop a common understanding of IFALE through training and consultation. Working conditions must be democratic, transparent and comfortable, and continuously improved. An efficient institutional system depends on a reasonable budget to cover all necessary costs. Good working conditions and a reasonable budget will contribute positively to the motivation of all involved. When the strategic direction, shared values, collaborative action, and reasonable running costs and remuneration exist, officials at all levels, experts and facilitators are likely to be well motivated to implement IFALE effectively.

The strategies of a relevant curriculum and an efficient institutional system permit a separate analysis of how productive IFALE teaching and learning is and how the IFALE organizational set up works. Each is essential for the success of IFALE. Therefore, the two strategies with their identified support factors have guided the development of the research instruments used for data collection regarding the implementation of the IFALE policy in Ethiopia (refer Chapter 5).

4.6 The Causation of Learner Empowerment and SD

Mackie's definition of the INUS condition as employed by Cartwright and Hardie (2012, pp. 63-64) is relevant to analyze the IFALE policy in relation to the causal cake for improving learner empowerment (Figure 4.5). In this case the IFALE policy is an INUS condition, namely an Insufficient part to produce a contribution (improved learner empowerment and SD) as it requires other ingredients (support factors), such as agriculture and rural development, health, women's affairs, youth and sports, and labor and social affairs and other sectors. These sectors have signed a memorandum of understanding to support adult education in Ethiopia (MoE, 2010a, p. 3). Without these support factors, the contribution of IFALE to learner empowerment may be very small, but when combined the expected outcome could be met. However, the IFALE policy must exist as a Necessary ingredient. It and its corresponding cake is, however, Unnecessary as regards a contribution to the effect (learner empowerment and eventually SD) since other cakes with other policies and support factors, for example training farmers in Farmers' Training Centers (FTCs), can play a role in learner empowerment and SD. At the same time, the IFALE policy and its support factors are Sufficient to generate a contribution to the effect of learner empowerment and SD in their own right.

IFALE should help rural adult learners acquire the basic skills of reading, writing and calculating, as well as life skills (MoE, 2010c). In the literature and in Ministry documents, the knowledge and skills relevant to daily life of the learners covers a range of issues on participation in politics; community development activities; gender equality; family cohesion; family planning, personal hygiene and child and maternal care; HIV/AIDS control; environmental protection; agricultural technology use; social justice; democratic culture; conflict resolution; harmful traditional practices; and self-concept and self-efficacy (Burchfield et al., 2002a; 2002b; Carion et al., 1989; Kagitcibasi et al., 2005; Vasumathi, 2008; Stromquist, 2009; MoE, 2010c; TBE, 2010; ABE, 2011a, b; ABE 2012a, b).



Figure 4.5: A cake for the IFALE policy and support factors for learner empowerment and SD

IFALE should also enable learners to apply their knowledge and skills in their everyday life. Other support factors, such as the Ministries of agriculture and rural development, health, women's affairs, youth and social affairs should also contribute to equipping the learners with knowledge and practices required for empowerment and SD.



Figure 4.6: A cake for learners' empowerment and SD: FTCs and support factors

However, IFALE is not alone in contributing to learner empowerment and SD since other causal cakes and their ingredients might have an influence as well. The policy of training farmers in FTCs is an imperative *INUS* condition (Figure 4.6). Amongst the support factors are: self-directed learning efforts of farmers (personal motivation and initiative to learn new

things), local media sources (especially radio, TV, and newspapers), local administration leaders, and farmers' and women's associations.

Unless policy makers explore and support the most relevant policies and also recognize informal ways of learning, e.g. indigenous knowledge from local culture, elders, parents and friends, the IFALE policy may be inadequate in terms of the expected effect on empowerment and SD in Ethiopia. Learning must include all forms of learning and be life-long.

The two strategies with their support factors are brought together in the analytical model presented in Figure 4.7.

4.7 Analytical Model for the Study

In Figure 4.7 the causal chain and pathways (or direction) of IFALE, empowerment and SD are displayed. The model underlies the research design and guides the analysis of the data on the implementation of IFALE in Tigray and Amhara regions. It focuses on the two main strategies of a relevant curriculum and an efficient institutional system with their support factors. The support factors for each of the two main strategies have been identified based on the review of successful literacy programs and the Ethiopian context. These will be used to examine the relative success of the IFALE policy for learner empowerment and SD in the selected regions understood under the influence of other external factors that have been identified through contextual analysis. The IFALE policy is understood to have its own support factors as expressed in the nature of collaboration across different ministries.

In the dissertation, empowerment refers to a set of feelings, knowledge, skills and practices that enable learners to participate in their social environment and thereby lead to social transformation. This is in accordance with Stromquist (1995; 2009) and supported by Olaleye and Adeyemo (2012). While Stromquist identified cognitive, political, economic, and psychological dimensions of empowerment, this study excludes the cognitive dimension but includes the socio-cultural dimension. The dimensions are understood as follows in the dissertation:

Political empowerment refers to IFALE participants' involvement in institutions and policies at the community or national level in accordance with Stromquist (2009). The indicators have been adapted from Burchfield et al. (2002a; 2002b) and Carion et al. (1989) and include knowledge of voting age and political parties, participation in voting and decision making in public institutions.

Economic empowerment denotes the contribution of IFALE to individuals' capacity to support themselves economically, have an improved life and make autonomous decisions. This is based on the definition by Stromquist (2009). The indicators have been adapted from Stromquist (2009), Cameron and Cameron (2005), and the development packages of Ethiopia embraced in the IFALE Curriculum Framework (MoE, 2010c). They include: access to credit and income generation schemes, financial contributions to productive investment and/or mutual support, as well as importance and practices of agricultural technologies, environmental protection and saving.



Figure 4.7: A model of IFALE causal pathways for learner empowerment and SD

Socio-cultural empowerment represents an individual's ability to enhance social capital (such as maintaining good health and living longer, learning throughout life, controlling reproductive behavior, raising healthy children, and addressing gender equality), develop and promote new values as well as transform other values, attitudes and behaviors. This definition is based on UNESCO (2005c) and Vasumathi (2008, pp. 121-122, p. 129). The indicators of socio-cultural empowerment have been adapted from Kagitcibasi et al. (2005), Burchfield et al. (2002a; 2002b), Farah (2005), UNESCO (2005c), Vasumathi (2008) and the development packages of Ethiopia included in the IFALE Curriculum Framework (MoE, 2010c). They include family planning, gender equality, HIV/AIDS prevention, maternal and child care, value of children, modern practices, and democratic culture.

The psychological dimension has been adapted from Stromquist (2009) as the learners' feeling of being competent, worthy of better conditions as well as capacity to take action. The indicators of self-concept and self-efficacy knowledge and practices have been adapted from Kagitcibasi et al. (2005) and Stromquist (2009). *Self-concept* is a general term used to refer to how someone thinks about, evaluates or perceives him/herself. To be aware of oneself is to have a concept of oneself. According to Baumeister (1999), *self-concept* is 'the individual's belief about himself or herself, including the person's attributes and who and what the self is'. *Self-efficacy* refers to belief in one's general capacity to handle tasks or succeed in tasks.

In the dissertation, IFALE's contribution to SD through empowerment is understood based on its approach to teaching and learning ideals and principles underlying SD. These include fundamental issues, such as livelihood improvement, environmental and natural resource conservation, gender equality, inclusive societies and social cohesion. These are understood as interlinked and rest on a holistic approach to teaching and learning. The definition of IFALE for SD has been adapted from the conceptualization of UNESCO (2006b, 2007), Huckle (2008), Ndoye and Walther (2011), and Hoppers and Yekhlef (2012). The analysis of IFALE investigates the extent to which IFALE fulfills the main features of ESD in the two selected regions. Indicators to explore IFALE's role for empowerment also serve as indicators for SD since the domains of empowerment (political, economic, socio-cultural and psychological) are analyzed in relation to the four components of SD, namely economy, society, environment and culture. In addition to the factors for learner empowerment, other external factors, for example agricultural and rural development, health, and energy policies contribute to the SD of a country. However, these external factors are not of concern in this dissertation.

The model displayed in Figure 4.7 helps to explore what support factors might be missing, be partially at work, or adequately implemented in Tigray and Amhara. Each list of factors in the model also helps to understand whether the IFALE policy is evidence-based. An evidence-based policy must be rooted in local conditions, implemented with proper follow up, and integrated with other relevant policies to bring about the expected results. How the research was designed and methods selected to collect the data for investigating IFALE's contributions to empowerment and SD is discussed in Chapter 5.

Chapter 5 Research Design and Methodology

In this chapter, the underlying philosophical assumptions or worldview informing the research design, analysis and interpretation of the study is presented. The chapter also outlines the methodologies for collecting and processing the data. It brings up a range of methodological and ethical issues related to the research process in order to establish the validity and reliability of the research findings.

5.1 Realism as the Epistemological Stance

Realism is the epistemological stance of the study because of its affinity with causation and evidence-based policy. For Phillips (1987, p. 205), realism is 'the view that entities exist independently of being perceived, or independently of our theories about them'. Schwandt (1997, p. 133) defines scientific realism as 'the view that theories refer to real features of the world. 'Reality' here refers to whatever it is in the universe (i.e., forces, structures, and so on) that causes the phenomena we perceive with our senses'.

Chakravartty (2015, p. 1) has a more comprehensive definition: 'Scientific realism is a positive epistemic attitude towards the content of our best theories and models, recommending belief in both observable and unobservable aspects of the world described by the sciences.' In brief, in realism, several entities (both the observable and the unobservable) described by theories certainly exist. An adult learner, a textbook, IFALE centers, and regions in a country are real entities and so are literacy, empowerment, and SD.

Antirealists do not accept the existence of an unobservable (intangible) aspect: the only real entity for them is what is observable (can be seen or touched). Antirealism is an epistemic attitude that includes any position against one or more of the three dimensions of realism. These dimensions include 'the metaphysical commitment to the existence of a mind-independent reality; the semantic commitment to interpret theories literally or at face value; and the epistemological commitment to regard theories as constituting knowledge of both observables and unobservables' (Chakravartty, 2015, p. 10). According to the realists, both observable and unobservable entities represent scientific truth, whereas for antirealists they are not. In this study, theories represent instruments for intellectual thinking and knowledge production which combine observable and unobservable entities.

5.2 Comparative Case Study and Study Locations

The study is a comparative case study of IFALE policy implementation and its contribution to empowerment and SD in rural areas of two regions of Ethiopia: Tigray and Amhara (Figure 5.1). A case study is an in-depth study of one or more instances of a phenomenon in its real life context that reflects the perspective of the participants involved in the phenomenon (Gall, Gall, & Borg, 2007). For Anderson (1998), case studies principally deal with how and why things occur and, thereby, provide an opportunity to explore the realities about an event, a process, or a problem relating to the case. Under a similar condition, there can be two or more cases for comparison. The most common geographical level of analysis in educational comparative studies is a country; however, world regions/continents, states/provinces/regions, districts, schools/institutions and classrooms can also be compared (Bray & Thomas, 1995).



Figure 5.1: Study locations in Tigray and Amhara

Ethiopia is a very large country that has much variation between regions in various spheres. Of the nine regional states and two city administrations in Ethiopia, the neighboring Tigray and Amhara regions were selected purposefully for comparison. This is partly because the source of funding for the study is the joint Collaboration Project of the University of Life Sciences (Ås) in Norway, and Mekelle and Hawassa Universities in Ethiopia that focuses on

Source: http://www.ethiodemographyandhealth.org/Amhara.html; http://www.tigraytube.com/article_read.php? a=1; https://commons.wikimedia.org/wiki/File:Ethiopia_regions_english.png (Images copied 26 January 2017)

northern Ethiopia. More importantly, Tigray and Amhara are two of only four regions (together with Oromia, and Southern Nations, Nationalities and Peoples regions) representing the four parties founding the EPRDF, the front that is currently leading the country. The four regions have stronger capacity than the other regions to implement the development endeavors of the various sectors, including the education sector. Thus, the study was likely to produce more solid results if conducted in any of these four regions. Oromia and Southern Nations, Nationalities and Peoples region could not be selected because of the restrictions on the use of the funding. Furthermore, since regions have their own separate administrations, the importance attached to and the understanding and implementation of IFALE strongly depends on the local and regional context. The two selected regions could, therefore, potentially differ despite their proximity and similarities.

CSA (2013) projected the population of Tigray and Amhara to be 5,247,005 and 21,134,988 respectively in 2016¹⁵. The Tigray Regional State (Tigray) in the northern part of Ethiopia had a population of 4,314,456 in 2007. The majority (80.5%) was agrarian of whom 50.8% were females (Federal Democratic Republic of Ethiopia Population Census Commission, 2008). Tigray has seven zones. The study was done in the Eastern Zone and its two corresponding weredas (Saesea-Tseada Emba and Kilte-Awlaelo) (Figure 5.1).

The Amhara Regional State (Amhara) is located in the northwestern part of Ethiopia. Based on the 2007 census, the region had a population of 17,214,056 almost equally divided between women and men (8,636,875 males and 8,577,181 females). The urban population was 2,112,220 people or 12.6% of the total. The region comprises 11 administrative zones. The study was done in Northern Wello and its two weredas (Meqhet and Wadla).

Historically, the two selected regions are interesting since political power has been concentrated until recently with the Amhara elite and in the capital of Addis Ababa, whereas Tigray led the political opposition movement that is now in power. Most Amhara elites have been proponents of a unitary government while the government in power has implemented federalism in the country on an ethno-linguistic basis.

¹⁵ The latest census was conducted in 2007.

S.No.	Item	Tigray	Amhara
1	Location*	Northern region of Ethiopia	Northwestern region of Ethiopia
2	Population projection for 2016*	5,247,005 (2,587,003 male; 2,660,002 Female)	21,134,988 (10,585,995 Male; 10,548,993 Female)
3	Zones*	Western, Northwestern, Central, Eastern, Southeastern, Southern and Mekelle Special Zone (7 zones)	Waghemira, North Wollo, North Gondar, South Gondar, South Wollo, North Shewa, Oromia, East Gojjam, West Gojjam, Awi and Bahir-Dar Special Zone (11 zones)
4	Adult literacy rate in 2014*	51% (68% male; 34% female)	40% (54% male; 25% female)
5	Household heads with higher education in 2011**	0.03 %	0.2%
6	Linguistic background and religious beliefs*	Semitic and Cushitic people; dominantly orthodox believers; dominant language is Tigrigna	Semitic and Cushitic people; dominantly orthodox believers; dominant language is Amharic
7	General livelihoods*	Small scale farming dependent on rain; poor and vulnerable rural people that supplement their food shortages from safety net programs	Small scale farming dependent on rain; poor and vulnerable rural people that supplement their food shortages from safety- net programs
8	Household heads working as farmers/family farm workers***	90%	89%
9	Total area transferred to commercial agriculture***	5%	7%
10	Improved seed user households***	18%	34%
11	Engagement in soil conservation by household***	85%	88%
12	Average number of livestock in a household***	3	3
13	Household heads with thatched roof houses***	39%	40%
14	Household heads with floor houses made of earth, and cow dung mixed with soil***	99%	100%
15	Households that own radio/TV***	26%	20%
16	Annual revenue of household from the sale of livestock ***	1001 birr (44 USD)	1053 birr (46 USD)
17	Annual revenue of a household from the sale of chicken***	30.6 birr (1 USD)	24.9 birr (1 USD)
18	Percentage of households with access to a vehicle during the rainy season***	39%	31%
19	Percentage of households with access to a vehicle during the dry season***	59%	51%
20	Distance to the nearest daily market***	23 km	22 km
21	Proportion of households with access to electricity***	29%	23%
22	Proportion of households with access to cell phone***	69%	80%
23	Proportion of households with access to tap water***	55%	39%
24	Proportion of households with access to radio broadcast***	98%	98%
25	Proportion of households with nearby primary schools***	42%	47%
26	Average distance to the nearest government clinic/health post***	12 km	10 km

Source: *CSA (2013); **Oakland Institute (2011) as cited in Bekele, et al. (2015); ***AGP Baseline Survey 2011 (Berhane et al., 2011)

Note: Items 4, 8, 10-26 are rounded figures.

While Amhara has had a relatively peaceful existence throughout history, Tigray was a war zone during 1965-1991 and again during the Ethio-Eritrean war during 1998-2000. This led to air-raids, destruction, migration and displacement and severely affected educational

opportunity. It also led to strong support for the Tigray Peoples' Liberation Front (TPLF) which together with four other freedom movements formed the EPRDF that ousted the socialist military regime ruling during 1965-1991. With the coming to power of the EPRDF, many non-Tigrayan elites consider that Tigray has benefited disproportionally from the dominance of the TPLF in the country's politics, economy, security and military (OPride, 2015).

Each of the two regions has its own political autonomy and is also exposed to different sociocultural and economic conditions which may potentially affect the implementation and contribution of IFALE in the respective region. The literacy rate is comparatively higher for the lesser populated region of Tigray despite the fact that Amhara hosts more highly educated people and that educational opportunities have traditionally been more limited in Tigray because of the unrest and wars affecting the region (Table 5.1). Each region has a distinct mother tongue and language of instruction, namely Tigrigna and Amharic, despite sharing the same alphabet (Ge'ez). In terms of services, more people in Tigray own a radio/TV, and have more access to electricity, tap water, and transport than is the case in Amhara. On the other hand, Amhara has a higher proportion of households with access to a primary school and cell phones, as well as shorter distances to the nearest government clinic/health post. More households in Amhara than in Tigray participate in soil conservation and improved seed use, and receive a higher annual income from livestock. Although Amhara has a higher total land area for commercial agriculture than Tigray, the two regions are largely similar in terms of farm workers, average number of livestock, and annual revenue from the sale of chicken per household. They are also similar in terms of roof and floor quality of their houses, and distance to the nearest daily market.

5.3 Exploratory Sequential Mixed Methods Design

In order to conduct the study, an exploratory sequential mixed methods design was chosen. This means that qualitative data were explored before complementary quantitative data were gathered to explain relationships in the qualitative data (Creswell, 2012). According to Creswell and Clark (2007, p. 5):

Mixed methods research is a research design with philosophical assumptions as well as methods of inquiry. As a methodology, it involves philosophical assumptions that guide the direction of the collection and analysis and the mixture of qualitative and quantitative approaches in many phases of the research process. As a method, it focuses on collecting, analyzing, and mixing both quantitative and qualitative data in a single study or series of studies. Its central premise is that the use of quantitative and
qualitative approaches, in combination, provides a better understanding of research problems than either approach alone.

There are three main reasons for the choice of mixed methods in this dissertation. Firstly, it was necessary to have qualitative information or stories about the implementation of IFALE and the way IFALE embraces SD, as well as numbers (quantitative data) to analyze the role IFALE plays in learners' empowerment and support of SD. Creswell (2012, p. 35) states that the use of stories and numbers in a study presents 'an alternative perspective in a study'. Secondly, since interviews, observations and document analysis were insufficient to answer all the research questions, a survey was done to address the knowledge and practice of empowerment. The survey was developed in consideration of the provisional analysis from the qualitative data in order to extend, elaborate on and explain these data, and to discover causal relationships (Creswell, 2012; Greene, Caracelli, & Graham, 1989; King, 1994; Robson, 2011). Thirdly, the use of mixed methods permits triangulation, and integrating and embedding the data to understand the research issues fully (Roseman & Wilson, 1985; Creswell, 2012).

In the exploratory sequential design, the researcher's role can vary depending on the method of data collection. In this study, knowledge construction through interaction and collaboration with the participants was an integral part of the research process in order to understand the participants' experiences. I, the researcher, was an insider in the sense of being an Ethiopian from Tigray. However, I was an outsider as a student of the University of Oslo. According to Dwyer and Buckle (2009, p. 61) the researcher 'occupies the space between, with the costs and benefits this status affords'. During the qualitative data collection, I had direct contact with the participants. Their stories were immediate and real, their voices were not lost numbers and their understandings came out in the transcriptions of the data (Dwyer & Buckle, 2009, p. 61). During the survey, there was limited contact with the respondents since other interviewers had the direct contact with and provided the necessary background information to the respondents.

5.4 Sampling and Instruments of Data Collection

There were two phases of data collection. During the first phase (April-May 2013), qualitative data were collected based on interviews, observation protocols and checklists, and field notes were taken about the status of IFALE. Important documents were examined regarding the implementation of IFALE and its potential impact, in particular the ETP, the NAES, the

IFALE Curriculum Framework, the Master Plan for Adult Education, the facilitators' manuals and the primers.

The second phase took place five months later (November 2013) and concerned quantitative data. A survey was prepared based on the general results of the qualitative data. According to Creswell (2012, p. 543), 'a researcher uses the exploratory design when existing instruments, variables, and measures may not be known or available for the population under study'.

Type of sample	Selection
Purposive sampling	Tigray and Amhara
	91 interviewees representing the lower (weredas education offices and IFALE centers), middle (regional bureaus and zonal education offices) and upper (MoE) levels of the IFALE system
	2 zones (Eastern Zone in Tigray; Northern Wello in Amhara)
	4 weredas (2 in Eastern Zone of Tigray; 2 in Northern-Wello of Amhara)
Purposive random sample	16 centers (8 of 10 in Saesea-Tsada Emba and Kilte-Awlaelo weredas of Tigray; 8 of 11 in Meqhet and Wadla weredas of Amhara)
Simple random sample	184 of 619 IFALE participants in 16 centers that graduated from the program in May 2013 (Tigray 91; Amhara 93); 194 of 793 non-IFALE participants that registered to attend IFALE in the 16 centers in the academic year 2013/14 (Tigray 97; Amhara 97) However, 2 IFALE participants from Tigray could not attend the survey

 Table 5.2: Sampling procedures for study area and population

After having selected the study area, 8 of the 10 literacy centers in Tigray and 8 of the 11 centers in Amhara were randomly purposefully sampled (Table 5.2). In both regions, the sample areas were assumed to be 'information rich' (Patton (1990, p. 169) and credible (Gall et al., 2007, p. 183). A purposive sampling procedure was used to recruit interviewees representing upper (MoE), middle (regional bureaus and zonal education offices) and lower (wereda education offices and IFALE centers) levels of the IFALE system. Ninety-one interviewees involved in planning and executing IFALE were purposefully selected. The categories of interviewees are further discussed below.

5.4.1 Semi-structured Interviews

In all 91 semi-structured interviews were conducted with officials at the different levels of administration as well as the supervisors, coordinators, facilitators, learners and collaborating institutions, such as NGOs and ministries other than the MoE (Table 5.3). Following an interview protocol (Appendix A) that was partly adapted from Rogers and Kramer (2008), the interviews aimed at examining how the interviewees understood the implementation and contributions of IFALE in terms of their own experiences, expectations, motivations and gains. Data from the interviews also helped identify a range of issues that the subsequent survey addressed in terms of the contribution of IFALE to learners' empowerment and SD.

Table 5.3: Category and number of interviewees

Category	Number	Total
Policy maker	2 Males	2
Federal expert	3 Males	3
Bureau head/Deputy and regional official	3 Males (1 Head of bureau, 1 curriculum and implementation official in Tigray, 1 Deputy Head of Bureau in Amhara)	3
Zonal official	2 Males (1 from each region)	2
Zonal IFALE expert/Focal person	3 Males (2 experts in Amhara, 1 focal person in Tigray)	3
Wereda official	4 Males (2 per region)	4
Wereda IFALE expert	2 Males (Amhara; no IFALE expert in Tigray)	2
Supervisor (including IFALE)	4 Males (2 Tigray, 2 Amhara)	4
School principal (IFALE coordinators)	13 Males (5 Tigray, 8 Amhara)	13
Facilitator	11 Males, 5 Females (8 in each region; 4 females from Tigray)	16
Learner	16 Females, 16 Males (half of each gender per region)	32
Federal NGO	3 Males (1 local, 2 international)	3
Collaborating bureau	1 Female, 1 Male (Amhara*)	2
Regional NGO	1 Male (Tigray)	1
Collaborating ministry	1 Male (Ministry of Health)	1
	Grand total	91

*There are no collaborating bureaus (sectors) for IFALE in Tigray

Procedures to record the information from the field followed the six steps developed by Creswell (2009, p. 12) for analyzing and interpreting qualitative data: collecting data; preparing data for analysis; reading through the data, coding the data; coding the text for description; and coding the text for themes. The interview responses from the participants were audiotaped and notes were taken as backup. The interviews were organized and stored based on the administrative level and function of the interviewees: policy makers, regional officials, wereda officials, center coordinators, facilitators, and adult learners. The researcher and other professionals transcribed and translated the interview data from Tigrigna/Amharic into English. The data was coded manually based on themes. This required self-reflection in order to minimize bias. The thick description obtained from the interviews was an essential component of the study.

5.4.2 Observations

Observations served to examine how IFALE was implemented with respect to the strategies of a relevant curriculum and building an efficient institutional system. There were three kinds of observations based on observation protocols: observations of 16 IFALE sessions (8 in each region); observations of 16 IFALE centers, and observations of 4 IFALE participants' practices in their households (2 from Tigray and 2 from Amhara) (Table 5.4; Appendices B, C, D). The purpose of the observations was, in accordance with Creswell & Clark, 2011, to record detailed information on how the teaching and learning took place, what the learning environment was like, and how the IFALE learners applied their learning experiences in daily

life. The data from the observations, pictures and other visual materials, and reflective notes based on the information were organized according to the category of observation made.

Instrument	Region	Number	Total
Session observation	Tigray	8	
	Amhara	8	16
Center observation	Tigray	8	
	Amhara	8	16
Observation of individual IFALE	Tigray	2 (1 Female, 1 Male)	
participants	Amhara	2 (1 Female, 1 Male)	4

Table 5.4: Center and participant observation by region and number

The session observations focused on the objective(s) of the lesson, lesson contents, learner motivation, the role of the learners and facilitators, the time allotted for a session, the kinds of active or learner centered methods used in class, time distribution amongst facilitators and learners, and learner interaction amongst themselves and with facilitators. This gave a clear picture of classroom dynamics and how teaching and learning takes place. The observation form was filled in both by the researcher and by an experienced professional in adult education who served as a research assistant and who received training beforehand. After each session, agreement was reached on one observation form which was signed based on thorough discussion of the individual observation protocols.

The purpose of the center observation protocol was to examine the enrollment figures on the learners and the suitability of the learning environment. Information on enrollment was obtained from official records in the centers while information on the learning conditions were based on the observations.

The observation of IFALE participant practices in their own homes covered housekeeping, use of latrines, compost preparation, and mastery of other life skills taught in the program. The observation data were recorded in words and pictures.

5.4.3 **Survey**

The study examined the contribution of IFALE to learners' empowerment and SD based on survey participants' self-reports. The survey was distributed to a total of 376 individuals in the two regions of whom 182 were IFALE participants and 194 were non-IFALE participants. IFALE participants were the treatment group while non-participants were the comparison group (Table 5.5).

According to exploratory interviews with education officials, experts, facilitators and IFALE learners (Appendix A), IFALE participants in the two regions (Tigray 89; Amhara 93) had no

prior primary school or adult literacy education. According to the interviews, participants registered for the program after having been targeted by the IFALE center and the local administration. They enrolled in the program partly out of fear of being fined or of being dismissed from the safety-net program. Thus, self-selection seems not to have influenced their enrollment in the program. According to the same interviews, non-IFALE participants (Tigray 97; Amhara 97) were newly registered learners who also had no previous exposure to primary education or adult literacy education and who were also targeted by the IFALE center and the local administration.

Region	Number	Total		
	IFALE participants:	186		
	49 Females, 40 Males=89			
Tigray	Non-participants:			
	55 Females, 42 Males=97			
	IFALE participants:	190		
	29 Females, 64 Males=93			
Amhara	Non-participants:			
	49 Females, 48 Males=97			
	Grand Total			

Table 5.5: Number of survey participants in Tigray and Amhara

The survey participants in each of the 16 centers (8 in Tigray; 8 in Amhara) were selected through simple random sampling based on official lists from the centers. Simple random sampling ensured that each individual had an equal chance of being selected (Tashakkori & Teddlie, 2010). Based on Yemane's formula¹⁶ (1967, p. 886), the sample amounted to 153 IFALE participants (out of 619) and 162 non-participants (out of 793). Another 20% (31 participants and 32 non-participants) were added to compensate for individuals that could potentially not be contacted. This gave a final total of 184 participants and 194 nonparticipants. The IFALE participants (Tigray 91; Amhara 93) were selected from the total IFALE participant population size that completed the program in May 2013 (619 IFALE graduates) through simple random sampling. The 194 non-IFALE participants (97 from Tigray; 97 from Amhara) were also selected through simple random sampling from the total population of 793 who were registered as new entrants to the IFALE program in 2013/14 and, therefore, had not been exposed to the program at the time of collecting the data. Both groups

16 n= Ν

 $1+N(e)^{2}$

$$1- 019 -155.4$$

 $1+619 (0 07)^2$

The sample size of the comparative group (non-IFALE participants) that were registered to join the IFALE in 2013/14 was: n= $\frac{793}{1+793(0.07)^2}$ = 162.3 = 162 non-IFALE participants

Where n is the sample size, N is the population size, e is the level of precision, and 1 is the probability of the event Occurring. Based on the formula, the sample size of the IFALE participant population was: =153.4 =153 IFALE participants. 610

of participants were called to the nearby IFALE centers to join the survey. In all, 182 IFALE participants and all non-participants responded to the survey, giving a response rate of 99.4%.

The survey used a 5-point Likert rating scale to record the views of participants on the importance and practice of or their interest in particular areas. Participants provided answers to questions according to scales ranging from *very important* to *totally unimportant*, *very good* to *very poor*, *very high* to *very low*, and *always* to *not at all*. They also provided *yes/no* answers for factual information.

The survey addresses the empowerment dimensions, i.e. the political, economic, sociocultural, and psychological dimensions. Specific indicators were developed for each domain as displayed in Table 5.6. Twelve qualified interviewers were trained in recording the answers from the survey participants. The survey was in Tigrigna and Amharic and pilot-tested for quality assurance. The survey instrument was distributed to each literacy center in the two regions using transport from Mekelle University. The costs of distribution were covered by the research fund from the Institutional Collaboration between Mekelle and Hawassa Universities, and the Norwegian University of Life Sciences.

S.				
No	Domain	Indicator	Measure	Value
		Importance of participation in voting	5-Likert Scale: Very important	Mean
			Totally unimportant*	
		Practice of participation in voting	5-Likert Scale: Very goodVery poor**	Mean
		Interest in becoming a local election representative	5-Likert Scale: Very highVery low***	Mean
		Participation in decision making in public institutions	5-Likert Scale: Very highVery low	Mean
		Knows the minimum voting age	Yes/No	Percent
		Can name a political party/ political parties	Yes/No	Percent
		Do you think that you could become a local representative in an election at any level?	Yes/No	Percent
		Do you know a law on electing representatives (men/women)?	Yes/No	Percent
1	Political	Are you a member of a community organization?	Yes/No	Percent
		Do you know any community/village development committees in your area?	Yes/No	Percent
		Did you participate in community development activities over the last two years?	Yes/No	Percent
		Importance of agricultural technologies	5-Likert Scale:	Mean
			Very importantTotally unimportant	
		Practice of agricultural technologies	5-Likert Scale: Very goodVery poor	Mean
		Importance of measures of environmental protection	5-Likert Scale:	Mean
			Very importantTotally unimportant	
		Practices of measures of environmental protection	5-Likert Scale: Very goodVery poor	Mean
2	Economic	Importance of participation in saving	5-Likert Scale:	Mean
			Very importantTotally unimportant	
		Practice of participation in saving	5-Likert Scale: Very goodVery poor	Mean
		Did you contribute money to an organization for	Yes/No	Percent
		productive investment and/or mutual support over the		
		last two years?		
		Do you have access to credit for additional income	Yes/No	Percent
		generation in your community?		

Table 5.6: Domains of empowerment by indicator, measure and value

S.				
No	Domain	Indicator	Measure	Value
		Do you think that males and females contribute	Yes/No	Percent
		equally to development?		
		Do you know whether there are laws against gender	Yes/No	Percent
		discrimination?		
		Do you know of laws protecting women and children?	Yes/No	Percent
		Importance of family cohesion	5-Likert Scale:	Mean
			Very important Totally unimportant	
		Practice of family cohesion	5-Likert Scale: Very highVery low	Mean
		Importance of harmful traditional activities	5-Likert Scale:	Mean
			Very important Totally unimportant	
		Use of harmful traditional activities		Mean
			sometimes, rarely, not at all	
		Importance of reasons for wanting an additional child	5-Likert Scale:	
			Very important Totally unimportant	
		Practice of the reasons for wanting an additional child	5-Likert Scale: Very highVery low	Mean
		Importance of family planning	5-Likert Scale :	Mean
			Very important Totally unimportant	
3	Socio-	Practice of family planning	5-Likert Scale: Very goodVery poor	Mean
	Cultural		5-Likert Scale:	Mean
		Importance of personal hygiene	Very importantTotally unimportant	
		Practice of personal hygiene	5-Likert Scale: Very goodVery poor	Mean
		Importance of maternal and child care	5-Likert Scale:	Mean
			Very importantTotally unimportant	16
		Practice of maternal and child care	5-Likert Scale: Very goodVery poor 5-Likert Scale	Mean
		Importance of HIV/AIDS control measures		Mean
			Very important Totally unimportant 5-Likert Scale: Very good Very poor	Maria
		Practice of HIV-AIDS control measures	5-Likert Scale: Very good Very poor 5-Likert Scale:	Mean Mean
		Importance of social justice activities		Mean
		Practice of social justice activities	Very important Totally unimportant 5-Likert Scale: Very goodVery poor	Mean
		Importance of democratic culture	5-Likert Scale: Very good Very poor	Mean
		importance of democratic culture	Very important Totally unimportant	Ivicali
		Practice of democratic culture	5-Likert Scale: Very goodVery poor	Mean
		Importance of conflict resolution methods	5-Likert Scale:	Mean
		importance of conflict resolution methods	Very importantTotally unimportant	Ivicali
		Practice of conflict resolution methods	5-Likert Scale: Very goodVery poor	Mean
		Importance of self-concept	5-Likert Scale:	Mean
		importance of sen-concept	Very importantTotally unimportant	wican
		Practice of self-concept	5-Likert Scale: Very goodVery poor	Mean
4	Psycholog-		5-Likert Scale: Very good Very poor	Mean
'	ical	Importance of self-efficacy	Very important Totally unimportant	wicall
		Practice of self-efficacy	5-Likert Scale: Very goodVery poor	Mean
		I TACHEC OF SETT-ETHEACY	J-LINGIT SCALE. VELY BOOUVELY POOL	IVICAII

Note: *Very important (5), Important (4), Moderately important (3), Not important (2), Totally unimportant (1); **Very good (5), Good (4), Average (3), Poor (2), Very poor (1); ***Very high (5), High (4), Average (3), Low (2), Very low (1)

Note: The mean value of the each italicized indicator of empowerment was calculated from the scales of a single item, while the mean values of each unitalicized sub-theme (indicator) were calculated from the scales of two or several items considered together.

After the data collection, a code was given to each survey and all data was entered into STATA Version 14 software, edited and cleaned. Afterwards, the 5-point Likert scale self-reports of knowledge and practice, the *yes* and *no* responses, as well as the frequency responses to survey participants' sources of knowledge and practice were analyzed. Descriptive statistics (including mean, frequencies and percent) and inferential statistics (t-test and chi-square) were important data analysis tools for the self-reports. The survey data led to a deeper understanding of the likely contribution of IFALE to empowerment and SD in Ethiopia.

5.4.4 Document Analysis

The study also included systematic analysis of policy documents (ETP and NAES), the National IFALE Curriculum Framework, the IFALE facilitators' manuals, primers, as well as different kinds of field records (Table 5.7).

Туре	Number	Total
Education and Training Policy (ETP); National Adult Education Strategy (NAES); National	3	
IFALE Curriculum Framework		
Tigray IFALE Facilitator Manual	1	
Amhara IFALE Facilitator Manual	2	
Tigray IFALE Learner Textbook	0	0
Amhara IFALE Learner Textbook	2	8
IFALE related field notes, letters, pictures collected from relevant offices (MoE, Regional	n/a	n/a
Bureaus, Zonal and Wereda Education offices, Collaboration Sectors, i.e. Agriculture and		
Rural Development, and Health, and IFALE Centers)		

Table 5.7: Sources for document analysis by type and number

Document analysis is a qualitative method used to give voice, discover meaning and insights, develop understanding, and produce knowledge about a particular theme (Merriam, 1998; Bowen, 2009). The document analysis was used to explore how the government was planning to implement IFALE to contribute to sustainable development through learner empowerment.

5.5 Validity and Reliability

Throughout the process of conducting the research, issues of validity and reliability were considered. Validation is a major concern in mixed methods studies because of the difficulty to conceptualize, report and discuss it (Creswell and Clark, 2007; Venkatesh, Brown, and Bala, 2013). Teddlie and Tashakkori (2003, 2009, 2010) and Tashakkori and Teddlie (2008) have developed two important concepts for the assessment of validity in mixed methods research: *inference quality* and *data quality*. For Venkatesh et al. (2013), inference quality refers to interpretations and conclusions from mixed methods research, while data quality is the degree to which collected data (results of measurement or observation) meet the standards of validity in terms of *trustworthiness* and *reliability* or *dependability*. Inference quality includes both *design quality* and *interpretive rigor* (Teddlie and Tashakkori, 2009). According to Teddlie and Tashakkori (2009), *design quality* concerns whether mixed methods research follows commonly accepted best practices, while *interpretive rigor* is about the standards for judging the accuracy or authenticity of the conclusion. These are discussed below in the context of the dissertation.

As has appeared above, inference quality addresses issues of internal validity. *Internal validity* concerns the degree to which a study makes good inferences about a causal

relationship between two variables (Trochim, Donnelly, & Arora, 2015). This was applied in the current study in different ways. For the survey, a quasi-experimental design with randomly selected IFALE participants (treatment) and non-participants (comparison) were used in order to find answers to the research questions on IFALE's contributions to empowerment and SD in Tigray and Amhara regions. The IFALE participants (sampled literacy graduates from the IFALE centers' official lists) and non-IFALE participants (sampled non-literate individuals from the official lists of the IFALE centers who were ready to enroll in the IFALE program) were asked about their prior and current sources of knowledge and practice related to the major indicators of empowerment. None mentioned primary or other education. The data collected from the two groups allowed for causal inferences since, as illiterate small-scale farmers, they had a similar background, economic status, and access to other opportunities, but differed in terms of exposure to IFALE.

Internal validity was also addressed using construct validity and statistical validity. *Construct validity* has been ensured through the definition of the important concepts of adult literacy education, empowerment, and SD, and through the development of an analytical model based on the theory of evidence for use, literacy models, and the concepts of empowerment and SD. According to Trochim (2006), construct validity refers to the extent to which legitimate inferences can be made from the operationalization of the theoretical constructs that serve as a foundation. In this study, the theoretical constructs were translated into operational definitions and key indicators in relation to adult literacy education. For example, the political, economic, socio-cultural and psychological dimensions of empowerment relate to adult literacy education and each has specific indicators. Indirect measures or self-reports (verbal reports) by the survey participants were used as data sources for investigating IFALE's role in empowerment and SD. During the survey data collection, each participant had a detailed explanation of how to give true responses in order to avoid 'social-desirability bias'; i.e. that survey participants answer the questions to portray themselves as socially acceptable.

Statistical validity was achieved through adequate sample size and appropriate statistical data. According to Trochim et al. (2015), statistical validity refers to the extent to which conclusions drawn from a statistical test are accurate and reliable. For the quantitative data collected through the survey, two-sample t-test with equal variance was appropriate. The t-test compared the mean values of the IFALE participant group and non-participant group attached to scales of self-reports. The chi-square compared the frequency of answers to *yes* and *no* questions.

In terms of *credibility* concerning interpretations and inferences, clear explanations of all the stages of the study have been provided (who the participants were, how they were selected, how the different data sets were analyzed, and how the conclusions were drawn). The exploratory mixed design was appropriate to and consistent with the research questions. The viewpoints of the interview participants were interpreted carefully. The methods helped to capture meanings, associations, and effects with regard to the implementation and contributions of IFALE focusing on the comparison of Tigray and Amhara regions. After careful interpretation of data from the various sources, the results were compared not only within the study but also with results from previous research.

Confirmability of findings was guaranteed through triangulation in accordance with Duffy (1987). According to Torrance (2012), and Bekhet and Zauszniewski (2012), triangulation attempts to validate research findings by generating and comparing different data types and different perspectives of respondents on the phenomenon under examination. The study ensured validity by comparing the findings obtained from different respondents (e.g. policy makers, collaborators, officials, facilitators, IFALE participants and non-participants) and different sources (e.g. interviews, observations, document analysis, and survey) which helps to understand complex problems.

Transferability refers to the degree to which the findings of the research and/or initially developed concepts can be applicable to other situations and populations (Bitsch 2005; Tobin & Begley, 2004). It requires caution because of the need to consider contextual factors (Hammersley, Gomm & Foster, 2000). This study on the implementation and contributions of IFALE in Ethiopia is important in its own right but is also part of the wider research on ALE. This makes cautious transferability possible. Policy makers in Ethiopia may consider the design and outcomes of the research in the context of future policy making and implementation in the country. ALE researchers may utilize the knowledge and concepts developed in this study, for example the identification of support factors, the specific understanding of empowerment (with its dimensions and indicators), to investigate the contribution of ALE in other contexts.

In order to ensure reliability, utmost efforts have been made to explain clearly the different research processes and phases, including its rationale, the design, and the nature of the data collected. Overall, this permits inferences to be made and conclusions to be drawn.

5.6 Ethical Issues

All research needs to consider a range of ethical issues (Creswell, 2012). Ethical conduct of research was followed before the start of the fieldwork (Cosby, 2012; Creswell, 2012), in the field (Creswell, 2012; Silverman, 2006) and during data analysis, write up and dissemination (Creswell, 2012; Schwandt, 2007; Maykut & Morehouse, 1994). The study is registered with the Norwegian Social Science Data Services (Norsk Samfunnsvitenskapelig Datjeneste AS-NSD) (Appendix F) and was conducted with a research permit from the University of Oslo and Mekelle University (Appendices G and H). The responsible officials at the different levels of the education system gave their consent to the conduct of the research corresponding to what Creswell (2012) suggests: a researcher needs to consult with different gatekeepers at multiple levels of an organization.

The participants of the study agreed to provide information and the researcher and the interviewers explained clearly and fully the purpose of the study and the implications of a free and 'informed consent'. By obtaining permissions and clearly communicating the purpose of a study, the researcher can better make participants accept his/her presence in their own setting (Creswell, 2012). Each participant submitted an informed consent in writing (Appendix I). They were informed that all information would remain confidential and would only be used for the research purpose. During the fieldwork, utmost efforts were made to ensure that the participant involvement was as easy and pleasant as possible and that the classroom observations caused the least possible disruption to the ongoing activities. All sources of information were given codes to preserve anonymity and all information that may affect the confidentiality of the sources will be destroyed after the approval of the dissertation.

Ethical issues were also addressed during analysis and reporting. First, the data was analyzed systematically and fairly to minimize bias. The research uses language recommended in the American Psychological Association (APA) style (APA, 2010) and scholarly terms from a qualitative dictionary of terms (Schwandt, 2007). In addition, attention was paid to avoid personal bias and to voice the feelings and meanings of the participants in the qualitative data interpretations. According to Maykut and Morehouse (1994), the perspective of qualitative researchers is perhaps paradoxical: it is reflect the experiences and meaning systems of others while at the same time recognizing one's own biases and preconceptions. Equally important, utmost effort was made to employ appropriate language and research formalities since

encoding the language of research, in the view of Creswell (2012), will enhance the acceptability of the research to graduate committees, publishers and conference attendees.

Verbal or written feedback was given to all participants who requested it and reporting was done in an ethical and honest way in accordance with Creswell (2012). Copies of the dissertation will be distributed to the MoE, the regional bureaus, wereda education offices and IFALE centers that formed part of the research in order to share the results.

5.7 Conclusion

Framed within realism, the study is an exploratory, sequential comparative case study of two regions in Ethiopia, Tigray and Amhara, using a mixed methods design. It is based on extensive interviews with relevant officials and participants in different organizational settings. It also includes observations of individual participants at their homes and of the teaching and learning process in the centers. Furthermore, the potential effect of IFALE on participants—compared with non-participants—was analyzed through data from a survey designed to investigate the issues of empowerment and SD. Analysis of the curriculum materials and other documents on IFALE provided evidence on SD features in the program.

The outline of the research process has served as a basis to judge the scientific value of the study related, amongst others, to issues of reliability and validity of the data and adherence to ethical principles for the conduct of research. The huge amounts of data gathered and the complexities of the issues addressed in the research made all steps of the process challenging. The results of the data analysis appear in the following three chapters, starting with an analysis of the core strategies for the implementation of IFALE.

Chapter 6 IFALE and the Implementation of a Relevant Curriculum and an Efficient Institutional System

The analysis in this chapter of the strategies and support factors related to a relevant curriculum and an efficient institutional system is undertaken within an evidence-based policy frame. The factors for analysis are displayed in Figure 4.7 but are here organized slightly differently. They are understood as important for successful implementation of IFALE. The analysis is the foundation for the analysis of IFALE's contribution to empowerment of learners and SD in Tigray and Amhara in the next chapters.

The analysis is based on the contents of a range of ministerial policy documents, primers and facilitator manuals supplemented with information provided by policy makers, learners, facilitators and others on the relevance of the curriculum and on institutional factors. The analysis, therefore, combines the results of document analysis and relevant data from interviews, observations and the survey undertaken in 2013. It aims at highlighting similarities and differences between Amhara and Tigray regions in terms of understandings and implementation of IFALE.

Of the support factors displayed in Figure 4.7, the ones that have contextual relevance to the study are analyzed below. These include for the strategy of a relevant curriculum: primers and other materials; qualifications of officials, experts and facilitators; learning time and use of active learning methods; learner motivation and accessibility to IFALE; and learning environment. For building an institutional system, they include: policy direction; understanding of IFALE; collaboration among stakeholders; working conditions; and budget.

6.1 Implementing the Strategy of a Relevant Curriculum

6.1.1 Understanding of a Relevant Curriculum

According to the MoE (2008, p. 2), adult education programs should be problem solving and relevant to the everyday life of the learners who are directly involved in production activities. Besides, the curriculum should be based on various development packages. According to the

IFALE Curriculum Framework (MoE, 2010c, p. 3), the contents should relate to the life of the learners, improve their health and income, and eventually support the achievement of SD. The success of the strategy depends on the primers and other materials, well-trained officials, experts and facilitators, and active learning methods.

According to the interviews, regional officials and policy makers in the MoE believe that IFALE is relevant to the learners. As stated by one policy maker:

....the execution of a program starts from designing a strategy. This means, the current ALE program is a tool to ensure the continuity of the development strategies of the country. First, the education provision should relate to the life of the learner. Teaching to read, write and calculate should not be the only focus of the program. Second, through adult education we can build a literate and skilled workforce that can implement the development strategies and policies of the country. The IFALE material preparations carried out at the regional level originate from the growth and transformation plan of the country. The learning materials have also been prepared by involving various stakeholders, who are directly engaged in the developmental activities. It is prepared based on the livelihood and environment of the learners. This means, the agriculture sector is involved in ALE. The health sector, under ALE, will work to ensure a healthy environment and society. This is why the health minister, agriculture and rural development minister, women and social affairs minister are working as members of the technical committee and board. All of the stakeholders have a specific part in the program.

....the relevance of education derives from the needs of the adult learners. It should also focus on the development strategy of adult learners. ...We decided that the materials should build on the common interest of the learner and the development strategies of the country. This is how it is delivered. The program is provided in the first language of the learners....When we see the methodology, as it is stated in the implementation guideline, facilitators carry out the learning process based on the principles of andragogy and the interest of the participants. Adult learners already have some knowledge when they come to join the program. Thus, the program considers their interest, bearing their psychology in mind, respecting the culture and tradition of the area. This is not a teaching learning process; it is rather learning from each other. Unlike the traditional method, learners come up with the topic, discuss and solve problems (policy maker interviewee 2, April 2013).

According to the interview, the preparation of the NAES, the National IFALE Curriculum Framework and the IFALE teaching materials should take into consideration the interests of the learners and the relevant stakeholders. The curricula should include teaching of the most important development packages in addition to basic literacy and numeracy and traditional teaching methodologies should be replaced by active learning methods. According to the survey results, respondents in Tigray rated the relevance of IFALE to their day-to-day realities and needs of the community as nearly very good (mean=4.67) while in Amhara it was found to be almost good (mean=4.18) (Appendix E, survey question 82.1). The analysis

of primers and facilitator manuals showed that contents from the government's development packages which are believed to improve livelihoods are indeed included, for example beekeeping, dairy farming, fruit and vegetable production, and poultry (refer 8.1.1). However, teaching practices in the literacy centers during observations showed that traditional lecture, rather than participatory methods were predominant and that learners were in practice not encouraged to reflect their ideas and propose topics for learning.

6.1.2 Primers and Other Materials

In terms of curriculum materials, there were differences between the two regions. In Tigray it was restricted to the preparation of the facilitator manual in the Tigrigna language, while in Amhara both primers and facilitator manuals were developed in Amharic. In addition to the lack of primers in Tigray, it appeared from one of the observations undertaken in center G that IFALE learners had no other supporting materials as well, such as exercise books and pens. The facilitator instead distributed paper for the learners to write notes. By contrast, in Amhara learners had primers, exercise books, and pens or pencils. This is likely to make IFALE classes in Amhara more effective than in Tigray and may contribute to empowering learners beyond the mastery of basic literacy and numeracy.

6.1.3 Qualification of Officials, Experts and Facilitators

In terms of qualifications, interviews with bureau officials showed that there are clear differences between the two regions. The head of the Tigray Bureau of Education (TBE) and his deputy have no background in (adult) education and no one was assigned to specifically administer IFALE neither in the education bureau nor in the wereda education offices. Coordination and management of IFALE was instead an additional duty carried out by officials or experts employed in the formal education system at different levels.

In Amhara, there were IFALE positions in the regional education bureau and in the weredas. This helped to plan, execute, monitor and evaluate IFALE more effectively. Both the head and the deputy of the Amhara Bureau of Education (ABE) had master's degrees in relevant fields of education. The bureau has an IFALE program team consisting of three members with master's degrees in related education fields—of which one is in adult and lifelong learning. In addition, each of the weredas in Amhara had at least two IFALE experts. One of the two selected weredas had three IFALE experts, one with a bachelor's degree in education, who coordinated the other two members that were pursuing their diploma level training in adult education through the summer in-service program.

A wereda IFALE expert from Amhara stated:

The Curriculum, Assessment and Evaluation Division in the education office of the wereda is led by one coordinator. Within this division, three of seven staff members are adult literacy experts. One has a bachelor degree in psychology and the two of us are pursuing our education in adult education in the diploma program. The government has given attention to adult literacy. We put this into action, considering the adult literacy education as one program (expert interview 1 in Amhara, May 2013).

This, however, also implies that some IFALE officials and experts in the Amhara region and weredas had no specialized training in adult education.

In terms of the qualification of facilitators there were more similarities than differences between the two regions since almost all were recruited for their jobs after they had completed grade 10. They received specialized training in teaching adults focusing on ways of motivating people to join the IFALE program, adult learners and their behavior, lesson planning, and methods of teaching. The length of this training varied somewhat in the two regions. In Amhara, six of the eight facilitators had received 10 days of training (in Flaqhit town by Meqhet Wereda IFALE experts). In Tigray, 7 out of 8 facilitators had received three weeks to a month's training (one week training by supervisors and other officials in the weredas and two to three weeks at Abbiyi-Addi Teacher Training College).¹⁷

The qualifications of the IFALE teachers are somewhat inadequate in both regions, but more so in Tigray than Amhara. In Amhara, primary school teachers do the actual teaching in the IFALE sessions even if they are trained in teaching children. They are occasionally supplemented with selected agricultural and health experts. The facilitator and the primary school principal coordinate the IFALE program in each center and the facilitator invites primary school teachers and agricultural and health extension workers to teach IFALE learners. A wereda IFALE expert from the region said:

The role of the facilitator is to facilitate. Based on the choice of the adult learners, the facilitator invites a particular professional and arranges class schedules.... After the lesson is over, the facilitator may fill a gap by covering some points or explain things to latecomers (IFALE expert 1, Amhara, May 2013).

The facilitator from the same region said:

I work as a facilitator, not as a teacher. I do not teach. School teachers, health and agriculture experts should take turns teaching. I attend the session but I only teach in the absence of the invited person. The facilitator's job is to check learners' attendance and invite teachers (facilitator interviewee 3, Amhara, May 2013).

¹⁷ One of the new facilitators had completed grade 12 and had received no training in teaching adults.

The facilitator's main duty is to coordinate the program, such as setting the time plan, identifying and inviting relevant professionals, and providing supplementary teaching to the learners.

In contrast, in Tigray the facilitator conducts the teaching in each session, besides inviting professionals from other sectors. It appeared from the observations and interviews with the facilitators and school principals that professionals from other sectors never actually came to teach despite the invitations. In the absence of the facilitator, it is school teachers who teach the IFALE sessions.

6.1.4 Use of Active Learning Methods

The analysis of active learning methods includes time allotment for each session and the teaching and learning methods. According to the observation results, the plan to teach 2 hours per session and three times a week (6 hours per week) was not followed in any of the regions, although the sessions in Amhara were closer to the planned time than in Tigray. The use of active learning methods was limited in both regions.

Methods of teaching and	IFALE centers in Tigray (centers A-H) IFALE centers in Amhara (centers A-H) and time used for the method used for the method															
learning	Α	B	С	D	E	F	G	Η	Α	B	С	D	Е	F	G	H
Brainstorming	3'	0	10'	7'	0	15'	0	0	0	0	0	0	5'	0	0	0
Group discussion	40'	0	0	0	0	0	8'	0	4'	0	5'	3'	0	7'	15'	0
Pair work	0	0	0	0	5'	0	0	0	0	0	0	0	0	3'	0	0
Gapped lecture	0	0	30'	10'	35'	25'	10'	0	25'	0	0	35'	30'	0	0	0
Lecture (only teacher talking)	0	37'	0	0	0	0	0	37'	0	35'	55'*	0	20'	25'	18'*	35'
Question and answer	0	2'	30'	30'	0	25'	50'	3'	25'	10'	10'	50'	30'	0	26'	2'
Independent work	0	0	0	13'	30'	40'	0	0	0	0	0	5'	20'	4'	0	0
Problem solving	0	0	0	0	0	0	0	0	15'	0	0	0	0	0	0	0
Reflection	0	2'	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Total session time in each center	43'	41'	1hr & 10'	1hr	1hr & 10'	1 hr & 45'	1hr & 8'	40'	1hr & 9'	45'	1hr & 10'	1hr & 33'	1 hr & 45'	39'	59'	37'

Table 6.1: Time allotted by method in IFALE classes in Tigray and Amhara by center

Note: *The facilitator read aloud from the textbook and learners listened

Source: Session Observations (April & May 2013)

In Tigray, the session time varied from 43 minutes (center A), to one hour (center D), and to 1 hour 45 minutes (center F). In Amhara, the time ranged from 1 hour 45 minutes (center E), to 1 hour 9 minutes (center A) to 37 minutes (center H) (Table 6.1). The predominant methods

were question and answer, gapped lecture¹⁸, independent work and group discussions. The traditional lecture method was also widely used, especially in Amhara (Table 6.1).

Despite the clearly stated direction for a multi-method approach, classroom observation results show that this is not the case. As appears in Table 6.1, facilitators or teachers generally used the same kinds of methods which did not include most of the active learning methods, for example buzz group, visits, role play, case study, demonstration, simulation, presentation, debate, cooperative work, matching exercises, drawing and picture, picture analysis, pyramiding,¹⁹ experiment, experience sharing, drama and game. Furthermore, the primers in Amhara and the facilitator manuals in both regions do not underline the variety of methods/approaches but mostly recommend group discussion. This may affect the capacity of learners for independent thinking and reflection.

6.1.5 Learner Motivation and Access to IFALE

As to learner motivation, statistical data on enrollment and drop out collected at the centers in both regions indicate a high dropout rate over two years (2011/12 and 2012/13), namely 36% in Tigray and 28% in Amhara (Table 6.2). The dropout rate in Tigray is below the universally acceptable level reported by Cameron and Cameron (2005), i.e. 25-30% while that of Amhara is within this acceptable range.

Table 6.2: Enrollment, completion, drop out and dropout rate of IFALE learners in Tigray and Amhara, 2011/12-2012/13

Region	Enrollment 2011/12	Completion 2012/13	Drop outs (2011/12 and 2012/13)	Dropout rate, %
Tigray	554	354	200	36
Amhara	366	265	101	28
Total	920	619	301	33

Source: Records in the Centers (April & May 2013)

Session observations in Tigray showed that only half of the learners enrolled in year II of the IFALE program actually attended the observed sessions. According to interview and lesson observations results, learners did not participate in decisions about why, how, where and when classes should be conducted. This may explain the low interest of the IFALE learners. In contrast, in Amhara most IFALE learners enrolled in year II attended the session, many learners were highly motivated and had been involved in decisions on the time and place of

¹⁸ Gapped lecture is a type of lecture that is interspersed with other types of activities, e.g. work in pairs.

¹⁹ In pyramiding, each learner in a class first works on an issue for five minutes to collect his/her thoughts. Then the learners work on the issue for five to ten minutes in pairs. After that, they work in small groups and finally discuss the issue as a class. Such activities are often accompanied by detailed instructions and briefings.

the learning, and had access to primers. They indicated that having the primers encouraged them to study outside of the classroom.

In terms of access, there are IFALE centers in most villages of the weredas in both regions and potential learners can join IFALE quite easily. In the survey, IFALE participants in Tigray rated access as *very good* (mean=4.61) while in Amhara it was rated as *good* (mean=4.05). The Educational Statistics Annual Abstracts of the MoE (2011, 2012, 2013, 2015, 2016) show that enrollments in adult and non-formal basic education, including IFALE (15 years and above), has increased over the years in all regions in the country during ESDP IV, 2010/11-2014/15 (Table 6.3).

Table 6.3: Adult and non-formal basic education enrollments during ESDP IV, 2010/11-2014/15

Area	2010/11	2011/12	2012/13	2013/14	2014/15
Tigray	77	93,780	173,972	276,217	504,757
	M*=20	M=54,192	M=95,942	M=143,251	M=234,672
	F**=57	F=39,588	F=78,030	F=132,966	F=270,085
Amhara	1,644	224,651	1,046,623	2,519,998	2,602,896
	M=1,133	M=168,219	M=741,755	M=1,575,365	M=1,619,239
	F=511	F=56,432	F=304,868	F= 944,633	F=983,657
All regions, including	334,410	2,092,234	3,415,776	6,506,310	5,990,409
Tigray and Amhara	M=197,843	M=1,281,951	M=2,116,620	M=3,722,306	M=3,427,604
	F=136,567	F=810,283	F=1,299,156	F=2,784,004	F=2,562,805

Source: MoE Educational Statistics Annual Abstract (September 2011; September 2012; November 2013; June 2015; July 2016)

Key:*M=Male; F**=Female

Enrollment in adult and non-formal basic education increased by 95% in the country as a whole from 2010/11 to 2014/15. During the same period, enrollment in Tigray and Amhara increased by almost 100%. Enrollment in IFALE has led to a rise in the adult literacy rate in Ethiopia from 39% in 2007 to 49% in 2015 (UNESCO, 2015; World Bank, 2016b).

6.1.6 Appropriate Environment and Time for Learning

Lesson observations and interview results indicate that the only learning facility in the IFALE centers of the two regions is the chalkboard. There were no charts, pictures or models or other means that could help enhance learning.

IFALE classes are conducted under different conditions and times in the two regions. Only two of the eight IFALE sessions observed in Tigray took place in a primary school classroom, while two sessions were under trees and four in a $dass^{20}$ that could not properly protect the learners from the sun or the wind (Figure 6.1). In one center (center C), the IFALE sessions

²⁰ A dass is a shelter made of wood and branches of trees used as a classroom for learning. In dass classes, stones are often seats.

were in a *dass* prepared inside the primary school compound even though the school classrooms were free. Two IFALE sessions (in centers B and G) were carried out after the learners finished the day's demanding terracing duty. In general, the teaching and learning in Tigray was not conducted in a conducive environment.



Figure 6.1: IFALE session (Center G) in Tigray conducted under trees after a terracing chore *Source*: Session Observations (April 2013)

In Amhara, IFALE classes were well planned and taught in well-equipped formal primary school classrooms. This was done either before (in the early morning between 6 am and 8 am) or after (late in the afternoon between 5.30 pm and 8 pm) the regular teaching program of the primary schools (Figure 6.2). The learning environment in Amhara was therefore more comfortable for the learners.



Figure 6.2: IFALE session in Amhara (Center A) conducted in a classroom *Source*: Session Observations (May 2013)

6.1.7 Summary

The findings make clear that the strategy of implementing a relevant curriculum was unsuccessful in Tigray, but partly successful in Amhara. Of the support factors for the strategy, only the facilitators' manual and participants' access to the IFALE program (i.e. 2 of 9) were at work. By contrast, in Amhara, most of the support factors (6 of 9) were present, namely provision of primers and facilitators manuals, qualified officials and experts, motivated learners, appropriate environment and time for learning, and access to IFALE. Absent in the strategy were qualified facilitators, use of active teaching and learning methods, and adequate learning aids/facilities.

6.2 The Strategy of Building an Efficient Institutional System

6.2.1 Understanding of an Efficient Institutional System

An efficient institutional system often requires a clear organizational structure (Ingram, 2018; Nedović & Božinović, 2013). In the case of IFALE, the organizational set up runs from the MoE down to the IFALE center. The Coordination Office of the Department of Non-Formal

and Adult Education in the MoE provides leadership, coordination and technical assistance to all levels of the IFALE system. It has 5 experts and 4 support staff. In accordance with the NAES, partner ministries signed a memorandum of understanding to ensure their participation in IFALE. A National Adult Education Board was established in 2008 which in 2012 produced a guideline for the formation of adult education boards and technical committees at regional and wereda levels as well. All regions have received monitoring and reporting templates. Evaluation materials and instruments are based on quality standards and outcomes for IFALE.

6.2.2 Leadership

In contrast to the stipulation in the guidelines of the National Adult Education Board, Tigray has no functioning regional IFALE board or technical committee representing the different sectors that were to be engaged in the implementation of the program. IFALE also has no independent institutional system or separate organizational structure in the regional education bureau and the weredas. This has prevented monitoring and evaluation of the programs although the reporting system is operational at the federal level. Nevertheless, a provisional team from the relevant sectors produced the IFALE facilitator manual in 2010.

In contrast, in Amhara, a regional adult education board headed by the Vice-President of the region and a regional adult education technical committee has been set up and an IFALE organizational system was set up from the region to the center. The existence of this system has facilitated monitoring, evaluation and reporting in the region. In addition, representatives from all relevant sectors prepared the primers and facilitator manuals that were distributed in 2012/13. At the wereda and kebele levels, adult education boards and technical committees were not fully operational until November 2013. This means that there was more inclusion of other partners in IFALE at the regional than at the wereda and kebele levels.

6.2.3 Staff Motivation

The issue of staff motivation was examined based on the perceptions of IFALE officials and school principals during interviews. In Tigray, education personnel in the regional bureau, the weredas and the centers (school teachers and school principals) were not strongly motivated to participate in IFALE activities since it was an additional duty to their work in formal education. They lacked incentives. In contrast, in Amhara experts and officials are fully engaged in implementing, monitoring, evaluating and reporting on IFALE. Their main concern was that the collaborating institutions did not participate as planned in the

implementation of IFALE. In contrast, the voluntary school teachers who conducted IFALE sessions mostly lacked incentives for participating in IFALE.

The facilitators in both regions regarded both their training and their salary as inadequate. In Tigray, they are paid 636 birr (about 23 USD) a month while in Amhara they receive a monthly salary of 500 birr (about 18 USD). This is very low compared to the Ethiopian primary school teacher's monthly salary when first employed, i.e. 1719 birr (about 62.80 USD). This, according to the wereda education officials, leads to a high turnover of facilitators who leave IFALE when they find better paying jobs.

6.2.4 Policy Direction and Understanding of IFALE

There is no clear definition of adult and non-formal education or IFALE in the ETP which was published in 1994 (MoE, 1994). According to the newly introduced NAES (MoE, 2008), the focus of adult education should be on integrated functional adult education and a number of documents have been produced to implement the program effectively.

According to the interviews, officials at the different levels in Tigray had no shared understanding of IFALE. Policy maker interviewee 1 and regional and wereda official interviewees 1 and 2 in Tigray stated that IFALE was important to teach farmers in order to support Ethiopia to become a middle-income country by 2025. Nevertheless, many high-ranking officials could not explain the essence of this. One official in the wereda and two facilitators referred to IFALE as an 'illiteracy eradication program'. This may be because they had not been informed of its meaning, mission, values and principles. Officials and experts generally considered IFALE as an additional duty. In the absence of officials who clearly understand the strategic direction of adult education, IFALE may not adequately prepare rural learners for improved livelihoods or empower them to support SD.

In contrast, education and IFALE officials in the Amhara regional education bureau and wereda education offices explained in detail what IFALE is, its assumptions, values, and principles, why it is needed, and how it could be implemented. For example, one regional education official in Amhara stated:

In this program everything is integrated. All shareholders of the program must be integrated to bring the desired change. The government, society, sector bureaus, and participants have their share of responsibility for the success of the program. The program is integrated in terms of responsibility, content and philosophy. The content is linked with life skills of the learner. As we teach the life skills, reading, writing and numeracy skills are integrated. The owner of the program is the society. Therefore,

every stakeholder should be responsible for the development of the program. There is integration among different sectors, i.e. health, agriculture, small scale business, women's affairs, etc. The principle is that the program is governed by integration. It is also flexible: the participant can learn at any time they prefer and anything they want. Unlike formal education, it does not have opening and closing seasons. The learners can learn at any time based on their interests, be it during the day, night, summer, or winter, or before or after working hours (Regional official interviewee 2, Amhara, May 2013).

Another regional official claims:

...We assume that the development strategies will be internalized by the community. We assume that there is community mobilization and the awareness of people from regional to kebele levels. There is a bid for community mobilization and awareness raising (Regional official interviewee 2, Amhara, May 2013)

As one wereda educational official reports, the main objective of the ALE is to produce farmers who know their environment, are economically independent and healthy, can read and write, and have a modern life style (Wereda official interview 1, Amhara, May 2013).

These are examples of thorough knowledge of IFALE, perhaps because of the officials' background in education and adult education, and the fact that they are full-time employed to run the program. Their knowledge is expected to have a bearing on the provision of the IFALE program.

6.2.5 Collaboration among Stakeholders

Wereda educational officials and IFALE center coordinators in both regions reported that the contribution of other sectors to the implementation of IFALE was not in place in 2013. A wereda educational official from Tigray stated:

IFALE was planned to be implemented through the joint efforts of the education, the agricultural and rural development, health and other sectors, but the sectors do not consider it as part of their duty and responsibility. They have left the duty to the education sector. It is the education officials, the supervisors, the principals of the schools hosting the centers and the facilitators who currently carry out the IFALE activities. None of the other sectors is working together with us in the teaching and learning process of IFALE (Wereda official interviewee 2, Tigray, May 2013).

A regional IFALE team leader from Amhara similarly reported:

Each sector and board members are to share the responsibility. I remember, the chairperson of the board once told each member to prepare a checklist and assess what had been done at the grassroots level and report on it. However, this was not fully done. It is inconsistent. It is the education sector that works responsibly. The board members are not fully committed to take responsibility. There is a gap. Nevertheless, it is a good beginning and should be strengthened (Regional IFALE official interviewee, Amhara, May 2013).

This means that although the curriculum contents have been integrated, the teaching is not done as planned by involving other sectors. In November 2013, the offices of agriculture and rural development as well as health sent letters to the extension workers at grassroots level instructing them to perform their duty of facilitating relevant topics in IFALE sessions (Appendix J and K). This may affect IFALE in the future.

6.2.6 Working Conditions

The working conditions in the two regions differ. In Tigray, the regional and wereda officers do not have enough time to execute IFALE activities because of their main duty in formal education. Besides, the learning environment is not ideal with facilitators teaching in *dass* classes and having to move the chalkboard from their home every time they teach. There are no tables or chairs, charts, pictures or other learning materials. This probably has a negative impact on the quality of learning in IFALE. In Amhara, officials and experts have clear job descriptions and carry out IFALE activities in primary school classrooms with benches, chairs and tables, probably enhancing learning outcomes.

6.2.7 Budget

The MoE has allocated a budget of about 12 billion birr (approximately USD 545 million) for IFALE and a total of USD 20 per learner per year in ESDP IV (MoE, 2010b). This accounts for almost 9% of the total budget for education at the national level and increased significantly from ESDP III when it was only 0.5% of the total education budget.

The rough budget allocated by the government amounts to approximately 3 USD per participant per year in Tigray. Tigray is, thus, severely lacking in terms of the planned annual budget of 20% per participant. This may indicate poor commitment concerning the IFALE program on behalf of the regional government and education bureau. Furthermore, the money distributed by the government was not spent on publication of primers, teaching materials and other important expenses during ESDP IV, but some money was spent on training of facilitators and some officials which, according to the interviews, was insufficient. It was a local NGO, the Relief Society of Tigray (REST), which instead supported the production of the facilitators in the centers.

In Amhara, the government budget amounted to about 16 USD per learner per year which was closer than in Tigray to the government's annual budget plan for each learner, i.e. 20 USD. A budget of about 32 million birr (approximately USD 1.4 million) was allocated for

the preparation and printing of primers and facilitator manuals during ESDP IV. The regional government also covered the salary of all regional and wereda officials and experts, facilitators, and the costs of training the experts for a diploma in adult and non-formal education. The budget allotted for IFALE in Amhara, though not enough to fulfill all the basic requirements, shows higher commitment to take the program forward than in Tigray.

6.2.8 Summary

The strategy of building an efficient institutional system was unsuccessful in Tigray, but partly successful in Amhara. All the support factors of the strategy (8) were practically absent in Tigray while in Amhara all support factors but three were at work. As the results show, strong leadership, motivated officials and experts, good working conditions, adequate budget and shared understanding of IFALE among officials and experts were in place in Amhara while a clear policy frame, motivated facilitators, and collaboration among stakeholders were missing.

6.3 IFALE Participants' Proposed Solutions

According to the survey results, IFALE participants in Tigray and Amhara suggested five important solutions for the problems related to the implementation of IFALE (refer Appendix E, survey question 84). The majority of the IFALE participants in Tigray proposed the following ones: an adequate budget (90%); increased participation of women (80%); more training of facilitators (76%), primers (66%); and overcoming attitudinal problems (self-defeating beliefs) of participants (such as 'I am too old to go to an IFALE center') (47%). IFALE participants in Amhara identified largely the same solutions although with a difference in relative importance: an adequate budget (67%); increased participation of women (62%); primers (61%); training of facilitators (56%); and commitment of leaders of the sector (55%).

Thus, in the views of the IFALE participants in both regions, increasing the IFALE budget is a key factor to improve the implementation of the IFALE policy. The money should be spent on recruiting and training officials/experts and facilitators who play a key role in moving the IFALE policy forward and on material supplies.

6.4 Conclusion

The implementation of the strategies of a relevant curriculum and an efficient institutional system was unsuccessful in Tigray but partly successful in Amhara (Tables 6.4 and 6.5). In

Tigray, most of the support factors for the strategy of a relevant curriculum (7 of 9) were not at work (Table 6.4). The facilitators' manual was prepared but the primer and other supporting materials, and teaching and learning aids were missing. Active learning methods were not in use. The learning environment was often inappropriate. This led to low motivation and dropout amongst the learners. The region has no staff specifically assigned to the program. The bureau head and his deputy have no specific background in education or adult education. IFALE was an additional responsibility to the work in formal education for which particularly the wereda officials and school principals received neither additional training nor appropriate remuneration. In short, the absence of most of the support factors influenced the implementation of IFALE in Tigray.

	Support factors at work					
	Tigray	Amhara				
Support factors	Yes /No	Yes /No				
Primers	X	\checkmark				
Facilitator manual(s)						
Qualified officials & experts	X					
Qualified facilitators	X	X				
Use of active learning methods	X	X				
Learners' motivation	X					
Access to IFALE	\checkmark					
Adequate learning aids/facilities	X	X				
Appropriate environment and time for learning	X					
Implementation of the strategy	Unsuccessful	Partly successful				

Table 6.4: Support factors for the strategy of a relevant curriculum

Key: Yes:√; No:X

In Amhara, there was more attention to the strategy and it was partly successful (Table 6.4). Most of the support factors are at work (6 of 9). The learners had primers and the facilitators had manuals. IFALE centers shared classrooms and other resources with primary schools. The education bureau head and his deputies as well as IFALE officials and experts could lead the program partly because of their background in education although some officials and experts requested further training. IFALE was easily accessible since teaching took place in nearby primary schools. The learners were generally motivated to learn because of the provision of primers, the learning environment and the timing of the sessions. The dropout rate (28%) was lower than in Tigray (36%) and within the internationally suggested acceptable range for dropout, i.e. 25%-30% (Cameron & Cameron, 2005). However, school teachers who act as facilitators need training in adult education and the use of active learning methods.

The implementation of the strategy of building an efficient institutional system was unsuccessful in Tigray but partly successful in Amhara (Table 6.5). NAES and the IFALE Curriculum Framework have a clear strategic direction, but no policy for IFALE. However,

the IFALE officials at regional and wereda levels lack a shared understanding of IFALE in theoretical and practical terms. The regional and wereda IFALE board and technical committees representing all the collaborating sectors were not functioning although a provisional committee representing the relevant sectors had prepared the facilitator's manual. The TBE allocated no budget for textbook preparation, training of facilitators and other important service provisions. The approximate total annual budget for each learner was three USD. In brief, the strategy of building an efficient IFALE institutional system in Tigray was deficient in all respects (0 of 8) which impacted implementation.

	Support factor at work						
	Tigray	Amhara					
Support factors	Yes /No	Yes /No					
Strong leadership roles	Х	\checkmark					
Motivated officials and experts	Х	\checkmark					
Motivated facilitators	X	X					
Clear policy direction	X	X					
Shared understanding of IFALE	Х	\checkmark					
Collaboration among stakeholders	Х	X					
Good working condition	Х	\checkmark					
Adequate budget	Х						
Implementation of the strategy	Unsuccessful	Partly successful					

Table 6.5: Support factors for the strategy of an efficient institutional system

Key: Yes: $\sqrt{}$; No: X.

In Amhara, most of the vital supporting factors for the strategy of building an efficient IFALE institutional system were at work (5 of 8) and the strategy was therefore partly successful. IFALE officials and experts in the regional and wereda educational offices coordinated activities at regional and wereda levels based on NAES but there was no clear adult literacy policy frame. Working conditions were generally good and staff was motivated to execute IFALE activities despite lack of support from the collaborating sectors in teaching and learning and the poor pay for facilitators. IFALE officials and experts had a shared understanding of the essence of adult education and of IFALE. There was a budget for staffing, textbook and facilitator manual preparation, and for training of experts and facilitators. The total annual budget for each learner was approximately 16 USD which is closer to the planned budget (20 USD) than in Tigray. Initiatives to involve collaborators were in progress in November 2013.

It seems clear, therefore, that there are deficiencies in terms of implementation of IFALE in both regions, but more so in Tigray than in Amhara. Whether IFALE nevertheless managed to promote some degree of empowerment of learners is analyzed in Chapter 7.

Chapter 7 IFALE's Contribution to Learner Empowerment

This chapter focuses on the empowerment that the rural-based IFALE learners in Tigray and Amhara may have gained from IFALE. It is based on the self-reports of the survey participants, occasionally supported by interviews with education officials and/or IFALE participants in the two regions. The chapter explores the contributions IFALE may have made to rural adult learners' political, economic, socio-cultural and psychological empowerment by comparing them with rural adults who had not been exposed to IFALE. Conclusions are made based on an analysis of each indicator, and on a consolidated analysis of each empowerment dimension based on all its indicators. In the survey, each indicator is a scale or a yes/no item that has equal value in representing the dimensions of empowerment.

7.1 IFALE and Political Empowerment

As underlined by Stromquist (2009), the political dimension of empowerment addresses adult learners' capacity to involve themselves in local institutions and policy-making in their communities and to help change institutions and policies. In this study, the political dimension of empowerment refers to the IFALE participants' knowledge and practice of participation in political affairs and local institutions or citizenship education gained from IFALE. The indicators of political empowerment include: knowledge of importance and practice of participation in voting during elections; knowledge of voting age, political parties and the national policy on electing representatives; membership in a union/community development organization; engagement in community self-help initiatives; and community level decision-making.

7.1.1 Importance and Practice of Participation in Voting

According to Sharma (2016), good citizens are supposed to vote during elections. In line with this, participation in an election is part of the IFALE curriculum as will be discussed later (refer 8.1.1). The responses of the survey participants to the importance and their practice of participating in voting appear in Table 7.1 (Appendix E, Survey questions 11 & 12). IFALE participants and non-participants in Tigray (mean=4.98 & 4.93, respectively) and Amhara (mean=4.65 & 4.57 respectively) rated the importance of their participation in voting to be

almost *very important*. The difference between the two groups in both regions was statistically insignificant. From the results, it appears that attending IFALE had no significant contribution to IFALE participants' knowledge of the importance of participation in voting.

In terms of practice of voting, IFALE participants (mean=4.80) and non-participants (mean=4.62) in Tigray rated this as almost *very good*. The difference was statistically insignificant, indicating that IFALE might not have contributed to IFALE participants' voting practice. In contrast, IFALE participants (mean=4.30) and non-participants (mean=3.99) in Amhara reported their practice of voting as approximately *good*. The difference between the mean values was statistically significant. In this case, IFALE seems to have enabled adult participants to elect representatives for political leadership in local councils which may have contributed to participants' capability for control of their environment (Nussbaum, 2000) through election of their representatives.

As regards the importance of participation in voting, the mean values of the IFALE participants in Tigray (mean=4.98) were higher than those in Amhara (mean=4.65). Likewise, the mean values of non-IFALE participants in Tigray (mean=4.93) was higher than in Amhara (mean=4.57). In these cases, the difference between the mean values was statistically significant. The mean values for voting practice by IFALE participants in Tigray (mean=4.80) and Amhara (mean=4.30) were respectively very good and average, and those by non-IFALE participants in Tigray (mean=4.62) and Amhara (mean=3.99) were nearly *very good* and *average*, respectively. In both comparisons, the differences were statistically significant. This may suggest that other external factors, such as local leaders and media, have made more important contributions in Tigray than in Amhara.

7.1.2 Interest in Becoming an Elected Representative

As regards each survey participant's interest in becoming an elected representative (Table 7.1; Appendix E, survey question 13), IFALE participants in Tigray rated it *very high* (mean=3.56) while for non-participants it was *average* (mean=2.89). In Amhara, IFALE participants' interest was almost *average* (mean=2.67) while that of non-participants was *low* (mean=2.24). The difference between the mean values in each region was statistically significant. The higher mean values for IFALE participants in the two regions may result from the role IFALE has played since the curriculum contents include issues of democracy and rule of law, nationalism, election and community participation. This probably helped IFALE participants develop their interest in becoming an elected representative.

IFALE participants in Tigray showed almost *high* interest (mean=3.56) in becoming an elected local representative. In Amhara the interest was nearly *average* (mean=2.67). Non-IFALE participants in Tigray (mean=2.89) and Amhara (mean=2.24) had nearly *average* and *low* interest in being an elected local representative. The difference between the mean values of the two groups in each region was statistically significant. In each region, IFALE participants, therefore, appear to show significantly higher interest in being an elected representative than the non-participants. This is probably because of the efforts of the IFALE program.

	significa	ance						
					Statistical significance of the difference based on two- sample t-test ²¹ with equal variance (t) & probability value (pr) ²²			
						T-tests for regional		
					T-tests for comparing IFALE	differences, conducted		
S.					participants and non-participants	separately for each		
No.	Indicator	Region	Group	Mean	in each region	group		
	Importance of participation in voting		IFALE participants	4.98	t=1.421	IFALE participants		
		Tigray	Non-participants	4.93	pr=0.157	t=5.215***pr=0.000		
		Amhara	IFALE participants	4.65	t=0.901	Non-participants		
1			Non-participants	4.57	pr=0.369	t=5.231*** pr=0.000		
1	Practice of	Tigray	IFALE participants	4.80	t=1.662*	IFALE participants		
	participation in voting		Non-participants	4.62	pr=0.098	t=4.248***pr=0.000		
		Amhara	IFALE participants	4.30	t=2.008**	Non-participants		
			Non-participants	3.99	pr=0.046	t=4.271*** pr=0.000		
	Interest in Becoming an elected representative	Tigray	IFALE participants	3.56	t=3.318***	IFALE participants		
2			Non-participants	2.89	pr=0.001	t=4.703*** pr=0.000		
2		Amhara	IFALE participants	2.67	t=2.319**	Non-participants		
			Non-participants	2.24	pr=0.022	t=3.288*** pr=0.001		
	Participation in	on-making Tigray	IFALE participants	3.66	t=2.508**	IFALE participants		
	decision-making in public		Non-participants	3.21	pr=0.013	t=0.158 pr =8.74		
2			IFALE participants	3.63	t=-2.740***	Non-participants		
3	institutions	Amhara	Non-participants	4.09	pr=0.007	t=-5.246*** pr=0.000		

Table 7.1: Self-reported political empowerment gains of survey participants by mean and statistical significance

***=level of significance at 1%, ** level of significance at 5 % and *level of significant at 10%

7.1.3 Participation in Decision Making in Public Institutions

IFALE participants in Tigray rated the extent of their participation in decision making in public institutions (Table 7.1; Appendix E, survey question 14) to be nearly *high* (mean=3.66) while non-IFALE participants rated it *average* (mean=3.21). The difference between the mean values of the two groups was statistically significant. Interviews with regional, zonal and wereda education officials, and with IFALE participants revealed that IFALE participants

²¹ The t-static or t-test (t) is the ratio of the mean of the difference to the standard error of the difference. In a t-test the level of significance corresponding to the difference between two mean values can be put as follows: when t \geq 2.58 or t \leq -2.58, the significance level is at 1% level, when t= between [1.96, 2.58) or [-1.96, -2.58), the significance is at 5% level, when t= between [1.65, 1.96) & [-1.65, -1.96), the significance is at 10% level, but when t \leq 1.64 or \geq -1.64, the difference is insignificant.

²² Probability value (pr) is the probability value of getting a difference of this size or more by pure chance.

who had finished year one and had started to read and write were encouraged to be involved in local institutions, such as schools, as parental representatives and as members of local administration councils. This might be the reason for the higher mean value by IFALE participants. This is in line with Burchfield et al. (2002a) who report that women in Nepal who participate in literacy programs become more involved in community groups and community decisions.

In the case of Amhara, IFALE participants (mean=3.63) and non-participants (mean=4.09) regarded their participation in decisions as *high*. The difference between the mean values was statistically significant. This probably implies that the non-IFALE participants were more involved in decision-making positions.

The extent of participation in decision making in public institutions by IFALE participants in Tigray (mean=3.66) and in Amhara (mean=3.63) was almost *high*. The difference was statistically insignificant. Participation in decision making in public institutions by non-IFALE participants in Tigray (mean=3.21) was *average* while for the similar group in Amhara (mean=4.09) it was *high*. The difference between the two groups was statistically significant in favor of the group in Amhara.

7.1.4 Role in Local 2013 Election

As indicated in Table 7.2, survey participants reported on the roles they played during the election for the wereda and kebele councils in 2013 (Appendix E, survey questions 16 & 17). In Tigray, 83 IFALE participants (93%) and 86 non-participants (89%) were voters while in Amhara the number was 81 IFALE participants (87%) and 76 non-participants (78%). IFALE may have played a crucial role in making participants cast their vote in both regions, but less so in Amhara.

	_	Tig	ray	Amhara		
S. No.	Participation in election	Number of IFALE participants involved	Number of non- participants involved	Number of IFALE participants involved	Number of non- participants involved	
1	Candidate for wereda/kebele assembly	0	0	0	0	
2	Election observer	3 (3%)	3 (3%)	2 (2%)	2 (2%)	
3	Election facilitator/officer	17 (19%)	8 (8%)	5 (5%)	6 (6%)	
4	Voter	83 (93%)	86 (89%)	81 (87%)	76 (78%)	

 Table 7.2: Survey participants' self-reported participation in the local election in 2013 by frequency and percent

A comparison of IFALE participants in Tigray and Amhara regions shows that only 17 (19%) and five (5%), respectively were election facilitators/officers, fewer were election observers, and none were candidates for the assemblies.

7.1.5 Knowledge of Minimum Voting Age and Political Parties

As to the knowledge of minimum voting age, 82 of the IFALE participants (92%) and 73 nonparticipants (75%) in Tigray, and 71 IFALE participants (76%) and 52 non-participants (54%) in Amhara stated the correct minimum voting age as 18 years (Table 7.3; Appendix E, survey question 18). The IFALE curriculum includes teaching the voting age for federal, regional and local elections. In both regions, the number of IFALE participants who knew the voting age was higher than for non-participants. A significantly higher number of IFALE participants and non-participants in Tigray knew the correct voting age than in Amhara.

Each survey participant was to name the political parties he/she knows (Appendix E, survey question 19). In Tigray, 71 IFALE participants (80%) and 52 non-participants (54%) did so while in Amhara the numbers were 52 IFALE participants (56%) and 25 non-participants (26%). The difference between the knowledge of IFALE participants and non-participants was statistically significant in both regions. This suggests that IFALE has made an important contribution in both regions, but less so in Amhara than in Tigray, in terms of participants' advanced political participation. This is in line with Burchfield et al. (2002a) who reported that women in Nepal who participated in literacy programs had better political knowledge of minimum voting age and names of political parties than non-participants. The difference between both the participants and the non-participants in the two regions is statistically significant in favor of Tigray, probably owing to the role of local leaders and the media.

7.1.6 Confidence to be a Locally Elected Representative

In terms of having confidence to be a locally elected representative (Appendix E, survey question 20), 61 IFALE participants (69%) and 49 non-participants (51%) in Tigray as well as 44 IFALE participants (47%) and 42 non-participants (43%) in Amhara answered positively (Table 7.3). The difference between the two groups was statistically significant in Tigray while it was insignificant in Amhara. The difference in terms of IFALE participants (61 in Tigray and 44 in Amhara) was statistically significant. Self-confidence is a necessary precondition for participation in political affairs and ensuring a political space, in this case for rural-based poor people to get their voices heard. In the case of Nepal, female participants in

literacy programs had more self-confidence than female non-participants in terms of believing that they could serve as political representatives (Burchfield et al., 2002a).

Tigray Amhara Statistical significance of							ifference based		
			IIg	ıay	Annara		Statistical significance of difference based on chi-square ²³ (chi2) & probability value (pr)		
			IFALE	Non-			Chi-squares for comparing Chi-squares for		
			partici-	partici-	partici-	partici-		regional differences,	
S			pants	pants	pants	pants	non-participants in each	conducted separately	
No.	Item/Question	Answer	(N=89)	(N=97)	(N=93)	(N=97)	region	for each group	
	What is the	Yes	82	73	71	52	Tigray	IFALE participants	
	minimum voting age?		(92%)	(75%)	(76%)	(54%)	pr=0.002***chi2=9.519	pr=0.004*** chi2=8.466	
1		No	7	24	22	45	Amhara	Non-participants	
			(8%)	(25%)	(24%)	(46%)	pr=0.001***chi2=10.751	pr=0.002*** chi2=9.919	
	Can you name a	Yes	71	52	52	25	Tigray	IFALE participants	
	political party/		(80%)	(54%)	(56%)	(26%)	pr=0.000***chi2=14.189	pr=0.001***chi2=11.819	
2	parties? Please	No	18	45	41	72	Amhara	Non-participants	
	do.		(20%)	(46%)	(44%)	(74%)	pr=0.000***chi2=17.896	pr=0.000***chi2=15.698	
	Could you	Yes	61	49	44	42	Tigray	IFALE participants	
	become a local		(69%)	(51%)	(47%)	(43%)	pr=0.012** chi2=6.240	pr=0.004*** chi2=8.396	
	elected representative?	No	28	48	49	55	<u>Amhara</u>	Non-participants	
3			(31%)	(49%)	(53%)	(57%)	pr=0.579 chi2=0.309	pr=0.314 chi2=1.014	
	Is there a law on electing representatives (men/women)?	Yes	73	66	43	39	<u>Tigray</u>	IFALE participants	
			(82%)	(68%)	(46%)	(40%)	pr=0.028**chi2=4.805	pr=0.000***chi2=25.198	
4		No	16	31	50	58	<u>Amhara</u>	Non-participants	
			(18%)	(32%)	(54%)	(60%)	pr=0.402 chi2=0.704	pr=0.000***chi2=15.134	
	Are you a member of a community	Yes	80	90	73	70	Tigray	IFALE participants	
			(90%)	(93%)	(78%)	(72%)	pr=0.482 chi2=0.495	pr=0.036** chi2=4.407	
		No	9	7	20	27	<u>Amhara</u>	Non-participants	
5	organization?		(10%)	(7%)	(22%)	(28%)	pr=0.312 chi2=1.02	pr=0.000***chi2=14.265	
	Do you know	Yes	87	92	64	58	Tigray	IFALE participants	
	any community		(98%)	(95%)	(69%)	(60%)	pr=0.298 chi2=1.083	pr=0.000***chi2=26.945	
	development	No	2	5	29	39	Amhara	Non-participants	
6	committees in	110	(2%)	(5%)	(31%)	(40%)	pr=0.195 chi2=1.682	pr=0.000***chi2=33.979	
	your area?				· · ·				
	Did you	Yes	87	88	88	93	<u>Tigray</u>	IFALE participants	
	participate in		(98%)	(91%)	(94%)	(96%)	pr=0.042**chi2=4.124	pr=0.273 chi2=1.204	
	community	No	2	9	5	4	Amhara	Non-participants	
7	development		(2%)	(9%)	(6%)	(4%)	pr=0.685 chi2=0.165	pr=0.151 chi2=2.061	
/	activities over								
the last 2 years?									

 Table 7.3: Self-reported political empowerment gains of survey participants by frequency, percent and level of significance

***=significance level at 1%, **significance level at 5%, * significance level at 10%

7.1.7 Knowledge of a Law on Electing Representatives

In response to whether they know any law on electing representatives (Appendix E, survey question 21), 73 IFALE participants (about 82%) and 66 non-participants (about 68%) in Tigray responded positively (Table 7.3). The difference was statistically significant. In Amhara, only less than half of the IFALE participants (43=46%) and non-participants (39=40%) knew of a law. The difference was statistically insignificant. The difference between the two groups in the regions was statistically significant in favor of Tigray. The fact

²³ In chi-square, when the probability (p value) is $\leq 1\%$, it is significant at 1% level; when the probability (p value) is between (1, 5]%, it is significant at 5% level; and when the probability (p value) is between (5, 10]%, it is significant at 10% level.

that a significantly higher number of IFALE participants than non-participants in Tigray knew of a law may indicate that IFALE contributed its share to enhancing participants' knowledge, including why and how representatives are chosen.

7.1.8 Membership of Community Organizations

Each survey participant reported whether he/she is a member of a community organization (Table 7.3; Appendix E, survey question 24). In Tigray, 80 IFALE participants (90%) and 90 non-participants (93%) were members of, for example, a farmers' association, a women's association, a youth association, a cooperative organization, or of local social institutions, such as the edir²⁴. In Amhara, 73 IFALE participants (78%) and 70 non-participants (72%) were involved in community organizations. The difference between participants and non-participants in each region was statistically insignificant, but across the two regions it was significant in favor of Tigray. According to Stromquist (2009), literacy learners in the US and the UK improved contact with local people because of their involvement in community organizations as a result of the program.

7.1.9 Knowledge of Community Development Committees

As regards community development committees (Appendix E, survey question 27), in Tigray 87 IFALE participants (98%) and 92 non-participants (95%) responded positively knowing, for example, the 1-5 network and development teams²⁵ (Table 7.3). Likewise, 64 IFALE participants (69%) and 58 non-participants in Amhara (60%) knew of village development committees in their areas. The difference in knowledge between the two groups in each region was statistically insignificant, but across the regions it was significantly higher in Tigray for both groups.

7.1.10 Participation in Community Development Activities

Participation in community development activities is another indicator of political empowerment (Table 7.3; Appendix E, survey question 29). In Tigray, 87 IFALE participants (98%) and 88 non-participants (91%) stated that they had participated in activities, such as water and soil conservation, planting trees, digging water wells, and school and road construction activities. The difference between the two groups was statistically significant and

²⁴ *Edir* is a traditional community organization whose members assist each other during mourning. Members contribute monthly to the fund.

²⁵ Development teams in rural Ethiopia are organized as follows: one *kebele* comprises 1000 households; and one network is organized in 5 members for which the leader is chosen by the community based on the performance of health and agricultural extension packages. One development team has 5 networks and one development team has 20-30 households in which case one *kebele* has 30-33 development teams. Each household is expected to be a member of one 1-5 network.

may be due to the lessons which the IFALE program provided to its participants. Increased environmental protection and involvement in social services benefit both children and the community at large and also supports livelihoods improvements.

By contrast, 88 IFALE participants (94%) and 93 non-participants (96%) in Amhara had participated in similar community development activities. The difference between the two groups was statistically insignificant. Comparing the two regions, more IFALE participants had participated in community development activities in Tigray than in Amhara while the reverse was the case for the non-participants. In both cases, the difference was statistically insignificant.

7.1.11 Sources of Political Knowledge and Practice

In terms of knowledge and practice of political participation as well as involvement in public institutions (Appendix E, survey question 23), the majority of IFALE participants in Tigray (75%) and Amhara (67%) indicated that IFALE was the most important source (Table 7.4). Local leaders and the local media (local radio) came next. The local leaders and the local media were more important knowledge and practice sources of IFALE participants and non-participants in Tigray than in Amhara.

			IFALE participants		Non-participants	
S.			Tigray	Amhara	Tigray	Amhara
No	Item	Sources	(N=89)	(N=93)	(N=97)	(N=97)
	Knowledge and practices of	IFALE	67 (75%)	62 (67%)	n/a	n/a
	for political participation as	Friends	8 (9%)	11 (12%)	13 (13%)	14 (14%)
1	well as decision-making of	Local leaders	64 (72%)	47 (51%)	70 (72%)	53 (55%)
	public organizations	Media	47 (53%)	20 (22%)	28 (29%)	12 (12%)
		Others	5 (6%)	3 (3%)	2 (2%)	4 (4%)
	Knowledge and practices	IFALE	63 (71%)	59 (63%)	n/a	n/a
	of involvement in	Agricultural extension workers	27 (30%)	23 (25%)	30 (31%)	17 (18%)
	community organizations,	Friends	3 (3%)	9 (10%)	6 (6%)	8 (8%)
	development committees	Local leaders	77 (87%)	60 (65%)	84 (87%)	70 (72%)
2	and development	Media	14 (16%)	9 (10%)	6 (5%)	4 (4%)
	activities	Others	5 (6%)	4 (4%)	5 (5%)	5 (5%)

 Table 7.4:
 Self-reported sources of survey participants' knowledge and practice of political activities by frequency and percent

n/a=not applicable

As regards their involvement in community organizations, development committees and development activities (Appendix E, survey question 32), local leaders were the most important sources for IFALE participants (Tigray=87%; Amhara, 60%) and non-participants (Tigray=87%; Amhara, 72%) (Table 7.4). The IFALE participants and non-participants in Tigray seemed to have stronger gains than was the case in Amhara. The enhanced knowledge
of community development committees and higher membership in community organizations may be due to more efforts of local leaders at grassroots levels.

7.1.12 Summary

The comparisons between IFALE participants and non-participants indicate that IFALE has contributed to some degree of political empowerment gains of participants, but more so in Tigray than in Amhara. In both Tigray and Amhara, IFALE played an important role in supporting participants' knowledge of the voting age and the names of different political parties. In Tigray, IFALE has meaningfully improved its participants' confidence in terms of becoming an elected representative, participating in public institutions' decision-making, knowing the law on electing representatives, and participating in community development activities. In Amhara, IFALE is likely to improve the participants' practice of participation in voting.

Consistent with Nussbaum (2000), IFALE is therefore likely to have resulted in improved capabilities in terms of knowledge and practice of what is sensible or practical political behavior as well as ability to participate in politics through playing different roles in the community. However, this was more so in Tigray than in Amhara. By contrast, IFALE seems not to have led to significant political gains in terms of its participants' membership of community organizations, or knowledge of community development committees in the two regions.

IFALE participants and non-participants in Tigray had significant political gains compared to Amhara. The specific areas concerned: practice of participation in voting in different roles (voters, facilitators, and observers); awareness of the voting age and names of different political parties; interest in being a local representative for elections; knowledge of the law on electing representatives; membership of community organizations; and knowledge of community development committees. This can be attributed to endeavors made by, for example local leaders, agricultural extension workers and local media.

7.2 IFALE and Economic Empowerment

Economic empowerment concerns the contribution of adult literacy to economic improvements that enable individuals to support themselves, have a better quality of life and make decisions independently. According to Lauglo (2001) and Stromquist (2009), access to both credit services and income generation schemes are also crucial if adult literacy education

is to bring about economic empowerment. In this dissertation, other factors are also seen as important to empower small-scale rural farmers. In the study, the indicators of economic empowerment are: knowledge and practice of agricultural technologies; knowledge and practice of environmental protection; access to credit; use of income generation schemes; knowledge and practice of savings; and knowledge and practice of financial contributions for investment or mutual support services.

7.2.1 Importance and Practice of Agricultural Technologies

The survey participants rated the importance and their practices of agricultural technologies that increase agricultural production, namely improved agricultural tools (such as ploughs, tools for sowing crops in lines), selected seeds, fertilizers, compost, manure, pesticides, herbicides, seed planting in lines, well-timed crop gathering, drip-irrigation, and the use of a motor pump (Appendix E, survey question 33).

As appears in Table 7.5, IFALE participants (mean=4.96) and non-participants (4.97) in Tigray and IFALE participants (mean=4.65) and non-participants (mean=4.61) in Amhara rated the agricultural technologies as nearly *very important*. The difference between the two groups in each region was statistically insignificant. This implies that IFALE seems not to have played a significant role in developing its participants' awareness of the importance of the technologies.

In terms of practices of agricultural technologies, IFALE participants (mean=3.53) and nonparticipants (mean=3.43) in Tigray rated them above *average*. The difference between them was statistically insignificant. The ratings of IFALE participants (mean=3.32) and nonparticipants (mean=3.11) in Amhara were also statistically insignificant. IFALE, therefore, seems not to have contributed to what Cameron and Cameron (2005) consider as the capacity to employ technological advances and precise specification requirements for new technology. Nevertheless, studies on adult literacy education in Kenya (Carron et al., 1989) and Tanzania (Carr-Hill et al., 1991) report that adult literacy programs played an important role in spreading modern agricultural techniques.

In relation to the importance of agricultural technologies, the difference between IFALE participants in Tigray (mean=4.96) and Amhara (mean=4.65) was statistically significant and so was the difference between non-participants of IFALE in Tigray (mean=4.97) and Amhara (mean=4.61). In terms of the practices of agricultural technologies, the difference between IFALE participants in Tigray (mean=3.53) and Amhara (mean=3.32) was statistically

insignificant while the difference between non-participants of IFALE in Tigray (mean=3.43) and Amhara (mean=3.11) was statistically significant.

	Significe				Statistical significance level	of difference based on	
					t-test and pr	of unterence based on	
					T-tests for comparing	T-tests for regional	
					IFALE participants and	differences, conducted	
S.					non-participants in each	separately for each	
No.	Item	Region	Group	Mean	region	group	
		- 8 -	IFALE participants	4.96	t=-0.551	IFALE participants	
	Importance of	Tigray	Non-participants	4.97	pr=0.582	t=6.062* **pr=0.000	
	agricultural		IFALE participants	4.65	t=0.659	Non-participants	
	technologies	Amhara	Non-participants	4.61	pr=0.511	t=7.840*** pr=0.000	
1			IFALE participants	3.53	t=0.821	IFALE participants	
	Practice of	Tigray	Non-participants	3.43	pr=0.413	t=1.882* Pr=0.061	
	agricultural		IFALE participants	3.32	t=1.9325*	Non-participants	
	technologies	Amhara	Non-participants	3.11	pr=0.055	t=2.570** pr=0.011	
	Importance of		IFALE participants	4.98	t=0.137	IFALE participants	
	measures of	ental	Non-participants	4.98	pr=0.890	t=6.452***pr=0.000	
	environmental		IFALE participants	4.73	t=1.245	Non-participants	
	protection	Amhara	Non-participants	4.66	pr=0.215	t=7.815***pr =0.000	
2	Practices of		IFALE participants	4.34	t=0.617	IFALE participants	
	measures of	Tigray	Non-participants	4.29	pr=0.538	t=6.908***pr=0.000	
	environmental		IFALE participants	3.76	t=0.667	Non-participants	
	protection	Amhara	Non-participants	3.70	pr=0.506	t=6.996***pr=0.000	
	Importance of		IFALE participants	4.90	t=2.709***	IFALE participants	
	participation	Tigray	Non-participants	4.64	pr=0.007	t=3.046*** pr=0.003	
	in saving		IFALE participants	4.61	t=1.308	Non-participants	
		Amhara	Non-participants	4.44	pr=0.192	t=1.512 pr =0.132	
3	Practice of		IFALE participants	3.66	t=2.145**	IFALE participants	
	participation	Tigray	Non-participants	3.23	pr=0.033	t=2.368** pr=0.019	
	in saving		IFALE participants	3.20	t=2.112**	Non-participants	
		Amhara	Non-participants	2.80	pr=0.036	t=2.131** pr=0.034	
444	1 0	(10/	**1 1 0		0/ and *lawal of gignificant	. 100/	

 Table 7.5:
 Self-reported economic empowerment gains of survey participants by mean and statistical significance

***=level of significance at 1%, ** level of significance at 5 % and *level of significant at 10%

7.2.2 Importance and Practice of Environmental Protection

There were a number of indicators for the importance and practices of environmental protection measures (Appendix E, survey question 35). They included, amongst others, contour ploughing, terracing, check dams, gabion walls, daget²⁶, sediment traps, tree planting, enclosures, fuel-wood saving stoves and latrines.

As shown in Table 7.5, ratings of environmental protection measures by IFALE participants (mean=4.98) and non-participants (mean=4.98) in Tigray were *very important*. The mean values attached by IFALE participants (mean=4.73) and non-participants (mean=4.67) in Amhara suggest that environmental protection measures were approximately *very important*. The difference between them was statistically insignificant. This could imply that IFALE has

²⁶ Daget is a traditional soil and water conservation technique used by north Ethiopian farmers. It is a 50 cm to 3 metres high lynchet between fields (farmlands) and has grass strips of different sizes on its shoulder (Nyssen, 1998).

not played a significant role in raising the awareness of the survey participants as far as knowledge of the importance of environmental protection measures is concerned.

The practices of environmental protection measures of IFALE participants (mean=4.34) and non-participants (mean=4.29) in Tigray were reported as above *high*, while for the IFALE participants (mean=3.76) and non-participants (mean=3.70) in Amhara they were nearly *high*. The differences in both regions were statistically insignificant, indicating that IFALE has not made a difference in enhancing the practices of participants. IFALE, therefore, seems not to play a significant role in helping the participants in the two regions select superior technologies to protect the natural environment. This is in contrast to other studies, for example literacy programs in Uganda which helped participants reduce the amount of wood fuel they used (Katahoire, 2001as cited in Cameron & Cameron, 2005), implying wiser use of environmental resources and improved environmental protection.

The differences in the importance and the practice of the environmental measures were significant in favor of Tigray both when comparing IFALE participants, and when comparing non-participants across the two regions.

7.2.3 Importance and Practice of Saving

In respect of the importance and practice of saving (Appendix E, survey questions numbers 37 & 38), IFALE participants (mean=4.90) and non-participants (mean=4.64) in Tigray ratings were almost *very important* (Table 7.5). The difference between the groups was significant. This probably implies that participation in IFALE in Tigray plays an important role in increasing participants' knowledge of the importance of saving. In Amhara, IFALE participants (mean=4.61) and non-participants (mean=4.44) rated saving as *very important* and *important*. The difference was statistically insignificant.

The saving practice of IFALE participants (mean=3.66) and non-participants (mean=3.23) in Tigray was rated nearly *good* and *average*, respectively, while in Amhara IFALE participants (mean=3.20) and non-participants (mean=2.80) rated themselves as approximately *average*. In each region, the difference between participants and non-participants was statistically significant. It implies that IFALE possibly played an important role in improving the saving practices of the participants which could help them combat economic vulnerability, diversify their income, for example for additional income generation activities, and eventually enhance livelihoods sustainably. This is in accordance with Cameron and Cameron (2005, p. 8) who claim that literacy gains may lead to saving opportunities. The Women Empowerment

Program (WEP), a literacy training program in the Terai region of Nepal, also reinforced prevailing practices of saving and credit associations (Ashe and Parrott, 2001).

The difference in the importance of saving between IFALE participants in Tigray and Amhara was statistically significant while in the case of non-IFALE participants in the two regions it was statistically insignificant. As regards saving practice, the differences between IFALE participants and non-participants in Tigray and Amhara were statistically significant. The difference between the two regions may be explained by the coordinated efforts in Tigray of the Dedebit Credit and Savings Institution (DECSI), agricultural extension workers and local leaders on the importance of saving.

7.2.4 Sources of Knowledge and Practices of Agricultural Technologies and Environmental Protection Measures

The responses by the survey participants on factors that support their awareness of importance and practice of agricultural technologies and environmental protection measures appear in Table 7.6 (Appendix E, survey question 34). IFALE participants and non-participants in Tigray rated local leaders and agricultural extension workers higher as sources of knowledge and practices of agricultural technologies than in Amhara. This may explain the significant difference in mean values between non-participants in both regions in terms of practices of agricultural technologies (refer 7.2.1).

	teen	lologies and environmental	1	2 1	1 3 1		
			IFALE pa	articipants by	Non-participar		
S.			frequency	y and percent	frequency and	percent	
No.	Item	Sources	Tigray (N=89)	Amhara (N=93)	Tigray (N=97)	Amhara N=97)	
		IFALE	73 (82%)	65 (70%)	n/a	n/a	
	Knowledge	FTCs	77 (87%)	65 (70%)	74 (76%)	70 (72%)	
1	and practices	Agricultural extension workers	72 (81%)	37 (40%)	79 (81%)	47 (48%)	
	of agricultural	Health extension workers	20 (22%)	16 (17%)	12 (12%)	14 (14%)	
	technologies	Local leaders	59 (66%)	25 (27%)	75 (77%)	39 (40%)	
		Media	29 (33%)	10 (11%)	20 (21%)	8 (8%)	
		Others	0	8 (9%)	1 (1%)	7 (7%)	
	Knowledge	IFALE	80 (90%)	76 (82%)	n/a	n/a	
	and practices	FTCs	76 (85%)	51 (55%)	77 (79%)	55 (57%)	
	of	Agricultural extension workers	62 (70%)	27 (29%)	73 (75%)	37 (38%)	
	environmental	Health extension workers	53 (60%)	54 (58%)	55 (57%)	61 (63%)	
2	protection	Local leaders	56 (63%)	30 (32%)	69 (71%)	47 (48%)	
	measures	Media	36 (40%)	10 (11%)	20 (21%)	8 (8%)	
		Others	1 (1%)	11 (12%)	3 (3%)	5 (5%)	

 Table 7.6: Survey participants' self-reported sources of knowledge and practice of agricultural technologies and environmental protection measures by frequency and percent

As regards environmental protection measures (Appendix E, survey question 36), IFALE followed by FTCs, health extension workers, agricultural extension workers and local leaders are the most important sources of knowledge and practice for IFALE participants in both Tigray and Amhara. In Tigray, the agricultural extension workers, the local administrators

and the FTCs are more important sources than in Amhara. This may explain the difference regarding environmental protection measures in favor of Tigray (refer 7.2.2).

7.2.5 Financial Contributions to Productive Investment and/or Mutual Support

As regards contributions of money for an organization or productive investment in 2011and 2012 (Appendix E, survey question 42), 84 IFALE participants (94%) and 93 non-participants (96%) in Tigray had done so (Table 7.7). Eighty-two IFALE participants (88%) and 84 non-participants (87%) in Amhara had contributed money to different organizations. In both regions, the difference between the two groups was statistically insignificant. This means that IFALE has not contributed to improving services of mutual support or, as stated by Cameron and Cameron (2005, p. 7), 'seeking new ways of financing productive investment'.

The difference between financial contributions by IFALE participants in the two regions was statistically insignificant while for the non-participants it was statistically significant. Most of the survey participants contributed money for the construction of schools and health centers, as well as for the purchase of an ambulance. They also bought bonds for the construction of the Ethiopian Grand Renaissance Dam.

							Statistical significance leve	
			Tigray		Amhara		based on pr and ch2 value	
			IFALE	Non-	IFALE	Non-	Chi-squares for comparing	Chi-squares for
			partici-	partici-	partici-	partici-	IFALE participants and	regional differences,
S.			pants	pants	pants	pants	non-participants in each	conducted separately
No.	Question	Answer	(N=89)	(N=97)	(N=93)	(N=97)	region	for each group
	Did you	Yes	84	93	82	84	Tigray	IFALE participants
	contribute		(94%)	(96%)	(88%)	(87%)	pr=0.635 chi2=1.225	pr=0.139 chi2=2.187
	money for							Non-participants
1	investment or a	27	-	4	11	10	Amhara	pr=0.022** chi2=5.222
	mutual support	No	5	4	11	13	pr=0.744 chi2=0.107	pi=0.022 ** emi2=3.222
	service in 2011,		(6%)	(4%)	(12%)	(13%)	pi=0.744 cm2=0.107	
	2012 & 2013?							
	Do you have	Yes	84	89	81	79	Tigray	Participants
	access to credit		(94%)	(92%)	(87%)	(81%)	pr=0.482 chi2=0.494	<u>p</u> r=0.091* chi2=2.850
	for additional							
2	income	No	5	8	12	18	<u>Amhara</u>	Non-participants
	generation in		(6%)	(8%)	(13%)	(19%)	pr=0.285 chi2=1.141	pr=0.035**chi2=4.441
	your community?							
	Do you have	Yes	82	80	72	71	Tigray	IFALE participants
	any means of		(92%)	(82%)	(77%)	(73%)	pr=0.050**chi2= 3.854	pr=0.006***chi2= 7.565
3	income	No	7	17	21	26	Amhara	Non-participants
	generation?	1 (10)	(8%)	(18%)	(23%)	(27%)	pr=0.500 chi2=0.455	pr=0.120 chi2=2.420

 Table 7.7:
 Self-reported credit access gains and financial contributions of survey participants by frequency, percent, and statistical significance

***=significance level at 1%, **significance level at 5 %, * significance level at 10%

7.2.6 Sources of Knowledge and Practice of Financial Contributions

As shown in Table 7.8, in terms of financial contributions for schools, purchase of ambulances etc. (Appendix E, survey question 44), local leaders were the most important source of encouragement (Tigray=87%; Amhara=82%), and more important than IFALE (Tigray=51%; Amhara=52%). Health extension workers, the local media and agricultural extension workers were more important for non-IFALE participants in Tigray than in Amhara.

contr	ibutions by frequency and per	cent				
		IFALE partie	cipants	Non-participants		
Item	Sources	Tigray (N=89)	Amhara (N=93)	Tigray (N=97)	Amhara (N=97)	
	IFALE	45 (51%)	48 (52%)	n/a	n/a	
Knowledge and	Agricultural extension workers	22 (25%)	14 (15%)	20 (21%)	14 (14%)	
practice of	Health extension workers	22 (24%)	17 (18%)	21 (22%)	11 (11%)	
financial	Friends	5 (6%)	5 (5%)	3 (3%)	4 (4%)	
contributions	Local leaders	77 (87%)	76 (82%)	92 (95%)	88 (91%)	
	Media	20 (22%)	11 (12%)	13 (13%)	6 (6%)	
	Others	5 (6%)	4 (4%)	4 (4%)	2 (2%)	

 Table 7.8: Self-reported sources of survey participants' knowledge and practice of financial contributions by frequency and percent

n/a=not applicable

7.2.7 Access to Credit

In Tigray, 84 IFALE participants (94%) and 89 non-participants (92%) had access to credit for income generation compared to 81 IFALE participants (87%) and 79 non-participants (81%) in Amhara (Table 7.7; Appendix E, survey question 39). The difference between the two groups within each region was statistically insignificant. This shows that access to microcredit is not related to IFALE in contrast to the study on women in Nepal (Ashe and Parrott, 2001). However, the differences in access to credit between both IFALE participants and nonparticipants across the two regions were statistically significant.

Regarding credit service (Appendix E, survey question 40), most survey participants reported that a rural credit and savings bank was the most important source (Table 7.9), specifically DCSI in Tigray and Amhara Credit and Savings Institution (ACSI) in Amhara. More opportunities for credit access can accelerate additional income generation that lead to more economic empowerment.

More non-IFALE participants in Tigray than Amhara have access to credit services. This may suggest that rural community members in Tigray also had awareness raising training by local leaders and agricultural extension workers who play a key role in the village community concerning awareness of agricultural technologies and environmental protection measures.

S.		Number of IFA	LE participants	Number of non-participants		
No.	Sources of credit	Tigray (N=89)	Amhara (N=93)	Tigray (N=97)	Amhara (N=97)	
1	A rural credit and saving bank	77 (86%)	60 (65%)	80 (82%)	55 (57%)	
2	Women's association	19 (21%)	4 (%)	18 (19%)	2 (2%)	
3	NGOs	5 (6%)	2 (4%)	0	1 (1%)	
4	Local money lenders	5 (6%)	11 (12%)	3 (3%)	13 (13%)	
5	Others*	10 (11%)	18 (19%)	13 (13%)	23 (24%)	

Table 7.9: Survey participants' self-reported sources of credit by frequency and percent

*=Friends, relatives, neighbors etc.

7.2.8 Income Generation Schemes

As indicated in Table 7.7 (Appendix E, survey question 41), 82 IFALE participants (92%) and 80 non-IFALE participants (82%) in Tigray have a means of income generation. The difference between them was statistically significant. This shows that IFALE meaningfully enabled program participants to have a means of income generation or income diversification through participation, for example in handicraft, vegetable growing and dairy production.

In Amhara, 72 IFALE participants (77%) and 71 non-IFALE participants (73%) had a method of income generation. The difference between both groups was statistically insignificant, perhaps suggesting that IFALE did not support program participants in having a means of income generation. The difference between IFALE participants across the two regions was statistically significant in favor of Tigray.

The number of IFALE participants in Tigray who participated in income generation schemes (301) was significantly higher than the number of the non-participant group (247) (Table 7.10; Appendix E, survey question 41). On average an IFALE participant was involved in approximately three income generation schemes and a non-participant in 2.5 income generation schemes. In Amhara, IFALE participants (179) also slightly outnumber non-participants (164) in terms of involvement in income-generation schemes which is on average two schemes for each group. This may suggest that literacy gains helped the participants use credit facilities to invest in income generation schemes. According to Cameron and Cameron (2005), adult literacy can help participants engage themselves in revenue making productive investment.

In Tigray, the most common income generation scheme for IFALE participants was poultry (79%) whereas in Amhara it was vegetable growing (49%). Generally speaking, IFALE participants and non-participants in Tigray were more involved in income generation schemes than in Amhara—with the exception of vegetable growing and small businesses.

		Tigr	ay	Amha	ara
S. No	Income generation schemes	IFALE participants (N=89)	Non-participants (N=97)	IFALE participants (N=93)	Non-participants (N=97)
1	Vegetable growing	32 (36%)	26 (27%)	46 (49%)	49 (51%)
2	Small business	17 (19%)	13 (13%)	27 (29%)	14 (14%)
3	Handicraft	24 (27%)	14 (14%)	16 (17%)	11 (11%)
4	Poultry	70 (79%)	63 (65%)	28 (30%)	28 (29%)
5	Dairy production	34 (38%)	27 (28%)	13 (14%)	8 (8%)
6	Fattening (oxen)	43 (48%)	40 (41%)	28 (30%)	22 (23%)
7	Bee keeping	25 (28%)	17 (18%)	11 (12%)	8 (8%)
8	Supplying spices	10 (11%)	7 (7%)	1 (1%)	3 (3%)
9	Working as a seasonal daily laborer	44 (49%)	40 (41%)	4 (4%)	12 (12%)
10	Others (for example, hair doing, masonry)	2 (2%)	0 (0%)	5 (5%)	9 (%)
	al frequency of participation in ous income generation schemes	301	247	179	164

Table 7.10: Survey participants' self-reported involvements in income generation schemes by frequency and percent

7.2.9 Summary

Although empowerment gains from IFALE are relatively better in Tigray than Amhara, these gains appear to be generally low in both regions. As the results show, IFALE plays an important role in enhancing participants' saving practice in both Tigray and Amhara. It also contributes to the involvement of its participants in income generation schemes in Tigray. However, IFALE does not result in improved practices of agricultural technologies and environmental protection measures, access to credit for income generation, and practice of financial contributions for productive investment and mutual support in either region.

IFALE participants and non-participants in Tigray have more meaningful gains than their comparison groups in Amhara on most economic empowerment indicators, including better practices of agricultural technologies, measures of environmental protection, access to credit, income generation schemes and saving. But this appears to be because of other factors than IFALE, such as agricultural extension workers and local administrators.

7.3 IFALE and Socio-Cultural Empowerment

The understanding of socio-cultural empowerment in this dissertation builds on two definitions in the literature. According to Vasumathi (2008), in the context of adult literacy education, social empowerment refers to the processes that help establish social networks, norms and social trust and lead to common benefit among people. Cultural empowerment enables learners to replace a traditional lifestyle with a modern one (Vasumathi, 2008). Socio-cultural empowerment refers to the ability of literacy education to contribute to enhancing learners' social capital and changing particular cultural traditions.

Socio-cultural empowerment is closely interlinked with political and economic empowerment. The indicators for socio-cultural empowerment in this study originate from the IFALE curriculum and primers and include: gender equality in decision making; understanding gender equality for development; knowledge of laws protecting women and children and of laws against gender discrimination; importance and practices of HIV/AIDS control measures; family cohesion; wanting an additional child; family planning; specific traditional activities; personal hygiene; maternal and child care; social justice and democratic culture activities; and conflict resolution methods.

7.3.1 Gender Equality in Decision Making

In order to understand gender equality at household level (Appendix E, survey question 45), survey participants reported on who is the main decision maker in their family as regards the eight areas listed in Table 7.11.

The concept of family was understood broadly. In Tigray, the survey included 6 single, 10 divorced, 4 widows/widowers and 69 married IFALE participants and 4 single, 12 divorced, 4 widows/widowers and 77 married non-participants. In Amhara, there were 11 single, 5 divorced, 4 widows/widowers and 73 married IFALE participants, and 5 single, 8 divorced, 1 widow and 83 married non-participants. Since those who are single, divorced or a widow/er live without a partner, only responses from married survey participants were analyzed regarding gender equality of husband and wife in decision making within their families (Table 7.11; Appendix L).

In Tigray, the number of IFALE participant married couples that took decisions together was smaller than for the non-participants in the same region. This implies that IFALE did not contribute to learner practices about gender equality in decision-making. In contrast, more married IFALE participants than non-participants in Amhara attested to equal participation in decision making concerning selling expensive things, using a method of birth control, disciplining a child, purchasing agricultural inputs, and deciding what crops to sow. This may mean that IFALE played a role in household decision making in Amhara, probably because civics, ethics and gender equality within the family are included as contents in the primer (ABE, 2011a, ABE, 2012a). This is in contrast to a study by Kagitcibasi et al. (2005) which shows no significant positive changes among functional adult literacy participant women in terms of household decision making practices in Turkey.

		frequency and p		Tio	gray			Am	hara	
			Married		Marrie	d Non-	Marrie	d IFALE	Marrie	d Non-
			participa		particip		particip		participants	
S.			(Number		(Number=77)		(Number=73)		(Number=83)	
No.	Item	Options	Freq.	%	Freq.	%	Freq.	%	Freq.	%
	Main decision	Husband	4	5.8	3	3.90	9	12.33	15	18.07
	maker in your	Wife	4	5.8	0	0	6	8.22	2	2.41
1	family for buying	Husband and wife	61	88.41	74	96.1	53	72.60	62	74.70
	expensive things?	Others	0	0	0	0	5	6.85	4	4.82
		Total	69	100	77	100	73	100	83	100
	Main decision	Husband	3	4.35	4	5.19	6	8.22	16	19.28
	maker in your	Wife	3	4.35	0	0	5	6.85	2	2.41
2	family for selling	Husband and wife	63	91.3	73	94.81	58	79.45	60	72.29
	expensive things?	Others	0	0	0	0	4	5.48	5	6.02
		Total	69	100	77	100	73	100	83	100
	Main decision	Husband	1	1.45	0	0	1	1.37	2	2.41
	maker in your	Wife	3	4.35	0	0	6	8.22	4	4.82
3	family for having	Husband and wife	65	94.20	77	100	66	90.41	77	92.77
	another child?	Others	0	0	0	0	0	0	0	0
		Total	69	100	77	0	73	100	83	100
	Main decision	Husband	0	0	0	0	0	0	3	3.61
	maker in your	Wife	6	8.70	3	3.90	8	10.96	12	14.46
4	family for using	Husband and wife	63	91.30	72	93.51	64	87.67	68	81.93
	birth control?	Others	0	0	2	2.60	1	1.37	0	0
		Total	69	100	77	100	73	100	83	100
	Main decision	Husband	2	2.90	4	5.19	21	28.77	31	37.80
	maker in your	Wife	4	5.80	2	2.60	3	4.11	1	1.22
5	family for	Husband and wife	63	91.30	71	92.21	48	65.75	50	60.98
	disciplining the	Others	0	0	0	0	1	1.37	0	0
	child?	Total	69	100	77	100	73	100	82	100
	Main decision	Husband	1	1.45	1	1.30	9	12.33	13	15.66
	maker in your	Wife	3	4.35	0	0	5	6.85	2	2.41
6	family for arranging		65	94.20	76	98.70	52	71.23	62	74.70
	marriage for a	Others	0	0	0	0	7	9.59	6	7.23
	daughter/son?	Total	69	100	77	100	73	100	83	100
	Main decision	Husband	0	0	5	6.58	14	19.18	29	34.94
_	maker in your	Wife	6	8.70	0	0	3	4.11	5	6.02
7	family for	Husband and wife	63	91.30	71	93.42	54	73.97	49	59.04
	purchasing	Others	0	0	0	0	2	2.74	0	0
	agricultural inputs?	Total	69	100	76	100	73	100	83	100
	Main decision	Husband	1	1.45	2	2.63	19	26.03	36	43.37
0	maker in your	Wife	5	7.25	0	0	3	4.11	6	7.23
8	family for deciding	Husband and wife	63	91.30	74	97.37	51	69.86	41	49.40
	what crops to sow?	Others	0	0	0	0	0	0	0	0
		Total	69	100	76	100	73	100	83	100

Table 7.11: Married survey participants' self-reports on gender equality in decision making in the family by frequency and percent

7.3.2 Understanding Gender Equality for Development

The survey participants were asked whether males and females contribute equally to development (Table 7.12; Appendix E survey question 47). In Tigray, 83 IFALE participants (93%) and 94 non-participants (97%) responded positively. The responses in Amhara were similar: 88 IFALE participants (95%) and 86 non-participants (89%) said yes. The difference between IFALE participants and non-participants in each of the two regions was statistically insignificant. The difference between IFALE participants within the two regions was also

statistically insignificant while that of the non-participants was significant. The latter may be due to the endeavors of women's associations in Tigray (refer Table 7.13).

7.3.3 Knowledge of Laws against Gender Discrimination

Survey participants were asked if they know of any law against gender discrimination (Table 7.12; Appendix E, survey question 50). Eighty-one IFALE participants (91%) and 83 non-participants (86%) in Tigray, and 42 IFALE participants (45%) and 38 non-participants (39%) in Amhara did. The difference between IFALE participants and non-participants in each region was statistically insignificant.

In both regions, the survey respondents indicated that the Constitution assures equal rights of women and men in every sphere. It guarantees affirmative actions to remedy the sufferings of women because of past inequalities and the rights of women to own and administer property, to family planning services and to paid pre- and post-delivery maternity leaves. The regional differences between IFALE participants and between non-participants of both regions were statistically significant.

						Statistical significance of t	the difference based on			
		Tig	ray	Aml	nara	pr and ch2 values				
		IFALE	Non-	IFALE	Non-	Chi-squares for comparing	g Chi-squares for			
		partici-	partici-	partici-	partici-	FALE participants and	regional differences,			
		pants	pants	pants	pants	non-participants in each	conducted separately			
Question	Answer	(N=89)	(N=97)	(N=93)	(N=97)	region	for each group			
Do you think	Yes	83	94	88	86	Tigray	IFALE participants			
that males and		(93%)	(97%)	(95%)	(89%)	pr=0.247 chi2=1.342	pr=0.699 chi2(1)=0.149			
females						_				
contribute	No	6	3	5	11	<u>Amhara</u>	Non-participants			
equally to		(7%)	(3%)	(5%)	(11%)	pr=0.139 chi2=2.190	pr=0.026** chi2=4.927			
development?		()	()	()	()	-	-			
Do you know	Yes	81	83	42	38	Tigray	IFALE participants			
whether there		(91%)	(86%)	(45%)	(39%)	pr=0.251 chi2=1.319	pr=0.000***chi2=43.638			
are laws against) T	0	1.4	51	50	Ambana				
gender	No	•		-			Non-participants			
discrimination?		(9%)	(14%)	(55%)	(61%)	p1-0.403 cm2-0.698	r=0.000***chi2=44.475			
Do you know	Yes	88	95	60	51	<u>Tigray</u>	IFALE participants			
of laws		(99%)	(98%)	(65%)	(53%)	pr=0.612 chi2=0.258	pr=0.000***chi2=35.344			
protecting	No	1	2	33	46					
women and	110	(1%)	_			<u>Amhara</u>	Non-participants			
children?		(1/0)	(270)	(3370)	(4770)	pr=0.095* chi2=2.786	pr=0.000***chi2=53.594			
	Question Do you think that males and females contribute equally to development? Do you know whether there are laws against gender discrimination? Do you know of laws protecting women and	QuestionAnswerDo you think that males and femalesYescontributeNoequally to development?NoDo you know whether there are laws against genderYesDo you know discrimination?YesDo you know of laws protecting women and children?No	QuestionAnswerTig IFALE participantsDo you think that males and females contribute equally to development?Yes83 (93%)Do you know whether there are laws against gender discrimination?No6 (7%)Do you know of laws protecting women and 	UQuestionAnswerTigrayIFALENon- partici- pantspartici- pantsDo you think that males and femalesYes8394(03%)(93%)(97%)contributeNo63equally to development?(7%)(3%)Do you know whether there are laws against genderYes8183Mo63(14%)Do you know of laws protecting women and children?Yes8895No12(2%)	QuestionAnswerIFALE partici- pantsNon- partici- pantsIFALE partici- pantsDo you think that males and females contributeYes839488(93%)(97%)(97%)(97%)(95%)equally to development?No635Do you know gender discrimination?Yes818342No81451(95%)(99%)(14%)(55%)Do you know of laws protecting women and children?Yes889560No1233(1%)(1%)(2%)(35%)(35%)	TigrayAmharaTigrayAmharaIFALENon- partici- pantsIFALENon- partici- pantsQuestionAnswer $(N=89)$ $(N=97)$ $(N=93)$ $(N=97)$ Do you think that males and females contributeYes 83 94 88 86 (analysis) (97%) (97%) (97%) (95%) (89%) Do you know whether there are laws against genderYes 81 83 42 38 (No 8 14 51 59 discrimination? (99%) (14%) (55%) (61%) Do you know of laws protecting women and children?No 1 2 33 46 No 1 2 33 46 (No 1 (2%) (35%) (47%)	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $			

 Table 7.12: Self-reported socio-cultural empowerment gains of survey participants by frequency, percent, and statistical significance

***=level of significance at 1%, ** level of significance at 5 % and *level of significant at 10%

7.3.4 Knowledge of Laws Protecting Women and Children

In Tigray, 88 IFALE participants (99%) and 95 non-participants (98%) knew of laws protecting women and children (Table 7.12; Appendix E, survey question 55). In Amhara, 60 IFALE participants (65%) and 51 non-participants (53%) were aware of the laws. The difference was statistically insignificant within each region.

Generally speaking, the survey participants reported that they know the Constitution which affirms the rights of all persons including men, women and children alike, recognizes the vulnerable position of children, and guarantees their right. In the case of both participants and non-participants, differences were statistically significant in favor of Tigray.

7.3.5 Knowledge Sources of Gender Equality, and Laws against Gender Discrimination and for Protection of Women and Children

IFALE seems to be the most important source for equal gender decision making among IFALE participants (rated by 78% in Tigray, 72% in Amhara) (Table 7.13; Appendix E, survey question 46). Local leaders, agricultural extension workers and the local media were also influential. In Tigray, more IFALE participants and non-participants referred to local leaders as sources of knowledge and practice about gender equality of decision-making than was the case in Amhara.

S.			IFALE par	J	Non-parti	
No	Item	Sources	Tigray	Amhara	Tigray	Amhara
	Knowledge of	IFALE	69 (78%)	67 (72%)	n/a	n/a
	gender equality	Agricultural extension workers	35 (39%)	14 (15%)	34 (35%)	6 (6%)
1	in decision	Local leaders	55 (62%)	24 (26%)	71 (73%)	39 (40%)
	making within	Media	27 (30%)	13 (14%)	18 (19%)	9 (9%)
	the family	Others	9 (10%)	58 (62%)	12 (12%)	81 (84%)
	Knowledge of	IFALE	79 (89%)	75 (81%)	n/a*	n/a
	gender equality	Agricultural extension workers	15(17%)	11 (12%)	20 (21%)	11 (11%)
	in development	Women's association	49 (55%)	22 (24%)	55 (57%)	17 (18%)
2		Friends	8 (9%)	13 (14%)	7 (8%)	11 (11%)
		Local leaders	68 (76%)	51 (55%)	88 (91%)	74 (76%)
		Media	32 (36%)	20 (22%)	40 (41%)	21 (22%)
		Others	5 (6%)	0 (0%)	2 (2%)	6 (6%)
	Knowledge of	IFALE	76 (85%)	71 (76%)	n/a	n/a
	laws against	Women's association	55 (62%)	27 (29%)	56 (58%)	20 (21%)
3	gender	Friends	9 (10%)	7 (8%)	10 (10%)	12 (12%)
	discrimination	Local leaders	65 (73%)	62 (67%)	81 (84%)	65 (67%)
		Media	46 (52%)	22 (24%)	31 (32%)	15 (15%)
		Others	0 (0%)	3 (3%)	6 (6%)	5 (5%)
	Knowledge of	IFALE	79 (89%)	58 (62%)	n/a	n/a
	laws for	Agricultural extension workers	15 (17%)	10 (11%)	11 (11%)	5 (5%)
4	protecting	Women's association	58 (65%)	27 (29%)	53 (55%)	16 (16%)
	women and	Friends	4 (4%)	12 (13%)	5 (5%)	12 (12%)
	children	Local leaders	61 (69%)	41 (44%)	75 (77%)	55 (57%)
		Media	33 (37%)	16 (17%)	26 (27%)	13 (13%)
		Others	3 (3%)	2 (2%)	2 (2%)	7 (7%)

 Table 7.13:
 Survey participants' self-reported knowledge of gender equality, and laws against gender discrimination and protection of women and children by frequency and percent

As to gender equality for development (Appendix E, survey question 49), IFALE participants in both regions attached the highest importance to IFALE followed by local leaders and women's associations. Local leaders, women's associations and local media (the local radio)

seem to be more important for both IFALE participants and non-participants in Tigray than in Amhara.

IFALE also seems to be the most important source of knowledge of laws against gender discrimination among IFALE participants (Tigray 85%; Amhara 76%) followed by local leaders (Tigray 73%; Amhara 67%) and women's associations (Tigray 62%; Amhara 29%) (Appendix E, survey question 54). IFALE participants and non-participants in Tigray appeared to benefit more from women's associations as regards their knowledge of laws against gender discrimination than in Amhara. This may explain the significant difference between both participants and non-participants in the two regions in favor of Tigray.

IFALE participants ranked IFALE (Tigray 89%; Amhara 58%) first amongst the knowledge sources of laws protecting women and children (Appendix E, survey question 57). Local leaders and women's associations were also of importance for both groups in the two regions. The number of IFALE participants and non-participants in Tigray who reported that women's associations, local leaders and the media were their sources of knowledge of the laws was significantly higher than in Amhara.

Thus, in all three areas local leaders, women's association and local media appear to play a more significant role in Tigray than in Amhara concerning the knowledge of both IFALE participants and non-participants of gender equality and laws against gender discrimination and for protection of women and children.

7.3.6 Importance and Practice of Family Cohesion

There were a number of factors to assess family cohesion. These included, amongst others, whether the mother and father/husband and wife do things together, closeness between the mother/father and the child, closeness between siblings in a family/members of an extended family, and the quality of the relationship among all members of a nuclear family/extended family members (Appendix E, survey question 58).

In terms of family cohesion within the close and extended family, IFALE participants (mean=4.86) and non-participants (mean=4.77) in Tigray rated this as nearly *very important* (Table 7.14). In Amhara, the ratings by IFALE participants (mean=4.39) and non-participants (mean=4.32) was approximately *important*. In both regions, the difference was statistically insignificant and may imply that participation in IFALE played no role in this.

Survey participants also rated their family cohesion practice within their nuclear and extended family. The responses from IFALE participants in Tigray (mean=4.42) was almost *high* which was also the case of non-participants in the same region (mean 4.30). IFALE participants (mean=3.71) and non-participants (mean=3.62) in Amhara also responded nearly *high*. The differences were statistically insignificant. This suggests that the contents and the approach of IFALE play no role because of the strong social bond in Ethiopian culture. By contrast, Kagitcibasi et al. (2005) identify a significant positive contribution to family cohesion because of women's empowerment gained from functional literacy in Turkey.

In relation to the importance of family cohesion, the difference between IFALE participants and between non-participants across regions was significant in favor of those in Tigray. Regional differences between IFALE participants in Tigray (mean=4.42) and Amhara (mean=3.71) as well as non-IFALE participants in Tigray (mean=4.30) and Amhara (mean=3.62) as regards family relation practices were statistically significant.

7.3.7 Importance and Practice of Specific Traditional Activities

To understand empowerment in terms of addressing specific cultural practices, survey participants were asked about the importance and their personal practices in the following areas: breaking up immediately after marriage if the bride is not a virgin; considerable age difference in marriage, underage marriage; marriage by abduction; levirate marriage²⁷; sexual harassment (unwanted and deliberate touching, leaning over, cornering, or pinching); genital mutilation; rape; milk teeth extraction; uvulectomy; engagement of children in hard work; physical abuse of children; sexual abuse of children; high spending in feasts; increased number of holidays; scratching one's face and beating one's chest after the death of a beloved one; and visits to local fortune-tellers (Appendix E, survey question 60).

IFALE participants (mean=1.50) and non-participants (mean=1.70) in Tigray rated the list of specific traditional activities as almost *totally unimportant* and *not important*, respectively (Table 7.14). In Amhara, the ratings of IFALE participants (mean=1.52) and non-participants (mean=1.73) was approximately *not important*. Within each region, the differences were statistically significant. IFALE appears to make a difference in empowering its participants in terms of understanding the disadvantages of the particular traditional activities since practices, such as genital mutilation, milk teeth extraction, uvulectomy, high spending in feasts, and

²⁷ According to the Concise Oxford Dictionary of World Religions, levirate marriage is a type of marriage in which a brother is obliged to marry his brother's widow, and the widow is obliged to marry her deceased husband's brother.

increased number of holidays, affect health, livelihoods and individual rights in one way or another.

Survey participants self-reported how often they practice the mentioned specific traditional activities. IFALE participants in Tigray (mean=1.45) seemed *not* to practice the activities *at all*, while the non-participants in the same region (mean=1.61) practiced them only *rarely*. IFALE participants (mean=1.51) and non-participants (mean=1.62) in Amhara practiced them almost *rarely*. The difference between the mean values of the two groups in each region was statistically significant. This may be explained by IFALE which focuses its contents on harmful traditional practices, such as marriage by abduction and genital mutilation that have been rooted in Ethiopia for generations (MoE, 2010c; TEB, 2010; AEB, 2011a, 2011b). This may suggest that IFALE contributed to changing attitudes towards these practices through its contents and approaches.

In terms of the importance of the specified traditional activities, regional differences were insignificant between both IFALE participants in Tigray (mean=1.50) and Amhara (mean=1.52) and between non-IFALE participants in Tigray (mean=1.70) and Amhara (mean=1.73). In terms of the practices of specified traditional activities across the two regions, there was insignificant statistical difference between IFALE participants in Tigray (mean=1.45) and Amhara (mean=1.51), and between non-IFALE participants in Tigray (mean=1.62) and Amhara (mean=1.62).

7.3.8 Importance and Practice of Wanting an Additional Child

An additional²⁸ child here refers to the wish to have another child in addition to what the survey participants already had. Survey participants rated the importance and their practices of wanting an additional child for the following seven reasons: financial support in old age (economic value); old age security (psychological value); support to the parents when the child is young (economic value); love (psychological and social value); closeness (psychological and social value); parental name (social value); and fun (psychological value) (Appendix E, survey question 62). The lower the values attached to each of the reasons, the more empowered the survey participants are considered to be.

IFALE participants (mean=4.67) and non-participants (mean=4.80) in Tigray rated having an additional child as almost *very important* (Table 7.14). The difference between the two groups was statistically significant. In this case, IFALE seems to make a difference for the

²⁸ The total fertility rate in rural Ethiopia is 5.5 children per family (Ethiopian Central Statistics Agency, CSA, 2012).

participants. In contrast, in Amhara, IFALE participants (mean=4.63) and non-participants (mean=4.62) rated having an additional child as almost *very important* with insignificant difference between the two groups and, therefore, apparently no impact from IFALE.

Survey participants also assessed their practices of wanting an additional child. IFALE participants (mean=4.57) and non-participants (mean=4.72) in Tigray rated them as almost *very high*. The difference between the mean values was statistically significant. The lower mean values may be due to behavioral changes gained from exposure to the IFALE program. Similar to this, Kagitcibasi et al. (2005) report that functional literacy participating women in Turkey valued children positively. The ratings by IFALE participants (mean=4.03) and non-participants (mean=4.00) in Amhara were *high* and the difference was statistically insignificant.

The difference between IFALE participants in Tigray (mean=4.67) and Amhara (mean=4.63) was statistically insignificant while in the case of non-participants in Tigray (mean=4.80) and Amhara (mean=4.62) it was statistically significant. The non-participants in Amhara seem to have more modern attitudes than in Tigray. This is probably because the group in Tigray is more highly influenced by local culture and parents. As regards the practice of wanting an additional child, the differences between IFALE participants in Tigray (mean=4.57) and Amhara (mean=4.03) as well as between the non-participants in Tigray (mean=4.72) and Amhara (mean=4.00) were statistically significant in favor of the groups in Amhara. The groups in Amhara seem to be more empowered probably because of lesser influence of the local culture.

7.3.9 Importance and Practice of Family Planning

IFALE participants (mean=4.99) and non-participants (mean=4.98) in Tigray rated family planning methods (birth spacing and methods of contraception) as almost *very important* (Table 7.14; Appendix E, survey question 64). This was also the case in Amhara (mean values of IFALE participants=4.67 and non-participants=4.54). The difference in each region was statistically insignificant, indicating that participation in IFALE hardly contributes to better knowledge of the importance of family planning. This is in contrast to other studies, for example Burchfield et al. (2002b) reporting that literacy program participants in Bolivia had more knowledge of family planning than non-participants.

In terms of family planning practice, the responses of IFALE participants (mean=4.19) and non-participants (mean=4.05) in Tigray were statistically insignificant. In Amhara,

(mean=3.93 and 3.48) the difference was statistically significant. This may mean that the IFALE program in Amhara probably contributes substantially to the empowerment of its participants in terms of the practice of controlling the number and spacing of children. This could be because of population and family planning which is included in the contents of the primer (ABE, 2011a; ABE, 2012a).

With respect to regional comparison, the difference in the importance of family planning between IFALE participants in Tigray (mean=4.99) and Amhara (mean=4.67) and between non-participants in Tigray (mean=4.98) and Amhara (mean=4.54) was statistically significant. As regards, family planning practices of IFALE participants in Tigray (mean=4.19) and Amhara (mean=3.93), the difference was statistically insignificant while for non-participants in Tigray (mean=4.05) and Amhara (mean=3.48) it was statistically significant.

7.3.10 Importance and Practice of Personal Hygiene

Survey participants rated the importance of personal hygiene, including regular baths, cleaning of teeth and cutting of nails (Table 7.14; Appendix E, survey question 65). In Tigray, IFALE participants (mean=5) and non-participants (mean=5) rated this as *very important*. In Amhara, IFALE participants (mean=4.53) and non-participants (4.34) rated it *very important* and *important*, respectively. The difference was statistically insignificant in both regions, but significant in favor of Tigray when comparing the two regions. IFALE participants (mean=4.05) and non-participants (mean=3.91) in Tigray rated their personal hygiene practices as *good*. IFALE participants (mean=3.05) and non-participants (mean=2.86) in Amhara rated their practices as *average*. The difference was statistically insignificant in both regions, but significant in favor of Tigray when comparing the two regions. This shows that IFALE does not make a difference in terms of improving knowledge of importance and practices of personal hygiene. IFALE does not, therefore, contribute to participants' essential capability of bodily health or having good health (Nussbaum, 2000). In contrast, adult literacy programs in India have resulted in positive changes in terms of hygiene practices (Vasumathi, 2008).

	statisti	ical sign	ificance			
					Statistical significance of the test	he difference based on t-
					T-tests for comparing IFALE	T-tests for regional
S.					participants and non-	differences, conducted
No.	Item	Region	Group	Mean	participants in each region	separately for each group
			IFALE participants	4.86	t=1.840*	IFALE participants
		Tigray	Non-participants	4.77	pr = 0.067	t=7.873*** pr=0.000
	family		IFALE participants	4.39	t=0.866	Non-participants
1	cohesion	Amhara	Non-participants	4.32	pr= 0.387	t=6.404*** pr=0.000
1			IFALE participants	4.42	t=1.617	IFALE participants
	Practice of	Tigray	Non-participants	4.30	pr=0.108	t=7.886*** pr=0.000
	family		IFALE participants	3.71	t=0.915	Non-participants
	cohesion	Amhara	Non-participants	3.62	pr=0.361	t=7.849*** pr=0.000
	Importance of		IFALE participants	1.50	t=-3.396***	IFALE participants
	particular	Tigray	Non-participants	1.70	pr=0.001	t=-0.470 pr=0.639
	traditional		IFALE participants	1.52	t=-3.146***	Non-participants
	activities	Amhara	Non-participants	1.73	pr=0.002	t=-0.390 pr=0.697
2	Practice of		IFALE participants	1.45	t=-3.127***	IFALE participants
	particular	Tigray	Non-participants	1.62	pr=0.002	t=-1.074 pr=0.284
	traditional		IFALE participants	1.51	t=-2.037**	Non-participants
	activities	Amhara	Non-participants	1.62	pr=0.043	t=-0.033 pr=0.974
	Importance of		IFALE participants	4.67	t=1.995**	IFALE participants
		Tigray	Non-participants	4.80	pr=0.048	t=0.543 pr=0.588
	wanting an		IFALE participants	4.63	t=0.201	Non-participants
2	additional child	Amhara	Non-participants	4.62	pr=0.841	t=3.068*** pr=0.003
3	Practice of the	L.	IFALE participants	4.57	t=-2.209**	IFALE participants
	reasons for	Tigray	Non-participants	4.72	pr=0.028	t=5.466*** pr=0.000
	wanting an additional child		IFALE participants	4.03	t=0.277	Non-participants t=8.442*** pr=0.000
	additional child	Amnara	Non-participants	4.00	pr=0.782	_
	T	T.	IFALE participants	4.99	t=0.749	IFALE participants
	Importance of family	Tigray	Non-participants	4.98	pr=0.455	t=4.218*** pr=0.000 Non-participants
	planning	A1	IFALE participants	4.67	t=1.244	t=6.099*** pr=6.099
4	plaining	Amhara	Non-participants	4.54	pr=0.215	÷
		m.	IFALE participants	4.19	t=0.904	IFALE participants
	Practice of	Tigray	Non-participants	4.05	pr=0.368	t=1.563 pr=0.120
	family planning	A 1	IFALE participants	3.93	t=2.629***	<u>Non-participants</u> t=3.435*** pr=0.001
	plaining	Amhara	Non-participants	3.48	pr=0.009	<u>^</u>
		 .	IFALE participants	5.00	t=0.958	IFALE participants
		Tigray	Non-participants	5.00	pr=0.340	t=6.442*** pr=0.000
	personal hygiene	Amhara	IFALE participants	4.53	t=1.915*	<u>Non-participants</u> t=8.903*** pr=0.000
5	nygiene	Amhara	Non-participants	4.34	pr=0.057	-
5	Durations	T	IFALE participants	4.05	t=1.218	IFALE participants
	Practice of	Tigray	Non-participants	3.91	pr=0.225	t=9.660*** pr=0.000 Non-participants
	personal hygiene	A	IFALE participants	3.05	t=1.954*	t=9.691*** pr=0.000
		Amhara	Non-participants	2.86	pr=0.052	1
	Importance of	Ti an	IFALE participants	4.98	t=0.483	IFALE participants
	maternal and child care	Tigray	Non-participants	4.97	pr=0.630	t=4.266*** pr=0.000 Non-participants
	cillu cale	Amhara	IFALE participants	4.58	t=0.670	$t=5.291^{***}$ pr=0.000
6	Draatiac of	Amhara	Non-participants	4.49	pr=0.504 t=1.539	_
0	Practice of maternal and	Tigray	IFALE participants	4.68	t=1.539 pr=0.126	<u>IFALE participants</u> t=6.945**** pr=0.000
	child care	Tigray	Non-participants IFALE participants	4.49 3.69	t=2.780***	Non-participants
		Amhara	Non-participants	3.69	pr=0.006	t=9.0830*** pr=0.000
		1 milai a	Non-participants	3.21	P1 0.000	•

Table 7.14: Self-reported socio-cultural empowerment gains of survey participants by mean and statistical significance

					Statistical significance of the difference based on t-					
					test					
					T-tests for comparing IFALE					
S.					participants and non-	differences, conducted				
No.	Item	Region	Group		participants in each region	separately for each group				
	Importance of		IFALE participants	4.93	t=1.292	IFALE participants				
	HIV/AIDS	Tigray	Non-participants	4.88	pr =0.198	t=4.679*** pr=0.000				
	control		IFALE participants	4.63	t=1.520	Non-participants				
	measures	Amhara	Non-participants	4.50	pr=0.130	t=5.657*** pr=0.000				
~	Practice of	Tigray	IFALE participants	3.80	t=0.747	IFALE participants				
7	HIV-AIDS		Non-participants	3.74	pr=0.456	t=5.568*** pr=0.000				
	control	Amhara	IFALE participants	3.20	t=0.205	Non-participants				
	measures		Non-participants	3.18	pr=0.838	t=5.797*** pr=0.000				
			Non-participants	4.09						
			IFALE participants	3.23	t=0.781					
		Amhara	Non-participants	3.10	pr=0.436					
			IFALE participants	4.43	t=1.060	IFALE participants				
	Importance of	Tigray	Non-participants	4.35	pr=0.290	t=3.947*** pr=0.000				
	social justice		IFALE participants	4.12	t=2.586***	Non-participants				
	activities	Amhara	Non-participants	3.91	pr=0.011	t=5.493*** pr=0.000				
8			IFALE participants	3.85	t=1.177	IFALE participants				
	Practice of	Tigray	Non-participants	3.72	pr=0.241	t=7.410*** pr=0.000				
	social justice		IFALE participants	3.09	t=3.421***	Non-participants				
	activities	Amhara	Non-participants	2.78	pr=0.001	t=9.307*** pr=0.000				
			IFALE participants	4.90	t=2.498**	IFALE participants				
	Importance of	Tigray	Non-participants	4.80	pr=0.013	t=7.992*** pr=0.000				
	democratic	Amhara	IFALE participants	4.45	t=2.450**	Non-participants				
	culture		Non-participants	4.26	pr=0.015	t=8.199*** pr=0.000				
0			IFALE participants	4.55	t=1.720*	IFALE participants				
9	Practice of	Tigray	Non-participants	4.41	pr=0.087	t=9.215*** pr=0.000				
	democratic		IFALE participants	3.67	t=2.780***	Non-participants				
	culture	Amhara	Non-participants	3.37	pr=0.006	t=11.445*** pr=0.000				
	Importance of		IFALE participants	4.68	t=-0.518	IFALE participants				
	conflict	Tigray	Non-participants	4.72	pr=0.605	t=-0.956 pr=0.340				
	resolution		IFALE participants	4.74	t=1.517	Non-participants				
10	methods	Amhara	Non-participants	4.65	pr=0.131	t=1.061 pr=0.290				
10	Practice of	L.	IFALE participants	4.09	t=-0.002	IFALE participants				
	conflict	Tigray	Non-participants	4.09	pr=0.998	t=5.088*** pr=0.000				
	resolution		IFALE participants	3.23	t=0.781	Non-participants				
	methods	Amhara	Non-participants	3.10	pr=0.436	t=5.994*** pr=0.000				

***=level of significance at 1%, ** level of significance at 5 % and *level of significant at 10%

7.3.11 Importance and Practice of Maternal and Child Care

Maternal and child health care in this study refers to the efforts made to promote therapeutic, rehabilitation or care in order to reduce mortality and morbidity among mothers, infants and pre-school children. IFALE participant and non-participant groups in both regions rated the importance of maternal and child care, i.e. pre- and post-natal care (Table 7.14; Appendix E, survey question 66). IFALE participants (mean=4.98) and non-participants (mean=4.97) in Tigray rated this as nearly *very important*. IFALE participants (mean=4.58) and non-participants (mean=4.49) in Amhara rated it as roughly *very important* and *important*, respectively. In both regions, the differences were statistically insignificant. This may imply that IFALE did not contribute to participants' understanding of the importance of maternal and child care. This is contrary to other studies showing the importance of adult literacy for

raising awareness of health problems (Cameron & Cameron, 2005; Robinson-Pant, 2005; Farah, 2005). Vasumathi (2008) reports the positive contribution of adult literacy programs to knowledge of the importance of pre- and post-natal care in India.

When comparing maternal and child care practices of IFALE participants (mean=4.68) and non-participants (mean=4.49) in Tigray, the difference was statistically insignificant. In Amhara, the difference between IFALE participants (mean=3.69) and non-participants (mean=3.27) was statistically significant which may be due to the IFALE efforts.

With respect to the importance of maternal and child care across the regions, the differences between IFALE participants in Tigray (mean=4.98) and Amhara (mean=4.58) as well as of non-participants of IFALE in Tigray (mean=4.97) and Amhara (mean=4.49) were statistically significant. As to practices of maternal and child care, differences between IFALE participants in Tigray (mean=4.68) and Amhara (mean=3.69) as well as of non-participants in Tigray (mean=4.49) and Amhara (mean=3.69) as well as of non-participants in Tigray (mean=4.49) and Amhara (mean=3.27) were statistically significant in favor of the groups in Tigray.

7.3.12 Importance and Practice of Measures of HIV/AIDS Control

Survey participants self-reported on the importance and practices of abstinence, sterilization of shared sharp objects before use, condom use and faithfulness as HIV/AIDS control measures (Table 7.14; Appendix E, survey question 68). The ratings of IFALE participants (mean=4.93) and non-IFALE participants (mean=4.88) in Tigray and of IFALE participants (mean=4.62) and non-IFALE participants (mean=4.50) in Amhara were *very important*, with insignificant statistical difference in each region. It seems that IFALE did not contribute to the knowledge of the importance of the HIV/AIDS measures.

The ratings given to the practice of HIV/AIDS measures by IFALE participants (mean=3.80) and non-participants (mean=3.74) in Tigray were nearly *good* while those by IFALE participants (mean=3.20) and non-participants (mean=3.18) in Amhara were approximately *average*. The difference between the two groups in each region was statistically insignificant indicating the likely absence of an IFALE role. This is contrary to the claim that literacy improves awareness of health threats in the physical environment (Cameron & Cameron, 2005), can contribute to improved knowledge (Burchfield et al., 2002a) and practice of measures of HIV/AIDS (DFID, 2002) as ways of improving livelihoods.

The difference between the ratings of IFALE participants in Tigray (mean=4.93) and Amhara (mean=4.63), and of non-IFALE participants in Tigray (mean=4.88) and Amhara (mean=4.50) on the importance of HIV/AIDS control measures was statistically significant. As to practices of HIV/AIDS control mechanisms, the differences between IFALE participants in Tigray and Amhara and between the non-IFALE participants in both regions were statistically significant in favor of the groups in Tigray.

7.3.13 Importance and Practice of Social Justice

Social justice here refers to fairness, equality, equity, and opportunities for participation in personal development, social and community life and decision making for all members of a society. The survey participants self-reported on the importance of a number of social justice activities, including, amongst others, marriage arrangements with other social groups in their communities, fair treatment of all people irrespective of religious differences, encouragement of female participation in all spheres of life, male participation in traditional female areas, female participation in traditional male activities, engagement in social activism or social change in the community, and supporting the needy (Appendix E, survey question 70).

In Tigray, IFALE participants (mean=4.43) and non-participants (mean=4.35) rated social justice activities as *important* (Table 7.14). The difference was statistically insignificant. This means that IFALE seems not to play a role in raising participants' awareness of the importance of social justice activities. In Amhara, IFALE participants (mean=4.12) and non-participants (mean=3.91) rated the activities as approximately *important*. The difference was statistically significant. In Amhara IFALE appears to contribute to participants' understanding of the importance of social justice activities in their everyday life.

Survey participants also reported their practices of social justice activities. IFALE participants (mean=3.85) and non-participants (mean=3.72) in Tigray represented them as approximately *good*. The difference was statistically insignificant. IFALE participants (mean=3.09) and non-participants (mean=2.78) in Amhara rated their own practice of social justice activities as *average*. The difference was statistically significant. This may show the vital role played by IFALE in participants' practices of fair treatment of all kinds of people, male participation in traditional female areas and vice versa, and supporting the needy.

Concerning the importance of social justice activities, the difference between IFALE participants in Tigray (mean=4.43) and Amhara (4.12), as well as between non-participants in Tigray (mean=4.35) and Amhara (3.91) was statistically significant. In terms of practice of

social justice activities, the difference between IFALE participants in Tigray (mean=3.85) and Amhara (mean=3.09), and between non-participants in Tigray (mean=3.72) and Amhara (mean=2.78) was statistically significant in favor of Tigray.

7.3.14 Importance and Practice of Democratic Culture

A number of activities concern the importance and practice of democratic culture. They include, amongst others, ability to distinguish facts from opinions, use of participatory decision-making, willingness to listen to others and to reflect on one's actions, communication at an eye level regardless of age, respect for difference in perceptions and beliefs, for diversity, and for everyone's civil and legal rights (Appendix E, survey question 72).

IFALE participants (mean=4.90) and non-participants (mean=4.80) in Tigray rated democratic culture activities as nearly *very important* whereas in Amhara IFALE participants (mean=4.45) and non-participants (mean=4.26) rated the activities as approximately *important* (Table 7.14). The differences were statistically significant and may be due to the IFALE program.

In terms of practice of democratic culture activities, the mean value difference between IFALE participants (mean=4.55) and non-participants (mean=4.41) in Tigray was statistically insignificant, indicating that IFALE was not a contributing factor. The difference between IFALE participants (mean=3.67) and non-participants (3.37) in Amhara was statistically significant. IFALE in Amhara possibly served as a contributing factor to participants' enhanced practices of democratic culture activities, such as use of participatory decision-making, willingness to listen to others and to reflect on one's actions, respect for difference in perceptions and beliefs, for diversity, and for everyone's civil and legal rights. This is important to change autocratic ways of solving problems.

Concerning the importance of democratic culture activities, the difference between the mean values of IFALE participants in Tigray (mean=4.90) and Amhara (mean=4.45), as well as of non-participants of IFALE in Tigray (mean=4.80) and Amhara (mean=4.26) was statistically significant. In terms of the practices of democratic culture activities, the differences between IFALE participants (mean=4.55 in Tigray and mean=3.67 in Amhara) and non-participants (mean=4.4 in Tigray and mean=3.37 in Amhara) were also statistically significant.

7.3.15 Importance and Practice of Conflict Resolution

The importance and practice of conflict resolution included: direct discussion with the other party; negotiation (compromise); apology; and arbitration by elders or formal litigation (Appendix E, survey question 74). The IFALE participants (mean=4.68) and non-participants (mean=4.72) in Tigray found conflict resolution techniques to be *very important*. IFALE participants (mean=4.74) and non-participants (mean=4.65) in Amhara also rated them as *very important* (Table 7.14). The difference in each region was statistically insignificant. The findings imply that IFALE did not play a role in enhancing participants' knowledge of the importance of conflict resolution techniques.

In terms of practices, in Tigray the ratings of IFALE participants (mean=4.09) and nonparticipants (mean=4.09) were *good*. In Amhara IFALE participants (mean=3.23) and nonparticipants (mean=3.10) rated their practices as *average*. The difference in each region was statistically insignificant, meaning that IFALE was probably inconsequential to conflict resolution practice.

For the importance of conflict resolution methods, the differences between IFALE participants (mean=4.68 in Tigray and 4.74 in Amhara) and between non-participants (mean=4.72 in Tigray and 4.65 in Amhara) were statistically insignificant. Comparisons between IFALE participants in Tigray (mean=4.09) and Amhara (mean=3.23) as well as between non-participants in Tigray (mean=4.09) and Amhara (mean=3.09) showed a statistically significant difference in favor of the groups in Tigray, probably implying that local leaders and local media played stronger roles in Tigray than in Amhara (refer Table 7.15).

7.3.16 Sources of Socio-Cultural Empowerment Gains

The survey groups reported their sources of knowledge and practice of the selected indicators for socio-cultural empowerment: family cohesion, harmful traditional activities, wanting an additional child, personal hygiene, family planning, maternal and child care, HIV/AIDS control mechanisms, social justice activities, democratic culture activities, and conflict resolution methods (Table 7.15).

Regarding knowledge and practice of family cohesion (Appendix E, survey question 59), local culture and IFALE were the most important sources for IFALE participants in both regions. While IFALE was most important for IFALE participants in Amhara (77%), IFALE

participants in Tigray ranked local culture highest (96%). Local culture inherited over generations seems to be the most important source of knowledge and practice of family cohesion for both IFALE participants and non-participants in Tigray. This may also be the reason for the substantial difference in favor of Tigray between IFALE participants and non-participants of the two regions.

As to the sources of knowledge and practices of traditional harmful activities (Table 7.15; Appendix E, survey question 61), IFALE participants rated IFALE (Tigray 93%; Amhara 90%), local leaders (Tigray 76%; Amhara 71%), health extension workers (Tigray 72%; Amhara 65%), and the media (Tigray 38%; Amhara 33%) first, second, third and fourth. This shows IFALE as the most important source of knowledge on harmful traditional practices as well as on measures taken against them.

Table 7.15 also helps to understand the crucial factors that contributed to increasing the knowledge and practice of survey participants for wanting an additional child (Appendix E, survey question 63). Ratings by IFALE participants in both regions show that IFALE (Tigray 46%; Amhara 41%) had a less important role than parents (Tigray 96%; Amhara 76%), and local culture (Tigray 97%; Amhara 69%). Local culture and parents in Tigray play stronger roles than in Amhara. This may explain that both groups in Tigray want significantly more to have an additional child than in Amhara.

In terms of personal hygiene, family planning and maternal and child care (Table 7.15; Appendix E, survey question 67), IFALE in Amhara was the most important source of knowledge and practice for participants (85%) followed by health extension workers (75%). IFALE in Tigray is the second most important source (84%) following health extension workers (94%). IFALE participants and non-participants in Tigray rated local leaders, the media and health extension workers higher than in Amhara. This probably explains the significantly higher practices of personal hygiene, family planning and maternal and child care among IFALE participants and non-participants in Tigray than in Amhara.

As to the sources of knowledge and practice of HIV/AIDS control mechanisms (Table 7.15; Appendix E, survey question 69), IFALE participants in the two regions rated IFALE (Tigray 91%; Amhara 86%) and health extension workers (Tigray 93%; Amhara 83%) as the most important. IFALE participant and non-participant groups in both regions stressed health extension workers as the main source of knowledge and practice of HIV/AIDS control measures. IFALE participants and non-participants in Tigray claimed that local leaders and

local media were more important sources than in Amhara which may explain the significant difference in favor of IFALE participants and non-participants in Tigray.

		requercy and percent	IFALE parti		Non-partici	
S.			frequency an	d percent	frequency a	nd percent
No	Item	Sources	Tigray	Amhara	Tigray	Amhara
		IFALE	51 (57%)	72 (77%)	n/a	n/a
	Knowledge and	Health extension workers	22 (25%)	14 (15%)	20 (21%)	12 (12%)
1	practice of	Local culture	85 (96%)	54 (58%)	94 (97%)	80 (82%)
	family cohesion	Media	19 (21%)	11 (12%)	14 (14%)	6 (6%)
		Others	6 (7%)	7 (8%)	13 (13%)	28 (29%)
		IFALE	83 (93%)	84 (90%)	n/a	n/a
	Knowledge and	Agricultural extension workers	21 (24%)	16 (17%)	17 (18%)	9 (9%)
	practice of	Health extension workers	64 (72%)	60 (65%)	62 (64%)	67 (69%)
2	harmful	Local leaders	68 (76%)	66 (71%)	80 (82%)	76 (78%)
	traditional	Media	34 (38%)	31 (33%)	23 (24%)	20 (21%)
	activities	Others	3 (3%)	6 (6%)	6 (6%)	28 (29%)
		IFALE	41 (46%)	38 (41%)	n/a	n/a
	Knowledge and	Health extension workers	16 (18%)	1 (1%)	11 (11%)	3 (3%)
	practice of	Local culture	86 (97%)	64 (69%)	95 (98%)	74 (76%)
3	wanting an	Local leaders	18 (20%)	8 (9%)	21 (22%)	17 (18%)
	additional child	Parents	85 (96%)	71 (76%)	95 (98%)	72 (74%)
		Media	8 (9%)	2 (2%)	4 (4%)	0 (0%)
		Others	0 (0%)	2 (2%)	0 (0%)	7 (7%)
	Knowledge and	IFALE	78 (84%)	79 (85%)	n/a	n/a
	practice of	Agricultural extension workers	15 (17%)	7 (8%)	13 (13%)	8 (8%)
	personal hygiene,		84 (94%)	70 (75%)	88 (91%)	76 (78%)
4	family planning,	Local leaders	43 (48%)	22 (24%)	62 (64%)	37 (38%)
	and maternal and	Media	35 (39%)	17 (18%)	28 (29%)	14 (14%)
	child care	Others	2 (2%)	2 (2%)	1 (1%)	15 (15%)
		IFALE	81 (91%)	80 (86%)	n/a	<u>n/a</u>
	Knowledge and	Agricultural extension workers	15 (17%)	10 (11%)	11 (11%)	9 (9%)
	practice of	Health extension workers	83 (93%)	77 (83%)	88 (91%)	83 (86%)
5	HIV/AIDS	Local leaders	47 (53%)	27 (29%)	66 (68%)	44 (45%)
	control	Media	38 (43%)	24 (26%)	33 (34%)	14 (14%)
	mechanisms	Others	2 (2%)	3 (3%)	2 (2%)	8 (8%)
		IFALE	78 (88%)	64 (69%)	n/a	n/a
	Knowledge and	Agricultural extension workers	35 (39%)	11 (12%)	28 (29%)	6 (6%)
	practice of	Health extension workers	37 (42%)	21 (23%)	37 (38%)	15 (15%)
6	social justice	Local leaders	64 (72%)	38 (41%)	71 (73%)	66 (68%)
0	activities	Media	37 (42%)	16 (17%)	28 (29%)	16 (16%)
		Others	11 (12%)	14 (15%)	26 (27%)	24 (25%)
		IFALE	79 (89%)	81 (87%)	n/a	n/a
	Knowledge and	Agricultural extension workers	24 (27%)	13 (14%)	19 (20%)	11 (11%)
	practice of		34 (38%)	20 (22%)	24 (25%)	17 (18%)
7	democratic	Health extension workers Local leaders	59 (66%)	43 (46%)	77 (79%)	72 (74%)
'	culture				25 (26%)	
	Culture	Media Others	<u>36 (40%)</u> 9 (10%)	14 (15%) 3 (3%)	<u>25 (26%)</u> 13 (13%)	12 (12%)
					. ,	20 (21%)
	Unovilates or 1	IFALE	78(88%)	69 (74%)	n/a	$\frac{n/a}{10(100/)}$
0	Knowledge and	Agricultural extension workers	21 (24%)	12 (13%)	16 (16%)	10 (10%)
8	practice of conflict	Health extension workers	19 (21%)	11 (12%)	13 (13%)	10 (10%)
	resolution	Local leaders	59 (66%)	49 (53%)	79 (81%)	68 (70%)
	resolution	Media	30 (34%)	12 (13%)	20 (21%)	12 (12%)
		Others	10 (11%)	8 (9%)	14 (14%)	27 (28%)

 Table 7.15:
 Survey participants' self-reported sources of knowledge and practice of socio-cultural gains by frequency and percent

In respect of the sources of knowledge and practice of social justice activities (Table 7.15; Appendix E, survey question 71), IFALE participants in both regions rated IFALE (Tigray 88%; Amhara 69%) and local leaders (Tigray 72%; Amhara 41%) as the first and second

most important sources. Agricultural extension workers, health extension workers and local leaders in Tigray contributed more to the knowledge and practice of IFALE participants and non-participants than was the case in Amhara. Stronger engagement of the three sources can explain the significant difference in knowledge and practice of social justice activities in favor of the IFALE participant and non-participant groups in Tigray.

The majority of IFALE participants rated IFALE (Tigray 89%; Amhara 87%) as the main source of knowledge and practices of democratic culture activities (Table 7.15; Appendix E, survey question 73), followed by local leaders (Tigray 66%; Amhara 46%). The local media seems to play a more important role in Tigray than in Amhara and may explain the significant difference in favor of the IFALE participants and non-participants in Tigray.

As regards the sources of knowledge and practice of conflict resolution methods (Table 7.15; Appendix E, survey question 75), IFALE participants in both Tigray and Amhara rated IFALE highest. Local leaders (81% of the non-IFALE participants in Tigray; 70% of non-IFALE participants in Amhara) and the local media (34% of IFALE participants in Tigray; 13% in Amhara) seem to play a more important role in Tigray than in Amhara and could explain the significant difference between the survey groups in the two regions.

In sum, for both IFALE participants and non-participants local leaders, local media and health extension seem to be more important sources of knowledge and practice in many areas of socio-cultural empowerment in Tigray than in Amhara. This may explain the significant difference between the two regions.

7.3.17 **Summary**

The comparison between IFALE participants and non-participants in each region with regard to socio-cultural empowerment shows that IFALE makes an important contribution in terms of practices of traditional activities and democratic culture activities among participants in both Tigray and Amhara. In Tigray, IFALE has played a substantial role in changing its participants' reasons for wanting an additional child. In Amhara, IFALE has contributed significantly to participants in terms of practices of maternal and child care, family planning and social justice activities. Likewise, IFALE results in considerable changes in achieving gender equality of most decisions within the family among its married participants in the region.

However, in the two regions IFALE has not made an important contribution to improving socio-cultural empowerment related to practices of personal hygiene, HIV/AIDS control

measures, family cohesion, and conflict resolution or to knowledge of the role of gender equality for development, and knowledge of laws against gender discrimination.

The regional comparison shows that both IFALE participants and non-participants in Tigray had higher practice of family cohesion, personal hygiene, maternal and child care, measures of HIV/AIDS control, social justice activities, democratic culture actions, and conflict resolution methods, as well as knowledge of laws protecting women and children and against gender discrimination. But this knowledge cannot be categorically attributed to the IFALE program since other sources, such as local leaders and local media, were influential.

7.4 IFALE and Psychological Empowerment

In this study, psychological empowerment is related to the concepts of self-concept and self-efficacy discussed in other literature (for example Kagitcibasi et al., 2005). The indicators for the self-concept are: being pleased with oneself; being a good husband/wife; being pleased with one's appearance; being a good mother/father; being a good son/daughter; being knowledgeable, skillful and successful in life; and being a resourceful contributor to the community (Appendix E, survey question 76). As regards self-efficacy, the indicators are: buying items in the market; buying food in a restaurant; going to the church/mosque; going to the clinic/hospital; taking the bus/taxi/rickshaw (*bajaj*); making transactions in a (rural) bank; paying bills; and crossing the bridge (Appendix E, survey question 78).

7.4.1 Importance and Practice of Self-Concept

IFALE participants (mean=4.95) and non-participants (mean=4.96) in Tigray rated the selfconcept as almost *very important* (Table 7.16). The difference is statistically insignificant. In Amhara, IFALE participants (mean=4.64) and non-participants (mean=4.54) rated the selfconcept as nearly *very important*. The difference is statistically insignificant.

The practice of self-concept was rated above *good* by IFALE participants (mean=4.42) and non-participants (mean=4.28) in Tigray. In Amhara, IFALE participants (mean=3.87) and non-participants (mean=3.70) found it nearly *good* (Table 7.16). The difference in both regions was statistically insignificant. This indicates that IFALE did not enhance self-concept practices meaningfully. By contrast, the impact assessment study of functional adult literacy for women in Turkey shows that their self-concept improved significantly (Kagitcibasi et al., 2005).

In relation to the importance of self-concept, the differences between IFALE participants in Tigray (mean=4.95) and Amhara (mean=4.64) as well as between non-IFALE participants in Tigray (mean=4.96) and Amhara (4.54) were statistically significant. The difference between IFALE participants in Tigray (mean=4.42) and Amhara (mean=3.87) as well as the difference between non-IFALE participants in Tigray (mean=4.42) and Amhara (mean=3.87) as well as the difference between non-IFALE participants in Tigray (mean=4.28) and Amhara (mean=3.70) corresponding to self-concept practices were also statistically significant. This means that local culture and parents probably played a stronger role in Tigray (refer Table 7.17).

7.4.2 Importance and Practice of Self-Efficacy

In respect of the importance of self-efficacy, the ratings of IFALE participants and non-participants in Tigray (mean=4.18 and 3.96, respectively) were *important*, while in Amhara (mean=3.24 and 3.11, respectively) they were roughly *average* (Table 7.16). There was a statistically insignificant difference between IFALE participant and non-participant groups in each region.

In terms of self-efficacy practices in Tigray, IFALE participants rated these as nearly *high* (mean=3.80) whereas the non-participants set them above *average* (mean=3.46). In the case of Amhara, IFALE participants and non-participants reported their self-efficacy practice to be nearly *average* (mean=2.74) and approximately *poor* (mean=2.47), respectively. The differences in both regions were statistically significant, indicating that IFALE contributed much in terms of building participants' confidence. Similar studies, for example in El Salvador, Bangladesh and Uganda (Archer & Cottingham, 1996), report that adult literacy education contributes to considerable changes in self-efficacy. However, an impact assessment study of functional adult literacy on women's empowerment in Turkey (Kagitcibasi et al., 2005) did not result in improved self-efficacy.

As regards the importance of self-efficacy, the difference between IFALE participants in Tigray (mean=4.18) and Amhara (mean=3.24) and between non-participants in Tigray (mean=3.96) and Amhara (mean=3.11) was statistically significant. Regarding self-efficacy practice, the differences between IFALE participants in Tigray (mean=3.80) and Amhara (mean=2.74) as well as between non-participants in Tigray (mean=3.46) and Amhara (mean=2.47) were also statistically significant. This may be explained by the efforts of local leaders and health extension workers in Tigray (refer Table 7.17).

					Statistical significance of the difference based on t-test and p-value		
S.					T-tests for comparing IFALE participants and non-	T-tests for regional differences, conducted	
No.	Item	Region	Group	Mean		separately for each group	
	Importance of self- concept	Tigray	IFALE participants	4.95	t=-0.766	IFALE participants	
			Non- participants	4.96	pr =0.445	t=6.379*** pr=0.000	
			IFALE participants	4.64	t=1.542	Non-participants	
		Amhara	Non-participants	4.54	pr=0.125	t=8.967*** pr=0.000	
1	Practice of		IFALE participants	4.42	t=1.836*	IFALE participants	
1	self- concept	Tigray	Non-participants	4.28	pr=0.068	t=5.994*** pr=0.000	
			IFALE participants	3.87	t=1.712*	Non-participants	
		Amhara	Non-participants	3.70	Pr=0.089	t=6.644*** pr=0.000	
	Importance		IFALE participants	4.18	t=1.554	IFALE participants	
	of self- efficacy	Tigray	Non-participants	3.96	Pr=0.122	t=7.155*** pr=0.000	
			IFALE participants	3.24	t=1.038	Non-participants	
		Amhara	Non-participants	3.11	Pr=0.301	t=6.506*** pr=0.000	
2			IFALE participants	3.80	t=2.327**	IFALE participants	
	Practice of self- efficacy	Tigray	Non-participants	3.46	Pr=0.021	t=8.108*** pr=0.000	
			IFALE participants	2.74	t=2.492**	Non-participants t=7.728*** pr=0.000	
		Amhara	Non-participants	2.47	Pr=0.014		

 Table 7.16:
 Self-reported psychological empowerment gains of survey participants by mean and statistical significance

***=level of significance at 1%, ** level of significance at 5 %, *level of significant at 10%

7.4.3 Sources of Knowledge and Practice of Self-Concept and Self-Efficacy

IFALE participants in Tigray ranked local culture (83%) followed by parents (82%) and IFALE (75%) as the most important sources of their self-concept knowledge and practice (Table 7.17; Appendix E, survey question 77). In Amhara, IFALE participants rated IFALE (73%), followed by parents (58%) and local culture (56%) as the most important sources. The role played by local culture and parents in raising the practice of the self-concept of the survey participants seems to be higher in Tigray than in Amhara and may explain the significant self-concept practice results in Tigray.

IFALE and local leaders seem to be the most important sources of the self-efficacy knowledge and practice (Appendix E, survey question 79). Most IFALE participants in both regions found IFALE (Tigray 73%; Amhara 60%) to be an important source of self-efficacy knowledge and practice. Local leaders were also important for IFALE participants and non-participants, but more so in Tigray than in Amhara. The efforts made, amongst others, by the local leaders may explain the more highly improved self-efficacy practice in Tigray compared to Amhara. A greater number of non-IFALE participants than IFALE participants in Amhara had other self-efficacy knowledge and practice sources, such as friends, parents, and neighbors.

		Sources	IFALE participants		Non-participants	
S. No	Item		Tigray	Amhara	Tigray	Amhara
1	Knowledge and	IFALE	67 (75%)	68 (73%)	n/a	n/a
	practice of self-	Agricultural extension	22 (25%)	11 (12%)	16 (16%)	9 (9%)
	concept	workers				
	_	Local leaders	33 (37%)	37 (40%)	38 (39%)	28 (29%)
		Local culture	74 (83%)	52 (56%)	91 (94%)	62 (64%)
		Parents	73 (82%)	54 (58%)	83 (86%)	72 (74%)
		Media	13 (15%)	8 (9%)	6 (6%)	6 (6%)
		Others	3 (3%)	7 (8%)	3 (3%)	15 (15%)
2	Knowledge and	IFALE	65 (73%)	60 (65%)	n/a	n/a
	practice of self-	Health extension workers	25 (28%)	7 (8%)	16 (16%)	5 (5%)
	efficacy	Local leaders	54 (61%)	33 (35%)	50 (52%)	39 (40%)
		Media	23 (26%)	16 (17%)	13 (13%)	7 (7%)
		Others*	17 (19%)	21 (23%)	26 (27%)	50 (52%)

 Table 7.17:
 Survey participants' self-reported sources of knowledge and practice of self-concept and self-efficacy by frequency and percent

*= Friends, parents, neighbors, etc.

7.4.4 Summary

The comparison between self-reports of IFALE participants and non-participants shows that IFALE has made a significant contribution to increasing its participants' practices of self-efficacy in Tigray and Amhara, but not considerably to their self-concept. Both IFALE participants and non-participants in Tigray have better practices of both self-concept and self-efficacy as indicators of psychological empowerment than the groups in Amhara.

7.5 Conclusion

As discussed in Chapter 6, the supporting institutions of the IFALE policy, e.g. agriculture and rural development, health, women's affairs, youth and social affairs, participated in the preparation of curriculum materials, but largely not in the actual teaching and learning sessions of IFALE. The findings in this chapter regarding the contributions of IFALE to various forms of empowerment should, therefore, be related to the education sector.

The political empowerment gains of IFALE participants in Tigray are higher than in Amhara. The results of the IFALE participant and non-participant self-reports in Tigray show important gains in most of the major political empowerment indicators (7 of the 10, 70%) used in the study. These were: knowledge of the voting age; knowledge of different political parties; interest in becoming an elected representative; knowledge of a law on electing representatives; confidence in terms of becoming an elected representative; participation in decision making in public institutions; and participation in community development activities. In Amhara, IFALE made a meaningful contribution in less than half of the political empowerment indicators (4 of 10, i.e. 40%): knowledge of the voting age; knowledge of

different political parties; interest in becoming an elected representative; and voting participation.

These results imply that there was more emphasis on learning political issues in Tigray than in Amhara. The political empowerment gains in the two regions represent two of Nussbaum's (2000) important capabilities: affiliation to others; and control over one's environment. Affiliation to others manifests as IFALE participants' significant concern for issues that help them live with others in their surroundings, for example participation in pulic institutions and community development activities. Control over the environment is expressed in terms of their ability to participate in politics, such as their interest and participation in voting.

As regards economic empowerment gains, the results for IFALE participants of the two regions are generally poor, but comparatively better in Tigray than in Amhara. The results of the IFALE participants' and non-participants' self-reports show considerable gains in only less than half of the major economic empowerment indicators. These bring important changes among IFALE participants in terms of saving practice and involvement in income generation schemes in Tigray (2 of 6, i.e. 33%), but only of saving practice in Amhara (1 of 6, i.e., 17%). This suggests that immediate economic gains from IFALE are low in both regions. However, other expressions of economic empowerment (improved livelihoods) to which IFALE significantly contributed included improved saving and involvement in income generation schemes. These could result in the capability of participants to own property (Nussbaum, 2000) and, therefore, according to Sen's (1999) capabilities approach, to development that entails increased income and wealth.

In terms of socio-cultural empowerment, the results of self-reports by IFALE participants and non-participants show that the IFALE program played a role in achieving significant gains in only 2 of 14 indicators among participants in Tigray (14%) and in 6 of 14 indicators (43%) in Amhara. In Tigray, IFALE showed positive results in participants' practices of traditional activities, and reasons for wanting an additional child. In Amhara, IFALE contributed positively to its learners in respect of the following indicators: practices of traditional activities, democratic culture activities, maternal and child care, family planning and social justice activities; and achievement of gender equality in most decision making areas within the family. IFALE has advanced more in Amhara than in Tigray with curriculum materials and IFALE teachers in Amhara emphasizing social and cultural issues in their teaching.

As a means to an end for some essential socio-cultural empowerment gains, IFALE represents some of Nussbaum's (2000) vital capabilities. IFALE appears to have helped participants, especially in Amhara, to think, reason and use thought connection, form a conception of the good and start critically reflecting on it (e.g. actions against specific traditional practices), achieve bodily health for mothers and children (e.g. maternal and child care and family planning), and have concern for and live with others in the environment and have practical reason (e.g. democratic culture and social justice activities).

S.		Empowerment gains			
No.	Dimension	Tigray	Amhara		
1	Political	7 of 10 indicators (70%)	4 of 10 indicators (40%)		
		High	Moderate		
2	Economic	2 of 6 indicators (33%)	1 of 6 indicators (17%)		
		Limited	Very limited		
3	Socio-cultural	2 of 14 indicators (14%)	6 of 14 indicators (43%)		
		Very limited	Moderate		
4	Psychological	1 of 2 indicators (50%)	1 of 2 indicators (50%)		
		Moderate	Moderate		
	Overall	12 of 32 indicators (38%)	12 of 32 indicators (38%)		
		Limited	Limited		

Table 7.18: IFALE's contribution to empowerment in Tigray and Amhara

Key: Vey high ≥80%; High 60-79%; Moderate 40-59%; Limited 20-39%; Very limited ≤19%

As to psychological empowerment, IFALE contributes considerably to developing participants' self-efficacy but not self-concept in Tigray and Amhara (1 of 2, i.e., 50%). IFALE played a vital role in participants' general development of self-confidence in a way that, according to Stromquist (2009), is a precondition for the enhancement of other domains of empowerment. Improved self-efficacy or confidence gains of IFALE participants in the two regions is an important capability enhancement that can be associated with one of Nussbaum's (2000) core capabilities: the capability to imagine, think, reason and use thought connection.

In sum, IFALE made important contributions to the empowerment of IFALE participants in 12 of the 32 indicators (about 38%) in each of the two regions which is considered to be limited (Table 7.18). Further advancement would depend on the concerted efforts of all concerned providers. The deficiencies in the implementation process, in particular the absence of the support factors in terms of a relevant curriculum and of building an efficient institutional system (as discussed in Chapter 6), have undoubtedly affected the contribution of IFALE to empowerment with possible implications for SD as well.

The difference between IFALE participants and between non-participants shows significant gains in Tigray in 5 of 10 political empowerment indicators (50%), in 4 of 6 economic empowerment indicators (67%), in 10 of 14 socio-cultural empowerment indicators (71%),

and in 2 of 2 psychological empowerment indicators (100%). In total, the IFALE participants and non-IFALE participants in Tigray had empowerment gains in 66% of the indicators. This may be because other parties than IFALE, including local leaders, agricultural extension workers, local media (especially the radio), FTCs, and women's associations, local culture and parents also played important roles. They served as independent sources of political and economic empowerment gains. This underlines that, in addition to IFALE, other non-formal and informal ways of learning are vital sources of empowerment. It confirms the claim by Walters et al. (2012), and Cooper and Walters (2011) that all forms of education and learning often work together, making it difficult to distinguish one from the other.

Having clarified that IFALE has led to empowerment gains in some of the vital areas for each of the empowerment dimensions, although differently in the two regions, the next chapter examines whether such empowerment has contributed to SD as well.

Chapter 8 IFALE as Education for Sustainable Development (ESD)

This chapter analyzes the interrelationship amongst IFALE, empowerment and SD. It is based on an analysis of the understanding of ESD in the policy and curriculum documents seen in relation to the empowerment gains from IFALE (Chapter 7). The document analysis is based on the ETP, the NAES, the National IFALE Curriculum Framework, the IFALE primers in Amhara, and the IFALE facilitator manuals in Amhara and Tigray. The document analysis is supplemented with data from observations of individual learners in order to examine the extent to which IFALE teaching of ESD is applied in practice.

8.1 IFALE and ESD

According to Huckle (2012), learning for sustainability can ensure a more viable future for humans:

.....learning for sustainability is essentially about learning to value sustainable relations between people (social relations); between people and the rest of the bio-physical world (environmental relations); and between the elements that make up that non-human world (ecological relations). It is also about considering dominant and alternative forms of technology and social organization (political economy) and their potential to foster such sustainable relations. Changing social relations, of economic, political, and cultural power, shape (and are shaped by) environmental and ecological relations, and people are more likely to realize their common interest in sustainable relations if social relations are just and democratic (pp. 35-36).

Learning for sustainability puts the understanding of social, environmental, and ecological relations as well as the principal and possible forms of political economy at the center of the teaching and learning process in order to fulfill the needs of present and future generations. ESD is, thus, about learning for sustainability or abandoning unsustainable ways of life. The provision of IFALE to citizens who have received no other educational opportunity is a necessary contribution to SD and addresses equity and human rights concerns (MoE, 2010, p. 2).

As discussed in Chapter 3, UNESCO acknowledges eight value elements (main features) of ESD: an interdisciplinary and holistic curriculum; values and principles underpinning SD; critical thinking and problem solving skills; a multi-method approach in teaching and

learning; use of learning experiences in personal and professional life; participatory decisionmaking; local relevance (addressing local/global issues in the language(s) of the learners); and critical respect for others, for difference and diversity, for the environment and for the resources of the planet (UNESCO, 2006b, pp. 4-5; UNESCO, 2007, p. 6). These eight features are used to guide the analysis below of IFALE's contribution to ESD although their order has been changed to suit the focus of the analysis.

8.1.1 IFALE and the Design of an Interdisciplinary and Holistic Curriculum

The ETP (MoE, 1994, p. 16) underlines that 'basic education will focus on literacy, numeracy, environment, agriculture, crafts, home science, health services and civics'. According to the MoE (2008, p. 6), functional literacy enables peasants, pastoralists, women and youth to make sustainable use of knowledge and skills gained from the various development packages. Adult education builds on the needs of the beneficiaries and requires multi-sectoral activities (MoE 2008, p. 20).

The IFALE Curriculum Framework, as the blueprint of the country's curriculum contents, emphasizes that IFALE (unlike the curriculum for school children) should integrate contents closely linked with the life of adults and give immediate response to their needs (MoE, 2010c, p. 7). It should focus on agriculture, health, income generation, critical thinking, civics and ethics, environmental protection, gender, and social life (MoE, 2010c). The IFALE curriculum is, therefore, to be understood holistically rather than as a single subject, with contents related to the everyday life of the participants. This should increase their motivation to learn. Basic literacy and numeracy should be taught together with the contents of the development packages of different sectors of the country and, therefore, be interdisciplinary in nature.

The content analysis of the curricula materials in the Tigray and Amhara regions supports the interdisciplinary nature of IFALE laid out in the policy documents. The materials integrate topics on agriculture, health, traditional practices, the environment, etc. Although primers were not prepared until the end of 2015 in Tigray, the facilitator's manual (TBE, 2010) includes, for example, the following contents:

 $\sqrt{}$ Leading one's personal life in a planned way (Managing oneself and accomplishing activities timely based on a plan);
- $\sqrt{}$ Problem solving skills; increasing production (division of labor, the importance of training in increasing production, and wise use of money and resources);
- $\sqrt{}$ Democratic system (democracy and the rule of law, election, equality among people and love of one's country or patriotism, as well as the right of children and the physically disabled);
- $\sqrt{}$ Equality (in the family, in the work place) and the role of women in social organizations;
- $\sqrt{}$ Social and economic problems, such as fast development, population explosion and its effects on the natural environment, as well as harmful practices;
- $\sqrt{}$ Caring for the aged;
- $\sqrt{}$ Road traffic safety;
- √ Agriculture (domestic animals and animal products, modern animal production, dairy farming, poultry, keeping the health of domestic animals, preparing animal fodder, bee keeping, and irrigation—vegetable and fruit production) and preparation of food;
- $\sqrt{}$ Health (family planning, maternal and child care, personal hygiene and environmental cleanliness, and prevention of diseases, including HIV/AIDS); and
- $\sqrt{}$ Environmental care and climatic change.

The facilitator's manual in Tigray focuses on how adults should lead a quality life without compromising the planet, much in accordance with the eight features of ESD as understood by UNESCO (2006, 2007b).

The primers and facilitator manuals of Amhara (ABE, 2011a, 2011b; 2012a, 2012b) also have interdisciplinary contents aiming at a sustainable life:

- $\sqrt{}$ Awareness of oneself and one's environment; agricultural development (crop production, fruit and vegetable production, irrigation, poultry, bee-keeping, fishing, animal production, animal fodder production dairy, hides and skins, cooperatives);
- $\sqrt{}$ Natural resources and environmental protection (environmental pollution, potable water and water source care);
- $\sqrt{}$ Health care (personal and environmental health care, food and water, maternal and child care, HIV/AIDS, tuberculosis, malaria, addiction, physical exercise);
- $\sqrt{}$ Medicine usage;
- $\sqrt{}$ Population and family planning;

- $\sqrt{}$ Enhancing useful cultural practices (like *edir* and *equb*, hospitality, culture of supporting one another, arbitration to resolve conflict by elders, tolerance, supporting the aged and orphans), avoiding harmful cultural practices, and people demanding special support;
- $\sqrt{}$ Critical thinking and problem solving skills; gender (gender equality, traditional practices);
- $\sqrt{}$ Income generation schemes;
- $\sqrt{\text{Saving}};$
- $\sqrt{}$ Job creation skills, small and medium scale industries;
- $\sqrt{}$ Civics and ethics (building a democratic system, equity and equality, nationalism, election, community participation, rule of law);
- $\sqrt{}$ First aid and road traffic safety; and
- $\sqrt{}$ Sports.

Therefore, the curriculum contents in both Tigray and Amhara was designed in an interdisciplinary and holistic way to enable learners to perform in their daily life, to eradicate unsustainable practices, and to help the country achieve sustainable development towards 2025. However, absence of primers (in Tigray) and the lack of involvement of relevant experts from different sectors in teaching IFALE in the two regions (refer Chapter 6), negatively affect the program's capacity to contribute to empowerment and SD.

8.1.2 IFALE and Critical Respect for Others, for Difference and Diversity, for the Environment, and for the Resources of the Planet We Inhabit

In accordance with the NAES, the IFALE Curriculum Framework underlines that IFALE should enable participants to develop values, including respect for all human beings, working together for the common benefit despite differences, and using one's own skills and capacity to protect the environment (MoE, 2010c, pp. 5-6). More specifically, the contents of civics and ethics, environmental protection and wise use of resources in the IFALE Curriculum Framework aim to empower adult participants to respect difference and diversity among people (democratic culture) and to care for the environment and its resources. These contents are also included in the facilitator's manual of Amhara and Tigray and in the primers of Amhara. They are important requirements to create a sustainable society. The contents on caring for the environment aims to achieve one of Nussbaum's (2000) essential capabilities: 'being able to have concern for and live with others in the environment'.

According to the survey results, IFALE played a central role in improving practices of democratic culture in Amhara, but not in Tigray (refer 7.3.14). These concerned areas such as: respect for other individuals; respect for difference in perceptions and beliefs about an idea; respect for diversity due to difference in ethnicity, language or religion; and respect for everyone's civil and legal rights. This is essential for 'learning to live together' (Delors, 1996) and to achieve a sustainable society.

Enhancing IFALE's contribution to participants' respect for the environment and for the resources of the world at a practical level seems to require more efforts. According to survey participants' self-reports in both regions, learning in the program does not result in notable changes in practices of environmental protection measures although this was included in the IFALE curricula (refer 7.2.2). IFALE, therefore, has probably made no significant contribution to supporting practices of environmental sustainability.

8.1.3 IFALE and Sharing the Values and Principles Underpinning SD

According to Huckle (2008, p. 350), a major outcome of ESD should be moral and social responsibility deriving from moral and values education. Whether delivered by formal or non-formal means, citizenship education should support social justice and help present and future generations enjoy a decent life. Huckle (2008, p. 350) argues:

Citizenship education should therefore promote political virtues (reasonableness, a sense of fairness, a spirit of compromise and a readiness to meet others half way) designed to ensure intragenerational justice and sustainability virtues (essentially the duty of the current generation to maintain the 'circumstances of justice' for future generations) designed to ensure intergenerational justice.

In line with this, IFALE must support positive attitudes towards sustainability and participation in sustainable decisions through developing adult participants' knowledge and skills of sustainability. ESD relates to eco-pedagogy²⁹, a discourse and an approach to education that seeks to educate planetary citizens to care, respect and take action for all life. Contents in IFALE curriculum materials in both Tigray and Amhara about democracy, rule of law, equality among people, gender equality in social organizations, caring for the aged, the physically disabled and orphans, tolerance, conflict resolution, environmental protection, and harmful practices are to enhance political and sustainability virtues in accordance with Huckle (2008).

²⁹ Eco-pedagogy is associated with Latin American educators, including Paulo Freire, Moacir and Leonardo Boff.

According to UNESCO (2001, p. 1), the final goal of ESD is to empower participants with the knowledge, skills, attitudes and values to produce the changes required to live in peaceful and sustainable societies. According to the survey results, IFALE participants considered the knowledge, skills, attitudes, and values that encourage and support citizens to lead sustainable lives as *very good* in Tigray (mean=4.58) and *good* in Amhara (mean=4.22) (Appendix E, survey question 82). This implies that IFALE appears to share many values and principles of SD. IFALE contributes to participants' improved practices of democratic culture activities and to better practices of social justice activities (for example, fair treatment of people irrespective of religious, gender and social background differences) in Amhara, but not so in Tigray.

IFALE seems to have meaningfully contributed to a range of areas (refer 7.1): knowledge of the voting age and of different political parties, and improved interest in becoming an elected representative (in Tigray and Amhara); knowledge of a law on electing representatives, confidence in terms of becoming an elected representative, and involvement in public institutions' decision-making and community development activities (in Tigray); and participation in voting (in Amhara). These are all characteristics of being a good citizen. This implies that in both regions IFALE embraces particular citizenship education elements that enhance important virtues of civilized life, a vital element of a sustainable society. However, in other areas, such as practices of conflict resolution (i.e., compromise, apology, direct discussion with the other party, and arbitration by elders or formal litigation), IFALE seems to have had no impact (refer 7.3.15).

8.1.4 IFALE and Endeavors to Address Critical Thinking and Problem Solving Skills of Its Participants

ETP (1994, p. 16) states that 'non-formal education will be concrete in its contents focusing on enabling the participants develop problem solving attitudes and abilities'. According to the IFALE Curriculum Framework (MoE, 2010b, p. 33), the source of many problems in Ethiopia is lack of critical thinking which is necessary to solve disagreements and everyday problems of society through discussion and consultation. The development of democratic culture can be speeded up through skills and capability enhancement and knowledge of democratic values, i.e. improvement of innovation skills, ability to distinguish facts from opinions, quest for problem solving ways, recognition of the decisiveness of team work, emancipation from exploitation, protection of unfair treatment, etc. (MoE, 2010c, p. 33). The IFALE curriculum has contents proposed in the National Curriculum Framework, such as problem solving skills, conflict resolution through discussion about what needs to be done to achieve development, and self-confidence enhancement skills.

The facilitator's manual in Tigray addresses problem solving skills (TBE, 2010, pp. 11-16). The lesson objectives are, among others, to enable the participants to speak about the challenges they face in their life, identify methods of problem solving and solutions to their challenges, and compare common and new methods of problem solving. The manual also advises facilitators to make IFALE participants discuss the source of conflicts and the ways of resolving them in order to develop their problem-solving skills. Well-trained facilitators play a decisive role in making this happen.

Similarly, in the learner textbook (ABE, 2011a, p. 93; 2011b) of Amhara, adult participants are instructed in group discussions on the reasons for conflicts, types of conflict caused because of lack of awareness (critical thinking), and the way conflicts are resolved in order to develop their problem solving skills. As stated in the document, the main objective is to enable participants to understand problems, including conflicts between partners, divorce and problems related to children after divorce, revenge, theft, harassment/verbal abuse, and looking down on others, as well as to develop skills to solve such problems. Problem solving and critical thinking skills are necessary to achieve social or behavior change. They are vital elements for what Torres (2003) and Ouane (2008) call 'learning to change', a pillar of LLL. They can help learners be proactive in order to bring about dynamic changes necessary for human well-being or social, environmental, cultural and economic sustainability.

Although critical thinking and problem solving skills are what the curricula intend to enhance, the capacity of the facilitators to support their development was virtually non-existing in both regions probably due to lack of training in active learning methods and in teaching adults or andragogy (refer 6.1.3).

8.1.5 IFALE and the Application of a Multi-Method Approach in the Teaching and Learning Process

UNESCO (2007, p. 6) recommends the use of a multi-method approach in the teaching and learning process, for example art, drama, debate, experience, and other pedagogies. The NAES (MoE, 2008) aims:

to facilitate the launch of a practice-focused adult education program, which supports the productive part of the society (adults) by mastering knowledge and skills to competently participate in the national development, and which complies with and serves as a tool for the different development strategies of the government (p. 10).

This implies a practice-focused adult literacy program to implement the various strategies that enable sustained and accelerated development of the country. According to the MoE (2008, p. 19), improving the methods and techniques of adult education delivery is essential. This necessitates identification and use of appropriate delivery methods and approaches for adult participants, skills development training for managers, supervisors and facilitators as well as preparation of training manuals based on adult education delivery approaches.

In Tigray, according to the IFALE facilitator's manual, the teaching and learning methods start from the experience, knowledge and skills of participants in order to achieve improved personal, community and national development (TBE, 2010). As stated in the introduction to the manual, the methods to be used are group discussion, practical observation, demonstration, question and answer, experience sharing, individual assignments, brainstorming, visits, role-play and drama. In the body text of the manual the use of group discussion is generally recommended.

In the introduction to the primers and facilitator's manual of Amhara, it is suggested that facilitators integrate literacy and numeracy with life skills and use, among others, discussions, question and answer, and field visit as methods in the teaching and learning process (ABE, 2011a, 2011b; ABE, 2012a, 2012b). However, the most frequently recommended method in the manuals is group discussion. Group discussion was used to some extent in the sessions in both Tigray and Amhara, but the teaching and learning method was overwhelmingly traditional question and answer (refer 6.1.3).

In both regions, the lack of extensive use of active learning methods affects learners' selfconfidence and makes them less able to act and engage in solving everyday challenges and, thereby, lead sustainable lives.

8.1.6 IFALE and Implementation of Participatory Decision Making in Teaching and Learning

Within a sustainability frame of mind, IFALE should encourage its participants to participate in decision-making. According to the IFALE Curriculum Framework, IFALE participants should actively participate in content selection and lesson design (MoE, 2010c, p. 6). Although the facilitator's manual generally serves as the starting point for what should be taught, the main learning contents is to be initiated by the participants during their discussions of the problems encountered in their daily life and the solutions they propose (see, e.g., TBE, 2010, p. v). While promoting 'learning to do' (adaptation to the learners' future work environments) that is a component of LLL (Delors, 1996), participatory decision making in the classroom gives learners the opportunity to express their opinions, and develops their sense of ownership of the teaching and learning process. This may enhance their sense of well-being and comfort with the way the classroom functions (Stefanou, Perencevich, DiCintio, & Turner, 2004). Participatory decision-making is important for a sustainable society.

The observation of the lesson sessions in Tigray shows that facilitators took the decisions (Appendix B, question 18). In Amhara, school teachers (who served as facilitators) also strictly followed the contents in the primers and manuals. Learners in the two regions were not encouraged to participate.

8.1.7 IFALE and Integration of Learning Experiences in Daily Life

The NAES and the IFALE Curriculum Framework expect participants to apply what they have gained from IFALE in their daily activities if they are to contribute to accelerated and sustained development for Ethiopia to become a middle-income country by 2025. Access to information enables participants to learn new skills and to utilize the gains to improve their life and productivity (MoE, 2008, p. 6). Besides, the facilitator manuals and primers encourage adult participants to use what they have learned to improve their livelihoods and quality of life, among which are health and use of latrines. These can be important to improve the capability of 'bodily health' (Nussbaum, 2000) and support well-being and environmental sustainability.

Some of the survey results reveal that IFALE participants in both Tigray and Amhara applied what they learned. The significant empowerment gains in saving practices, and practices of specific traditional activities and democratic culture activities are examples (refer Chapter 7). IFALE participants in both regions have improved, amongst others, their practice of saving, self-efficacy, and in changing harmful traditional activities. Comparisons between IFALE participants and non-participants show that the former applied maternal and child care, and some democratic culture and social justice activities more than non-participants (in Amhara) and displayed more 'modern' values of wanting an additional child (in Tigray) (refer 7.3).

This means that IFALE participants have integrated their learning experiences in their everyday life in some of the critical areas.

Four household observations of IFALE participants (two in each region) were part of the study in order to establish participants' practice in terms of housekeeping, personal hygiene and latrine use. According to the participants in both regions, their practices changed positively after joining the IFALE program. However, the observations showed that the participants in Tigray live in cleaner and better-organized houses, and have relatively better latrines than was the case in Amhara. Figure 8.1 shows that in Tigray kitchen items are better ordered, cleaner and of better quality than in Amhara (Figure 8.2). The latrine in Tigray (Figure 8.3) is also better than the one in Amhara (Figure 8.4).

Health and hygiene also concern the living conditions of the participants. In the case of Tigray, the family and domestic animals were separated either by a separate shelter (Figure 8.5) or by a door. In Amhara, the families shared the room with the domestic animals which, they believe, helps keep the family warm during the cold nights. In another, there was a simple wooden partition between the family and their cattle and sheep (Figure 8.6). Based on the observation results, participants in Tigray do relatively better than in Amhara in terms of application of knowledge from IFALE in their own lives. However, considering the findings from both the survey and the observations, participants in the two regions all applied some, but different knowledge from the IFALE program which reflects important features of ESD.



Figure 8.1: Kitchen items in the house of an IFALE participant in Tigray *Source*: Observation (November 2013)



Figure 8.2: Kitchen items in the house of an IFALE participant in Amhara *Source*: Observation (November 2013)



Figure 8.3: IFALE participant's latrine in rural Tigray *Source*: Observation (November 2013)



Figure 8.4: IFALE participant's latrine in rural Amhara *Source*: Observation (November 2013)



Figure 8.5: IFALE participant's house in Tigray with a separate shelter for domestic animals *Source*: Observation (November 2013)



Figure 8.6: IFALE participant's house in Amhara partitioned between the family and domestic animals *Source*: Observation (November 2013)

8.1.8 IFALE and Local Relevance: Addressing Local and Global Issues and the Language of the Participants

One of the general objectives of ETP is 'to bring up citizens who can take care of and utilize resources wisely, who are trained in various skills, by raising the private and social benefits of education' (MoE, 1994, p. 7). A specific objective refers to promoting 'relevant and appropriate training through formal and non-formal programs' (MoE, 1994, p. 8). These objectives endorse empowerment of participants for sustainability.

In line with the ETP, NAES articulates the importance of the relevance of adult education by stressing that adult participants will be motivated to participate if the program is related to their needs and daily activities, and is useful for solving their problems (MoE, 2008). According to the IFALE Curriculum Framework, all regions in the country should emphasize locally relevant contents. That is why basic reading, writing and computing are integrated with the teaching of the contents of agriculture, health, income generation, critical thinking, civics and ethics, environmental protection, gender, and social life issues that are understood as essential life skills. It means that IFALE should have the same emphasis on life skills as on basic literacy and numeracy skills.

The IFALE Curriculum Framework and the primers plan for contents that are to be relevant to both lowlanders—whose life mainly depends on animal husbandry—and highlanders who

live from crop production. Officials at the MoE and regional educational officials stated in the interviews (MoE policy maker interview 2, Addis Ababa April 2013; regional IFALE focal person interviewee, Tigray, May 2013) that learning materials should be designed taking regional and sub-regional relevance into consideration that meet the demands of the adult participants, for example of the pastoralists in the lowlands and the farming communities in the highlands. This study was done in the highlands of Ethiopia and interviews with IFALE participants revealed that the contents of the materials is relevant to them since they believe that it helps improve their lives (refer Appendix A, 1, III, questions 3 & 6).

According to the ABE (2011a, b; 2012a, b), the contents of the primers integrate life skills with literacy and numeracy through the assessment of participants' needs and problems. IFALE enables participants to understand themselves and their surroundings and develops their attitudes. It contributes to personal and national development and to personal initiative in the fight against poverty, eventually providing them with opportunities to be productive citizens. The contents in the teaching materials of Amhara correspond with the development packages of the country, revealing its relevance to the participants' lives. It addresses, for example, agriculture and rural development, health, natural resources and environmental protection, and change of specific practices. The facilitator's manual in Tigray integrates similar contents in addition to contents on global warming which can have both local and international significance.

The language of instruction in both regions is the participants' mother tongue, i.e. Tigrigna in Tigray and Amharic in Amhara. The contents in both regions seem to be authentic and relevant to the lives of the participants. Relevant contents delivered in the mother tongue can help learners engage in overall efforts to achieve economic, social, environmental, and cultural sustainability for Ethiopia to become a middle-income country by 2025.

8.2 Conclusion

The document analysis and survey results show that in Amhara IFALE has integrated five of the eight UNESCO (2006b, 2007) ESD features. In Tigray IFALE embraced only four (half) of them. In particular, the use of critical thinking and problem solving skills, a multi-method approach in the teaching and learning process, and participatory decision-making were not apparent in the teaching and learning of IFALE neither in Tigray nor in Amhara. In addition, in Tigray IFALE did not enhance critical respect for others, for difference and diversity, for the environment and for the resources of the planet (Table 8.1). This is partly due to

weaknesses in the implementation of the strategies of a relevant curriculum and an efficient institutional system (refer Chapter 6).

S.		Tigray	Amhara
No.	ESD features	Yes /No	Yes /No
1	Design of an interdisciplinary and holistic curriculum		
2	Critical respect for others, for difference and diversity, for the	X	
	environment, and for the resources of the planet we inhabit		
3	Sharing the values and principles underpinning SD		
4	Endeavors to address critical thinking and problem solving skills	Х	Х
5	Application of a multi-method approach in the teaching and learning	X	Х
	process		
6	Implementation of participatory decision making in teaching and	X	Х
	learning		
7	Integration of learning experiences in daily life		
8	Addressing local and global issues and the language of the		
	participants		
Overall		Fulfillment of 4 of 8	Fulfillment of 5 of
		(half) features	8 (most) features

Table 8.1: ESD features embraced in IFALE in Tigray and Amhara

Key: Yes: √; No: X

Nevertheless, the implementation of IFALE did lead to improvements in some areas of political, economic, socio-cultural and psychological empowerment domains that could support the realization of SD in some ways. This particularly concerns participants' psychological empowerment in terms of self-efficacy (in both Tigray and Amhara). Enhanced self-efficacy can lead to gains in other dimensions of empowerment and thus contribute to the development of a sustainable economy, society, environment and culture in the country.

However, the fact that IFALE made no considerable contribution to most of the other indicators of empowerment (only 12 of 32 indicators) and to only a restricted number of ESD features in the two regions shows that the program's overall contribution to SD is limited, although somewhat stronger in Amhara than in Tigray.

Chapter 9 Major Findings, Key Conclusions, and Significance of the Study

In this final chapter, the most important findings are first discussed in view of the findings of previous studies on adult education (Chapter 3). This is followed by brief answers to each of the research questions (Chapter 1) and some reflections on the conclusion. The underlying assumptions and the analytical model for the study (Figure 4.7) are considered. The chapter finishes with a brief discussion of the significance of the study and its implications for further research and for policy-making.

9.1 Implementation and Contribution of IFALE

The study explores the implementation and contribution of IFALE to empowerment and SD in Tigray and Amhara regions using mixed methods design. The qualitative methods include interviews, observations, and document analysis. There were 91 informants at different levels and in different functions. In addition to classroom and center observations, a few (four) IFALE participants were observed in terms of their practice at their homes. The quantitative analysis was based on a survey undertaken with 376 survey participants. Of these 89 were IFALE participants and 97 were non-participants in Tigray compared to 93 IFALE participants and 97 non-participants in Amhara.

9.1.1 Implementation of IFALE

According to NAES, the IFALE curriculum contents should be based on different development packages, including from agriculture and health, in order to be relevant to and improve the lives of the adult participants. Besides, efforts should be made to build an efficient institutional system. In contrast to previous research, both strategies are in this study considered to be essential for the accomplishment of the IFALE policy in Ethiopia. The understanding of the strategies and their supporting factors appear in the analytical model displayed in Figure 4.7. The model was developed based on a critical review of existing research and of policy documents in Ethiopia.

According to the results of interviews, observations and document analysis, the strategy of a relevant curriculum in Tigray lacked most of its support factors and was thus unsuccessful.

These were: primers and other additional materials; qualified and autonomous (independent) officials and experts; well-trained facilitators; use of active learning methods; motivation of IFALE participants; use of learning aids (facilities); and learning conditions. In Amhara, the strategy of a relevant curriculum had most of its support factors, except for lack of active learning methods in the teaching and learning, the presence of unqualified and unmotivated facilitators, and the availability of some officials and experts that need training in adult and literacy education. Amhara, therefore, was partly successful with regard to the strategy of implementing a relevant curriculum for IFALE.

The results of interviews, observations and document analysis also show that practically all the support factors of the strategy of building an efficient institutional system were missing in Tigray. These are: strong leadership and motivated officials, experts and facilitators; shared understanding of IFALE by responsible officials and experts; clear policy direction; favorable working conditions; involvement of stakeholders; and an adequate budget. In Amhara, some of the support factors were also not at work: facilitators were not motivated because of a low salary; there was no clear non-formal and adult education policy frame; collaborators did not participate in the teaching and learning process; and the budget was insufficient to train officials and experts who lacked qualifications in adult and literacy education, provide additional training for facilitators and school teachers that conduct IFALE sessions, and prepare additional reading materials for participants.

The comparison of the two regions shows that the implementation of IFALE in terms of the two strategies was partly successful in Amhara while it was not in Tigray. According to Comings et al. (1995) weak program implementation causes inefficiency, for example high dropout rates and low skill levels of participants. The deficient implementation of the strategies also affects the potential contribution of IFALE to participants' empowerment and SD.

9.1.2 IFALE and Empowerment

The study shows that the support factors, including agriculture and development, health, women affairs, and youth and social affairs sectors, participated to a limited degree in implementing the IFALE policy. Although the IFALE Curriculum Framework and the regional curriculum were developed in collaboration with various sector ministries as a way to support SD, only the education sector participated in its implementation through teaching. Empowerment gains from IFALE can, therefore, only be attributed to the teaching and

learning by IFALE staff, regular facilitators, and voluntary primary school teachers. The collaborating sectors did, however, contribute independently (or as external factors) to the empowerment of IFALE participants (and non-IFALE participants) together with other external factors, such as FTCs, local media, local leaders, farmers' and women's associations.

In accordance with Page and Czuba (1999) and Kagitcibasi et al. (2005, p. 476), empowerment in this study are at the following levels: social (higher social participation scores); family (changes in family dynamics and attitude towards fertility); and personal (more positive self-concept and self-efficacy). At the societal level, IFALE has contributed to participation in voting (in Amhara) and to developing an interest in becoming an elected representative (in Tigray and Amhara). At the personal level, knowledge of the minimum voting age and political parties (in both regions), confidence to be a local representative for an election, and knowledge of a law on electing representatives (in Tigray) are outcomes of IFALE which can result in actions at the community level.

At the societal level, IFALE participants have political empowerment gains, including participation in decision making in public institutions, membership of community organizations, and participation in community development activities (in Tigray) which reflect citizenship obligations and likely wider contact with local people. The findings are consistent with adult literacy impact studies conducted in Kenya (Carron et al., 1989), in the US and the UK (Stromquist, 2008), and in Turkey (Kagitcibasi et al., 2005). Burchfield et al. (2002a, 2002b) also found the knowledge and practices of political participation of women in literacy programs in Nepal and Bolivia to be higher than for non-participant women.

The economic empowerment benefits from IFALE in terms of increase in saving practice (in Tigray and Amhara) and involvement in income generation schemes (in Tigray) are gains at the personal level that can also have implications at societal level. These results support Cameron and Cameron (2005) who claim that literacy is a contributor to livelihoods improvements from saving opportunities and creating supplementary income.

The changes that IFALE has brought in terms of practices of traditional activities in both Tigray and Amhara affect the culture and have value at both individual and societal levels. At the family level, these also include the wish to have fewer children for economic, social or psychological reasons (in Tigray), gender equality in decision making in the household as well as practices of family planning and maternal and child care (in Amhara). Other key benefits of IFALE at the societal level are enhanced democratic culture activities and social

justice activities (in Amhara). Egbo (2000) in his study on literate women in Nigeria, found that women gained more knowledge of their rights, making them participate in community matters and household decisions. In the case of Turkish women, functional adult education led to positive changes in the value of children, but not in household decision making (Kagitcibasi et al., 2005).

Psychological empowerment at the personal level is a major ingredient of social capital. IFALE can help since it can break down barriers of isolation, particularly for illiterate women whose lack of social networks hampers their participatory practices in development (Stromquist, 2009). In Tigray and Amhara, the self-efficacy gains from IFALE at the personal level appear to facilitate social engagement, political behavior, and economic opportunities of participants. The gains may produce what Rodger and Street (2012) refer to as a new sense of identity or the confidence to contribute significantly to the public sphere. However, IFALE's lack of contribution to self-concept is in opposition to the study on a Turkish functional adult literacy program (Kagitcibasi et al., 2005) where important change was reported in the self-concept among female participants, but none as far as self-efficacy is concerned. Other studies (e.g. Prins, 2008; Ash & Parrot, 2001; Burchfield, 1977; Archer & Cottingham, 1996) also show considerable contributions of adult literacy programs to developing self-confidence, self-efficacy and self-respect.

The findings regarding IFALE learners' empowerment gains in the two regions represent some of Nussbaum's (2000) core capabilities and the pillars of education that underpin LLL (Delors, 1996; Ouane 2008; Torres, 2003; Preece, 2013). Better self-efficacy and improved traditional practices in Tigray and Amhara enhance participants' capabilities to imagine, think, reason, use thought connection, form a perception of the good and start critical reflection. This supports learning to be, i.e. the development of IFALE participants' mind, body, intelligence and sensitivity.

9.1.3 IFALE for Empowerment and Sustainable Development

The political, economic, socio-cultural, and psychological empowerment achievements from IFALE can contribute to SD in Ethiopia. IFALE as ESD supports SDG 4 (UNESCO, 2015), i.e. 'sustainable lifestyles, human rights, gender equality, culture of peace and non-violence, global citizenship and appreciation of cultural diversity and of culture's contribution to SD by 2030'. In the two regions IFALE helped participants develop knowledge, skills, attitudes and

values needed to shape a sustainable future to a certain degree. These gains address the four components of SD, namely the environment, society, the economy, and culture.

The analysis of the NAES, the National IFALE Curriculum Framework, and the teaching and learning materials of IFALE as well as the self-reported survey results show that most of the major features of ESD articulated in DESD (UNESCO, 2006b, pp. 4-5) have been embraced in Amhara. These include: the design of an interdisciplinary and holistic curriculum; critical respect for others, for difference, and diversity, for the environment, for the resources of the planet; the values and principles underpinning SD; integration of learning experiences in day-to-day life; and discussions of local and global issues in the language the participants understand or use. In Tigray, four of the features embraced in Amhara were supported, excluding critical respect for others, for difference, and diversity, for the environment, for the resources of the planet. In both regions, IFALE gives low emphasis to critical thinking and problem solving skills, the participation of participants in decisions about the way they learn, and the use of a multi-method approach in the teaching and learning process which affects the capacity of participants to support SD.

In conclusion, IFALE has contributed to improving areas of empowerment at individual, family and community levels through enhancing learning and capabilities. IFALE fulfills most of the ESD features in Amhara but only half of them in Tigray. The contribution of IFALE to SD is higher (though moderate) in Amhara than in Tigray (where it is limited). Therefore, the ambitious plan for IFALE to support SD in order for Ethiopia to be a middle income country by 2015 is not matched by the reality on the ground.

9.2 Answering the Research Questions

While the previous section discussed the general findings in light of former research, the purpose in the following is to answer each of the research questions concisely.

9.2.1 How successful is the implementation of IFALE in terms of the strategy of a relevant curriculum?

In Tigray, the strategy of implementing a relevant curriculum was not successful since almost all the support factors were missing (refer 9.1.1). IFALE participants did, however, have good access to the program since there are IFALE centers in every village, a facilitator manual guided the teaching, and the availability of trained facilitators was better (but not adequate) in Tigray than in Amhara. By contrast, in Amhara most of the support factors for the strategy were present, except for the use of active learning methods, supply of learning aids and qualified facilitators. In addition, some IFALE officials and experts needed training related to adult education. The implementation of IFALE in terms of the strategy of a relevant curriculum was unsuccessful in Tigray but partially successful in Amhara.

9.2.2 How successful is the implementation of IFALE in terms of the strategy to build an efficient institutional system?

There was an independent organizational set up for IFALE from the regional bureau down to the centers in Amhara, but not in Tigray. The implementation of IFALE in terms of the strategy of building an efficient institutional system lacked virtually all support factors in Tigray despite clear strategy direction in the NAES. In Amhara, most support factors were in place, except for a clear policy frame, the participation of collaborating sectors in teaching, and motivated facilitators to conduct the teaching and learning process (refer 9.1.1). Therefore, the implementation of IFALE as regards the strategy was unsuccessful in Tigray but partly successful in Amhara.

9.2.3 What is the contribution of IFALE to political empowerment?

IFALE contributed more to political empowerment in Tigray than in Amhara. In Tigray participants were empowered in most of the political indicators (7 of 10, i.e. 70%) in contrast to Amhara (4 of 10, i.e. 40%). In both regions, the gains were particularly in participants' knowledge of the voting age and political parties, and their interest in becoming an elected representative.

In Tigray, IFALE increased the confidence of the participants to become an elected representative, provided knowledge of a law on electing representatives, and promoted participation in community development activities and decision making in public institutions in Tigray. This was not the case in Amhara. In contrast, IFALE increased participants' voting participation in Amhara as opposed to Tigray. In none of the regions did IFALE lead to significant political gains in terms of participants' membership of community organizations, and knowledge of community development committees. The contribution of IFALE to political empowerment is higher in Tigray (high) than in Amhara (moderate).

9.2.4 What is the contribution of IFALE to economic empowerment?

IFALE's overall contribution to economic empowerment is small in both regions. It did, however, contribute to increasing the saving practice of IFALE participants in both regions and improved participant involvement in income generation schemes in Tigray as opposed to Amhara. IFALE made an insignificant contribution to improved practices of environmental protection measures and agricultural technologies for improving production, financial contributions for productive investment and mutual support, or credit access for income generation among participants in both regions. Overall, IFALE enhanced less than half of the indicators of economic empowerment (2 of 6, i.e. 33%) in Tigray and (1 of 6, i.e. 17%) in Amhara. The contribution of IFALE to participants' empowerment was higher, though limited, in Tigray than in Amhara where it was very limited.

9.2.5 What is the contribution of IFALE to socio-cultural empowerment?

IFALE's contribution to socio-cultural empowerment is very low in Tigray (2 of 14 key indicators, i.e. 14%) but relatively better in Amhara (6 of 14 indicators, i.e. 43%). IFALE has contributed to reducing traditional practices in both regions, but played a role in limiting the number of children only in Tigray. By contrast, IFALE improved practices of democratic culture, maternal and child care, family planning, social justice activities, and gender equality in family decision making in Amhara as opposed to Tigray.

IFALE has not enhanced practices of HIV/AIDS measures, family cohesion and conflict resolution or enhanced participants' knowledge of laws protecting women and children, laws against gender discrimination, or knowledge of gender equality in development in either of the two regions. Generally speaking, the contribution of IFALE to socio-cultural empowerment is better, though moderate, in Amhara than in Tigray where it is very limited.

9.2.6 What is the contribution of IFALE to psychological empowerment?

In both regions, IFALE played a moderate role in improving participants' psychological empowerment gains (1 of 2 indicators, i.e. 50%). IFALE contributed to improving self-efficacy, but not the self-concept.

9.2.7 How does IFALE support SD?

IFALE has been partly successful in Amhara but unsuccessful in Tigray in terms of implementing the strategies of a relevant curriculum and an efficient institutional system. This may explain the limited overall empowerment gains (12 of 32 indicators of empowerment in each region) and the failure to fulfill all ESD features (only 5 of 8 ESD features in Amhara and 4 of 8 features in Tigray). This has led to moderate support to SD in Amhara but limited support in Tigray.

IFALE's contribution to socio-cultural empowerment gains in terms of improved practices of traditional activities (in Tigray and Amhara), as well as democratic culture activities, social justice activities, maternal and child care, family planning and gender equality in household decisions (in Amhara) address unsustainable social and cultural challenges. The contribution of IFALE to psychological empowerment expressed in terms of self-efficacy (in Tigray and Amhara) enhances the self-confidence needed for moving towards sustainability. It is a precondition for improved empowerment in the political, economic and socio-cultural domains that support a sustainable economy, society, environment and culture.

The important contributions to participants' political empowerment in terms of political interest, knowledge and participation, and to economic empowerment in terms of saving practice (in Tigray and Amhara) and involvement in income generation schemes (in Tigray) respond to citizenship roles and economic growth demands that can lead to sustainable livelihoods. IFALE has helped in focusing on knowledge, skills, attitudes and values, encouraging and supporting citizens to lead sustainable lives in both regions.

Moreover, IFALE fulfills some of the important features of ESD in both regions. IFALE

- is interdisciplinary and holistic since it integrates various topics into one, for example, agriculture, health, environment, conflict resolution, traditional practices, and income generation;
- 2. is values-driven and capable of promoting political and sustainability virtues: reasonableness, a sense of fairness, and readiness to meet others half way;
- 3. integrates or applies learning experiences in the everyday life of its participants. IFALE participants appear to apply their gains, e.g. used latrines and changed specific practices;
- 4. addresses local as well as global issues and uses the language(s) most commonly used by its participants. IFALE treats contents such as natural resources and environmental

protection (in Tigray and Amhara) as well as global warming (in Amhara). IFALE uses Tigrigna in Tigray and Amharic in Amhara as the medium of instruction.

However, whereas IFALE develops participants' critical respect for others, for difference and diversity, for the environment, and for the resources of our planet in Amhara, this is not the case in Tigray. IFALE does not apply a multi-methods approach in teaching and learning, carry out participatory decision-making for its participants to decide on what, how and when they should learn, or address critical thinking and problem solving skills of participants in the two regions.



Figure 9.1: Summary of results of the implementation and contributions of IFALE to empowerment and SD

Overall, IFALE in Amhara has supported SD to a moderate extent through carrying out partly successful strategies of a relevant curriculum and an efficient institutional system, contributing to certain areas of empowerment, and fulfilling most features of ESD (Figure 9.1). In Tigray, IFALE has supported SD to a limited extent through the two strategies, contributing to restricted areas of empowerment and embracing only half of the features of ESD. IFALE's support to SD in Ethiopia, i.e. to becoming a middle income country by 2025, seems therefore to be limited.

9.3 Reflections on Conclusion

Ethiopia has witnessed rapid economic growth, with real gross domestic product (GDP) growth averaging nearly 11% between 2003/4 and 2014/2015 (World Bank, 2016c). Ethiopia also has an ambitious plan to become a middle-income country by 2025. Nevertheless, the

Ethiopian government and the ruling party (EPRDF) have openly declared that inefficiency and poor governance are major challenges for the country's development (Walta Information Center, 2015). Despite a well-articulated NAES at country level and corresponding regional plans, poor IFALE implementation of the strategies of a relevant curriculum and an efficient institutional system in Tigray is likely to be due to the same reason.

Ethiopia has no policy frame to support IFALE which reflects absence of political will to implement the program. In Tigray, there is neither involvement of relevant sectors in the teaching and learning process nor follow up on the functionality of the regional IFALE board and technical committee. IFALE is not adopted as an essential plan for the region, with qualified staff placed in key positions playing leadership roles and having an adequate budget. By contrast, the strategies of a relevant curriculum and an efficient institutional system were moderately successful in Amhara. The regional government bureau of education owns the IFALE program and has appointed a capable leadership in the bureau and weredas to implement the program.

Successful implementation of the two strategies could lead to increased self-confidence, and independent, critical thinking and reflection of the participants. This could facilitate empowerment, promote environmental, societal, economic, and cultural considerations for development and lead to a better quality of life. However, as reported in this dissertation, IFALE's contribution is limited in terms of empowerment gains, fulfilment of ESD features and, therefore, also SD efforts.

The political empowerment gains discussed earlier may, nevertheless, enhance participants' medium and long term capability to live with others and have more control over the environment. The economic empowerment gains may also develop the capability to control the environment or ability to own property, thus demonstrating improved income and wealth among IFALE participants. This could support national and human development goals.

The socio-cultural and psychological empowerment gains could expand the participants' capability to imagine, think, reason and use thought connection as well as form a perception of the good which could all help critical reflection. It could signify learning that would contribute to participants' development in terms of mind, body, intelligence and sensitivity. Socio-cultural empowerment gains could enhance the capability for bodily health of mothers and children and facilitate learning to proactively guide change among participants for their well-being and better livelihoods.

Thus, IFALE could achieve even more significant empowerment gains in most indicators at personal, family, and societal levels if there were a more genuine response to the NAES and the regional plans. Enhanced learning, capabilities and livelihoods could make IFALE an important program in support of SD in its own right. If adult literacy education is to empower learners and support SD, government bodies must show political commitment. This commitment should be expressed, amongst others, in the inclusion of adult (literacy) and non-formal education as part of the education and training policy, the attainment of educational quality, relevance, a strong institutional system, an adequate budget to provide well-trained officials, experts, and facilitators, and appropriate curriculum materials and facilities.

9.4 Significance of the Study

The topic for this dissertation is relatively under-researched in the field of adult literacy education. Previous research has more commonly focused on impact assessment than on the empowering potential of adult literacy education. This study shows that gains in political, economic, socio-cultural and psychological empowerment through an adult literacy education program have the potential to support a sustainable economy, society, environment and culture. This is an important contribution to comparative and international education and to policy making in Ethiopia and elsewhere.

9.4.1 Significance for Academia

The philosophy of causation or evidence-based policy grounded in the theory of evidence for use formed the basis of the analysis of the implementation and contribution of IFALE to empowerment and SD in Ethiopia. The horizontal search of literature and MoE documents served to identify key support factors that guided the analysis of the reality in the two regions. Its application in the study shows the relevance of the underlying philosophy for the examination of policy issues.

The study also contributes to the general field of international and comparative education through its comparative analysis of Tigray and Amhara. It helps fill the general gap on (recent) adult education studies in Ethiopia, in general, and on literacy education, in particular. In terms of its specific findings, IFALE's contribution to improved saving practices in Tigray and Amhara as well as participation in income generation activities represent new knowledge regarding economic benefits of ALE. The fact that some of the findings in this study are inconsistent with former studies may underline that studies and findings need to be contextualized and that the contributions of one program cannot necessarily be generalized to

others due to the differences pertaining to content and approach. Careful judgements should be made since *there* is not the same as *here*.

The findings that IFALE can contribute to developing essential capabilities are important to comparative and international education. In this dissertation, they include: practical reason or perception of the good and critical reflection on it (e.g. improved traditional cultural practices and better self-efficacy); bodily health for mothers and children (e.g. improved maternal and child care and family planning); control over one's environment (e.g. participation in politics, and better saving practice and participation in income generation schemes that result in property ownership); and affiliation to others (e.g. better democratic culture and social justice activities).

The dissertation focuses on knowledge and practices of learners in political, economic, sociocultural and psychological empowerment which was examined in terms of integrated knowledge of literacy, numeracy and life skills. This is an important contribution to the existing knowledge on adult education since most studies are limited to analyzing literacy and numeracy tests to explore literacy and numeracy outcomes. The focus on the various dimensions of empowerment is an area that is worthy of future research.

Furthermore, a large-scale assessment of IFALE's contributions to SD in all regions of Ethiopia could help understand factors that promote or limit the successful implementation of IFALE. A tracer study of neo-literates, for example those who graduated from the IFALE program in Tigray and Amhara in 2013, could help examine intermediate and long term impacts of empowerment gains from IFALE at personal, family, community and society levels.

The implementation of IFALE was limited to two major strategies believed to be vital for the purpose of the study. One could argue that the implementation of IFALE could have been investigated differently by considering other factors (for example learners' attendance, cost effectiveness, mastery of basic literacy and numeracy, SMART learning objectives), and literacy programs that were also part of IFALE but implemented by NGOs. Thus, other causal cakes with their supporting ingredients could lead to effective IFALE depending on the nature of the study.

Future studies on economic benefits of ALE could also focus on other areas than those included in this study, for example safe and efficient use of equipment, access to insurance,

use of appropriate technologies to sustain the natural environment, and capabilities in social cooperation leading to improved advocacy and/or ability to compete in markets.

9.4.2 Significance for Policy Makers

Important lessons can be gained from the analysis of the two regions of Tigray and Amhara because of their different implementation practices. The better implementation of IFALE in terms of the strategies of a relevant curriculum and an efficient institutional system in Amhara can serve as 'best practice' that can be carefully adapted not only to Tigray but also to other regions of Ethiopia. The political gains achieved among IFALE participants in Tigray can also serve as 'best practice' by considering relevant factors for Amhara and in other contexts. Tigray and other regions can learn from the socio-cultural gains of IFALE in Amhara.

The study results show that the federal and regional governments have not displayed strong political will in order to achieve a successful implementation of IFALE. While the preparation of NAES is an important step, there is no emphasis on IFALE in the wider ETP. This prevents appropriate provision of the program. Adult education and lifelong learning, in general and adult literacy education, in particular, should be central components of the education policy in Ethiopia in order to ensure that relevant strategies and plans can be successful in reality. This needs to be reflected in political will and support for important factors, such as an adequate budget, training and support in adult and non-formal education, sufficient learning materials, and an appropriate learning environment.

In addition, the appointment and recruitment of dedicated IFALE staff at all levels could form a critical mass to champion adult functional literacy and IFALE in order to support the country's advance towards becoming a middle-income nation by 2025. To help promote IFALE, fora at all levels and for all parts of Ethiopia could discuss the intentions and share the provisional experiences and outcomes of IFALE. Such initiatives could help bridge the gap identified in the study between the policy rhetoric and the reality on the ground.

References

- Abadzi, H. (2003a). Adult literacy: A review of implementation experience. *Operations Evaluation Department. Washington, DC: World Bank (forthcoming).*
- Abadzi, H. (2003b). Improving Adult Literacy Outcomes: Lessons from Cognitive Research for Developing Countries. Washington, DC,World Bank. (Directions in Development Series.) http://files.eric.ed.gov/fulltext/ED478961.pdf Retrieved October 19, 2017 from http://files.eric.ed.gov/fulltext/ED478961.pdf
- Abate, Y. (1993). The Society and Its Environment. In Ofcansky, T, & Berry, L. *Ethiopia: A Country Study*. USA: Federal Research Division. Library of Congress
- Aitchison, J. & Alidou, H. (2009). The state and development of adult learning and education in Subsaharan Africa. Regional synthesis report. Hamburg: UNESCO Institute for Lifelong Learning.
- Amare, G. (2006). An Appraisal of the Literacy Campaign in Ethiopia during the Military Regime, 1974-1991. *Journal of Ethiopian Studies*, 93-117.
- American Psychological Association (APA). (2010). *Publication Manual of the American Psychological Association* (6th ed.). Washington DC: Author.
- Amhara Bureau of Education (ABE). (2011a). *IFALE Learners' Textbook 1*. Baherdar: Amhara.
- ABE. (2011b). IFALE Facilitators' Manual 1. Baherdar: Amhara
- ABE. (2012a). IFALE Learners' Textbook 2. Baherdar: Amhara
- ABE. (2012b). IFALE Facilitators' Manual 2. Baherdar: Amhara
- Anderson, G., (1998). Fundamentals of Educational Research. Falmer Press, London
- Annan, K. (2005). *Message on UN International Literacy Day*. Retrieved December 6, 2016 from http://www.un.org/press/en /2005/sgsm10065.doc.htm.
- Archer, D., & Cottingham, S. (1996). Consolidation Research Report on REFLECT (Regenerated Freirean Literacy Through Empowerment Community Techniques). The Experiences of Three REFLECT Pilot Projects in Uganda, Bangladesh and El Salvador. London: ODA Educational Research Serial No. 17.
- Arko, A. D., & Addison, K. (2009). The Impact of Functional Adult Literacy on Socio-Economic lives of Beneficiaries: A case study of the Agona district of Ghana. *Edo Journal of Counselling*. 2(2):199-213.
- Arnove, R. (2013). Introduction: Reframing comparative education: The dialectic of the global and the local. In R. Arnove, C.A. Torres & S. Franz (Eds.) Comparative education. The dialectic of the global and the local. (pp. 1–20). Lanham, Maryland: Rowman and Littlefield.
- Ashe, J., & Parrott, L. (2001). *Impact Evaluation. PACT's Women's Empowerment Program in Nepal.* Waltham, MA: Heller School, Brandeis University, and Freedom from Hunger, manuscript.
- Ayodele, J.B., & Adedokum, M.O.(2012). Towards Achieving Functional Adult Literacy in Nigeria. *European Scientific Journal*. 8(5).
- Azariadis, C., & Drazen, A. (1990). Threshold externalities in economic development. *The Quarterly Journal of Economics*, 105(2), 501-526. Retrieved October 22, 2017 from http://www.jstor.org/stable/pdf/2937797.pdf
- Baku, J., Asiegor, I., & Boakyi-Yiadom, F. (2011). Thematic Case-study on Shaping Basic Education System to Respond to the National Vision for Sustainable Development in Ghana. ADEA Working Document.
- Barro, R. (1991). Economic growth in a cross section of countries. *Quarterly Journal of Economics*, No. 106, pp. 407–43. Retrieved October 22, 2017 from http://www.jstor.

org/stable/pdf/2937943.pdf?refreqid=excelsior%3Af89d7fadd42185228ed8997825fb3 a39

- Barton, D. (1994). Preface: Literacy Events and Literacy Practices. In M. Hamilton, D. Barton, & R. Ivanic, *Worlds of Literacy* (pp. vii-x). Clevedon, Avon: Multilingual Matters, and Toronto: Ontario Institute for Studies in Education.
- Barton, D. (2007). Literacy: An introduction to the ecology of written language. Wiley-Blackwell.
- Barton, D., Hamilton, M., & Ivanic, R. (eds). (2000). Situated Literacies: Reading and writing in Context. London: Routledge.
- Bashir, A. H. M., & Darrat, A. F. (1994). Human capital, investment and growth: some results from an endogenous growth model. *Journal of Economics and Finance*, 18(1), 67-80.
 Retrieved October 22, 2017 from https://link.springer.com/article/10.1007/BF02920223
- Batliwala, S. (1994). The meaning of women's empowerment: new concepts from action, in:
 G. Sen, M.German &L. Chen (eds). *Population Policies Reconsidered: health Empowerment, and Rights*. Cambridge: International Women's Health Coalition and Harvard Center for Population and Development Studies. pp. 127-138.
- Baumeister, R. F. (1999). The self in social psychology. Psychology Press.
- Beder, H. (1999). The Outcomes and Impacts of Adult Literacy Education in the United States (NCSALL Report No. 6). Cambridge, MA: National Center for the Study of Adult Learning and Literacy.
- Beder, H., & Medina, P. (2001). *Classroom Dynamics in Adult Literacy*. National Center for for the Study of Adult Learning and Literacy (NCSALL). Working document.
- Bekele, M. (1966). A Study of Modern Education in Ethiopia: Its foundations, its development, its future, with emphasis on primary education. PhD dissertation, Columbia University.
- Bekele, M., Tesfaye, Y., Mohammed, Z., Zewdie, S., Tebikew, Y., Brockhaus, M., & Kassa, H. (2015). *The context of REDD+ in Ethiopia: drivers, agents and institutions* (Vol. 127). CIFOR.
- Bekhet, A. K., & Zauszniewski, J. A. (2012). Methodological triangulation: an approach to understanding data. *Nurse researcher*, 20(2), 40-43.
- Berhane, G., Dereje, M., Hoddinott, J., Koru, B., Nisrane, F., Tadesse, F., ... & Yohannes, Y. (2011). Agricultural Growth Program (AGP) of Ethiopia: Baseline Report 2011. *Ethiopia Strategy Support Program II. Addis Ababa: IFPRI*. Retrieved December 8, 2017 from http://www.gafspfund.org/sites/gafspfund.org/files/Documents/AGP%20 Baseline%202011%20Report%20-%20Final%20(December%2031,%202012).pdf
- Betts, J. (2002). 'Literacies and livelihoods: the DFID Kathmandu Conference'. Adult *Education and Development*. Retrieved May 8, 2012 from http://www.eaea.info/doc/dvv/58.pdf.
- Bhola, H. S. (1987). Adult Literacy for Development in Ethiopia: A Review of Policy and Performance at Mid-Point.
- Bingham, M.B. (2009). The Tennessee Longitudinal Study of Adult Literacy Program Participants. In S. Reder & J. Bynner (eds.), *Teaching Adult Literacy and Numeracy Skills. Findings from Longitudinal Research*. New York: Routledge, pp. 296-311.
- Bingman, M., & Ebert, O. (2000). 'I've Come a Long Way.' Learner-Identified Outcomes of Participation in Literacy Programs. NCALL Report No. 13, February. Boston, Mass., National Center for the Study of Adult Learning and Literacy. Retrieved October 18, 2017, from http://files.eric.ed.gov/fulltext/ED440301.pdf.
- Bishaw, A., & Lasser, J. (2012). *Education in Ethiopia: Past, Present and Future Prospects*. Retrieved March 10, 2014 from http://www.nobleworld.biz/images/5-Lasser_s_paper.pdf.

- Bitsch, V. (2005). Qualitative research: A grounded theory example and evaluation criteria. *Journal of Agribusiness*, 23(1), 75-91.
- Bowen, G. A. (2009). Document analysis as a qualitative research method. *Qualitative research journal*, 9(2), 27-40.
- Bown, L. (1990). *Preparing the Future: Women, Literacy and Development*. ActionAid Development Report No. 4. London, ActionAid/ODA. Retrieved October 18, 2017, from http://files.eric.ed.gov/fulltext/ED340881.pdf
- Brandt, D., & Clinton, K. (2002). Limits of the local: Expanding perspectives on literacy as a social practice. *Journal of Literacy Research*, *34*(3), 337-356.
- Bray, M., & Thomas, R. M. (1995). Levels of comparison in educational studies: Different insights from different literatures and the value of multilevel analyses. *Harvard Educational Review*, 65(3), 472-491.
- Budge, E. A. W. (2001). The Queen of Sheba and Her Only Son Menyelek (I): Being the" Book of the Glory of Kings"(Kebra Nagast)... Routledge. Retrieved February 20, 2014 from http://thepunkwitch.com/pdf/Queen_of_Sheba.pdf
- Burchfield, S. (1997). An Analysis of the Impact of Literacy on Women's Empowerment in Nepal. Harvard Institute for International Development
- Burchfield, S., H. Hua, D. Baral, & V. Rocha. (2002 a). A Longitudinal Study of the Effect of Integrated Literacy and Basic Education Programs on Women's Participation in Social and Economic Development in Nepal. Washington, D.C. Office of Women in Development, United States Agency for International Development. Retrieved February 22, 2013 from http://www.dec.org/pdf_docs/PNACR861.pdf.
- Burchfield, S, H. Hua, T. Saxo, & V. Rocha. (2002b). A Longitudinal Study of the Effect of Integrated Literacy and Basic Education Programs on Women's Participation in Social and Economic Development in Bolivia. Washington, D.C. Office of Women in Development, United States Agency for International Development. Retrieved February 22, 2013 from http://www.dec.org/pdf_docs/PNACR860.pdf.
- Cameron, J. & Cameron, S. (2005). Economic benefits of literacy. Background paper for *EFA Global Monitoring Report 2006*, through the University of East Anglia, Norwich. Retrieved October 10, 2017 from http://fieldmarshamfoundation.org/wp-content/ uploads/tumblr/References/UNESCO%20economic%20benefits%20literacy%202006. pdf.
- Canieso-Doronila, M. L. (1996). Landscapes of Literacy: An Ethnographic Study of Functional Literacy in Marginal Philippine Communities. Hamburg, Germany, UIE and Luzac Oriental. Retrieved October 20, 2017 from http://files.eric.ed.gov/ fulltext/ED434986.pdf.
- Capra, F. (2003). *The Hidden Connections: A Science for Sustainable Living*. London: Harper Collins.
- Carr-Hill, R (ed.). (2001). *Adult Literacy Programs in Uganda*. Africa Region Human Development Series. Washington: The World Bank.
- Carr-Hill, R.A., Kweka, A.N., Rusimbi, M., & Chengelele, R. (1991) The Functioning and Effects of the Tanzanian Literacy Programme. IIEP Research Report No. 93. Paris: International Institute for Educational Planning. Retrieved October 20, 2017 from http://unesdoc.unesco.org/images/0009/000902/090286e0.pdf
- Carron, G., Mwiria, K., & Righa, G. (1989). The functioning and effects of the Kenyan Literacy Program: A view from the local level (Research Report No. 76). Paris: International Institute for Educational Planning. Retrieved October 20, 2017 from http://unesdoc.unesco.org/images/0008/000850/085078eo.pdf
- Cartwright, N., & Hardie, J. (2012). *Evidence-based policy: a practical guide to doing it better*. Oxford University Press.

Cartwright, N., & Stegenga, J. (2011). *A theory of evidence for evidence-based policy*. Retrieved January 26, 2017 from https:// philpapers .org/archive/CARATO-3.pdf.

- Cavestro, L. (2003). PRA-Participatory Rural Appraisal Concepts Methodologies and Techniques. *Padova, Universita'degli Studi di Padova*, 38.
- Chakravartty, A. (2015). Scientific Realism, *The Stanford Encyclopedia of Philosophy* (Fall 2015 Edition), Edward N. Zalta (ed.). Retrieved November 28, 2014 from URL = http://plato.stanford.edu/archives/fall 2015 /entries/scientific-realism/.
- Chambers, R. (1992). *Rural Appraisal: Rapid, relaxed and participatory*. Institute of Development Studies (UK).
- Collins, J., & Blot, R. K. (2003). *Literacy and Literacies: Texts, Power, and Identity. Cambridge*: Cambridge University Press.
- Collins, R. (1992). *People, Programs, and Politics: Two Case Studies of Adult Literacy Classes*. Unpublished doctoral dissertation. Portland State University, Oregon.
- Comings, J., Smith, C., & Shrestha, C. S. (1995). *Adult Literacy Programs: Design, implementation and evaluation*. USA: USAID working document.
- Comings, J., & Soricone, L. (2007). Adult Literacy Research: Opportunities and Challenges. NCSALL Occasional Paper. *National Center for the Study of Adult Learning and Literacy (NCSALL)*.
- Comings, J., Soricone, L., & Santos, M. (2006). *An Evidence-based Adult Education Program Model Appropriate for Research* (NCSALL Occasional Paper). Cambridge, MA: National Center for the Study of Adult Learning and Literacy.
- Coombs, P.H. (1985). *The World in Crisis in Education: The view from the eighties*. Trenston: Oxford University Press.
- Cope, B., & Kalantzis, M. (eds.) (2000). *Multiliteracies: Literacy learning and the design of social futures*. London: Routledge.
- Coulombe, S., Tremblay, J. F., & Marchand, S. (2004). *Literacy Scores, Human Capital and Growth Across Fourteen OECD Countries*. (Catalogue No. 89-552, No. 11). Ottawa, Ont., Human Resources and Skills Development Canada, Statistics Canada. Retrieved October 20, 2017 from http://annarodrigues.com/wp-content/uploads/2012/02/89-552-MIE2004011.pdf.
- Creswell, J. W. (2009). *Research Design: Qualitative, quantitative, and mixed-methods approaches* (3rd ed.). California: Sage Publications.
- Creswell, J. W. (2012). Educational Research: Planning, conducting, and evaluating quantitative and qualitative research (4th ed.). Boston: Pearson Education, Inc.
- Creswell, J. W., & Clark, V. L. P. (2011). *Designing and conducting mixed methods research*. Thousand Oaks, CA:sage.
- Creswell, J. W., & Clark, V. L. P. (2007). *Designing and conducting mixed methods research*. Thousand Oaks, CA:sage.
- CSA. (2012). Ethiopia demographic and health survey 2011. Addis Ababa, Ethiopia and Calverton, Maryland, USA: Central Statistical Agency and ICF International.
- CSA. (2013). Population Projection of Ethiopia for All Regions at Wereda Level from 2014 2017, Addis Ababa. Retrieved January 3, 2017 from http://www.csa.gov.et/images/general/news/pop_pro_wer_2014-2017_final.
- Culligan, N. (2005). Theoretical Understandings of Adult Literacy: A literature review. New Zealand: The Massey University Department of Communication and Journalism, Wellington and Palmerston North. Retrieved June 28, 2012 from http://women-leadership.massey.ac.nz/massey/fms/Colleges/College%20of%20Business/Communic ation%20and%20Journalism/Literacy/Lit%20review/Theoretical%20Understandings %20of%20Adult%20Literacy.pdf.

- Darkenwald, G. (1986). *Effective approaches to teaching basic skills to adults: A research synthesis.* Washington, DC: Office of Educational Research and Improvement, U.S. Department of Education. Contract No. OERI-P-86-3-15.
- Davidson, K. (1990). The Adult Learner. New Jersey: prentice Hall.
- Dejene, S. (2008). Ethiopian Traditional Values versus the Social Teaching of the Church. *Exchange*, *37*(2), 124-155.
- Delors, J. (1996). Learning: The treasure within Report to UNESCO of the International Commission on Education for the Twenty-first Century, UNESCO.
- Department for International Development (DFID). (2002). *Improving Livelihoods for the Poor: the Role of Literacy*. Background Briefing (March).
- Dighe, A. (2005). Pedagogical approaches to literacy acquisition and effective programme design. Background paper for *EFA Global Monitoring Report 2006*. Retrieved October 20, 2017 https://pdfs.semanticscholar.org/c9a2/50d0514c360f0a8486a41dfc 4c010ca96966.pdf.
- Duffy, M. E. (1987). Methodological triangulation: a vehicle for merging quantitative and qualitative research methods. *Image: The Journal of Nursing Scholarship*, 19(3), 130-133.
- Dwyer, S. C., & Buckle, J. L. (2009). The space between: On being an insider-outsider in qualitative research. *International Journal of Qualitative Methods*, 8(1), 54-63.European Commission. (2000) A Memorandum on Lifelong Learning. Brussels: European Commission.
- Egbo, B. (2000). *Gender, literacy and life-chances in sub-Saharan Africa*. Clevedon, UK: Multilingual Matters.
- European Commission. (2000). A Memorandum on Lifelong Learning, Brussels: Commission of the European Communities (Commission Staff Working Paper). Retrieved July 22, 2013 from http://tvu.acs.si/ dokumenti/LLLmemorandum Oct2000.pdf.
- Farah, I. (2005). The cultural benefits of literacy. Background paper for *EFA Global Monitoring Report 2006*, through the University of East Anglia, Norwich. Retrieved October 19, 2017 from http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.566. 1622&rep=rep1&type=pdf
- Farrell, G. (ed.). (2004). ICT and literacy. Who benefits? Experience from Zambia and India. Final Report on the COL Literacy Project (COLLIT). Vancouver, Canada: Commonwealth of Learning. Retrieved May 10, 2012 from URL: http://www.col.org/ Consultancies/04Literacy. htm
- Fransman, J. (2008). Conceptualizing Literacy for Policy and Practice. Adult Education and Development 71/2008: DVV Publications. Retrieved June 20, 2012 from http://www.iiz-dvv.de/index. php?article_id=801&clang.
- Friere, P. (1970) *Pedagogy of the Oppressed*. Translated by M.B. Ramos. New York: Seabury Press.
- Freire, P. (1995). Pedagogy of Hope: Reliving Pedagogy of the Oppressed: New York: Continuum.
- Freire, P., & Macedo, D. (1987). Reading the Word and the World. London: Routledge.
- Gall, M. D., Gall, J. P., & Borg, W. R. (2007). Collecting research data with questionnaires and interviews. *Educational research: An introduction*. Boston: Allyn and Bacon.
- Gee, J. P. (2007). Social Linguistics and Literacies: Ideology in discourses. Routledge.
- Global Campaign for Education (GCE). (2009). Ten Key Messages for the Delayed CONFINTEA Meeting. Global Campaign for Education.
- Gomm, R., Hammersley, M., & Foster, P. (2000). Case study and generalization. *Case study method*, 98-115. Retrieved March 4, 2018 from https://www.researchgate.net/profile/ Roger_Gomm/publication/42800720_Case_study_and_generalisation/links/5a67068fa 6fdcc72a589c4a5/Case-study-and-generalisation.pdf.

- Goody, J.R. (1977). *The Domestication of the Savage Mind*. Cambridge, UK: Cambridge University Press.
- Gray, W.S. (1990). *Teaching of Reading and Writing: An international survey*. Chicago: Scott Foresman and Co.
- Greaney, V. (1996). Promoting reading in developing countries. Newark, DE: International Reading Association
- Greene, J.C., Caracelli, V.J., & Graham, W.F. (1989).Toward a conceptual framework for mixed-method evaluation designs. *Education Evaluation and Policy Analysis*, 11(3), 255-274.
- Hable-Selassie, S., & Tamrat, T. (1970). *The Church of Ethiopia: A panorama of history and spiritual life*. Addis Ababa: EOTC
- Haile, A. (2008). *The Impact of Devaluation on Trade Balance: The Case of Ethiopia*. (MA Thesis, University of Oslo, Department of Economics).
- Hamilton, M., Hiller, Y., & Tett, L. (2006). Introduction: Social practice of adult literacy, numeracy and language. In Tett, L., Hamilton, M., and Hiller, Y. (eds.) (2006). Adult Literacy, Numeracy, and language: Policy, practice and research, pp. 1-18. Maidenhead: Open University Press.
- Hannum, E. & Buchman, C. (2003). The Consequences of Global Educational Expansion. Cambridge, Mass., American Academy of Arts and Sciences. Retrieved October 19, 2017 from https://scholar.google.com/scholar?hl=en&as_sdt=0%2C5&q=Hannum% 2C+E.+and+Buchman%2C+C.+2003.+The+Consequences+of+Global+Educational+ Expansion.+Cambridge%2C+Mass.%2C+American+Academy+of+Arts+and+Science s&btnG=
- Harvey, D. (1996). Justice, Nature & the Geography of Difference. Oxford: Blackwell.
- Hoppers, W., & Yekhlef, A. (2012). Common Core Skills for Lifelong Learning and Sustainable Development in Africa. Synthesis prepared on the occasion of the 2012 ADEA Triennial on Education and Training in Africa on the theme: Promoting Critical Knowledge, Skills and Qualifications for Sustainable Development in Africa, Ouagadougou, Burkina Faso, February 12-17, 2012. ADEA, Tunis, December 2013. Retrieved March 27, 2014 from http://commons.wikimedia.org/wiki/File:Ethiopia_ regions_english.png.
- http://www.ethiodemographyandhealth.org/Amhara.html; http://www.tigraytube.com/article_read.php? a=. Retrieved January 26, 2017.
- Huckle, J. (2008). Sustainable Development. In Arthur, J., Davies, I., & Hahn, C. (Eds.). (2008). Sage Handbook of Education for Citizenship and Democracy. Saw.
- Huckle, J. (2012). Towards greater realism in learning for sustainability. *Learning for Sustainability*, 35-48.
- Infante, M.I. (ed.). (2000). Alfabetismo Funcional en siete países de América Latina. Santiago: UNESCO Oficina Regional para America Latina y el Caribe.
- Ingram, D. (2018). *Why Is Organizational Structure Important?* Retrieved January 24, 2018 from http://smallbusiness.chron.com/organizational-structure-important-3793.html.
- Inquai, S. (1998). *Adult Education in Ethiopia: A historical sketch*. Addis Ababa: IIZ/DVV Project Office Ethiopia.
- Ishaq, A., & Ali, M. (2012). Effective planning as a Potent Tool for Implementing Adult Literacy Programmes: Implications for achieving the Millennium Development Goals in Gombe State, Nigeria. *Journal of Economics and Sustainable Development*, 3(6), 93-98. Retrieved November 22, 2014 from http://iiste.org/Journals/index.php/JEDS/ article/view File/1684/1646.
- Ito, H. (2013). Literacy for All: A Neglected" EFA" Goal?. Journal of International Education & Leadership, 3(3). Retrieved January 10, 2018 from https://scholar.

google.com/scholar?hl=en&as_sdt=0%2C5&q=Literacy+for+all%3A+a+neglected+% 27EFA%27+goal%3F&btnG=.

- Jurmo, P. (1987). Learner Participation Practices in Adult Literacy Education Efforts in the United States. Unpublished doctoral dissertation, University of Massachusetts, Amherst.
- Kagitcibasi, C., Goksen, F., & Gulgoz, S. (2005). Functional Adult Literacy and Empowerment of Women: Impact of functional literacy in Turkey. *Journal of Adolescent & Adult Literacy*. March 2005. 48 (6):473-489.
- Kebede, M. (2005). Adult education in Ethiopia: situation, statistics, trends and policies, in Adult Education No 21 (*dvv international*, Regional Office East Africa) pp. 13–23).
- Kebede, M. (2006). Ethiopia: Where and who are the world's illiterates? *Background paper* prepared for the Education for All Global Monitoring Report.
- Kenea, A. (2014). Adult basic literacy" initiatives" in Ethiopia: change and continuity. *Journal for Critical Education Policy Studies (JCEPS)*, 12(1).
- King, N. (1994). The qualitative research interview. In C.Cassell and G. Symon, eds, *Qualitative Methods in Organizational Research*. London: Sage 271.
- Kruidenier, J. (2002). Research-based Principles for Adult Basic Education Reading Instruction. Washington, DC: National Institute for Literacy. Retrieved August 20, 2013 from http://www.nifl.gov/partnershipforreading/publications/ adult. html.
- Kvernbekk, T. (2015). Evidence-based Practice in Education: Functions of evidence and causal presuppositions (Vol. 147). Routledge.
- Lauglo, J. (2001). Engaging with Adults: The Case for Increased Support to Adult Basic Education in sub-Saharan Africa. Washington, DC, World Bank. (Africa Region Human Development Working Paper Series.) Retrieved October 10, 2013 from http://www1.worldbank.org/education/ adultoutreach/Doc/ENGAGING.pdf.
- Lau, J. & Chan, J. (2018). [S04] Causation. *Scientific Methodology*. Retrieved March 3, 2018 from http://philosophy.hku.hk/think/sci/causation.php
- Lerche, R. (1985). *Effective Adult Literacy Programs: A practitioner's guide*. New York: Cambridge.
- Levine, D. (1974). *Greater Ethiopia: The evolution of a multiethnic society*. Chicago: University of Chicago Press.
- Levine, K. (1994). Functional literacy in a changing world. Functional Literacy, 113-31.
- Lind, A. (1985). *Adult Literacy in the Third World*. Stockholm: University of Stockholm. Institute of International Education.
- Lind, A. (2008). Literacy for all: Making a difference (Vol. 89). Paris: Unesco.
- Mackie, J. L. (1965). Causes and conditions. *American Philosophical Quarterly*, 2(4), 245-264.
- Maddox, B. (2001). Literacy and the Market: the economic uses of literacy among the peasantry in north-west Bangladesh. In: Street, B.V. (Ed) *Literacy and Development: Ethnographic Perspectives*. London: Routhledge, 137-151
- Maddox, B. (2007). What can ethnographic studies tell us about the consequences of literacy?. *Comparative Education*, 43(2), 253-271.
- Mammo, G. (1982). The National Literacy Campaign in Ethiopia. *Prospects Quarterly Review of Education* vol.xii, no. 2, 1982.
- Maykut, P., & Morehouse, R. (1994). Beginning Qualitative Researchers: A philosophical and practical guide. Washington, DC: Falmer.
- McCune, D., & Alamprese, J. (1985). *Turning Illiteracy Around: An agenda for national action*. New York: Business Council for Effective Literacy.
- McDonald, B., & Scollay, P. A. (2009). Outcomes of literacy improvement: A longitudinal view. In S. Reder & J. Bynner (Eds.), Tracking Adult Literacy and Numeracy Skills. Findings from Longitudinal Research. New York: Routledge, pp. 312-328.

- Mclachlan, K., Tett, L. & Hall, S. (2009). The More You Learn the Better You Feel. Research into Literacies, Learning and Identity in Scotland. In S. Reder & J. Bynner (eds.), *Teaching Adult Literacy and Numeracy Skills. Findings from Longitudinal Research*. New York: Routledge, pp. 329-348.
- Merriam, S.B. (1998). Case Study Research in Education: A qualitative approach. San Francisco: Jossey-Bass.
- Merriam, S. B., & Brockett, R. G. (1997). *The Profession and Practice of Adult Education*. San Francisco: Jossey-Bass.
- Merrifield, J. (1998). Contested Ground: Performance Accountability in Adult Basic Education. Cambridge, MA: National Center for the Study of Adult Learning and Literacy (NCSALL) Report# 1.
- Metcalf, H., & Meadows, P. (2009). Outcomes for Basic Skills Learners. A Four-Year Longitudinal Study. In S. Reder & J. Bynner (eds.), *Teaching Adult Literacy and Numeracy Skills. Findings from Longitudinal Research*. New York: Routledge, pp. 225-241.
- Mezirow, J., Darkenwald, G., & Knox, A. (1975). *Last Gamble on Education*. Washington, DC: Adult Education Association of the U.S.A.
- Ministry of Education (MoE). (1986). The Ethiopian National Literacy Campaign: Its achievements and prospects. Addis Abeba: Ministry of Education.
- MoE. (1994). *Education and Training Policy*. Federal Democratic Republic of Ethiopia (FDRE). Transitional Government of Ethiopia. Addis Ababa.
- MoE. (2002). Education Sector Development Program II (ESDP II). FDRE. Addis Ababa.
- MoE/ EMIS. (2005). Education Statistics Annual Abstract (1996 E.C.), 2003/4. Addis Ababa.
- MoE. (2005). Education Sector Development Program III (ESDP III). Program Action Plan (PAP). FDRE. Addis Ababa.
- MoE. (2008). National Adult Education Strategy. FDRE. Addis Ababa.
- MoE. (2010a). Learning for Life: Master Plan for Adult Education in Ethiopia (2010/11 2019/20). FDRE. Addis Ababa.
- MoE. (2010b). Education Sector Development Program IV (ESDP IV). PAP. FDRE. Addis Ababa.
- MoE. (2010c). National IFALE Curriculum Framework. Addis Ababa
- MoE. (2012). IFALE Implementation Guideline for 2011/12 and 2012/13. Addis Ababa.
- Molla, T. (2008). *Revitalizing Adult Education in Ethiopia: the importance of lifelong learning perspectives*. Arhus University. Retrieved March 20, 2014 from http://www.ioe.stir.ac.uk/.
- Nafukho, F. Amutabi, M. & Otunga, R. (2005). Foundations of Adult Education in Africa. *African Perspectives on Adult Learning*. Pearson Education South Africa: Capetown
- National Literacy Campaign Coordinating Committee (NLCCC). (1981, 1983). Every Ethiopian Will Be Literate and Remain Literate. Addis Ababa
- Naudé, W. A. (2004). The effects of policy, institutions and geography on economic growth in Africa: an econometric study based on crosssection and panel data. *Journal of International Development*, Vol. 16, pp. 821–49. Retrieved October 20, 2017 from http://onlinelibrary.wiley.com/doi/10.1002/jid.1129/pdf
- Ndoye, M., & Walther, R. (2011). Critical knowledge, skills and qualifications for accelerated and sustainable development in Africa.
- Nedović, M., & Božinović, D. (2013). The effects of organizational structure on the efficiency managers. *Economic Crisis and European Union*, 1(1) Retrieved January 24, 2018 October 20, 2017 from https://scholar.google.no/scholar?hl=en&as_sdt= 0%2C5&q=The+effects+of+organizational+structure+on+the+efficiency+managers& btnG=
- Negash, T. (1996). Rethinking education in Ethiopia. Uppsala: The Nordic Africa Institute

- Niehoff, R. O., & Wilder, B. D. (1974). Non-Formal Education in Ethiopia. Program of Studies in Non-Formal Education. Team Reports.
- Nussbaum, M. (2000). *Women and Human Development: the Capabilities Approach*. Cambridge: Cambridge University Press.
- Nyssen, J. (1998). Soil and water conservation under changing socio-economic conditions in the Tembien highlands (Tigray, Ethiopia). *Bull. de la Société géographique de Liège*, 35, 5-17.
- OECD. (2000). Literacy in the Information Age. Paris: OECD.
- OECD & Statistics Canada. (1995). *Literacy, economy and society: Results of the first International Adult Literacy Survey.* Paris: Organization for Economic Cooperation and Development.
- Okech, A., Carr-Hill, R. A., Katahoire, A. R., Kakooza, T., & Ndidde, A. N. (1999). An Evaluation of Literacy Programmes in Uganda. *Draft. Kampala: Ministry of Gender* and Social Development. Retrieved October 20, 2017 from http://web.worldbank.org/ archive/website00573/WEB/PDF/ADULT_LI.PDF.
- Okech, A., Carr-Hill, R.A, Katahoire, A. R., & Kakooza, T., (2001). Adult literacy programs in Uganda. Africa Region Human Development Series. Washington, D.C.: The World Bank. Retrieved October 21, 2017 from http://documents.worldbank.org/curated/ en/5291314687655 62765/ Adult-literacy-programs-in-Uganda.
- Okech, A. (2009). Functional adult literacy and poverty reduction. In B. Sandhass (Ed.). *Cross-National Encounters in Adult Education: Public-Private Partnership in Adult Literacy*. International conference. 5-7 December 2006, Addis Ababa. Ethiopia.
- Olaleye, F.O, & Adeyemo, C.W.(2012). Functional literacy a tool for empowering women for sustainable development. *European Journal of Educational Studies*. 4(2).
- Olson, D. (1977). From Utterance to Text: The bias of language in speech and writing. *Harvard Educational Review*, 47, 257-281.
- OPride. (2015). TPLF 40: A celebration of Tigrayan dominance over Ethiopia by OPride contributor based in the United States. Retrieved February 21, 2018 from https://www.opride.com/2015/03/01/a-celebration-of-tigrayan-dominance-over-ethiopia/.
- Ouane, A. (2008). Lifelong Learning Connections: the nodal role of diversified post-primary and post-basic approaches. UIL paper to session 6 (Quality and Relevance of Post-Primary Education: Demand and Supply of teachers for Post-Primary Education, Curriculum and Skills Development) of the ADEA Biennial Meeting, Maputo, Mozambique, May 5-9, 2008.
- Oxenham, J. (2008). Effective literacy programmes: options for policy makers (Vol. 91). Paris: Unesco. Retrieved October 15, 2017 from http://unesdoc.unesco.org/images/ 0016/001636/163607e.pdf
- Oxenham, J., Diallo, A., Katahoire, A. R., Petkova-Mwangi, A., & Sall, O. (2002). Skills and Literacy Training for Better Livelihoods: A Review of Approaches and Experiences. Washington, DC, World Bank. Africa Region Human Development Series. Retrieved October 15, 2017 from http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.194. 4431 &rep=rep1&type=pdf
- Page, N., & Czuba, C. E. (1999). Empowerment: What Is It?. *Journal of extension*, 37(5), n5. Retrieved October 4, 2012 from http://www. joe.org/joe/1999october/comm1.php/ index.php.
- Pankhurst, R. (1962). The Foundation of Education, Printing, Newspapers, Book Production, Libraries and Literacy in Ethiopia. *Ethiopian Observer*, (6)3.
- Pankhurst, R. (1998). History of Education, Printing and Literacy in Ethiopia. 9: Educational Advances in Menilek's Day. *Addis Tribune, Oct*, 2.
- Papen, U. (2005). Adult Education as Social Practice: More than skills. London: Routledge.

- Patton, M.Q. (1990). *Qualitative Evaluation and Research Methods* (2nd ed.). Newbury Park, CA: Sage.
- Philadelphia Public Health Management Corporation (PHMC). (2011). Retrieved September 25, 2014 from *http://www.sacredplaces.org/wp-content/uploads/2011/12/Adult Educationand Literacy.pdf*.
- Phillips, D. C. (1987). *Philosophy, Science and Social Inquiry: Contemporary methodological controversies in social science and related applied fields of research*. Pergamon Press.
- Power, C., & Maclean, R. (2011). Lifelong Learning for Poverty Alleviation and Sustainable Development. International Symposium on Lifelong Learning for Poverty Alleviation and Sustainable development: Developing a research agenda for the Asia-Pasific. Hong Kong: Hong Kong Institute of Education 12-13 January 2011.
- Preece, J. (2013). Africa and international policy-making for lifelong learning: textual revelations. *International Journal of Educational Development*, 33(1), 98-105. Retrieved September 20, 2014 from http://www.sciencedirect.com/science/article/ pii/S0738059312000260.
- Prins, E. (2008). Adult literacy education, gender equity and empowerment: Insights from a Freirean-inspired literacy programme. *Studies in the Education of Adults*, 40(1): 24-39.
- Psacharopoulos, G., & Woodhall, M. (1985). Education for Development: An analysis of investment choices. New York: Oxford University Press.
- Purcell-Gates, V., Degener, S., & Jacobson, E. (1998). US Adult Literacy Program Practice: A Typology across Dimensions of Life-Contextualized/Decontextualized and Dialogic/Monologic. NCSALL Report# 2.
- Rabinowitz, P. (2012). Planning an adult literacy program. *The Community Tool Box*. Retrieved July 20, 2012 from http://ctb.ku.edu/en/tablecontents/sub_section_main_ 1254.aspx.
- Reder, S. (2009). The Development of Literacy and Numeracy in Adult Life. In S. Reder & J. Bynner (eds.), *Teaching Adult Literacy and Numeracy Skills. Findings from Longitudinal Research*. New York: Routledge, pp. 59-131.
- Robinson-Pant, A. (2005). The social benefits of literacy. Background paper for *EFA Global Monitoring Report 2006*, through the University of East Anglia, Norwich. Retrieved October 21, 2017 from http://unesdoc.unesco.org/images/0014/001461/146106e.pdf.
- Robson, C. (2011). Real World Research. Third Edition. UK; Wiley
- Rogers, A. (1999). Improving the quality of adult literacy programs in developing countries: the real literacies' approach. *International Journal of Educational Development*, 19(3), 219-234.
- Rogers, A. (2001). Afterword: Problematizing literacy and development. In B. Street (Ed.), *Literacy and development: Ethnographic perspectives* (pp. 205-222). London, UK: Routledge.
- Rogers, A., & Street, B. V. (2012). Adult Literacy and Development: Stories from the field. Leicester: niace.
- Rogers, R., & Kramer, M. A. (2008). Adult education teachers: Designing critical literacy practices. Routledge.
- Roseman, G., & Wilson, B. (1985).Numbers and words: Combining quantitative and qualitative methods in a single large-scale evaluation study. *Evaluation Review*, 9(5), 627-643.
- Rowe, S. (1994). Ecocentrism: the Chord that Harmonizes Humans and Earth. *The Trumpeter* 11(2): 106-107. Retrieved April 12, 2016 from http://www.ecospherics.net/pages/RoweEcocentrism.html.
- Sachs, J. & Warner, A. (1997). Sources of slow growth in African economies. *Journal of African Economics*, 6(3), pp. 335–76.
- Sandhaas, B. (2009). Portrait of Adult education in Ethiopia. Internal Paper No76 (special edition), dvv International.
- Schaffer, J. (2016). The Metaphysics of Causation, *The Stanford Encyclopedia of Philosophy* (Fall 2016 Edition), Edward N. Zalta (ed.), Retrieved March 4, 2018 from URL = https://plato.stanford.edu/archives/fall2016/entries/causation-metaphysics/.
- Schwandt, T. (1997). Qualitative Inquiry: A dictionary of terms. Sage Publications, Inc.
- Schwandt, T. (2007). *The SAGE Dictionary of Qualitative Inquiry*. (3rd ed.). California: Sage Publications Inc.
- Scribner, S., & Cole, M. (1981). *The Psychology of Literacy* (Vol. 198, No. 1). Cambridge, MA: Harvard University Press.
- Selber S. (2004). *Multiliteracies for a Digital Age*. USA: South Illinois University Press. Sen, A. (1999). *Development as Freedom*. Oxford University Press, Oxford.
- Sen, A. (1985). Commodities and Capabilities. Amsterdam: Elsevier Science.
- Sen, A. (1999). Development as Freedom, Oxford: Oxford University Press.
- Sharma, V. (2016). 16 Ways You Can Follow to Be a Good Citizen. Retrieved January 20, 2018 from http://www.klientsolutech.com/be-a-good-citizen/
- Shor, I. (1993). Education is Politics: Paulo Freire's Critical Pedagogy in P. McLaren and P. Leonaard (eds). *Paulo Freire: A critical encounter*, pp. 169-176. New York: Routledge.
- Silverman, D. (2006). Interpreting Qualitative Data (3rd ed.). London: Sage Publications Ltd.
- Sineshaw, T. (1996). The Land of the Scribe and the Thumb Print: the Sociology and Politics of Literacy in Ethiopia. *Proceeding of World Conference on Literacy*. Retrieved April 25, 2017 from http://www.literacy.org/sites/literacy.org/files/publications/sineshaw lit_in_ethiopia_96.pdf
- Sjöström, M., & Sjöström, R. (1982). Literacy and development. Sweden: University of Umeå
- Stefanou, C. R., Perencevich, K. C., DiCintio, M., & Turner, J. C. (2004). Supporting autonomy in the classroom: Ways teachers encourage student decision making and ownership. *Educational Psychologist*, 39(2), 97-110.
- Street, B. (1995). Social Literacies: Critical approaches to literacy in development, education and ethnography. London: Longman. Retrieved April 22, 2017 from https://www. researchgate.net/profile/Julianne_Turner/publication/247522695_Supporting_Autono my_in_the_Classroom_Ways_Teachers_Encourage_Student_Decision_Making_and_ Ownership/links/0f3175388b1403925f000000.pdf.
- Street, B. (1984). Literacy in Theory and Practice. CUP: Cambridge.
- Street, B. (1988). *Literacy in Theory and Practice*. Cambridge, UK: Cambridge University Press.
- Stromquist, N. (1995). The theoretical and practical bases for empowerment. In C. Medel-Anonuevo (ed.), Women, Education, and Empowerment: Pathways towards autonomy. Hamburg: UNESCO Institute for Education, pp.13-22.
- Stromquist, N. (1997). Literacy for Empowerment: Gender and Grassroots Dynamics in Brazil. Albany: State University of New York Press.
- Stromquist, N. (2009). Literacy and empowerment Evidence of the link and its Implications for action. *Occasional Papers in Education and Lifelong Learning: An International Journal*, 4 (1-2), 63-81.
- Stromquist, N. (2015). Women's Empowerment and Education: Linking knowledge two transformative action. *European Journal of Education*, 50 (3), 307-324. Retrieved March 17, 2016 from http://onlinelibrary.wiley.com.
- Tashakkori, A., & Teddlie, C. (2008). Quality inferences in mixed methods research. In Bergman, M. (Ed.) Advances in Mixed Methods Research: Theories and Applications. London: Sage, pp. 101-119.

- Tashakkori, A., & Teddlie, C. (Eds.). (2010). Sage handbook of mixed methods in social & behavioral research. Sage.
- Technocentrism. (2016, September 16). In *Wikipedia, The Free Encyclopedia*. Retrieved 10:12, April 12, 2017 from https://en.wikipedia.org/w/index.php?title= Technocentrism & oldid=739683154.
- Teddlie, C., & Tashakkori, A. (2003). Major Issues and Controversies in the Use of Mixed Methods in the Social and Behavioral Sciences, in *Handbook of Mixed Methods in Social and Behavioral Research*, A. Tashakkori and C. Teddlie (eds.), Thousand Oaks, CA: Sage Publications, pp. 3-50.
- Teddlie, C., & Tashakkori, A. (2009). *Foundations of Mixed Methods Research*, Thousand Oaks, CA: Sage Publications.
- Teddlie, C., & Tashakkori, A. (2010). Overview of contemporary issues in mixed methods research. *Handbook of mixed methods in social and behavioral research*, 1-41.
- Teferra, D., & Altbach, P. G. (2003). African Higher Education: An International Reference Handbook.
- Tekleyesus, A. (2004). Unfinished Pledge: The Ethiopian literacy campaign: A comparative study for the period 1979-1990. Utrecht University.
- Tett, L., & St. Clair, R. (2010). Adult Literacy Education. In K. Rubenson (Ed.), *Adult Learning and Education* (pp.100-105). United Kingdom: Elsevier Limited.
- *The Concise Oxford Dictionary of World Religions.* (2005). Levirate marriage. Retrieved April 12, 2017 from Encyclopedia.com:http://www.encyclopedia.com/religion/dictionaries-thesauruses-pictures-and-press-releases/levirate-marriage_
- The National Committees for Research Ethics in Norway. (2006). *Guidelines for Research Ethics in the Social Sciences, Law and the Humanities.* Retrieved May 20, 2012 from http://www.etikkom.no/en/In-English/Publications/.
- Tobin, G. A., & Begley, C. M. (2004). Methodological rigour within a qualitative framework. *Journal of advanced nursing*, 48(4), 388-396. Retrieved March 5, 2018 form http://202.28.25.163/mis/download/course/lec_566823_Tobin%20-%20Jan%2022.pdf
- Tigray Bureau of Education (TBE). (2010). Facilitators' Manual. Meqhelle. Tigray
- Torrance, R. (2012). Triangulation, respondent validation, and democratic participation in mixed methods research. *Journal of Mixed Methods Research*, 6(2), 111-123.
- Torres, R. (2003). Lifelong learning: a new momentum and a new opportunity for adult basic learning and education (AVBLE) in the South. *Adult Education and Development Supplement 60*, 1–240.
- Torres, R. (2006). Literacy and Lifelong Learning: Linkages. ADEA Working Document.
- Torres, C. A. (2013). Adult Learning Education and Literacy as International Comparative Education. In *Political Sociology of Adult Education* (pp. 67-75). SensePublishers.
- Trochim, W. (2006). Research Methods Knowledge base. Web Centre for social research methods. Retrieved February 28, 2016 from http://www.socialresearch methods.net/kb/constval.php.
- Trochim, W., Donnelly, J., & Arora, K. (2015). *Research methods: The essential knowledge base*. Cengage Learning.
- Trudell, B. (2009). Local-language literacy and sustainable development in Africa. *International Journal of Educational Development, Elsevier.* 29(1), 73–79.
- United Nations. (1987). Report of the World Commission on Environment and Development: Our common future. Geneva: United Nations.
- United Nations. (2015). Transforming Our World: The 2030 agenda for sustainable development. A/Res/70/1, United Nations. New York. Retrieved February 10, 2016 from https:// sustainabledevelopment.un.org/content/documents/21252030 %20Agenda%20for%20Sustainable%20 Development%20web.pdf.

- United Nations Economic Commission for Africa. (2005). Sustainable Development Report on Africa. *Managing land-based resources for sustainable development*. Retrieved February 26, 2014 from http://www.uneca.org/sites/default/files/publications/sdra1full.pdf.
- United Nations Environment Program (UNEP). (2004). *Environment and Cultural Diversity*. Nairobi: United Nations.
- UNESCO. (1976). *Recommendation on the Development of Adult Education*. Adopted by the general assembly at its nineteenth session, Nairobi, 26 November, 1976. Retrieved August 22, 2013 from http://www.unesco.org/education/pdf/NAIROB-E.PDF.
- UNESCO. (2000a). The Dakar Framework for Action: Education for All: Meeting Our Collective Commitments: Including Six Regional Frameworks for Action. UNESCO. Paris.
- UNESCO. (2000b). The Earth Charter. Retrieved August 1, 2016 from http://www. unesco.org/education/tlsf/mods/ theme _a /img/02_earthcharter.pdf.
- UNESCO. (2001). The Luneburg Declaration of Higher Education for Sustainable Development. Retrieved May 18, 2016 from http://www.iau-hesd.net/sites/default/ files/documents/ Luneburgdeclaration_0.pdf.
- UNESCO. (2004). The Plurality of Literacy and its Implications for Policies and Programs (Position Paper, UNESCO Education Sector). Paris, France. Retrieved May 5, 2012 from from http://unesdoc.unesco.org/images/0013/001362/136246e.pdf.
- UNESCO. (2005a). Session on literacy and empowerment issues paper. Sixth Meeting Working Group on Education for All (EFA). Paris.
- UNESCO. (2005b). United Nations Decade of Education for Sustainable Development (2005-2014): International Implementation Scheme. Paris. Retrieved December 4, 2012 from http://unesdoc. unesco.org/images/0014/001486/148654E.pdf.
- UNESCO. (2005c). *EFA Global Monitoring Report* 2006. Education for All: Literacy for Life. Paris: UNESCO. Retrieved October 5, 2017 from http://unesdoc.unesco. org/images/0014/001416/141639e.pdf.
- UNESCO. (2006a). Education for Sustainable Development Toolkit. Paris: UNESCO Education Sector.
- UNESCO. (2006b). Framework for the UN DESD International Implementation Scheme. Retrieved December 4, 2012 from http://unesdoc.unesco.org/images/0014/ 001486/148650E.pdf.
- UNESCO. (2007). UN Decade of *Education for Sustainable Development (DESD 2005-2014: The First Two Years*. Retrieved October 15, 2012 from http://unesdoc.unesco. org/images/0015/ 001540/154093e.pdf.
- UNESCO. (2009a). UNESCO World Conference on Education for Sustainable Development. Bonn Declaration. Retrieved May 20, 2014 from http://www.esd-world conference-2009. org/en/ whats-new/news-detail/item/bonn-declaration-adopted. html.
- UNESCO. (2009b). *Global Report on Adult Learning and Education*. UNESCO Adult and Lifelong Learning: Hamburg.
- UNESCO. (2009c). United Nations Literacy Decade: International Strategic Framework for Action. Paris: UNESCO.
- UNESCO. (2015). Institute of Statistics. Retrieved January 12, 2016 from http://en.unesco. org/countries/ethiopia.
- UNESCO. (2017). *Literacy and Numeracy from Lifelong Perspective*. UNESCO Institute for Lifelong Learning. UIL Policy Brief 7. Retrieved January 12, 2018 from http://unesdoc.unesco.org/images/0024/002470/247094E.pdf

- UNESCO Institute for Statistics (UIS). (2012). Adult and Youth Literacy, 1990-2015 Analysis of data for Selected countries. Retrieved March 14, 2015 from http://www.uis.unesco.org/Education/Documents/UIS-literacy-statistics-1990-2015-en.pdf.
- Vasumathi, T. (2008). *The Impact of Total Literacy Programs on Neo-literate Adults of Malappuram District in Kerala.* Unpublished PhD Dissertation. Department of Education. University of Calicut, India.
- Venkatesh, V., Brown, S. A., & Bala, H. (2013). Bridging the qualitative-quantitative divide: Guidelines for conducting mixed methods research in information systems. *Mis Quarterly*, 37(1), 21-54.
- Wagaw, T. G. (1979). *Education in Ethiopia: Prospect and retrospect*. Ann Arbor, MI: University of Michigan Press.
- Walta Information Center. (2015, November 3). *Discussion on the Outcome of the National Good Governance Study Parts 1 and 2* [Video file].Retrieved from https://www.youtube.com/watch?v= LEktoA6NUIU
- Walters, S., & Cooper, L. (2011). Learning/work: Turning work and lifelong learning inside out. *International Review of Education*, 57(1-2), 27. Retrieved March 16, 2014 from http://www.jstor.org/stable/ 41480682?seq=1#page_scan_tab_contents.
- Walters, S., Yang, J., & Roslander, P. (2012). Study on Key Issues and Policy Considerations in Promoting Lifelong Learning in Selected African Countries Ethiopia, Kenya, Namibia, Rwanda and Tanzania. *Tunis: Association for the Development of Education in Africa (ADEA)*. Retrieved April 24, 2013 from http://uil.unesco.org /fileadmin/keydocuments/Africa/en/study-on-LLL_UIL_final_11-Jan-2012.pdf.
- Webster, M. (2006). Merriam-Webster online dictionary. Retrieved March 8, 2017 from https://www.merriam-webster.com/dictionary/fundamental%20 education.
- Werqneh, T. (2011). *The adult education practice in Ethiopia*. Addis Ababa: Mega Publishers.
- Windham, D.M. (1999) Literacy and Economic Development. In D.A. Wagner, R.L. Venezky and B.V. Street (eds.) Literacy: An International Handbook, ch. 55. Oxford: Westview Press.
- Wirtu, D. (2005). Provision and participants of adult education: an overview. Adult Education and Development (dvv International East African Office –Addis Ababa), pp. 197-211.
- Woldemeskel, Z. (2005). The legal basis of adult education: legislation. *Adult Education and Development* (#22). dvv International.
- Woodward, J. (2005). *Making Things Happen: A theory of causal explanation*. Oxford university press.
- World Bank. (2014). *Ethiopia/Population* Retrieved January 20, 2017 from https://www. google.no/publicdata/explore?ds=d5bncppjof8f9_&met_y=sp_pop_totl&idim=co untry:ETH:NGA:EGY&hl=en&dl=en
- World Bank. (2016a). Adult Literacy Rate, Population 15+ Years, both Sexes (%). Retrieved January 1, 2016 from http://data.worldbank.org/indicator/SE.ADT.LITR.ZS? Locations=ET.
- World Bank. (2016b). *The World Bank Group*. Retrieved January 1, 2016 from http://www.worldbank.org/en/country/ Ethiopia.
- World Bank. (2016c). *Ethiopia: Overview*. Retrieved April 25, 2017 from *http://www.worldbank.org/en/country/ethiopia/overview*.
- Yamane, T. (1967). *Statistics: An Introductory Analysis*, 2nd Ed., New York: Harper and Row.
- Yibing, L. (2000). Role of Literacy Education in Sustainable Development of Poor Rural Ethnic groups. Retrieved December 3, 2016 from http://www.accu.or.jp/litdbase/ break/pdf/CHNf911A.PDF.

- Youngman, F. (1998). 'Botswana'. In African Adult Education: Chronologies in Commonwealth Countries, ed., J.A. Draper, pp.26-32. Cape Town: CACE, University of Western Cape.
- Young, M. B., Fitzgerald, N., & Morgan, M. A. (1994). *National Evaluation of Adult Education*. Arlington, VA: Development Associates.
- Young, M. B., Hipps, J., Hanberry, G., Hopstoch, P., & Golsmat, R. (1980). An Assessment of the State-Administered Program of the Adult Education Act: Final Report. Arlington, Va., Development Associates. Retrieved April 25, 2017 from https://eric.ed.gov/?q=An+Assessment+of+the+State-Administered+Program+of +the+Adult+Education+Act.+Final+Report.&id=ED195700
- Zents, A. (2005). Gender, education and Pentecostalism: the women's movement within the sssemblies of God in Burkina Faso. In: Aikman, S., Unterhalter, E. (Eds.), Beyond Access: Transforming Policy and Practice for Gender Equality in Education. Oxfam GB, Oxford, pp. 212–226.
- Zimmerman, M.A. (1984). Taking Aim on Empowerment Research: On the distinction between individual and psychological conceptions. *American Journal of Community Psychology*, 18(1), 169-177.

Appendices

A.Interview Protocols

1. Interview Protocol for IFALE Participants

Different people can have a different understanding of literacy education due to various reasons. As I would like to know about your understanding the state of adult literacy education, as well as contributions to learner empowerment and SD, I will interview you for approximately one and half hours; I am going to take notes and audiotape the interview.

I. Personal Information

- 1. Please tell me about yourself
 - Name
 - Position
 - Center where you learn
 - How were you recruited to Adult Literacy Education (ALE)?
 - For how long have you attended the Adult Literacy education (ALE)?
 - Why have you joined the ALE classes?

II. Goal of ALE

2. What do you think is the purpose of ALE/ Integrated Functional Adult Literacy Education?

III. ALE Learning and Empowerment

3. What are your perceptions about the ALE? Explain in detail.

Important issues to be addressed include views on:

- How the motivation of the learner was in relation to ALE
- What expectations they have had about the program
- The extent to which ALE satisfies the learner's needs and expectations. Why the learner feels it satisfies/ doesn't satisfy his/her needs and expectations
- What the learner has gained by joining the ALE
- Why the learner thinks all people who don't read, write and count well in his/her village can't join ALE
- What the learner thinks is missing from ALE or what he/she would like to be added to the learning materials
- What should be excluded from the ALE
- Whether or not there were dropouts and why he/she thinks people drop out of ALE, Women? Men? Boys? Girls?
- What learning materials he/she uses in adult literacy classes
- Explanation on how accessible the materials are
- How suitable the available learning materials to the learner's learning needs are.

- Explanation on the reasons for the suitability/ non-suitability.
- The class-size of where he/she learns; how many participants there are
- How he/she evaluates the facilities and equipment at his/her ALE center
- How one feels about the quantity and quality of facilitators at his/her center
- How appropriate one thinks are the facilitators' methods used in adult literacy classes? (Why he/she says so)
- What challenges, difficulties one encounters as a learner? (Personal? Domestic? Financial? Psychological? Administrative? Or any other?)
- 4. Would you please tell us in detail about your learning?

Important issues to be addressed include:

- Talking about what he/she has learned in his/her adult literacy class this year.
- What kinds of writing the learner does and how the learner participates into the decision about topics
- Specific examples of what kinds of reading the learner does
- Specific examples of what kind of basic numeracy the learner does
- Specific examples of what kind of basic life skills the learner does (or gains)
- Explanation on how one learns (inside and/or outside the class using self-study) Specific example of how one learns basic reading and writing Specific examples of how one learns basic numeracy (calculation) Specific examples of how one learns basic life skills
- What materials, for example books, the learner reads
- How the learner chooses the materials he/she reads
- If there is a time when the learner has a very different understanding of the materials his/her facilitator chooses—what the facilitator does about that and what the learner does about that
- How the facilitator keeps the track of the student's development as a reader, writer, and life skills learner
- If the learner is involved in the planning of the lessons? How he/she is involved
- If the facilitator teaches alone or requests other partners to assist him/her in ALE and if the learner can tell us for which lessons the facilitator made the invitation?
- 5. Would you please explain the knowledge, skills (basic skills, life skills, problem solving skills), self-confidence building matters you are gaining out of ALE?
- 6. Would you talk about your personal life relating to ALE and explain on how it has become part of your life: how do you feel about the application of knowledge and skills in your everyday lives?

Important issues to be addressed include:

- **Political awareness and participation** (voting age, parties found in the region/country, your participation in any of the parties)
- **Community participation** (participation in public institutions, community level decision making; community development activities, donation)
- Awareness of gender discrimination and inequalities (if he/she has ever heard of gender discrimination and inequality in his/her village)

- Awareness of legal protections for women and children (if she/he knows laws against violence, law mandating equal participation of women in local, national plans and incorporating women's needs in such plans)
- Family cohesion (quality of relationship; degree of closeness between members and among family members),
- **Importance of Children** (for financial support in old age, having a son, old age security, help while the child is young, love, closeness, parent naming, fun)
- **Decision making at home** (husband/ wife decides/ together/ for purchasing expensive things, having another child, using birth control, disciplining a child, whose opinion prevails in an important decision)
- **Self-concept** (How pleased you are with yourself, and with your appearance, how good you are as a husband/wife and as a mother/father; how knowledgeable you find yourself; how smart you find yourself)
- Self-efficacy (If you have ever bought items yourself in the market, taken a bus/car by yourself, done a transaction in a rural bank by yourself, paid bills yourself, crossed the bridge (to the other side of the village) by yourself
- 7. Could you please mention some of the income generating skills you learned in ALE?
- 8. Do you think that there is fair and equal access/opportunity to ALE for all people (men and women) in your village?
- 9. Have you experienced/ heard of an example of inequality of participation in class when you were ALE learner? If yes, could you explain what the situation was like?
- 10. Have you seen/heard of any kind of marginalization in your village (regional, ethnic, social class or gendered)? If you have seen/heard of it, please explain this in detail.
- 11. What do you think are a person's civil and legal rights? How do you apply these rights in your everyday life (in relation to others)? (the importance here is to know whether or not she/he respects them)

IV. SD

- 12. Explain to us what should be done to protect your environment from danger? (Important issues the participants raise may include: issues preventing environmental pollution, issues of protecting deforestation, terracing, afforestation, reforestation, using technologies saving fuel wood). To what extent were you aware of these before and after you joined ALE?
- 13. Explain to us what we need to do in order to keep the economy in your region/country developing? To what extent were you aware of these ways of keeping the economy developing before and after you joined adult literacy education?
- 14. How can we make the society continue pleasantly? The extent to which you are aware of this before and after your exposure to ALE?
- 15. How do you think can SD be achieved in Ethiopia? To what extent were you aware of this before you joined adult literacy education?
- 16. What do you think are the challenges and prospects of SD in Ethiopia? Explain in relation to your village, zone or region? How can we solve the problems related to sustainable

development? To what extent were you aware of these challenges, prospects and solutions before and after you joined ALE?

17. In which language is ALE offered and what were its benefits for you?

V. General Question

- 18. What are the major problems of the current ALE?
- 19. How do you think would the ALE provision be improved (Quality issues, improving planning and management, improving and increasing learning materials, popularizing the program, strengthening certification)

Thank you for your cooperation and participation in this interview.

2. Interview Protocol for IFALE Facilitators

Everyone can think differently about adult literacy education due to several reasons. As I would like to know about how you understand the implementation and contributions of adult literacy education to learner empowerment and SD, I will interview you for approximately one and half hours; I am going to take notes and audiotape the interview.

I. Personal Information

- 1. Tell me about yourself.
 - Name
 - Center where you facilitate
 - Level
 - Why did you become an adult literacy education (ALE) facilitator?
 - What is your educational and professional background?

II. Primary Goals of ALE

- 2. What do you think is the purpose of ALE in Ethiopia?
- 3. What do you most hope to have accomplished when your students finish their adult literacy program?
- 4. How can you tell when they have achieved this?

III. Facilitation and Learning

- 5. How do you evaluate your facilitation? Explain in detail. Important Issues to be considered about the facilitation include:
 - For how long he/she has worked as a facilitator
 - How do you think were the learners recruited to Adult Literacy Education (ALE)?
 - How he/she learned to facilitate
 - For how long the facilitator has been trained
 - If there are key people, and readings (or theories if any) that influence his/her facilitation; in what ways he/she has to be specific.
 - What instructional beliefs and practices, specific to literacy instruction, the facilitator holds
 - Explaining to a new adult literacy education facilitator the structure of the day, including the amount of time spent on different activities, role of the learners and his/her role.
 - Talking about his/her classroom facilitation this year.
 - How she/he knows what to teach
 - Giving specific examples on how he/she facilitates (teaches) basic reading, writing, numeracy and life skills
 - How she/he learns how to teach people literacy (reading and writing), calculation, and basic life skills
 - Specific examples on what kinds of reading her/his students do
 - What kinds of writing her/his students do; how she/he decides on topics and genres
 - Specific examples on what kind of basic numeracy her/his students do

- Specific examples on the kind of life skills her/his students learn
- Whether it is learning basic skills, life skills, socio-economic skills that is given more emphasis (and why it is so)
- What materials, for example books, the students read
- How the facilitator chooses the materials her/his students read
- If there is a time when a student has a very different understanding of the materials she/he chooses; what he/she does about that
- How she/he keeps the track of the students' development as readers, writers, life skills learners
- How the students are expected to learn inside and/or outside the class—using self-study
- How the students learn in your classroom; How she/he knows about the way the students learn
- How the facilitator sees the interest and motivation of learners
- How do you deal with learners requiring special support—visually impaired, handicapped, etc. ; What his/her thoughts on this are
- What limitations of self-knowledge/skill the facilitator recognizes
- What additional information would be helpful to inform her/his instructional practices (facilitation); how she/he would get that information
- Why the facilitator thinks all people who don't read, write and count well in his/her village can't join ALE
- Whether or not there were drop outs and why he/she thinks people drop out of ALE, Women? Men? Boys? Girls?
- What learning materials he/she uses in adult literacy classes
- Explanation on how accessible the materials are
- Do you teach alone or request other partners (agricultural extension, or health extension workers) to assist you in ALE?
- If yes, can you tell us for why and which lessons you invited them?

IV. Empowerment

- 6. Would you please explain the knowledge, skills (basic skills, life skills, problem solving skills), self-confidence building matters the learners are gaining out of ALE?
- 7. Would you talk about your personal life relating to ALE and explain on how it has become part of your life: how do you feel about the application of knowledge and skills in your everyday lives?

How he/she evaluates ALE in terms of enhancing learners':

- **Political awareness and participation** (voting age, parties found in the region/country, your participation in any of the parties)
- **Community participation** (participation in public institutions, community level decision making; community development activities, donation)
- Awareness of gender discrimination and inequalities (if he/she has ever heard of gender discrimination and inequality in his/her village)

- Awareness of legal protections for women and children (if she/he knows laws against violence, law mandating equal participation of women in local, national plans and incorporating women's needs in such plans)
- **Family cohesion** (quality of relationship; degree of closeness between members and among family members),
- Self-confidence and self-concept
- Decision making by oneself or together with the partner
- Income generation
- 8. Do you think that there is fair and equal access/opportunity to ALE for all people (men and women) in this village?
- 9. During classroom facilitation (discussion), do you encourage female learners to participate? If yes, why?
- 10. What do you do when one or more learners bring up controversial issues, or when there is ambiguity about the correct answer?

V. SD

- 11. What environmental protection issues have you dealt with in ALE classes?
- 12. How do you think are the learners in applying issues of environmental protection in their day today activities?
- 13. Do you also teach issues of social and economic sustainability in class? If yes, how are the learners in making them as part of their life?
- 14. In your view, what should be the role of ALE with regard to environmental or/ and socioeconomic sustainability?
- 15. In which language is ALE offered and what were its benefits for the learners?

VI. Assessment

16. How do you assess your students?

Important Issues to be considered include:

- What assessments of ALE are in use
- What assessments the facilitator personally uses
- How she/he uses the results

VII. General

17. What are the major problems associated with ALE?

18. How do you think would the ALE provision be improved (Quality issues, improving planning and management, improving and increasing learning materials, popularizing the program, strengthening certification)

Thank you for your cooperation and participation in this interview.

3. Interview Protocol for Wereda IFALE Officials and Supervisors

The purpose of this interview is to understand the contribution of adult literacy education for sustainable development. Everyone can have a different understanding of literacy education due to various reasons. As I would like to understand the implementation of adult literacy education and adult literacy education contributions to learner empowerment and sustainable development, I will interview you for approximately one and half hours. I will write notes and record the interview.

I. Personal Information

- 1. Please tell us about yourself
 - Name
 - Education level and field of specialization
 - Current position

II. Possible Assumptions of Adult Literacy Education (ALE)

- 2. What is the historical perspective of ALE in Tigray/Amhara Region in particular in Ethiopia in general?
- 3. What are the lessons learned from the past ALE programs? How differently is the current ALE designed?
- 4. What are the major possible assumptions of the current ALE?
- 5. What are the major founding principles of the current ALE?
- 6. What are its major objectives (purposes)?
- 7. What is the role of regional education bureaus in the design and implementation of ALE?

III. Relevance of the ALE Assumptions

- 8. What is the relationship between the designed literacy, numeracy and life skills on the one hand and the economic, social and environmental characteristics of the areas?
 - Specificity of the ALE program to regional, social, cultural, religion and livelihood variations
 - How the ALE knowledge and literacy skills match with local institutions
 - How the ALE is linked to life skills used at home, at work ?
- 9. What do you most hope to have accomplished when the learners finish their ALE? How can you tell when they have achieved this?
- 10. How do you evaluate the overall design of the regional ALE major manuals (Preparation of the Facilitators' manual, mobilization document, etc.)?

- Who participated in its design
- Who the collaborators involved are; their role in the design and implementation
- What your role as people responsible for ALE in the region.
- If the potential learners' needs, demands and priorities have been assessed before the design and have been considered in the design
- What notions of literacy events, practices and domains have been considered in the design of the program, if there were any?
- What the role of the various levels of the ALE Education system at various levels are

(MoE and its collaborators at national level, the Regional State, Bureau of Education, Wereda Education, etc.)?

IV. Taking Informed Decision on ALE

- 11. Do you think that there were any research undertakings done, for example baseline survey at national and regional levels to design the current ALE program? If so, what were the major/key findings identified and used as basis to design the ALE and to develop the curriculum and the ALE manuals, etc.?
- 12. Has there been a benchmark (role model country) when you design the ALE strategy or program?
- 13. What were the lessons considered from previous ALE programs in Ethiopia?
- 14. What were the bases when designing the ALE, preparing the National Adult Literacy Education, the ALE curriculum and facilitators' manuals, recruiting the facilitators, setting up the ALE training centers and institutional system to run the ALE?
- 15. Why IFALE, why not functional or critical or some other model?
- 16. How did you define the role of learners and facilitators, the duration of the program (2 years), and continuity of the program?

V. State of ALE

17. Please tell us about the current status of ALE at regional level.

- His/her vision and perception about the implementation of ALE plan (what works, what doesn't work)
- The extent to which the learners are gaining what they expected from the ALE provision. Explanation about what they have gained by joining ALE
- The reasons why illiterate (people who don't read, write and count well) in the villages don't join ALE
- Dropout rates and why people drop out of ALE—Women? Men? Boys? Girls?
- Academic background and training of facilitators
- What instructional beliefs and practices, specific to literacy instruction, exemplary adult literacy education facilitators are expected to hold; how he/she thinks the facilitators are in light of this.
- Appropriateness of the facilitators' approaches and methods used in adult literacy classes are
- Overall adequacy of facilitators (quantity and quality of facilitators) at national level
- Availability of learning materials for ALE
- Suitability of the available learning materials for the learners' needs
- Distance of ALE centers from the learners' homes
- Average class size at national level
- Availability of facilities and equipment in the ALE centers
- The extent to which ALE satisfies the learners' needs
- Learners' involvement in the planning of the lessons
- What challenges, difficulties the learners of ALE encounter as learners; Personal? Domestic? Financial? Psychological? Administrative? Or any other?

- How ALE is linked to lifelong learning (Post literacy plan)
- How the students are expected to learn inside and/or outside the class—using selfstudy

VI. Empowerment

18. How empowering is ALE to the learners?

If he/she could say something about ALE corresponding to its potential to enhancing learners':

- **political awareness and participation** (voting age, parties found in the region/country, your participation in any of the parties)
- **Community participation** (participation in public institutions, community level decision making; community development activities, donation)
- Awareness of gender discrimination and inequalities (if he/she has ever heard of gender discrimination and inequality in his/her village)
- Awareness of legal protections for women and children (if she/he knows laws against violence, law mandating equal participation of women in local, national plans and incorporating women's needs in such plans)
- **Family cohesion** (quality of relationship; degree of closeness between members and among family members),
- Self-confidence
- Decision making by oneself or together with ones the partner
- Income generation skills
- 19. What does the application of knowledge and skills to learners' everyday lives looks like?
- 20. How do you see ALE in the country is in terms of achieving its promises?
- 21. Would you please reflect on the extent to which the rural adult learners' competences satisfy the competence profile as defined by the ALE?
- 22. How is the intention and implementation ALE currently in terms of fairness, equality and equity?

- Detail explanation in relation to addressing issues of regional, ethnic, social class, or gendered marginalization
- If he/she thinks that what they are learning integrates issues of gender (women's role) or *marginalized people* into the curriculum
- If there is equality of ALE opportunity to marginalized groups
- What gender equity looks like in ALE provision
- Whether or not there is the opportunity to become autonomous through self-reflection and educators' facilitation,
- If there has been provision of equal good education by the state, with more resources to those with less e.g. the blind, to ensure the same quality,
- If ALE provision really responds to the daily realities of the poor (to their needs, practices, hopes and aspirations? Explaining in detail if it does; if not, how it can be made so.
- If the provision avoids stigmatizing those who have not had the opportunity to learn to read and write? Detail explanation as to how this can be made successful (If not, why not?)

• If ALE he/she feels can be made more relevant, accessible, equitable, extending to the very hardest to reach? Explain in detail how it can be so? (If not, why not?)

VII. SD

18. How do you evaluate ALE in terms of its focus on developing the learners' understanding and awareness of sustainability (sustained environment, society and economy) as well as application of their knowledge?

Important Issues to be considered include:

- Does ALE address imparting knowledge, skills, perspectives and values that encourage and support learner to lead sustainable social, environmental and economic lives?—if it does, what is the most convincing piece of evidence to support this conclusion?
- What environmental protection issues are you dealt with in ALE?
- How do you think are the learners in applying issues of environmental protection in their day today activities?
- Are issues of social and economic sustainability addressed in ALE? If yes, how are the learners in making them as part of their life?
- In your view, what should be the role of ALE with regard to environmental and socioeconomic sustainability?
- 19. To what extent do you think ALE is expanded to include critical-thinking skills, skills to organize and interpret data and information, skills to formulate questions, and the ability to analyze issues that confront communities?
- 20. In which languages is ALE offered and what do you think are the benefits to the learners?

VIII. General

- 21. Although Ethiopia's development policy agenda has for more than 20 years been Agricultural Development Led Industrialization (ADLI), the ALE as a well-articulated strategy and education sub-sector with in the ESDP has been a very recent phenomenon. *The young and adult farmers have until recently been denied of basic education, which might otherwise have speeded up more the achievement of ADLI*
- 22. Would you please tell us how the monitoring and evaluation of ALE is carried out?
- 23. What are the overall problems of ALE in Ethiopia?
- 24. How he/she thinks can the ALE be improved (Quality issues, improving program planning and management, improving and increasing learning materials, popularizing the program, strengthening certification)

Thank you for your cooperation and participation in this interview.

4. Interview Protocol for Regional IFALE Officials and Focal Persons

The purpose of this interview is to investigate the implementation and contributions of adult literacy education for learner empowerment and SD. Everyone can have a different understanding of literacy education due to various reasons. As I would like to know how you think about the state of adult literacy education and adult literacy education contributions to learner empowerment and SD, I will interview you for approximately one and half hours.I will take notes and record the interview.

I. Personal Information

- 1. Please tell us about yourself
 - Name
 - Education level and field of specialization
 - Current position

II. Possible Assumptions of Adult Literacy Education (ALE)

- 2. What is the historical perspective of ALE in Tigray/Amhara Region in particular in Ethiopia in general?
- 3. What are the lessons learned from the past ALE programs?
- 4. How differently is the current ALE designed?
- 5. What are the major possible assumptions of the current ALE?
- 6. What are the major founding principles of the current ALE?
- 7. What are its major objectives (purposes)?
- 8. What is the role of regional education bureaus in the design and implementation of ALE?

III. Relevance of the ALE Assumptions

- 9. What is the relationship between the designed literacy, numeracy and life skills on the one hand and the economic, social and environmental characteristics of the areas?
 - Specificity of the ALE program to regional, social, cultural, religion and livelihood variations
 - How the ALE knowledge and literacy skills match with local institutions
 - How the ALE is linked to life skills used at home, at work and in the....?
- 10. What do you most hope to have accomplished when the learners finish their ALE? How can you tell when they have achieved this?
- 11. How do you evaluate the overall design of the regional ALE major manuals (Preparation of the Facilitators' manual, mobilization document, etc.)?

- Who participated in its design
- Who the collaborators involved are; their role in the design and implementation
- What your role as people responsible for ALE in the region.
- If the potential learners' needs, demands and priorities have been assessed before the design and have been considered in the design
- What notions of literacy events, practices and domains have been considered in the design of the program, if there were any?

• What the role of the various levels of the ALE Education system at various levels are (MoE and its collaborators at national level, the Regional State, Bureau of Education, Wereda Education, etc.)?

IV. Taking Informed Decision on ALE

- 12. Do you think that there were any research undertakings done, for example baseline survey at national and regional levels to design the current ALE program? If so, what were the major/key findings identified and used as basis to design the ALE and to develop the curriculum and the ALE manuals, etc.?
- 13. Has there been a benchmark (role model country) when you design the ALE strategy or program?
- 14. What were the lessons considered from previous ALE programs in Ethiopia?
- 15. What were the bases when designing the ALE, preparing the National Adult Literacy Education, the ALE curriculum and facilitators' manuals, recruiting the facilitators, setting up the ALE training centers and institutional system to run the ALE?
- 16. Why IFALE, why not functional or critical or some other model?
- 17. How did you define the role of learners and facilitators, the duration of the program (2 years), and continuity of the program?

V. Status

18. Please tell us about the current status of ALE at regional level.

- His/her vision and perception about the implementation of ALE plan (what works, what doesn't work)
- The extent to which the learners are gaining what they expected from the ALE provision. Explanation about what they have gained by joining ALE
- The reasons why illiterate (people who don't read, write and count well) in the villages don't join ALE
- Dropout rates and why people drop out of ALE—Women? Men? Boys? Girls?
- Academic background and training of facilitators
- What instructional beliefs and practices, specific to literacy instruction, exemplary adult literacy education facilitators are expected to hold; how he/she thinks the facilitators are in light of this.
- Appropriateness of the facilitators' approaches and methods used in adult literacy classes are
- Overall adequacy of facilitators (quantity and quality of facilitators) at national level
- Availability of learning materials for ALE
- Suitability of the available learning materials for the learners' needs
- Distance of ALE centers from the learners' homes
- Average class size at national level
- Availability of facilities and equipment in the ALE centers
- The extent to which ALE satisfies the learners' needs
- Learners' involvement in the planning of the lessons

- What challenges, difficulties the learners of ALE encounter as learners; Personal? Domestic? Financial? Psychological? Administrative? Or any other?
- How ALE is linked to lifelong learning (Post literacy plan)
- How the students are expected to learn inside and/or outside the class—using self-study

VI. Empowerment

19. How empowering is ALE to the learners?

If he/she could say something about ALE corresponding to its potential to enhancing learners':

- **political awareness and participation** (voting age, parties found in the region/country, your participation in any of the parties)
- **Community participation** (participation in public institutions, community level decision making; community development activities, donation)
- Awareness of gender discrimination and inequalities (if he/she has ever heard of gender discrimination and inequality in his/her village)
- Awareness of legal protections for women and children (if she/he knows laws against violence, law mandating equal participation of women in local, national plans and incorporating women's needs in such plans)
- Family cohesion (quality of relationship; degree of closeness between members and among family members),
- Self-confidence
- Decision making by oneself or together with ones the partner
- Income generation skills
- 20. What does the application of knowledge and skills to learners' everyday lives looks like?
- 21. How do you see ALE in the country is in terms of achieving its promises?
- 22. Would you please reflect on the extent to which the rural adult learners' competences satisfy the competence profile as defined by the ALE?
- 23. How is the intention and implementation ALE currently in terms of fairness, equality and equity?

- Detail explanation in relation to addressing issues of regional, ethnic, social class, or gendered marginalization
- If he/she thinks that what they are learning integrates issues of gender (women's role) or *marginalized people* into the curriculum
- If there is equality of ALE opportunity to marginalized groups
- What gender equity looks like in ALE provision
- Whether or not there is the opportunity to become autonomous through self-reflection and educators' facilitation,
- If there has been provision of equal good education by the state, with more resources to those with less e.g. the blind, to ensure the same quality,
- If ALE provision really responds to the daily realities of the poor (to their needs, practices, hopes and aspirations? Explaining in detail if it does; if not, how it can be made so.

- If the provision avoids stigmatizing those who have not had the opportunity to learn to read and write? Detail explanation as to how this can be made successful (If not, why not?)
- If ALE he/she feels can be made more relevant, accessible, equitable, extending to the very hardest to reach? Explain in detail how it can be so? (If not, why not?)

VII. SD

24. How do you evaluate ALE in terms of its focus on developing the learners' understanding and awareness of sustainability (sustained environment, society and economy) as well as application of their knowledge?

Important Issues to be considered include:

- Does ALE address imparting knowledge, skills, perspectives and values that encourage and support learner to lead sustainable social, environmental and economic lives?—if it does, what is the most convincing piece of evidence to support this conclusion?
- What environmental protection issues are you dealt with in ALE?
- How do you think are the learners' in applying issues of environmental protection in their day today activities?
- Are issues of social and economic sustainability addressed in ALE? If yes, how are the learners in making them as part of their life?
- In your view, what should be the role of ALE with regard to environmental and socioeconomic sustainability?
- 25. To what extent do you think ALE is expanded to include critical-thinking skills, skills to organize and interpret data and information, skills to formulate questions, and the ability to analyze issues that confront communities?
- 26. In which languages is ALE offered and what do you think are the benefits to the learners?

VIII. General

- 27. Although Ethiopia's development policy agenda has for more than 20 years been Agricultural Development Led Industrialization (ADLI), the ALE as a well-articulated strategy and education sub-sector with in the ESDP has been a very recent phenomenon. *The young and adult farmers have until recently been denied of basic education, which might otherwise have speeded up more the achievement of ADLI*
- 28. Would you please tell us how the monitoring and evaluation of ALE is carried out?
- 29. What are the overall problems of ALE in Ethiopia?
- 30. How he/she thinks can the ALE be improved (Quality issues, improving program planning and management, improving and increasing learning materials, popularizing the program, strengthening certification)

Thank you for your cooperation and participation in this interview.

5. Interview Protocol for Policy Makers

The purpose of this interview is to investigate the implementation of IFALE and its contributions to empowerment and SD. Everyone can have a different understanding of literacy education due to various reasons. As I would like to know how you think about the implementation of IFALE and its contributions to learner empowerment and SD, I will interview you for approximately one and half hours; I am going to take notes and audiotape the interview.

I. Personal Information

- 1. Please tell us about yourself
 - Name
 - Education level and field of specialization
 - Current position

II. Possible Assumptions of Adult Literacy Education (ALE)

- 2. What is the historical perspective of ALE in Ethiopia?
- 3. What are the lessons learned from the past ALE programs?
- 4. What is the general objective of the National Adult Education Strategy (NAES)?
- 5. What are the strategic objectives of the NAES?
- 6. Would you please mention the strategic directions of the strategy?
- 7. How differently is the current ALE designed?
- 8. What is the focus of ALE in Ethiopia at this time?
- 9. What is your vision as far as ALE is concerned?
- 10. What are the major principles of the current ALE master plan?
- 11. What are its major objectives (purposes)?
- 12. How should the nature, delivery and relevance of ALE in Ethiopia be?

III. Contextual Validity of the ALE Assumptions (Relevance)

- 13. How are the ALE strategy and plan related to the other policies of the country?
- 14. What is the relationship between the designed literacy, numeracy and life skills on the one hand and the economic, social and environmental characteristics of the areas?
 - Specificity of the ALE program to regional, social, cultural, religion and livelihood variations
 - How the literacy skills are matching with local institutions
 - How the ALE is linked to life skills used at home, at work and in the....?
 - How empowering ALE to the learners (In terms of political awareness and participation, Community participation, awareness of gender discrimination and inequalities, awareness of legal protections for women and children, family cohesion, self-confidence, decision making by oneself or together with the partner, income generation)
 - How is it in relation to achieving fairness, equity and equality (from design up to implementation)?

• How is it in terms of raising environmental, social, and economic sustainability of the learners?

IV. Change in Action (Taking Informed Decision)

- 15. Were there any research undertakings done, for example baseline survey at national and regional levels to design the current ALE program? If so, what were the major/key findings identified and used as basis to design the ALE and to develop the curriculum and the ALE manuals, etc.?
- 16. Have you had a benchmark (role model country) when you design the ALE strategy or program?
- 17. What were the bases when designing the ALE, preparing the National Adult Literacy Education, the ALE curriculum and facilitators' manuals, recruiting the facilitators, setting up the ALE training centers and institutional system to run the ALE?
- 18. Why IFALE, why not functional or critical or some other model of literacy?
- 19. How did you define the role of learners and facilitators, the duration of the program (2 years), and continuity of the program?
- 20. How do you demarcate the role of the duties and responsibilities of the federal and regional bodies?
- 21. Although Ethiopia's development policy agenda has for more than 20 years been Agricultural Development Led Industrialization (ADLI), the ALE as a well-articulated strategy and education sub-sector with in the ESDP has been a very recent phenomenon. Don't you think that the young and adult farmers have until recently been denied of basic education, which might otherwise have speeded up more the achievement of ADLI?

V. Monitoring and Evaluation

- 22. What are the current expected outcomes of IFALE (as put in ESDP IV)?
- 23. Would you please mention the key outcome targets of IFALE?
- 24. Please tell us about the status of ALE (including, strengths, weaknesses, the way forward).
- 25. Would you please tell us how the monitoring and evaluation of ALE is carried out?

Thank you for your cooperation and participation in this interview.

6. Interview Protocol for Collaborators of Adult Literacy Education

The purpose of this interview is to investigate the implementation and contributions of adult literacy education for empowerment and SD. Everyone can have a different understanding of literacy education due to various reasons. As I would like to know how you think about the implementation of IFALE and contributions to learner empowerment and SD, I will interview you for approximately one and half hours; I am going to take notes and audiotape the interview.

I. Personal Information

- 1. Would you please tell me about yourself?
 - Name
 - Education level and field of specialization
 - What is your current position in the organization'

II. Primary Goals and the Design of Adult Literacy Education (ALE)

- 2. What do you think is ALE or Integrated Functional Adult Education (IFALE)? What does it integrate?
- 3. What is its purpose in the context of Ethiopia?

Important Issues to be considered include:

- How the interviewee thinks about its importance
- How the program is funded
- What the interviewee hopes the learners to have accomplished most when the they finish their adult literacy education; how he/she can tell when they have achieved this
- Explanation as to what extent the interviewee thinks the program is explicitly linked to lifelong learning
- 4. Do you think that you are playing your role properly as a collaborator? What are your contributions?
- 5. How do you evaluate the overall design of ALE?

- If he/she knows the collaborators who participated in the design ALE (*National Adult Literacy Strategy (2008), Learning for Life: Master Plan for Adult Education in Ethiopia (2010/11-2019/20), the Integrated Functional Adult Literacy Program as a sub-sector within the Education Sector Development Plan IV (2010/11-2014/15)*)
- Their role as a collaborator in the design and/ or implementation
- If the potential learners' needs, demands and priorities have been heard before the ALE program design and accommodated during the design of ALE
- If ALE is based on full assessment of the pre-existing ALE practices
- What lessons have been learned from them
- What notions of literacy events, practices and domains have been considered in the design of the program

III. Empowerment

6. Would you please explain what knowledge, skills (basic skills, life skills, problem solving skills), self-confidence building matters the learners are gaining out of ALE?

Important Issues to be considered include:

If the representative of the collaborating institution could say something about ALE corresponding to its potential to enhancing learners':

- **political awareness and participation** (voting age, parties found in the region/country, your participation in any of the parties)
- **Community participation** (participation in public institutions, community level decision making; community development activities, donation)
- Awareness of gender discrimination and inequalities (if he/she has ever heard of gender discrimination and inequality in his/her village)
- Awareness of legal protections for women and children (if she/he knows laws against violence, law mandating equal participation of women in local, national plans and incorporating women's needs in such plans)
- **Family cohesion** (quality of relationship; degree of closeness between members and among family members),
- Self-confidence
- Decision making by oneself or together with the partner
- Income generation
- 7. Would you explain on how ALE has become part of the learners' life: how do you feel about the application of knowledge and skills in their everyday lives?
- 8. As a collaborator, how do you evaluate the ALE in terms of fairness, equality, equity and provision of opportunities to the learners for participating in personal development, social and community life, and decision-making, considering the plan and implementation?
 - Do you think that there is fair and equal access/opportunity to ALE for all people (men and women) in the country?
 - How do you think is the classroom facilitation (discussion), in encouraging female learners to participate?

IV. SD

- 9. How do you evaluate ALE in terms of its focus on developing the learners' understanding and awareness of sustainability (sustained environment, society and economy) as well as application of their knowledge?
 - What environmental protection issues are you dealt with in ALE?
 - How do you think are the learners in applying issues of environmental protection in their day today activities?
 - Are issues of social and economic sustainability included in ALE? If yes, how are the learners in making them as part of their life?
 - In your view, what should be the role of ALE with regard to environmental and socioeconomic sustainability?

- 10. To what extent do you think ALE is expanded to include critical-thinking skills, skills to organize and interpret data and information, skills to formulate questions, and the ability to analyze issues that confront communities?
- 11. In which languages is ALE offered and what do you think are the benefits to the learners?

V. General

- 12. Although Ethiopia's development policy agenda has for more than 20 years been Agricultural Development Led Industrialization (ADLI), the ALE as a well-articulated strategy and education sub-sector with in the ESDP has been a very recent phenomenon. *However, the adult farmers have until recently been denied of basic education, which might otherwise have speeded up more the achievement of ADLI*
- 13. How is the monitoring and evaluation of ALE carried out?
- 14. How she/he evaluates the overall status of ALE in the country so far (How effective is it?
- 15. How has the role of collaborators been in the overall implementation of the program?
- 16. What do you think ought to be done to improve ALE in the country, specifically in relation to the role played by partners?

Thank you for your cooperation and participation in this interview.

B. IFALE Session Observation Protocol

I. Background

- Name of the ALE Center: _____ Woreda:: _____
 Region: _____
 Level of the observed adult literacy class:
- 2. Level of the observed adult incracy ef
- 3. Sex of the facilitator :
- 4. Date:_____
- 5. Title of lesson:

II. Put a tick on Yes or No options to indicate the criteria achieved (questions 7-12).

7. Objectives	Yes	No
7.1 Clear		
7.2 Appropriate		
7.3 Communicated to students		
7.4 Achieved		
Additional comments:		

8. ActivitiesYesNo8.1 Well organized8.2 Varied8.3 Active learning8.4 Student centered8.5 Well-paced8.6 Positive challenge and manageable

Additional comments:

9. Assessment	Yes	No
9.1 Appropriate		
9.2 Continuous		
9.3 Students aware of assessment		
9.4 Students received constructive feedback		

Additional comments:

Yes	No
	Yes

Additional comments:

11. Learners	Yes	No
11.1 Clearly interested in the lesson		
11.2 Participated actively (both male and female)		
11.3 Were treated as individuals		
11.4 Understood what was expected		
11.5 Were encouraged by the facilitator		
11.6 Enjoyed the lesson		

Additional comments:

Yes	No	If Yes, for how long?
	Yes	Yes No

Additional comments:

13. Approximate teacher talk time in percentage_____

- 14. Resources used:
- 15. General comment:
 - a. Very successful:
 - b. Successful:
 - c. Fairly successful: _____
 - d. Unsuccessful:_____
 - e. Very unsuccessful:_____

16. Does the center teach adults the essential skills targeted by the curriculum?

17. Do the adults relate to what is being learned and connect it with their own experience?

18. General reflections:

C. IFALE Center Observation Protocol

- 1. Name of center:
- 2. Woreda: _____

5. Enrollment of ALE learners by year:

Year	1 st			2 nd			
	Μ	F	Total	Μ	F	Μ	Total
2010/11 (2003							
Ethiopian Calendar,							
E.C.)							
2011/12 (2004 E.C.)							
2012/13 (2005 E.C.)							

6. Number of groups (Sections): 1st year:_____ 2nd year:_____

7. Number of drop outs by sex and year

	1 st		2 nd				
Year	Μ	F	Total	Μ	F	Μ	Total
20010/11 (2003 E.C.)							
20011/12 (2004 E.C)							
2012/13 (2005 E.C)							

8. Number of facilitators in the center by sex and qualification:

Certificate			Diploma		
М	F	Total	М	F	Total

Comments:

9. Number of facilitators by service years:

1 year	2 years	3 years	3 years and above

10. Condition of the center :

S No.	Item	Very good	Good	Fair	Poor	Very Poor	NA
1	Location						
2	Spaces						
3	Offices and facilities						
4	Water supply						
5	Fence						
6	Electric supply						
7	Latrine						
8	Facilities						
9	Setting						
10	Staff rooms						
Common	ta:						

Comments:

11. Physical condition of the learning places :

S No.	Item	Very Good	Good	Fair	Poor	Very Poor	NA
1	Chalkboard						
2	Chairs						
3	Benches						
4	Walls						
5	Roof						

Comments:

12. Condition of the reading post (if any):

Item	Very	Good	Fair	Poor	Very	NA
	Good				Poor	
Number of						
supplementary books						
Budget allocated on						
books						
Availability of qualified						
personnel						
	Numberofsupplementary booksBudgetallocatedonbooksAvailabilityofqualified	Kumberofsupplementary books	GoodNumberofsupplementary books	GoodNumberofsupplementary books	GoodGoodNumberofsupplementary books	GoodGoodPoorNumberofImage: Second sec

Comments:

Detail Evaluation:

D. Individual IFALE Participant Observation Protocol

Region:_____

Wereda:_____

Center:_____

Checklist for observing neo-literates in their homes:

- Personal hygiene
- House keeping
- Presence and use of latrines
- Preparation of compost

E. Survey for IFALE Participants and Non-**Participants**

As we would like to know about the contribution of adult literacy education to sustainable development in Ethiopia, we will survey with you for approximately one and half hours.

I. Personal Information

- 1. Name:_____
- 2. Wereda:_____
- 3. Zone:_____
- 4. Region:_____
- 5. Age: _____
- 6. Sex:
- 7. Marital Status (Put a tick ($\sqrt{}$) after one). Single: _____ Married: _____ Married: _____ Widow/Widower: _____

- 8. If you are married, number of years in marriage:
- 9. Number of children:
- 10. IFALE graduate: Non-graduate:

II.The Contribution of ALE

Participation in election and decision-making

- 11. How do you rate the importance of participation in voting? (circle one)
 - 4. Important 3. Moderately important 2. Not important 5. Very important 1. Totally unimportant
- 12. How do you rate yourself in terms of participation in voting? (circle one) 5. Very good 4. Good 3. Average 2. Poor 1. Very poor
- 13. To what extent would you be interested in becoming a local representative for an election?
 - 3. Average 5. Very high 4. High 2. Low 1. Very low
- 14. To what extent do you participate in decision-making in public institutions at any level? 5. Very high 4. High 3. Average 2. Low 1. Very low
- 15. If you participated what was your role?
- 16. Did you participate in local council election in 2013? A.Yes B. No
- 17. If you participated, in what capacity did you participate? put a tick $-\sqrt{}$ after the following alternatives

17.1 A candidate for a regional/wereda/kebele general assembly:

- 17.2 An observer: _____
- 17.3 A voter:
- 17.4 An election facilitator/officer:
- 17.5 If other, please specify:
- 18. What is the minimum voting age in Ethiopia?
- 19. Can you name the political party/parties you know in the region/country? Please mention them.
- 20. Do you think you could become a local representative in an election at any level? B. No A. Yes
- 21. Is there any law on electing representatives (men/women)? A. Yes B. No

- 22. If yes, can you tell us about it?
- 23. Where does your knowledge and practices of political participation as well as decisionmaking of public organizations come from? It is possible to put a tick ($\sqrt{}$) more than once.
 - 23.1 IFALE:_
 - 23.2 Local leaders (wereda/kebele): _____
 - 23.3 Friends:
 - 23.4 Local media:
 - 23.5 If others, please mention:

Community participation

- 24. Are you a member of a community organization? A. Yes B. No
- 25. If yes, to which group do you belong?
- 26. If yes, at what level did you participate?
- 27. Do you know the community/village development committees in your area? A. Yes B. No
- 28. If yes, could you mention some of the community development committees in your area?
- 29. Did you ever participate in community development activities in the last two years? A. Yes B. No

30. If yes, which?

31. If yes, what was your level of participation?

- 32. If yes, what is/are the source of your knowledge and practices of involvement in community organizations as well as community development committees and their activities? It is possible to put a tick ($\sqrt{}$) more than once.
 - 32.1 IFALE:___
 - 32.2 Agricultural extension workers:
 - 32.3 Friends:
 - 32.4 Local leaders (wereda, or kebele):
 - 32.5 Media:
 - 32.6 If others, please mention:

55. Mg	Use the following scale for Use the following scale for											
				llowir	ng so	ale for			llowii	ng sc	ale for	
		column		column 2								
		5. v		5. very good								
		4. i		4. good								
			ortant	3. average								
		2. n		2. poor								
			ant	1. very poor								
			oortance	How would you rate yourself in								
		of each of the following					terms of using each of the					
		agricultural technologies for					following agricultural					
		increasing your production?					technologies for increasing your					
	Item	(circle one)					production? (circle one)					
33.1	Improved agricultural tools											
	(such as ploughs, a tool for											
	sowing crops by line)	5	4	3	2	1	5	4	3	2	1	
33.2	Improved seeds	5	4	3	2	1	5	4	3	2	1	
33.3	Fertilizers	5	4	3	2	1	5	4	3	2	1	
33.4	Compost	5	4	3	2	1	5	4	3	2	1	
33.5	Manure	5	4	3	2	1	5	4	3	2	1	
33.6	Pesticides	5	4	3	2	1	5	4	3	2	1	
33.7	Herbicides	5	4	3	2	1	5	4	3	2	1	
33.8	Sowing seeds by lines	5	4	3	2	1	5	4	3	2	1	
33.9	Well-timed gathering of crops	5	4	3	2	1	5	4	3	2	1	
33.10	Drip-irrigation	5	4	3	2	1	5	4	3	2	1	
33.11	Having access to water	5	4	3	2	1	5	4	3	2	1	
	sources (river water, dam-											
	water, pond-water)											
33.12	Motor pump	5	4	3	2	1	5	4	3	2	1	

33 Agricultural technologies for increasing agricultural production

34. From whom did you learn about agricultural technologies for increasing production? It is possible to put a tick ($\sqrt{}$) more than once.

34.1 IFALE:

- 34.2 Farmer Training Center (FTC):_____ 34.3 Agricultural extension workers:_____
- 34.4 Health extension workers:
- 34.5 Local leaders:

34.6 Media: ______ 34.7 If others, please mention: _____

35. Environmental protection measures

		Use the following scale for column 1 5. very important 4. important 3.moderately important 2. not important 1.totally unimportant How do you rate the importance of the following measures of Environmental					Use the following scale for column 2 5. very good 4. good 3. average 2. poor 1. very poor How would you <i>rate yourself</i> in terms of participating in the following measures of environmental						
25.1	Items	protection? (circle one)				protection? (circle one) 5 4 3 2 1							
35.1 35.2	Contour-plowing	5	4	3	2 2	1 1	5	4	3	2 2	1		
35.2	Terracing Stone bunds	5	4	3	$\frac{2}{2}$	1	5	4	3	2	1		
35.4	Constructing check dams	5	4	3	2	1	5	4	3	2	1		
35.5	Constructing gabion	5	4	3	2	1	5	4	3	2	1		
35.6	Daget system	5	4	3	2	1	5	4	3	2	1		
35.7	Sediment trap on the riverside	5	4	3	2	1	5	4	3	2	1		
35.8	Planting trees	5	4	3	2	1	5	4	3	2	1		
35.9	Using fuel-wood saving stove at home	5	4	3	2	1	5	4	3	2	1		
35.10	Using solar/wind energy	5	4	3	2	1	5	4	3	2	1		
35.11	Preparing latrines	5	4	3	2	1	5	4	3	2	1		
35.12	Utilizing latrines	5	4	3	2	1	5	4	3	2	1		
35.13	Preparing liquid and solid waste-disposal techniques	5	4	3	2	1	5	4	3	2	1		
35.14	Utilizing liquid and solid waste-disposal techniques	5	4	3	2	1	5	4	3	2	1		
35.15	Securing enclosures from domestic animal and human intrusion	5	4	3	2	1	5	4	3	2	1		
35.16	Wise use of fauna or flora in our surrounding	5	4	3	2	1	5	4	3	2	1		

36. From whom did you learn about the measures of environmental protection? It is possible to put a tick ($\sqrt{}$) more than once.

36.1 IFALE: _____

36.2 Farmer Training Center (FTC):_____

36.3 Agricultural extension workers:

36.4 Health extension workers: _____

36.5 Local leaders: _____

36.6 Media:

36.7 If others, please mention:

Saving

37. How would you rate the importance of savings?

5. Very important 4. Important 3. Moderately important 2. Not important

1. Totally unimportant

38. How would you rate yourself in terms of savings?

5. Very good 4. Good 3. Average 2. Poor 1. Very poor
Access to credit and additional income generation schemes

- 39. Do you have access to credit for additional income generation in your community?A. YesB. No
- 40. If yes, what are the sources of credit? It is possible to put more than one tick ($\sqrt{}$).

36.1 A rural credit and saving bank: _____

- 36.2 Women's Association:
- 36.3 NGOs: _____
- 36.4 Local money lenders: _____
- 36.5 Others, Please specify: ____
- 41. Do you have any means of additional income generation? A. Yes B. No If yes, which income generation scheme do you use? It is possible to put more than one tick ($\sqrt{}$).
 - 41.1 Vegetable growing:
 - 41.2 Small business:
 - 41.3 Handicraft:_____
 - 41.4 Poultry:_____
 - 41.5 Dairy products:_____
 - 41.6 Fattening (oxen) for sale:
 - 41.7 Bee-keeping:_
 - 41.8 Supplying spices:
 - 41.9 Working as a day laborer (out of the farming season):_____
 - 41.10 Others, please specify:

Financial contribution

- 42. Did you contribute money for a mutual support and/or productive investment in the last two years (2011/12, 2012/13)?
 - A. Yes B. No
- 43. If yes, which?
- 44. If yes, who made you aware of or encouraged you to contribute money for a mutual support and/or productive investment? It is possible to put a tick ($\sqrt{}$) more than once. 44.1 IFALE:
 - 44.2 Agricultural extension workers:
 - 44.3 Health extension workers:
 - 44.4 Friends:_
 - 44.5 Local leaders (wereda or kebele):
 - 44.6 Media: _
 - 44.7 If others, please mention:

45. Gender equality in household decision-making

	Who is the main decision maker in your		ecision maker		
	family the following instances?	Husband	Wife	Together	Others
	Put a tick ($$).				
45.1	Buying expensive things				
45.2	Selling expensive things				
45.3	Having another child				
45.4	Using birth control				
45.5	Disciplining the child				
45.6	Arranging the marriage for a daughter/son				
45.7	Purchasing agricultural inputs				
45.8	Deciding what crops to sow				

- 46. Who has informed you of the need for equal participation of husband and wife in decision-making? It is possible to put a tick ($\sqrt{}$) more than once.
 - 46.1 IFALE:____
 - 46.2 Agricultural extension workers:
 - 46.3 Local leaders:
 - 46.4 Media:
 - 46.5 If others, please mention:

Awareness of gender equality for development

- 47. Do you think that males and females can contribute equally to development?
 - B. No
- A. Yes 48. If no, why?_
- 49. Who has made you aware of gender equality in development? It is possible to put a tick $(\sqrt{})$ more than once.
 - 49.1 IFALE:
 - 49.2 Agricultural extension workers: _____
 - 49.3 Women's association:
 - 49.4 Friends:
 - 49.5 Local leaders (wereda, or kebele):
 - 49.6 Media:
 - 49.7 If others, please mention:

Knowledge of laws against gender discrimination

- 50. Do you know whether there are laws against gender discrimination?
 - Yes B. No
- 51. If yes, please mention them.
- 52. Do you make use of it?
 - Yes B. No
- 53. If yes, tell us how.
- 54. Who has informed you about the laws against gender discrimination? It is possible to put a tick ($\sqrt{}$) more than once.
 - 54.1 FALE:
 - 54.2 Women's association:
 - 54.3 Friends:
 - 54.4 Local leaders (wereda, or kebele):
 - 54.5 Media:
 - 54.6 If others, please mention:

Awareness of legal protection for women and children

- 55. Do you know of laws protecting women and children?
- 56. If yes, please mention the laws?
- 57. From whom do you know about laws concerning legal protections of women and children? It is possible to put a tick ($\sqrt{}$) more than once.

B. No

57.1 IFALE:__

Yes

- 57.2 Agricultural extension workers:
- 57.3 Women's association:
- 57.4 Friends:
- 57.5 Local leaders (wereda, or kebele): _____
- 57.6 Media: _
- 57.7 If others, please mention:

	Item	colum 5. very 4. imp 3. mod 2. not 1. tota How co of eac	in 1 y importation deration impo illy u do yc ch of	oortan nt ely in ortant nimp ou rat f the	nt nport t ortan e the follo		colu 5. v 4. h 3. a 2. lo 1. v Hov <i>fam.</i>	imn 2 ery hi igh verag ow ery lo v wou	i gh ge wuuld y	ou <i>rc</i> s in 1	cale for <i>ute your</i> terms of ne)
58.1	Mother and father /husband and wife doing things together	5	4	3	2	1	5	4	3	2	1
58.2	Closeness between mother and child	5	4	3	2	1	5	4	3	2	1
58.3	The time a mother spends with her child (other than meal times)	5	4	3	2	1	5	4	3	2	1
58.4	Closeness between father and child	5	4	3	2	1	5	4	3	2	1
58.5	The time a father spends with his child	5	4	3	2	1	5	4	3	2	1
58.6	Closeness between siblings in a family (between/ among brothers/ sisters or between a brother and a sister or among brothers and sisters)	5	4	3	2	1	5	4	3	2	1
58.7	Closeness among all members of a nuclear family	5	4	3	2	1	5	4	3	2	1
58.8	The quality of relationship among all the members of a nuclear family	5	4	3	2	1	5	4	3	2	1
58.9	Closeness among the members of extended family members	5	4	3	2	1	5	4	3	2	1
58.10	The quality of relationship among extended family members	5	4	3	2	1	5	4	3	2 2	1

59. From whom do you know about the importance of closeness between/among family members? It is possible to put a tick ($\sqrt{}$) more than once.

59.1 IFALE:_____ 59.2 Health extension workers:_____

59.3 Local culture:

59.4 Media: ______ 59.5 If others, please mention: _____

60. Traditional practices

		for 5. v 4. in 3.m 2. n	colu very i mpor oder ot in	mn 1 mpor tant ately nport	tant impo	g scale rtant ant	Use the following sc for column 2 5. always 4. often 3. sometimes 2. rarely 1. not at all							
		imp <i>foll</i>	importance of each of the						How often have you undertaken/ facilitated each of the following					
	Items		cle o						es? (ci					
60.1	Break-up with a bride who has lost her virginity	5	4	3	2	1	5	4	3	2	1			
60.2	Underage marriage/Arranged marriage	5	4	3	2	1	5	4	3	2	1			
60.3	Considerable age difference in marriage (old men with young women)	5	4	3	2	1	5	4	3	2	1			
60.4	Widow inheritance or levirate marriage	5	4	3	2	1	5	4	3	2	1			
60.5	Sexual harassment common in rural areas specifically unwanted deliberate touching, leaning over, cornering, or pinching, unwanted sexual looks or gestures)	5	4	3	2	1	5	4	3	2	1			
60.6	Genital mutilation	5	4	3	2	1	5	4	3	2	1			
60.7	Rape	5	4	3	2	1					_			
60.8	Uvulectomy	5	4	3	2	1	5	4	3	2	1			
60.9	Milk teeth extraction	5	4	3	2	1	5	4	3	2	1			
60.10	Marriage by abduction	5	4	3	2	1	5	4	3	2	1			
60.11	Engaging children in hard work	5	4	3	2	1	5	4	3	2	1			
60.12	Physical abuse for children	5	4	3	2	1	5	4	3	2	1			
60.13	Sexual abuse for children	5	4	3	2	1	5	4	3	2	1			
60.14	High spending on food and drink in feasts (marriage, <i>teskar</i> , graduation ceremony, etc.)	5	4	3	2	1	5	4	3	2	1			
60.15	Performing tasks that are normally done by a blacksmith/a weaver/a potter	5	4	3	2	1	5	4	3	2	1			
60.16	Increasing the number of holidays	5	4	3	2	1	5	4	3	2	1			
60.17	Scratching one's face, uprooting one's hair, and beating one's chest after the death of one's beloved one	5	4	3	2	1	5	4	3	2	1			
60.18	Visiting the local magicians to tell you about you	5	4	3	2	1	5	4	3	2	1			

61. Who has been important to make you aware of the disadvantages of practicing these traditional practices? It is possible to put a tick ($\sqrt{}$) more than once.

- 61.1 IFALE:
- 61.2 Agricultural extension workers:
- 61.3 Health extension workers:
- 61.4 Local leaders: _____
- 61.5 Media:

 61.6 If others, please mention:

02. V	alue of children/wanting an	addii	10112	ai cn	na									
		Use	the	follov	ving	scale for	Use	the	follov	ving	scale	for		
		colun	ın 1				colun	ın 2						
		5. ver	y im	porta	nt		5. very high							
		4. imj	porta	nt			4. high							
		3.moo	lerat	ely in	nport	ant	3. average							
		2. not	imp	ortan	t		2. low							
		1. tota	ally u	ınimp	ortar	nt	1. vei	ry lov	W					
		How	do	y yo	ou i	rate the	How would you rate yourself i							
		impor	tance	e of	each	n of the								
		follov	ving	reaso	ns fo	r wanting	reason	ıs	for	war	ting	an		
	Item	an ado	litior	nal chi	ild? (c	ircle one)) additional child? (circle one)							
62.1	Financial support in old age	5	4	3	2	1	5	4	3	2	1			
62.2	Old age security	5	4	3	2	1	5	4	3	2	1			
62.3	Get help when the child is young	5	4	3	2	1	5	4	3	2	1			
62.4	Love	5	4	3	2	1	5	4	3	2	1			
62.5	Closeness	5	4	3	2	1	5	4	3	2	1			
62.6	Parental name (continuing the	5	4	3	2	1	5	4	3	2	1			
	generation)													
62.7	Fun	5	4	3	2	1	5	4	3	2	1			

62. Value of children/wanting an additional child

63. Who is the source of your knowledge of the value of children? It is possible to put a tick

$(\sqrt{})$ more than once.

- 63.1 IFALE:_____
- 63.2 Health extension workers:
- 63.3 Local culture:
- 63.4 Local leaders:
- 63.5 Parents: _____
- 63.6 Media: _____
- 63.7 If others, please mention:

64. Family planning

01.	ranny piann														
		Use the	follov	ving s	scale f	for colu	mn 1	Use the following scale for column 2							
		5. very	impor	tant				5. very good							
		4. impo	rtant					4. good							
		3.mode	rately	impo	ortant			3. average							
		2. not i	mport	ant				2. poor							
		1.totall	y unin	iport	ant			1. very poor							
		How d	o you	rate	the <i>i</i>	importai	nce of	How would you rate yourself in terms of							
		each of	the fo	llowin	ig med	asures?	(circle	making use of each of the following							
	Item	one)						measures of family planning? (circle one)							
64.1	Birth spacing	5	4	3	2	1		5 4 3 2 1							
64.2	Contraceptions	5	4	3	2	1		5 4 3 2 1							

65. Personal hygiene

	-~														
		Use	the	follo	wing	scale for	Use	the	follow	ving	scale	for			
		colur	nn 1				column 2								
		5. vei	ry im	porta	nt		5. ve	ry goo	d						
		4. im	porta	ant			4. go	od							
		3. mo	odera	tely i	mpor	tant	3. av	erage							
		2. no	t imp	ortar	nt		2. po	or							
		1.tota	ally u	nimp	ortan	t	1. ve	ry poo	r						
		How	do yo	ou rat	e the i	importance	How	would	d you	rate	yourse	<i>lf</i> in			
		of e	each	of	the	following	terms	s of n	naking	use	of eacl	h of			
		meast	ures	for pe	ersona	l hygiene?	the	follo	wing	me	asures	of			
	Item	(circl	le one	e)			perso	onal hy	giene	? (cir	cle one))			
65.1	Bathing oneself regularly	5	4	3	2	1	5	4	3	2	1				
65.2	Cleaning one's teeth	5	4	3	2	1	5	4	3	2	1				
65.3	Cutting one's nails	5	4	3	2	1	5	4	3	2	1				
65.4	Hygiene for young children	5	4	3	2	1	5	4	3	2	1				

66. Maternal and child care

	Item	column 5. very 4. impo 3.mod 2. not i 1.total How d of each for per	n 1 y imp ortar erate impo ly un lo yo h of t rsona	ortan It It im Irtan impo u rat he fo I hy	nt porta t ortant e the llowin giene		Use the following scale for column 2 5. very good 4. good 3. average 2. poor 1. very poor How would you <i>rate yourself</i> in terms of making use of each of the following measures of maternal and child health? (circle one)
66.1	Antenatal care (caring of mother and baby during pregnancy)	5	4	3	2	1	5 4 3 2 1
66.2	Postnatal care (caring for mother and baby after birth)	5	4	3	2	1	4 4 3 2 1

67. Who has been important to make you aware of these measures for personal hygiene, family planning as well as maternal and child care? It is possible to put a tick ($\sqrt{}$) more than once.

67.1 IFALE:

- 67.2 Agricultural extension workers:
- 67.3 Health extension workers:
- 67.4 Local leaders:
- 67.5 Media:
- 67.6 If others, please mention:

68. HIV-AIDS control

					Use the column	Use colui			llowir	ng so	ale for						
					5.	. very	' impo	ortan	t	5.							
							ortan			4. good							
						3.moderately						erage					
					important						2. poor						
							impoı			1.	ver	y po	or				
					1	.total	ly uni	impo	rtant								
							-		e the			-			<i>urself</i> in		
									of the						of the		
									f HIV-	follo	wing	g m	leasur	es o	f HIV-		
		Iter	m		AIDS	contr	ol? (c	ircle	one)	AIDS	s co	ntrol	l? (cir	cle or	ne)		
68.1	Abstinence				5	4	3	2	1		5	4	3	2	1		
68.2	Condoms				5	4	3	2	1		5	4	3	2	1		
68.3	Sterilization	of	sharp	objects	5	4	3	2	1		5	4	3	2	1		
	before use																
68.4	Faithfulness				5	4	3	2	1		5	4	3	2	1		

69. From whom do you know about the importance and practice of the measures of HIV-AIDS control? It is possible to put a tick ($\sqrt{}$) more than once.

- 69.1 IFALE:
- 69.2 Agricultural extension workers:
- 69.3 Health extension workers:
- 69.4 Local leaders:
- 69.5 Media: ______ 69.6 If others, please mention: ______

70. Social justice activities

	Item	scale for column 1 5. very important 4. important 3.moderately important 2. not important 1. totally unimportant How do you rate the					colum 5. ver 4. goo 3. ave 2. poo 1. ver How	n 2 y goo od rage or y poo wo <i>elf</i> in	od or ould terms	you s of e	lle for rate ach of ?
70.1	Marriage arrangement with other social classes (smiths, potters, weavers) in your community	5	4	3	2	1	5	4	3	2	1
70.2	Fair treatment of all people irrespective of differences in religion in all spheres of life	5	4	3	2	1	5	4	3	2	1
70.3	Encouraging women or girls to participate fully in all spheres of life	5	4	3	2	1	5	4	3	2	1
70.4	A male preparing food in the house (baking <i>injera</i> , preparing sauce, etc.)	5	4	3	2	1	5	4	3	2	1
70.5	A male doing household chores	5	4	3	2	1	5	4	3	2	1
70.6	A male fetching water from a nearby source	5	4	3	2	1	5	4	3	2	1
70.7	A female plowing the farm land using oxen	5	4	3	2	1	5	4	3	2	1
70.8	A female participating in resolving conflicts (both private and public)	5	4	3	2	1	5	4	3	2	1
70.9	Active engagement in social activism or social change in the community (for example playing a leading role in using modern agricultural practices or overcoming issues such as underage marriage, genital mutilation, rape, abduction and human trafficking)	5	4	3	2	1	5	4	3	2	1
70.10	Supporting the physically handicapped	5	4	3	2	1	5	4	3	2	1
70.11	Taking care of the elderly	5	4	3	2	1	5	4	3	2	1
70.12	Helping the needy by providing moral/financial/ material support	5	4	3	2	1	5	4	3	2	1

- 71. Who has informed you about the importance of social justice (equality, equity and fairness)? It is possible to put a tick ($\sqrt{}$) more than once.

 - 71.1 IFALE: _____ 71.2 Agricultural extension workers: _____
 - 71.3 Health extension workers:
 - 71.4 Local leaders:

 - 71.5 Media: _____ 71.6 If others, please mention: _____

72. Democratic culture activities

		for 6 5. v 4. in 3.m 2. n 1. to How	colum ery in mpor odera ot im otally	nn 1 nport tant ately porta unin you	impor	rtant ant	colum 5. 4. 3. 2. 1. How	nn 2 very g good avera poor very g wo	good ge ge poor uld	you	le for rate ach of
	Item				ities?	the	the fol				
72.1	Distinguishing facts from opinions	5	4	3	2	1	5	4	3	2	1
72.2	Using participatory decision making	5	4	3	2	1	5	4	3	2	1
72.3	Respect for other individuals	5	4	3	2	1	5	4	3	2	1
72.4	Willingness to listen, even when confronted with a view you disagree with	5	4	3	2	1	5	4	3	2	1
72.5	Willingness to reflect on one's actions, recognize mistakes, and learn from them	5	4	3	2	1	5	4	3	2	1
72.6	Communication at eye level (as opposed to talking up or down to someone)—regardless of age	5	4	3	2	1	5	4	3	2	1
72.7	Respect for difference in perceptions, beliefs about an idea	5	4	3	2	1	5	4	3	2	1
72.8	Respect for diversity due to ethnicity, language, religious beliefs, etc.	5	4	3	2	1	5	4	3	2	1
72.9	Respecting everyone's civil and legal rights	5	4	3	2	1	5	4	3	2	1

73. Who has informed you about the importance of democratic culture? It is possible to put a tick ($\sqrt{}$) more than once.

73.1 IFÁLE: ____

- 73.2 Agricultural extension workers:
- 73.3 Health extension workers:
- 73.4 Local leaders:
- 73.5 Media:
- 73.6 Community policing:
- 73.6 Community policing:

 73.7 If others, please mention:

74. Ways of conflict resolution

		colum 5. 4. 3. 2. 1. How d	Use the following scale for column 1 5. very important 4. important 3.moderately important 2. not important 1. totally unimportant How do you rate the						ood e oor you <i>rc</i>	ate yo	ale for urself in			
	Item	<i>import</i> activiti		of the	follow	wing	terms of each of the following activities?							
74.1	Resolving controversial issues through discussion	5	4	3	2	1	5	4	3	2	1			
74.2	Resolving conflicts through9negotiations	5	4	3	2	1	5	4	3	2	1			
74.3	Resolving conflicts through apology	5	4	3	2	1	5	4	3	2	1			
74.4	Resolving conflicts through legal means	5	4	3	2	1	5	4	3	2	1			

- 75. Who has informed you about the importance of the measures of conflict resolution? It is possible to put a tick ($\sqrt{}$) more than once.

 - 75.1 IFALE: ______ 75.2 Agricultural extension workers: ______
 - 75.3 Health extension workers:
 - 75.4 Local leaders:
 - 75.5 Media: _____
 - 75.6 If others, please mention:

76. Self-concept

		Use the following scale for column 1 5. very important 4. important 3.moderately important 2. not important 1. totally unimportant How do you rate the <i>importance</i> of each of the following measures of self-perception?				Use the following scale for column 2 5. very good 4. good 3. average 2. poor 1. very poor How would you <i>rate yourself</i> <i>in</i> terms of each of the following measures of self-				<i>yourself</i> of the of self-	
	Item	(cir	(circle one)				perception? (circle one)				
76.1	Being pleased with oneself	5	4	3	2	1	5	4	3	2	1
76.2	Being a good husband/wife	5	4	3	2	1	5	4	3	2	1
76.3	Being pleased with one's appearance	5	4	3	2	1	5	4	3	2	1
76.4	Being a good mother/father	5	4	3	2	1	5	4	3	2	1
76.5	Being a good son/daughter	5	4	3	2	1	5	4	3	2	1
76.6	Finding oneself to be knowledgeable	5	4	3	2	1	5	4	3	2	1
76.7	Finding oneself to be skillful	5	4	3	2	1	5	4	3	2	1
76.8	Finding oneself to be successful in life	5	4	3	2	1	5	4	3	2	1
76.9	Finding yourself important contributor to the community	5	4	3	2	1	5	4	3	2	1

- 77. Who has been important in developing your self-concept. It is possible to put a tick ($\sqrt{}$) more than once.
 - 77.1 IFALE: _
 - 77.2 Agricultural extension workers:
 - 77.3 Health extension workers:
 - 77.4 Local leaders: _____
 - 77.5 Local culture:
 - 77.6 Parents:
 - 77.7 Media: _____
 - 77.8 If others, please mention:

78. Self-efficacy

		Use the following scale for column 1 5. very important 4. important 3.moderately important 2. not important 1. totally unimportant How would you rate the importance of doing each of the following oneself? (circle					Use the following scale for column 2 5. very good 4. good 3. average 2. poor 1. very poor How do you rate yourself in terms of carrying out each of the following activities yourself?				
	Items	one)					(circle one)				
78.1	Buying items in the market	5	4	3	2	1	5	4	3	2	1
78.2	Going to church/mosque	5	4	3	2	1	5	4	3	2	1
78.3	Going to clinic/ hospital	5	4	3	2	1	5	4	3	2	1
78.4	Buying food in a restaurant	5	4	3	2	1	5	4	3	2	1
78.5	Taking the bus/taxi/bajaj	5	4	3	2	1	5	4	3	2	1
78.6	Making transactions in a (rural) bank	5	4	3	2	1	5	4	3	2	1
78.7	Paying bills	5	4	3	2	1	5	4	3	2	1
78.8	Crossing the bridge (to the other side of the village)	5	4	3	2	1	5	4	3	2	1

79. Who helped develop your self-efficacy? It is possible to put a tick ($\sqrt{}$) more than once. 79.1 IFALE:

- 79.2 Agricultural extension workers:
- 79.3 Health extension workers:
- 79.4 Local leaders:
- 79.5 Media:
- 79.6 If others, please mention:
- 80. In your opinion, are more resources provided for those who are in greatest need? A. Yes B. No
- 81. If yes, could you provide an example?

02. I III	e effectiveness of IFALE (Only for IFALE)								
		Use the following scale for column 2 5. very good							
		4. good							
		3. average							
		2. poor							
		1. very poor							
		How would you rate IFALE in terms of							
		addressing each of the following issues?							
	Items	(circle one)							
82.1	Having relevance to the daily realities and needs	5 4 3 2 1							
	of the community								
82.2	Having access to people living in remote areas	5 4 3 2 1							
82.4	Giving equal opportunity to women to participate								
	in class	5 4 3 2 1							
82.5	Focusing on knowledge that encourages and								
	supports citizens to lead sustainable lives	5 4 3 2 1							
82.6	Focusing on skills that encourage and support								
	citizens to lead sustainable lives	5 4 3 2 1							
82.7	Focusing on attitudes that encourage and support								
	citizens to lead sustainable lives	5 4 3 2 1							
82.8	Focusing on values that encourage and support								
	citizens to lead sustainable lives	5 4 3 2 1							
	entitents to read sustainable inves	5 1 5 2 1							

82. The effectiveness of IFALE (Only for IFALE participants)

- 83. How would you agree that facilitators help to improve conditions in your community?5. Strongly agree 4. Agree 3. Moderately agree 2. Disagree 1. Strongly disagree
- 84. Which of the following do you consider as the five most important solutions to improve IFALE?
 - 84.1 Allocating adequate budget:
 - 84.2 Providing more training to facilitators:
 - 84.3 Improving the commitment of the leaders of the sector at various levels:
 - 84.4 Strengthening integration among stakeholders (in the agriculture, health sectors etc.): _____
 - 84.5 Solving attitudinal problem among learners ('I am too old to learn'; 'Learning at this age is a waste of time'): _____
 - 84.6 Increasing the participation of women:
 - 84.7 Supplying primers to the learners:
 - 84.8 Providing learning facilities to the learners:
 - 84.9 Decreasing the number of the meetings:
 - 84.10 Conducting classes in convenient times rather than after the tiring water and soil conservation chore (safety-net chore): _____
 - 84.11 Improving learning conditions (e.g. under the green wood trees without seats):_____
 - 84.12 Others (if any):

Thank you for your cooperation and participation in this survey!

F. Research Permit

NORWEGIAN SOCIAL SCIENCE DATA SERVICES



Lene Buchert Pedagogisk forskningsinstitutt Universitetet i Oslo Postboks 1092 Blindern 0317 OSLO

Norsk samfunnsvitenskapelig datatjeneste AS



Vår dato: 10.06.2013

Vår ref:34382 / 3 / AMS

Deres dato:

Deres ref:

TILBAKEMELDING PÅ MELDING OM BEHANDLING AV PERSONOPPLYSNINGER

Vi viser til melding om behandling av personopplysninger, mottatt 28.04.2013. Meldingen gjelder prosjektet:

34382

The Contribution of Adult Literacy Education to Sustainable Development in Ethiopia Universitetet i Oslo, ved institusjonens øverste leder Behandlingsansvarlig Lene Buchert Daglig ansvarlig Fesseha Abadi Weldemichael Student

Personvernombudet har vurdert prosjektet, og finner at behandlingen av personopplysninger vil være regulert av 🖇 7-27 i personopplysningsforskriften. Personvernombudet tilrår at prosjektet gjennomføres.

Personvernombudets tilråding forutsetter at prosjektet gjennomføres i tråd med opplysningene gitt i meldeskjemaet, korrespondanse med ombudet, ombudets kommentarer samt personopplysningsloven og helseregisterloven med forskrifter. Behandlingen av personopplysninger kan settes i gang.

Det gjøres oppmerksom på at det skal gis ny melding dersom behandlingen endres i forhold til de opplysninger som ligger til grunn for personvernombudets vurdering. Endringsmeldinger gis via et eget skjema, http://www.nsd.uib.no/personvern/meldeplikt/skjema.html. Det skal også gis melding etter tre år dersom prosjektet fortsatt pågår. Meldinger skal skje skriftlig til ombudet.

Personvernombudet har lagt ut opplysninger om prosjektet i en offentlig database, http://pvo.nsd.no/prosjekt.

Personvernombudet vil ved prosjektets avslutning, 28.12.2014, rette en henvendelse angående status for behandlingen av personopplysninger.

Vennlig hilsen lights Vu

Vigdis Namtvedt Kvalheim

Anne-Mette Somber

Kontaktperson: Anne-Mette Somby tlf: 55 58 24 10 Vedlegg: Prosjektvurdering Kopi: Fesseha Abadi Weldemichael, College of Social Sciences and Languages, Mekelle University, Mekelle, Ethiopia PoBox 451, -1 UKJENT

Avdelingskontorer / District Offices

OSLO: NSD. Universitetet i OSlo, Postboks 1055 Blindern, 031760s.
CSLO: NSD. Universitetet i OSlo, Postboks 1055 Blindern, 031760s.
TRONDHEIM: NSD. Norges teknisk-naturvirenskapelige universitet, 7491 Trondheim. Tel: +47-73 59 19 07. kyrre svarva@svt.ntnu.no
TROMSØ: NSD. SVF, Universitetet i Tromsø, 9037 Tromsø, Tel: +47-77 64 43 36. nsdmaa@sv.ut.no

G. University of Oslo's Confirmation Letter for Data Collection

UO Faculty of Educational Sciences University of Oslo To whom it may concern Date: 12.12.2012 Your ref: Our ref: Confirmation

As the Department of Educational Research at the University of Oslo is now sending the PhD student researcher for field work in Ethiopia, all relevant and concerned government officials at various levels, collaborating organizations, partners, facilitators and learners of the adult literacy education sub-sector are kindly requested to provide him the required information and also facilitate his endeavors for the smooth accomplishment of the data collection process.

We are very grateful in advance for your kind cooperation.

Sincerely yours

Cost 1992 Blindern

Emil Trygve Hasle Senior Executive Officer Department of Educational Research



Department of Educational Research Postal addr.: PO Box 1092, Blindern, 0317 Oslo Visiting addr.: Sem Sælands vei 7 Helga Engs hus, 5. etasje Phone: (+47) 22 84 44 75 Telefax: (+47) 22 85 42 50 postmottak@uv.uio.no www.uv.uio.no Org. no.: 971 035 854

H. Mekelle University's Confirmation Letter for Data Collection



I. Informed Consent Form for Interview

University of Oslo Faculty of Education Department of Institute for Educational research Postboks 1092 Blindern Mobile Phone:+4792558585 E-mail:f.a.weldemichael@ped.uio.no

I volunteer to participate in a research project conducted by a research fellow Mr Fesseha Abadi Weldemichael from the University of Oslo. I understand that the project is designed to gather information about the Role of Adult Literacy Education for sustainable Development. I will be one of approximately 90 people being interviewed for this research.

My participation in this project is voluntary. I understand that I will not be paid for my participation. I may withdraw and discontinue participation at any time without penalty. If I decline to participate or withdraw from the study, no one on my campus will be told.

- 1. I understand that most interviewees will find the discussion interesting and thoughtprovoking. If, however, I feel uncomfortable in any way during the interview session, I have the right to decline to answer any question or to end the interview.
- 2. Participation involves being interviewed by a researcher from Oslo University, Norway. The interview will last approximately 30-45 minutes. During the interview, notes will be written and the interview will be audiotaped, and may be videotaped upon your agreement.
- 3. I understand that the researcher will not identify me by name in any reports using information obtained from this interview, and that my confidentiality as a participant in this study will remain secure. Subsequent uses of records and data will be subject to standard data use policies which protect the anonymity of individuals and institutions.
- 4. No one except the researcher and his assistants will be present at the interview. Neither will any other person other than the researcher and his supervisors (Professor Lene Buchert and Dr Wim Hoppers) have access to raw notes or transcripts. This precaution will prevent my individual comments from having any negative repercussions.
- 5. I understand that the findings or the results of the study might be presented in national and international conferences, published in the form of articles or monograph as of 2013.
- 6. I have read and understood the explanation provided to me. I have had all my questions answered to my satisfaction, and I voluntarily agree to participate in this study.
- 7. I have been given a copy of this consent form.

My Signature

Date

My Printed Name

Signature of the Investigator

For further information, please contact Ato Fesseha Abadi Weldemichael through the address given at the beginning of this page

J.Letter from Kilte-Awlaelo Health Office (Tigray)

3 or - 7 5 515 ብሄራዊ ክልሳዊ *መንግ*ስቲ ትግራይ The Government of the National Regional, State of Tigray, Kilte Awalaelo Wereda ቤት ዕሕፉት ዋዕና ወረዳ ክልተ አውላዕሎ Health office 4:06 714/11/06 ሰለት <u>13/03/2006</u> Sr-47 ----- TOS 114 /0 TCI BUC/ S-A -----ዋኒን፡- ብዛሪባ ዝተዋደደ ተግባር ተኮር ትም/ቲ ንአባሪሕ (IFAE) ይምልክት፡ አብ ርአስ ዋኒኑ ከምዝተባለስ ሃገርና ዝሓዙቶ ስትራተጂ ልምዓት ዋዕና አብ ዝተቸመጠሉ ግዜ ስሌዳ <u> ታክኔወት ዝተምሃረ ሓይቪ ሰብ ቸወዳዳሪ ዘይብሉ ድርሻ ከምክለዎ ይአመን። መንካስትና ድማ አቲ ብሰራሑ</u> እኖንልንልናዮ ክርክብ ከይተምሃረ ሕ/ሰብና ኖብርኡ ንምምሕይሽን ቀዕንኡ ንምሕላውን ከሕግዞ ተግባራዊ ትምህርቲ ክረክብ ምእንቲ "ዝተዋደደ ተግባር ተኮር ትምህርቲ ንአባóሕ" ዝብል ፕሮግራም ቀሪው ከተግብርን ካብ ፌዴራል ክሳብ ማብደ በለው ሲከተራት ዋዕና ፣ ሕርሻ ፣ ማይሃፍትን ትምህርትን ብዋናነት ከዋስኤሌን አብ መምርሐ አቲ ፕሮግራም አርቂቁ እዩ። ስስኪ አብ ኩለን ጣብደታት አትርከቡ ሰብ ምደ ተፅና አኪ ፕሮግራም መደባትና ንምስላተ በለዎ ራብሓ ብሎን ተግንኪብኩም አብዙ ምስ በለመ ካልኦት መዳርግቲ ሴክተራት ብምውዳድን አብ ትልምኩም ብምክታትን ክትስርሐሴን ከምዝግባእ እንዳሓበርና ብወገንና አውን እንዳተኪታተልኖን እንዳግምገምኖን ዘልዎ ደረጃ ንምፍላጥ ይዜን ናይ ደገፍ ግብረ መልሲ ንምሃብ ምእንታን ከተሪም ግና አብዚ ዓብይ መደብ እዚ <u>ግስራሕኩም ሥገርዝር ስራሕቲ አብ ወርሓዊ ወብዓብኩም አካቲትኩም ክትላእኩልና ብላዕንኦት ነተሓሳስብ።</u> በክሔ ክፍለማዜ ----- በክሔ ዝብምሃሩ ተባ ----- አነ ----- ደምሮ ----- $H.2m\sigma_0 \theta \eta^0 \gamma \eta$ መፍትሔ ነቢ ሱፐርቫይዘር ምስ ወርሓዊ ሪፖርት እናጠቅለለ ይሰ<mark>ደ</mark>ቃቸ a-10.6 . 24 ወፍሪ መለስ ንምድ ሓን አደ. ታትን ሀፃናትን!" 5 and the Ċ., 4:0.9 ACT' 0/ALDE 446. 1.1. 046.1. 4. 4 48 h 395 X 7-486 0/h/Xo-100" 3.31 30.1 0.1.1. 1.1. 1. 1. 0/11/20-1000 2 ንኩሎም ክይዲ ስራ ሎΕ ቤት ዕሕራትና ø ችሮ 200 ወቅሮ ትግራይ አብዮጵያ 031-1-13-01-13 Wukro, Tigray, Ethiopia <u>ራአይኖ</u> ቁፁይ ስድራ አብ ሕድሕድ ገዛን ወረዳኖን ምርአይ T.IN. R.D

engen seger

²⁵³

K.Letter from Meqhet Agriculture and Rural Development Office (Amhara)

.k.I.P mu/m/ I/m/A../<1//00

ቀን 24/02/2006ዓ.ም

n----- 45 + 7° + nn a/41.8 8/1.4

ስ ----- ቀበሴ ስስተዳደር ጽ/ቤት

ንዳዩ፡- የተግባር ተኮር ጉስማሶች ትምህርት ስስጣም ስስማሳመቅ ይመስከታል//

the second second

ከሳይ በርዕሱ እንደተገለፀው በመረዳቸን ሰቡሱም ቀበሴዎች 227 ጣቢያ የስርሶ አደረን ንንዛቤ ሰማሳዲግ ሰቡሱም ፓኬጀች ሳይ ስማስሰጠን ብሎም ከመዛይምነት ሰጣውጣት በያዘው የመንግስት መርሃ ንብር ሳይ የቀበሴ ሰማት ጣቢያ ባስሙያዎች የሚጠበቀባቸውን ስመወጣት ሰቀድመ ዝግጅትና ስተግባር ምዕራፍ መከናወን ያሰባቸውን ዋና ዋና ነጥቦች እንደሚከተሰው ስናሳውቃሰን።

በቀድመ ዝንጅት ምዕራፍ

- ስጹሳ ትምህርት የሚሰጥባቸውን ጣቢያዎች ስመሰየት በህብረተሰቡ ስንዲስራ ማድሪግና ስሥስማሶች ምቹ ሁኔታ መፍጠር
- የገበፈዎች ማስሰጠኛ ጣቢያ የተመረጠ ከሆነ የሙስጥና የሙጭ ገፅታዎችን ማስተካከስ ስንሰማስች ምቹ ሁኔታ መፍጠር፣
- ኝሮግራሙን በተሳካ ሁኔታ ስመፈፀም ከስጋር ስካሳት ጋር የጋራ ስቅድ ማዘጋጅት፣
- ከተዘጋጀሙ የጋራ ስቅድ በመነሳት የግብርና ዕማት ፓኬጅ ማስፈፀሚያ ስቅድ ማዘጋጀት የሰዕጣኞችን የስሙቀት ደረጃ በማጤን ስቅድን ደረጃ በደረጃ ማሙጣት፣
- ስዕጣኞች ስውቀትና ክህሱትን በጥሞና እንዲጨብጡ በማድረግ የስዕጠና ዘዲ መመሰን/ ስድራማ፣በሰርቶ ማሳይ፣መከታተስና መደንዬ፣ ቪዲዮ ማሳዮት ስስፈሳጊም ከሆነ የስካባቢውን ችግር በደስሳ ጥናት በማፈጋንጥ ስስዕጣኞች ማቀረብ ተገቢ ይሆናል።
- የስሰጠና ኘሮግራም ማዘጋጀት ስንስም በስቀድ፣ በህግጅት ፣በተግባርና በማምንማ የተከፈስ ሲሆን ሁስም በሙጤት መስካት ይኖርባቸዋል።
- ከስጋር ስካሳት ጋር ይሰሙን ተስስርና ተብብር ማጠናከር
- ስጋር ስካሳት /የተምህርት ባስሙያዎች፣የጤና፣ ባስሙያዎች ፣ግብርና የፖሲስ ስባሳት ስና የስካባቢ ጥበቃ ባስሙያዎች / የስ/ስደረን ችግርና ፍሳንት በሙስ ተረድተሙ ስንደያባዙ ጣድረግ
- በስጠቃሳይ የቀበሴሙን መስረታዊ መረጃዎች ለሣጠናክር ስስዕጣኞች ስንዲጨብጡ ለማድረግ ማዘጋጅት ይጠበቃል።

በተግባር ምሳራፍ

- 🞍 የስዕመና ስሳማ ስዕጣዮች ከስዕመና በኋሳ ምን መስራት ስንደሚችሱ የሚገስፅ መሆኑን ማረጋገጥ
- 🕹 የስካባቢ ፔቴንሻስ መስረት ያደረን ስስጠና መስጠት
- 🞍 ዮሴቶችን ተላትቬ በተግባር ማረጋንጥና ተጠቃሚ ማድረግ
- 🖌 በሚስጠው ስልጠና ስልጣኞች የሚያላትፍ መሆን ስልበት
- 🞍 ስዕጣኞች የስካባቢሙን ችግር ስንዲሰዩ ቀድሚያ መስጠት
- የትምህርት ስለጣጡ 80% በተግባር ተኮር /በመስክ ስየስራ የሚጣንረበት ሲሆን 20% በትመሬ የፖኬጅችን ሁስንተናዊ ጉዳዩች ሳይ በስጭረ በማዘጋጀት ስስልጣኞች በቀሳል ስማራጭ ማስጨበጥ ስና ከምንሰጠው ስልጠና ርዕስ ጋር የሚዛመድ መረጃ ማቅረብ
- 🗼 ስልጣኞች ተሞክሮን የመቀባበል ልምድ አንዲያደብረ ጣድረግ
- 🗼 በተስጣቸው ትምህርት ሰልጣኞችን ግብረ መልስ በመጠየቅ የታዩ ክፍተቶችን ማስተካከል
- 🔺 ሰስጣኞችን ከ10-15 ስባሳት ያሱ ቡድን ሰሣደራጅት ተላትፏቸው ስንዲደብር ማድረግ ወላኝ ተግባር ነው።
- 🗼 ስስልጣኝ የስስት ስልጠናሙን ካጠናቀቀ በኋሳ የስልጠናሙን ስስማ በድጋሜ ማስታወስ ይኖርባቸዋል።

ሳኈስማስች ትምህርት ትኩረት ሰመስመት የበኩዕችንን በመወጣት የሃንራችንን ፊልይ ስናፋምናሰን aun Pha nate YRLOW SITO enyn Desale

7117

1m/m/w/hh+222

ስመ/መ/ት/ጽ/ቤት ስኩስጣሶች ስስተባባል

L. Table on gender equality in decision making in the family by number of IFALE participants and non-participants and by gender

Tigray Amhara											
			8	2							
	Married IFALE		Marrie				Married Non-				
Iter	participants (M*=34,F**=35,		particip		particip (M=53,		participants (M=44, F=39,				
Iter	11	Total=69		(M=42, F=35, Total=77)		(WI-55, Total ='		(M - 44, F - 59, Total = 83)			
			Freq %		/) %	Freq.	/3) %	Freq			
1		4	5.8	Freq 3	3.9	9	12.33	15			
	Husband								18.0		
Main	IIusbuilu	M=1 F=3	M=1.45	M=1	M=1.3	M=6	M=8.22	M=5	M=6.02		
decision	XV'C.	<u>Γ-3</u>	F=4.35	F=2	F=2.6	F=3	F=4.11	F=10	F=11.98		
maker in	Wife	-	5.8	0	0	6 M-5	8.22	2 M-1	2. 41		
your family		M=1 F=3	M=1.45			M=5	M=6.85	M=1	M=1.205		
in for buying expensive	II	_	F=4.35	74	0(1	F=1	F=1.37	F=1	F=1.205		
things	Husband	61 M 22	88.41	74	96.1	53	72.60	62	74.70		
times	and wife,	M=32	M=46.38	M=41	M=53.24	M=38	M=52.05	M=36	M=43.37		
	together	F=29	F=42.03	F=33	F=42.86	F=15	F=20.55	F=26	F=31.33		
	Others	0	0	0	0	5	6.85	4	4.82		
						M=4	M=5.48	M=2	M=2.41		
	TT 1 · ·					F=1	F=1.37	F=2	F=2.41		
	Husband	3	4.35	4	5.19	6	8.22	16	19.28		
2		M=1	M=1.45	M=1	M=1.30	M=4	M=5.48	M=5	M=6.025		
Main		F=2	F=2.90	F=3	F=3.89	F=2	F=2.74	F=11	F=13.255		
decision	Wife	3	4.35	0	0	5	6.85	2	2.41		
maker in		M=1	M=1.45			M=4	M=5.48	M=0	M=0		
your family		F=2	F=2.90			F=1	F=1.37	F=2	F=2.41		
for selling	Husband	63	91.3	7	94.81	58	79.45	60	72.29		
expensive	and wife,	M=32	M=46.37		M=53.25	M=42	M=57.53	M=36	M=43.37		
things	together	F=31	F=44.93		F=41.56	F=16	F=21.92	F=24	F=28.92		
	Others	0	0	0	0	4	5.48	5	6.02		
						M=3	M=4.11	M=3	M=3.61		
	II	1	1.45	0	0	F=1	F=1.37	F=2	F=2.41		
3	Husband	1	1.45	0	0	1	1.37	2	2.41		
5		M=0	M=0			M=1	M=1.37	M=0	M=0		
Main	XV'C.	F=1	F=1.45	0	0	F=0	F=0	F=2	F=2.41		
decision	Wife	3 M=1	4.35 M=1.45	0	0	6 M=2	8.22 M=2.74	4 M=1	4.82 M=1.21		
maker in		F=2	F=2.90			F=4	F=5.48	F=3	F=3.61		
your family	Husband	65	94.20	77	100	66	90.41	<u>1-5</u> 77	92.77		
for having another child			M=47.82		M=54.55		M=68.49				
another child	together	F=32	F=46.38	F=35	F=45.45	F=16	F=21.92	F=34	F=40.96		
	Others	0	0	0	0	0	0	0	0		
	Husband	0	0	0	0	0	0	3	3.61		
4	LIUSDAIIU		, v	Ū			, v	M=1	M=1.20		
								F=2	F=2.41		
Main	Wife	6	8.70	3	3.90	8	10.96	12	14.46		
decision		M=1	M=1.45	M=1	M= 1.3	M=5	M=6.85	M=5	M=6.03		
maker in		F=5	F=7.25	F=2	F=2.6	F=3	F=4.11	F=7	F=8.43		
your family for using	Husband	63	91.30	72	93.51	64	87.67	68	81.93		
birth control	and wife,	M=33	M=47.82	M=40			M=65.75	M=38	M=45.78		
	together	F=30	F=43.48	F=32	F=41.56	F=16	F=21.92	F=30	F=36.15		
	Others	0	0	2	2.60	1	1.37	0	0		
				M=1	M=1.30	M=0	M=0				
				F=1	F=1.30	F=1	F=1.37				

	Husband	2	2.90	4	5.19	21	28.77	31	37.80
	iiussunu	M=0	M=0	М=0	M=0	M=1	M=21.9	M=15	M=18.29
5		F=2	F=2.90	F=4	F=5.19	F=5	F=6.85	F=16	F=19.51
Main decision	Wife	4	5.80	2	2.60	3	4.11	1	1.22
maker in	,, ne	M=1	M=1.45	M=0	M=0	M=2	M=2.74	M=1	M=1.22
your family		F=3	F=4.35	F=2	F=2.60	F=1	F=1.37	F=0	F=0
for	Husband	63	91.30	71	92.21	48	65.75	50	60.98
disciplining	and wife,	M=33	M=47.82	M=42	M=54.55	M=34		M=28	M=34.15
the child	together	F=30	F=43.48	F=29	F=37.66	F=14	F=19.18	F=22	F=26.83
	Others	0	0	0	0	1	1.37	0	0
						M=1	M=1.37		
						F=0	F=0		
	Husband	1	1.45	1	1.30	9	12.33	13	15.66
(M=0	M=0	M=1	M=1.30	M=5	M=6.85	M=5	M=6.02
6 Main		F=1	F=1.45	F=0	F=0	F=4	F=5.48	F=8	F=9.64
decision	Wife	3	4.35	0	0	5	6.85	2	2.41
maker in		M=1	M=1.45			M=4	M=5.48	M=1	M=1.205
your family		F=2	F=2.90	-		F=1	F=1.37	F=1	F=1.205
for arranging	Husband	65	94.20	76	98.70	52	71.23	62	74.70
marriage for	and wife,		M=47.82		M=53.25	M=38		M=34	M=40.96
a daughter/son	together	F=32	F=46.38	F=35	F=45.45	F=14	F=19.18	F=28	F=33.74
uaughter/son	Others	0	0	0	0	7 M-6	9.59	6 M-4	7.23
						M=6 F=1	M=8.22 F=1.37	M=4 F=2	M=4.82 F=2.41
	Husband	0	0	5	6.58	г-т 14	19.18	<u>2</u> 29	<u>г-2.41</u> 34.94
	musbanu	U	U	M=1	M=1.32	M=10	M=13.70	M=15	M=18.07
7				F=4	F=5.26	F=4	F=5.48	F=14	F=16.87
Main	Wife	6	8.70	0	0	3	4.11	5	6.02
decision	whe	M=1	M=1.45	v	v	M=2	M = 2.74	M=2	M=2.41
maker in		F=5	F=7.25			F=1	F=1.37	F=3	F=3.61
your family for	Husband	63	91.30	71	93.42	54	73.97	49	59.04
purchasing	and wife,	M=33	M=47.82		M=53.95	M=39		M=27	M=32.53
agricultural	together	F=30	F=43.48	F=30	F=39.47	F=15	F=20.54	F=22	F=26.51
inputs	Others	0	0	0	0	2	2.74	0	0
						M=2	M=2.74		
						F=0	F=0		
	Husband	1	1.45	2	2.63	19	26.03	36	43.37
0		M=0	M=0	M=1	M=1.315				M=22.89
8 Main		F=1	F=1.45	F=1	F=1.315	F=8	F=10.96	F=17	F=20.48
decision	Wife	5	7.25	0	0	3	4.11	6	7.23
maker in		M=1	M=1.45			M=2	M=2.74	M=2	M=2.41
your family		F=4	F=5.80		0=	F=1	F=1.37	F=4	F=4.82
for what	Husband	63	91.30	74	97.37	51	69.86	41	49.40
crops to sow	and wife,	M=33	M=47.82		M=53.95	M=40		M=23	M=27.71
	together	F=30	F=43.48	F=33	F=43.42	F=11	F=15.07	F=18	F=21.69
	Others	0	0	0	0	0	0	0	0

Note: M*=male, F**=female